# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

	FORM 10-Q		
Mark One) ☑ QUARTERLY REPORT PURSUANT TO SEC	TION 13 OR 15(d) OF THE SECURITI	ES EXCHANGE ACT OF 1934	
For t	he quarterly period ended June 30, 2017		
	OR		
_			
☐ TRANSITION REPORT PURSUANT TO SEC	TION 13 OR 15(d) OF THE SECURITI	ES EXCHANGE ACT OF 1934	
C	Commission File Number: 001-38093		
(Exact n	Veritone, Inc. ame of registrant as specified in its charter)		
Delaware (State or other jurisdiction of incorporation or organization)		47-1161641 (I.R.S. Employer Identification No.)	
	5 Anton Blvd., Costa Mesa, CA 92626 s of principal executive offices, including zip code)		
(Regis	(888) 507-1737 strant's telephone number, including area code)		
Indicate by check mark whether the registrant (1) has filed all preceding 12 months (or for such shorter period that the registrant was lays. Yes □ No ⊠			
Indicate by check mark whether the registrant has submitted elubmitted and posted pursuant to Rule 405 of Regulation S-T ( $\S$ 232.4 equired to submit and post such files). Yes $\boxtimes$ No $\square$			
Indicate by check mark whether the registrant is a large acceler growth company. See the definitions of "large accelerated filer," "accexchange Act. (Check one):			
arge accelerated filer		Accelerated filer	
Non-accelerated filer $\Box$ (Do not check if a smaller reporting	company)	Smaller reporting company	Σ
	Emerging growth company ⊠		
If an emerging growth company, indicate by check mark if the inancial accounting standards provided pursuant to Section 13(a) of t		on period for complying with any new or revised	
Indicate by check mark whether the registrant is a shell compare	ny (as defined in Exchange Act Rule 12b-2). Yes [	□ No ⊠	
As of July 31, 2017, there were 14,957,268 shares of common	stock outstanding.		

# VERITONE, INC. QUARTERLY REPORT ON FORM 10-Q June 30, 2017

# TABLE OF CONTENTS

PART I.	FINANCIAL INFORMATION	
Item 1.	Financial Statements (Unaudited)	
	Condensed Consolidated Balance Sheets as of June 30, 2017 and December 31, 2016	2
	Condensed Consolidated Statements of Operations for the three and six months ended June 30, 2017 and 2016	3
	Condensed Consolidated Statement of Shareholders' Equity (Deficit) for the six months ended June 30, 2017	4
	Condensed Consolidated Statements of Cash Flows for the six months ended June 30, 2017 and 2016	5
	Notes to the Condensed Consolidated Financial Statements	6
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	18
Item 3.	Ouantitative and Qualitative Disclosures About Market Risk	26
Item 4.	Controls and Procedures	26
PART II.	OTHER INFORMATION	
Item 1.	Legal Proceedings	26
Item 1A.	Risk Factors	27
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	43
Item 3.	Defaults Upon Senior Securities	44
Item 4.	Mine Safety Disclosures	44
Item 5.	Other Information	44
Item 6.	<u>Exhibits</u>	44
Signatures		45

# FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q contains certain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934, and we intend that such forward-looking statements be subject to the safe harbors created thereby. For this purpose, any statements made in this quarterly report on Form 10-Q that are not historical or current facts may be deemed to be forward-looking statements. Without limiting the generality of the foregoing, words such as "anticipates," "seeks," "estimates," "expects," "intends," "continue," "can," "may," "plans," "potential," "projects," "should," "could," "will," "would" or similar expressions and the negatives of those expressions are intended to identify forward-looking statements. Such statements include, but are not limited to, any statements that refer to projections of our future financial condition and results of operations, capital needs and financing plans, competitive position, industry environment, potential growth and market opportunities, acquisition plans and strategies, compensation plans, governance structure and policies and/or the price of our common stock.

The forward-looking statements included herein represent our management's current expectations and assumptions based on information available as of the date of this report. These statements involve numerous known and unknown risks, uncertainties and other factors that may cause our actual results, performance or achievements to be materially different from any future results, performance or achievements expressed or implied by the forward-looking statements. Factors that could cause our actual results to differ materially from our forward-looking statements include, but are not limited to, those discussed in more detail in Item 2 (Management's Discussion and Analysis of Financial Condition and Results of Operations) of Part I and in Item 1A (Risk Factors) of Part II of this Quarterly Report on Form 10-Q. Readers should carefully review these risks, as well as the additional risks described in other documents we file from time to time with the Securities and Exchange Commission. In light of the significant risks and uncertainties inherent in the forward-looking information included herein, the inclusion of such information should not be regarded as a representation by us or any other person that such results will be achieved, and readers are cautioned not to place undue reliance on such forward-looking information, which speak only as of the date of this report.

Moreover, we operate in an evolving environment. New risks and uncertainties emerge fromtime-to-time and it is not possible for our management to predict all risks and uncertainties, nor can we assess the impact of all factors on our business or the extent to which any factor, or combination of factors, may cause actual future results to be materially different from those expressed or implied by any forward-looking statements.

Except as required by law, we assume no obligation to update any forward-looking statements publicly, or to update the reasons actual results could differ materially from those anticipated in these forward-looking statements, even if new information becomes available in the future. We qualify all of our forward-looking statements by these cautionary statements.

# PART I. FINANCIAL INFORMATION

# Item 1. Financial Statements

# VERITONE, INC. CONDENSED CONSOLIDATED BALANCE SHEETS (in thousands, except per share and share data) (Unaudited)

	As of	
	June 30, 2017	December 31, 2016
ASSETS		
Cash and cash equivalents	\$ 65,912	\$ 12,078
Accounts receivable, net of allowance for doubtful accounts of \$67 and \$0, respectively	12,243	4,834
Expenditures billable to clients	8,302 2,374	3,384 1,071
Prepaid expenses and other current assets		
Total current assets	88,831	21,367
Property, equipment and improvements, net	46	68
Capitalized software, net	225	321
Other assets	169	592
Total assets	\$ 89,271	\$ 22,348
LIABILITIES, REDEEMABLE CONVERTIBLE PREFERRED STOCK AND STOCKHOLDERS' EQUITY (DEFICIT)		
Accounts payable	\$ 18,564	\$ 12,321
Accrued media payments	8,983	6,777
Other accrued liabilities	3,110	3,060
Customer advances	3,089	1,841
Convertible notes payable, net of discounts and issuance costs of \$0 and \$6,898, respectively	_	13,388
Warrant liability		7,114
Total current liabilities	33,746	44,501
Other liabilities	18	22
Total liabilities	33,764	44,523
Commitments and contingencies (Note 7)		
Redeemable convertible preferred stock:		
Series B Preferred Stock, par value \$0.001 per share; no shares authorized, issued and outstanding at June 30 2017 and 3,092,781		
shares authorized, issued and outstanding at December 31, 2016 (aggregate liquidation preference of \$18,138 at December 31,		
2016)	_	17,897
Series A Preferred Stock, par value \$0.001 per share; no shares authorized, issued and outstanding at June 30 2017 and 5,666,667 shares authorized, 3,914,697 issued and outstanding at December 31, 2016 (aggregate liquidation preference of \$8,353 at		
December 31, 2016)		5,453
Total redeemable convertible preferred stock	_	23,350
Stockholders' equity (deficit):	<u> </u>	
Preferred stock, par value \$0.001 per share; 1,000,000 shares authorized at June 30, 2017; no shares issued and outstanding at June 30, 2017	_	_
Common stock, par value \$0.001 per share; 75,000,000 and 38,500,000 shares authorized at June 30, 2017 and December 31, 2016,		
respectively; 14,956,068 and 2,620,803 shares issued and outstanding at June 30, 2017 and December 31, 2016, respectively	15	4
Additional paid-in capital	132,640	(293)
Accumulated deficit	(77,148)	(45,236)
Total stockholders' equity (deficit)	55,507	(45,525)
Total liabilities, redeemable convertible preferred stock and stockholders' equity (deficit)	\$ 89,271	\$ 22,348

The accompanying notes are an integral part of these condensed consolidated financial statements.

# VERITONE, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (in thousands, except per share and share data) (Unaudited)

		Three Months Ended June 30,		Six Months Ended June 30,		led		
		2017		2016		2017		2016
Net revenues	\$	4,087	\$	2,011	\$	7,195	\$	4,087
Cost of revenues		336		299		532		619
Gross profit		3,751		1,712		6,663		3,468
Operating expenses:				<u>_</u>	<u> </u>			·
Selling and marketing		3,414		1,840		6,013		3,486
Research and development		2,883		1,646		6,147		3,176
General and administrative		5,302		4,754		8,982		7,264
Total operating expenses		11,599		8,240		21,142		13,926
Loss from operations		(7,848)		(6,528)		(14,479)		(10,458)
Other income (expense), net		(13,746)		4		(12,960)		(27)
Loss before provision for income taxes		(21,594)		(6,524)		(27,439)		(10,485)
Provision for income taxes		1		1		3		3
Net loss		(21,595)		(6,525)		(27,442)		(10,488)
Accretion of redeemable convertible preferred stock		(3,397)		(794)		(4,470)		(1,588)
Net loss attributable to common stockholders	\$	(24,992)	\$	(7,319)	\$	(31,912)	\$	(12,076)
Net loss per share attributable to common stockholders:								
Basic and diluted	<u>\$</u>	(2.94)	\$	(3.20)	\$	(5.82)	\$	(6.89)
Weighted average shares outstanding attributable to common stockholders:								
Basic and diluted	_8	,491,391	2,	284,098	5,4	184,379	_1	,753,629

The accompanying notes are an integral part of these condensed consolidated financial statements.

# VERITONE, INC. CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY (DEFICIT) (in thousands, except share data) (Unaudited)

	Common S	Stock	Additional Paid-in	Accumulated	
	Shares	Amount	Capital	Deficit	Total
Balance as of December 31, 2016	2,620,803	\$ 4	\$ (293)	\$ (45,236)	\$(45,525)
Common stock issued through public offering, net	2,500,000	3	32,531	_	32,534
Conversion of Series B preferred stock upon public offering	2,309,135	2	18,666	_	18,668
Conversion of Series A preferred stock upon public offering	2,922,798	3	8,595	_	8,598
Exercise of Primary Warrant upon public offering	2,150,335	2	29,261	_	29,263
Conversion of Acacia Note and accrued interest upon public offering	1,523,746	2	20,735	_	20,737
Conversion of Bridge Loan and accrued interest upon public offering	590,717	1	8,044	_	8,045
Common stock issued to Bridge Loan holders	300,000	0	3,666	_	3,666
Issuance of restricted stock	34,333	_	_	_	_
Repurchase of common stock	(7,500)	(0)	(56)	_	(56)
Exercise of options	1,580	(2)	5	_	3
Issuance of warrants	_	_	8,658	_	8,658
Beneficial conversion feature	_	_	554	_	554
Modification of warrants	_	_	371	_	371
Stock-based compensation expense	10,121	_	1,903	_	1,903
Accretion of redeemable convertible preferred stock	_	_	_	(4,470)	(4,470)
Net loss				(27,442)	(27,442)
Balance as of June 30, 2017	14,956,068	\$ 15	\$132,640	\$ (77,148)	\$ 55,507

The accompanying notes are an integral part of these condensed consolidated financial statements.

# VERITONE, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands) (Unaudited)

		Six Monti June	
Net loss         \$(27,42)         \$(10,488)           Alguisments to reconcile net loss to net cash provided by (used in)         Toperating activities:           Operating activities:         3         6           Deprecation and amortization         119         61           Intangible asset impairment charges         5,90         50           Costs of warrants issued at IPO         5,70         -           Write-off of debt discounts and issuance costs at IPO         10,132         -           Change in fair value of warrant liability         67         -           Chaige in fair value of warrant liabilities         1,90         1,52           Changes in fair value of warrant liabilities         (7,46)         3,73           Accounts receivable         (7,46)         3,73           Accounts receivable         (7,47)         3,73           Expenditures biliable to clents         (7,47)         3,73           Accounts payable         6,042         4,81           Accounts payable         6,042         4,81           Act cash used in operating activities         2,206         1,51           Cash flows from investing activities         3,0         -           Cash flows from financing activities         3,0         -		2017	2016
Adjustments to reconcile net loss to net eash provided by (used in) operating activities:  Depreciation and amortization  Depreciation and amortization  1 19 61 61 16 16 16 16 16 16 16 16 16 16 16			
Depreciating activities:		\$(27,442)	\$(10,488)
Depreciation and amortization   19   61   10   10   10   10   10   10   10			
Intangible asset impairment charges		440	
Amortization of debt discounts and issuance costs			
Costs of warrants issued at IPO			500
Write-off of debt discounts and issuance costs at IPO		- 2	
Change in fair value of warrant liability         (7,114)         —           Provision for doubtful accounts         6.7         —           Stock-based compensation expense         1,903         1,522           Changes in assets and liabilities         7,476         379           Expenditures billable to clients         (1,918)         (2,184)           Expenditures billable to clients         (1,903)         (2,84)           Accounts payable         6,042         (4,810)           Accrued media payments         529         127           Customer advances         1,248         0,94           Cet cash used in operating activities         2,06         1,512           Cash flows from investing activities         3,0         —           Vet cash used in investing activities         3,0         —           Net proceeds from public offering         3,0         —			_
Provision for doubtful accounts         1,903         1,522           Stock-based compensation expense         1,903         1,522           Changes in assets and liabilities:         (7,476)         (379)           Expenditures billable to clients         (4,918)         (2,818)           Prepaid expenses and other current assets         (1,303)         (284           Accounts payable         6,042         (4,810)           Accounts payable         529         127           Other accrued liabilities         529         127           Customer advances         1,248         04           Vet cash used in operating activities         (16,47)         (15,612)           Cash flows from investing activities         (30)         —           Vet cash used in investing activities         (30)         —           Vet cash used in investing activities         (30)         —           Vet cash flows from financing activities         (30)         —           Vet cash from public offering         33,199         —           Proceeds from public offering         33,199         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —		-, -	
Stock-based compensation expense         1,903         1,522           Changes in assets and liabilities:         4         379           Expenditures billable to clients         (4,918)         (2,818           Prepaid expenses and other current assets         (1,303)         (284           Accounts payable         6,042         (4,810)           Accound media payments         5,290         1,277           Other accrued liabilities         5,29         1,277           Customer advances         1,248         (94)           Vet cash used in operating activities         (16,477)         (15,612           Cash flows from investing activities         (30)         —           Net cash used in investing activities         (30)         —           Net cash used in investing activities         (30)         —           Net proceeds from financing activities         (30)         —           Net proceeds from gracing activities         (30)         —           Net proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Proceeds from exercise of stock options         (56)         —           Net cash provided by financing activities         (56)         —		( ) ,	
Changes in assets and liabilities:         Counts receivable         (7,476)         (379)         C379         Accounts receivable         (4,918)         (2,818)         C2,818         Prepaid expenses and other current assets         (1,303)         C284         Accounts payable         (6,042)         (4,818)         C2,818         Accounts payable         (6,042)         (4,518)         C2,906         1,051         C2,006         1,051         C2,006         1,051         C2,006         1,051         C2,006         C1,051         C2,006         C2,007         C2,006         C2,007         C2,006         C2,007         C2,006         C2,007         C2,006         C2,007         <			
Accounts receivable         (7,476)         379           Expenditures billable to clients         (4,918)         (2,818           Prepaid expenses and other current assets         (1,303)         (284           Accounts payable         6,042         (4,810)           Accrued media payments         2,206         1,051           Other accrued liabilities         529         127           Customer advances         1,248         0,94           Customer advances         (16,47)         (15,612           Cash flows from investing activities         30         —           Net cash used in investing activities         30         —           Net cash used in investing activities         33,19         —           Net cash flows from financing activities         33,19         —           Net cash flows from financing activities         33,19         —           Net cash used in investing activities         33,19         —           Net cash flows from financing activities         33,19         —           Net cash grow from financing activities         33,19         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         — <t< td=""><td></td><td>1,903</td><td>1,522</td></t<>		1,903	1,522
Expenditures billable to clients         (4,918)         (2,818)           Prepaid expenses and other current assets         (1,303)         (284)           A Accounts payable         6,042         (4,810)           Other accrued liabilities         529         127           Other accrued liabilities         529         127           Customer advances         1,248         (94           Net cash used in operating activities         (16,477)         (15,612)           Cash flows from investing activities         (30)         —           Net cash used in investing activities         33,199         —           Net cash used in investing activities         33,199         —           Proceeds from financing activities         33,199         —           Net proceeds from public offering         3,800         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of Primary Warrant         29,263         —           Pobet issuance costs         (56)         —           Repurchase of common stock         (56)         —           Net cash provided by financing activities         70,341         —           Net cash and cash equivalents, end of period         33,381         (15,612     <		(7.476)	(270)
Prepaid expenses and other current assets         (1,303)         (284)           Accounts payable         6,042         (4,810)           Accrued media payments         2,206         1,051           Other accrued liabilities         529         127           Customer advances         1,248         094           Net cash used in operating activities         (16,47)         (15,612)           Cash flows from investing activities         (30)         —           Net cash used in investing activities         (30)         —           Net cash used in investing activities         33,199         —           Net proceeds from public offering         33,199         —           Proceeds from public offering         33,199         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Repurchase of common stock         (56)         —           Net cash provided by financing activities         3         —           Net cash and cash equivalents, beginning of period         35,834         (15,612)           Cash and cash equivalents, beginning of period         35,855			` /
Accounts payable         6,042         (4,810           Accrued media payments         2,206         1,051           Other accrued liabilities         1,248         0,94           Customer advances         1,248         0,94           Net cash used in operating activities         (16,477)         (15,612           Cash flows from investing activities         (30)         —           Net cash used in investing activities         (30)         —           Vet cash flows from financing activities         (30)         —           Net cash used in investing activities         (30)         —           Vet cash flows from financing activities         3,199         —           Net cash used in investing activities         8,000         —           Proceeds from public offering         3,199         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Net cash provided by financing activities         (56)         —           Net cash provided by financing activities         53,834         (15,612           Cash and cash equivalents, beginning of period         53,834         (15,612           Cash and cash equivalents, end of period         56,912			
Accrued media payments         2,206         1,051           Other accrued liabilities         529         127           Customer advances         1,248         (94           Net cash used in operating activities         (16,477)         (15,612           Cash flows from investing activities         300         —           Net cash used in investing activities         300         —           Vet cash used in investing activities         33,199         —           Cash flows from financing activities         33,199         —           Cash flows from public offering         33,199         —           Proceeds from public offering         8,000         —           Proceeds received under the Bridge Loan Agreement         8,000         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Repurchase of common stock         (68)         —           Net cash provided by financing activities         70,341         —           Net careas (decrease) in cash and cash equivalents         53,834         (15,612           Cash and cash equivalents, beginning of period         \$65,912         \$3,835           Supplemental Disclosure of Cash Flow Information		( ) ,	
Other accrued liabilities         529         127           Customer advances         1,248         69           Net cash used in operating activities         (16,477)         (15,612           Cash flows from investing activities           Net cash used in investing activities         (30)         —           Net cash used in investing activities         33,199         —           Cash flows from financing activities         33,199         —           Proceeds from public offering         33,199         —           Proceeds received under the Bridge Loan Agreement         8,000         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Debt issuance costs         (68)         —           Repurchase of common stock         (56)         —           Net cash provided by financing activities         70,341         —           Vet increase (decrease) in cash and cash equivalents         53,834         (15,612           Cash and cash equivalents, beginning of period         56,912         3,858           Supplemental Disclosure of Cash Flow Information         56,912         3,858           Supplemental Disclosure of Cash Flow Informati			
Customer advances         1,248         (94           Net cash used in operating activities         (16,477)         (15,612           Cash flows from investing activities         (30)         —           Net cash used in investing activities         (30)         —           Net cash used in investing activities         33,199         —           Cash flows from financing activities         33,199         —           Proceeds from public offering         33,199         —           Proceeds received under the Bridge Loan Agreement         8,000         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of Primary Warrant         29,263         —           Pobet issuance costs         (68)         —           Repurchase of common stock         (68)         —           Net cash provided by financing activities         70,341         —           Net increase (decrease) in cash and cash equivalents         53,834         1(5,612           Cash and cash equivalents, beginning of period         2,07,80         1,919           Cash and cash equivalents, end of period         5,07,90         3,585           Supplemental Disclosure of Cash Flow Information         5,07,90         3,585           Conversio			/
Net cash used in operating activities         (16,477)         (15,612           Cash flows from investing activities:         (30)         —           Net cash used in investing activities         (30)         —           Net cash used in investing activities:         33,199         —           Cash flows from financing activities:         33,199         —           Net proceeds from public offering         8,000         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Debt issuance costs         (68)         —           Repurchase of common stock         (56)         —           Net cash provided by financing activities         70,341         —           Vet cash provided by financing activities         53,834         (15,612           Cash and cash equivalents, beginning of period         53,834         (15,612           Cash and cash equivalents, end of period         53,835         (15,612           Cash and cash equivalents, end of period         52,782         \$ 2,8782         \$ 2,8782         \$ 2,8782         \$ 2,8782         \$ 2,8782         \$ 2,7266         \$ 2,7266         \$ 2,7266         \$ 2,7266         \$ 2,7266         \$ 2,7266         \$ 2,7266 <td>V</td> <td></td> <td></td>	V		
Cash flows from investing activities:         (30)         —           Net cash used in investing activities:         (30)         —           Cash flows from financing activities:         33,199         —           Net proceeds from public offering         33,199         —           Proceeds received under the Bridge Loan Agreement         8,000         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Debt issuance costs         (68)         —           Repurchase of common stock         (56)         —           Net cash provided by financing activities         70,341         —           Net increase (decrease) in cash and cash equivalents         53,834         (15,612           Cash and cash equivalents, beginning of period         12,078         19,197           Cash and cash equivalents, end of period         \$5,912         \$3,585           Supplemental Disclosure of Cash Flow Information         \$6,912         \$3,585           Supplemental Disclosure of Cash Flow Information         \$2,8782         \$-           Non-cash investing and financing activities:         \$2,7266         \$-			
Addition to intangible assets         (30)         —           Net cash used in investing activities         (30)         —           Cash flows from financing activities:         Set proceeds from public offering         33,199         —           Proceeds from public offering         8,000         —           Proceeds from exercise of Primary Warrant         29,263         —           Proceeds from exercise of stock options         3         —           Debt issuance costs         (68)         —           Repurchase of common stock         (56)         —           Net cash provided by financing activities         70,341         —           Net increase (decrease) in cash and cash equivalents         53,834         (15,612           Cash and cash equivalents, beginning of period         12,078         19,197           Cash and cash equivalents, end of period         \$65,912         \$3,855           Supplemental Disclosure of Cash Flow Information         Set provided by Inancing activities:         Set provided by Inancing activities         Set provided by Inancing acti	1 6	_(10,177)	(13,012)
Net cash used in investing activities  Cash flows from financing activities:  Net proceeds from public offering Proceeds received under the Bridge Loan Agreement Proceeds received under the Bridge Loan Agreement Proceeds from exercise of Primary Warrant Proceeds from exercise of stock options Proceeds from exercise of stock options Bebt issuance costs Repurchase of common stock Repurchase of common stock Repurchase of common stock Ret cash provided by financing activities Net increase (decrease) in cash and cash equivalents Sah and cash equivalents, beginning of period Sah and cash equivalents, end of period Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities: Conversion of convertible notes payable, including accrued interest, to common stock San Conversion of redeemable convertible preferred stock to common stock San		(20)	
Cash flows from financing activities:         Net proceeds from public offering       33,199       —         Proceeds received under the Bridge Loan Agreement       8,000       —         Proceeds from exercise of Primary Warrant       29,263       —         Proceeds from exercise of stock options       3       —         Debt issuance costs       (68)       —         Repurchase of common stock       (56)       —         Net cash provided by financing activities       70,341       —         Net increase (decrease) in cash and cash equivalents       53,834       (15,612         Cash and cash equivalents, beginning of period       12,078       19,197         Cash and cash equivalents, end of period       \$65,912       \$3,585         Supplemental Disclosure of Cash Flow Information       Non-cash investing and financing activities:         Conversion of convertible notes payable, including accrued interest, to common stock       \$28,782       \$—         Conversion of redeemable convertible preferred stock to common stock       \$27,266       \$—			
Net proceeds from public offering Proceeds received under the Bridge Loan Agreement Proceeds from exercise of Primary Warrant Proceeds from exercise of Primary Warrant Proceeds from exercise of stock options Proceeds from exercise of Primary Warrant Proceeds from exercise of Primary Proceeds Proceeds from exercise of Stock options Proceed	<u> </u>	(30)	
Proceeds received under the Bridge Loan Agreement 8,000 — Proceeds from exercise of Primary Warrant 29,263 — Proceeds from exercise of stock options 3 — Debt issuance costs (68 — Repurchase of common stock (56 — Net cash provided by financing activities 70,341 — Net increase (decrease) in cash and cash equivalents 53,834 (15,612 Cash and cash equivalents, beginning of period 12,078 19,197 Cash and cash equivalents, end of period \$65,912 \$3,585 Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock \$28,782 \$— Conversion of redeemable convertible preferred stock to common stock \$27,266 \$—			
Proceeds from exercise of Primary Warrant  Proceeds from exercise of stock options  Debt issuance costs  Repurchase of common stock  Net cash provided by financing activities  Net increase (decrease) in cash and cash equivalents  Cash and cash equivalents, beginning of period  Cash and cash equivalents, end of period  Supplemental Disclosure of Cash Flow Information  Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock  Supplemental Conversion of redeemable convertible preferred stock to common stock  \$ 28,782 \$ - Conversion of redeemable convertible preferred stock to common stock  \$ 27,266 \$ - Conversion of stock convertible preferred stock to common stock		,	_
Proceeds from exercise of stock options  Debt issuance costs  Repurchase of common stock  Net cash provided by financing activities  Net increase (decrease) in cash and cash equivalents  Cash and cash equivalents, beginning of period  Cash and cash equivalents, end of period  Cash and cash equivalents, end of period  Supplemental Disclosure of Cash Flow Information  Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock  Conversion of redeemable convertible preferred stock to common stock  \$ 28,782 \$ —  Conversion of redeemable convertible preferred stock to common stock  \$ 27,266 \$ —		,	
Debt issuance costs Repurchase of common stock  Net cash provided by financing activities Net increase (decrease) in cash and cash equivalents Cash and cash equivalents, beginning of period Cash and cash equivalents, end of period Cash and cash equivalents, end of period Cash and cash equivalents, end of period Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock Conversion of redeemable convertible preferred stock to common stock Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible preferred stock to common stock Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible preferred stock to common stock Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:		· · · · · · · · · · · · · · · · · · ·	_
Repurchase of common stock  Net cash provided by financing activities  Net increase (decrease) in cash and cash equivalents  Such and cash equivalents, beginning of period  Cash and cash equivalents, end of period  Cash and cash equivalents, end of period  Supplemental Disclosure of Cash Flow Information  Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock  Conversion of redeemable convertible preferred stock to common stock  \$ 28,782 \$ - \  Conversion of redeemable convertible preferred stock to common stock  \$ 27,266 \$ - \]			
Net cash provided by financing activities  Net increase (decrease) in cash and cash equivalents  Cash and cash equivalents, beginning of period  Cash and cash equivalents, end of period  Cash and cash equivalents, end of period  Supplemental Disclosure of Cash Flow Information  Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock  Conversion of redeemable convertible preferred stock to common stock  \$ 28,782 \$ - \$ \]		· ,	_
Net increase (decrease) in cash and cash equivalents  Cash and cash equivalents, beginning of period  Cash and cash equivalents, end of period  Cash and cash equivalents, end of period  Supplemental Disclosure of Cash Flow Information  Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock  Conversion of redeemable convertible preferred stock to common stock  \$ 28,782 \$	1		
Cash and cash equivalents, beginning of period 12,078 19,197 Cash and cash equivalents, end of period \$65,912 \$3,585  Supplemental Disclosure of Cash Flow Information Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock \$28,782 \$— Conversion of redeemable convertible preferred stock to common stock \$27,266 \$—			
Cash and cash equivalents, end of period \$ 5,912 \$ 3,585  Supplemental Disclosure of Cash Flow Information  Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock \$ 28,782 \$ —  Conversion of redeemable convertible preferred stock to common stock \$ 27,266 \$ —		,	( ) /
Supplemental Disclosure of Cash Flow Information  Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock  Conversion of redeemable convertible preferred stock to common stock  \$ 28,782 \$ —  Conversion of redeemable convertible preferred stock to common stock  \$ 27,266 \$ —	Cash and cash equivalents, beginning of period	12,078	19,197
Non-cash investing and financing activities:  Conversion of convertible notes payable, including accrued interest, to common stock  Conversion of redeemable convertible preferred stock to common stock  \$ 28,782 \$	Cash and cash equivalents, end of period	<u>\$ 65,912</u>	\$ 3,585
Conversion of convertible notes payable, including accrued interest, to common stock \$28,782 \$—  Conversion of redeemable convertible preferred stock to common stock \$27,266 \$—	Supplemental Disclosure of Cash Flow Information		
Conversion of redeemable convertible preferred stock to common stock \$27,266 \$—			
	Conversion of convertible notes payable, including accrued interest, to common stock	\$ 28,782	\$ —
Unpaid deferred IPO costs \$ 665 \$ —		* ', '	\$ —
	Unpaid deferred IPO costs	\$ 665	\$ —

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these condensed consolidated financial statements}.$ 

# VERITONE, INC.

Notes to the Condensed Consolidated Financial Statements (in thousands, except share and per share data) (Unaudited)

# Note 1. Description of Business and Basis of Presentation

# **Description of Business**

Veritone, Inc., a Delaware corporation, ("Veritone") (together with its wholly-owned subsidiaries, collectively, the Company) is a cloud-based cognitive software company that extracts understanding from unstructured audio and video data. The Company's platform incorporates patented technology to manage and integrate a wide variety of artificial intelligence processes, through the orchestration of machine learning algorithms known as cognitive engines to mimic human cognitive functions such as perception, reasoning, prediction and problem solving to transform unstructured data. The Company's artificial intelligence ("AI") platform stores the cognitive engine results in a searchable, time-correlated index, creating an online, searchable library of audio and video data that enables analysis and automated business solutions. Because of its open architecture, additional cognitive engines can be readily added to the platform, and new applications can be added by the Company or third parties to leverage the platform for a broad range of industries that capture or use audio and video data, including, without limitation, media, politics, legal and other commercial and government vertical markets. In addition, the Company operates a full service advertising agency. The Company's expertise in media buying, planning and creative development, coupled with its proprietary technology platform, enables the Company to deliver and analyze the effectiveness of advertising in a way that is simple, scalable and trackable.

# Initial Public Offering

In May 2017, the Company completed an initial public offering ("IPO') of its common stock. In connection with the IPO, the Company sold 2,500,000 shares of common stock at \$15.00 per share for an expected aggregate net proceeds of \$32,534 after underwriting discounts, commissions and offering costs of \$4,966.

Concurrent with the closing of the IPO on May 17, 2017, the following transactions were completed in accordance with the related agreements (see Notes 3 and 4):

- (a) The Series B Preferred Stock, which had a liquidation preference of \$18,668 and a net book value of \$18,463, was automatically converted into 2,309,135 shares of the Company's common stock;
- (b) The Series A Preferred Stock, which had a liquidation preference of \$8,598 and a net book value of \$6,272, was automatically converted into 2,922,798 shares of the Company's common stock;
- (c) The Convertible Note Payable to Acacia, which had a principal balance of \$20,000 and accrued interest of \$737, was automatically converted into 1,523,746 shares of the Company's common stock;
- (d) The Primary Warrant was automatically exercised by Acacia, which resulted in the issuance of 2,150,335 shares of the Company's common stock to Acacia in exchange for cash proceeds of \$29,263;
- (e) The Lenders funded the remaining undrawn amounts under the Bridge Loan, which provided cash proceeds of \$4,000 to the Company and increased the outstanding principal balance of the Bridge Loan to \$8,000. In connection with the funding, the Company issued to the Bridge Loan Lenders (a) an aggregate of 90,000 shares of its common stock, and (b) fully vested warrants to purchase a number of shares of the Company's common stock equal to the greater of (i) 120,000 shares of common stock, and (ii) 0.75% of the Company's fully diluted shares outstanding following completion of the IPO. Such warrants have a term of ten years following the date of issuance and have an exercise price per share equal to the lower of \$13.6088 or the IPO price per share to the public;
- (f) The Bridge Loan, which had a principal balance of \$8,000 and accrued interest of \$45, was automatically converted into an aggregate of 590,717 shares of the Company's common stock; and
- (g) The warrants to purchase an aggregate of 240,000 shares of the Company's common stock that were issued in connection with the Bridge Loan were automatically adjusted upon completion of the IPO to be exercisable to purchase an aggregate of 313,446 shares of the Company's common stock (which was equal to 1.5% of the fully diluted shares of common stock outstanding immediately following the closing of the IPO).

# Basis of Presentation

The accompanying condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP") for interim financial statements and the rules and regulations of the Securities and Exchange Commission (the "SEC"). Accordingly, they do not contain all information and footnotes required by GAAP for annual financial statements. Such unaudited condensed consolidated financial statements and accompanying notes are the representations of the Company's management, who is responsible for their integrity and objectivity. These unaudited condensed consolidated financial statements should be read in conjunction with the audited consolidated financial statements presented in the Company's final prospectus dated May 11, 2017 (the "Prospectus") contained in the Company's Registration Statement on Form S-1 (Reg. No. 333-216726).

The accompanying condensed consolidated financial statements have been prepared on the same basis as the annual financial statements and, in the opinion of management, reflect all adjustments, which are normal and recurring, necessary to fairly state its financial position, results of operations and cash flows. All significant intercompany transactions have been eliminated in consolidation. The financial data and the other information disclosed in these notes to the condensed consolidated financial statements reflected in the three-month and six-month periods presented are unaudited. The December 31, 2016 balance sheet included herein was derived from the audited financial statements, but does not include all disclosures or notes required by GAAP for complete financial statements.

# Use of Accounting Estimates

The preparation of the accompanying condensed consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the accompanying condensed consolidated financial statements and the reported amounts of revenue and expenses during the reporting period. The principal estimates relate to the valuation of common stock, stock awards, and stock warrants. Actual results could differ from those estimates.

# NOTE 2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

There have been no material changes in the Company's significant accounting policies from those that were disclosed in the Prospectus.

### Revenue Recognition

Net revenues for the three- and six-month periods presented were comprised of the following:

		Three Months Ended June 30,		ths Ended ie 30,
	2017	2016	2017	2016
Media agency revenues	\$ 3,739	\$ 1,938	\$6,638	\$ 3,974
AI platform revenues	348	73	557	113
Total net revenues	<u>\$ 4,087</u>	\$ 2,011	\$7,195	\$ 4,087

During the three months ended June 30, 2017 and 2016, the Company made \$31,243 and \$16,962 in gross media placements, of which \$28,784 and \$14,585, respectively, were billed directly to customers. Of the amounts billed directly to customers, \$25,045 and \$12,647 represented media-related costs netted against billings during the three months ended June 30, 2017 and 2016, respectively. For the six months ended June 30, 2017 and 2016, the Company made \$57,485 and \$35,037 in gross media placements, of which \$48,207 and \$29,608, respectively, were billed directly to customers. Of the amounts billed directly to customers, \$41,569 and \$25,634 represented media-related costs netted against billings for the six months ended June 30, 2017 and 2016, respectively.

# Other Income (Expense), Net

Other income (expense), net for the three- and six-month periods presented were comprised of the following:

		Three Months Ended June 30,		Ended 0,
	2017	2016	2017	2016
Interest income (expense), net	\$ (1,880)	\$ 1	\$ (4,234)	\$ 4
Stock warrants issued	(5,790)	_	(5,790)	_
Write-off of deferred debt discounts and issuance costs	(10,132)	_	(10,132)	_
Gain on fair value change of warrant liability	3,996	_	7,114	_
Other	60	3	82	(31)
Other income (expense), net	\$ (13,746)	\$ 4	\$ <u>(12,960)</u>	\$ (27)

Interest expense, net for the three- and six-month periods ended June 30, 2017 included amortization of deferred debt discounts and issuance costs of \$1,686 and \$3,740, respectively, related to the Company's convertible notes payable.

# Earnings Per Share

The following table presents the computation of basic and diluted net loss per common share for the periods presented:

	Three Months Ended June 30,		Six Mont June	
	2017	2016	2017	2016
Net loss per share:				
Numerator				
Net loss	\$ (21,595)	\$ (6,525)	\$ (27,442)	\$ (10,488)
Accretion of redeemable convertible preferred stock	(3,397)	(794)	(4,470)	(1,588)
Net loss attributable to common stockholders	\$ (24,992)	\$ (7,319)	\$ (31,912)	\$ (12,076)
Denominator				
Weighted-average common shares outstanding	8,666,147	2,424,554	5,679,024	1,862,401
Less: Weighted-average shares subject to repurchase	(174,756)	(140,456)	(194,645)	(108,772)
Denominator for basic and diluted net loss				
per share attributable to common stockholders	8,491,391	2,284,098	5,484,379	1,753,629
Basic and diluted net loss per share attributable to common stockholders	\$ (2.94)	\$ (3.20)	\$ (5.82)	\$ (6.89)

Other potentially dilutive securities that were not included in the calculation of diluted net loss per share attributable to common stockholders because their effect would be anti-dilutive are as follows (in weighted-average common equivalent shares):

	Three Months Ended June 30,		Six Montl June	
	2017	2016	2017	2016
Common stock options	2,697,361	682,415	1,682,804	636,112
Warrants to purchase common stock	2,381,530	_	2,165,872	_
Shares issuable upon conversion of redeemable convertible preferred stock		4,828,800		4,783,800
Total	5,078,891	5,511,215	3,848,676	5,419,912

# Fair Value of Financial Instruments

Fair value is defined as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. Valuation techniques used to measure fair value must maximize the use of observable inputs and minimize the use of unobservable inputs. The fair value hierarchy, which is based on three levels of inputs, the first two of which are considered observable and the last unobservable, that may be used to measure fair value, is as follows:

- Level 1—Quoted prices in active markets for identical assets or liabilities.
- Level 2—Inputs other than Level 1 that are observable, either directly or indirectly, such as quoted prices for similar assets or liabilities; quoted prices in markets that are not active; or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.
- Level 3—Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

The Company's financial instruments, other than its stock warrants, consist primarily of cash and cash equivalents, accounts receivable, accounts payable and convertible notes payable. The Company has determined that the carrying values of these instruments for the periods presented approximate fair value due to their short-term nature and the relatively stable current interest rate environment.

The Company's stock warrants are categorized as Level 3 within the fair value hierarchy. Stock warrants have been recorded at their fair value using a probability-weighted expected return model. This model incorporates contractual terms, maturity, risk free rates and volatility. The value of the Company's stock warrants would increase if a higher risk free interest rate were used, and the value of the Company's stock warrants would decrease if a lower risk free interest rate were used. Similarly, a higher volatility assumption would increase the value of the stock warrants, and a lower volatility assumption would decrease the value of the stock warrants. The unobservable inputs for Level 3 fair value measurements and fair value calculations are developed and determined by the Company's management with the assistance of a third party valuation specialist.

The following table summarizes quantitative information with respect to the significant unobservable inputs used for the Company's stock warrants that are categorized as Level 3 within the fair value hierarchy:

	December 31, 2016
Volatility	80.0%
Risk free rate	1.84%
Discount for lack of marketability	20.0%

The following table represents a reconciliation of the Level 3 measurement of the Company's Primary Warrant (see related discussion in Note 3):

Balance, December 31, 2016	\$ 7,114
Less: Change in fair value of warrant liability	<u>(7,114</u> )
Balance, June 30, 2017	<u>\$ —</u>

In addition, in May 2017, upon exercise of the Primary Warrant, the Company issued to Acacia a five-year warrant to purchase 809,400 shares of the Company's common stock, or 10% Warrant. The fair value of the 10% Warrant under Level 3 measurement is \$5,790 (see related discussion in Note 3).

The following table summarizes quantitative information with respect to the significant unobservable inputs used for the Company's 10% Warrant that are categorized as Level 3 within the fair value hierarchy:

	May 17, 2017
Volatility	70.0%
Risk free rate	1.44%
Discount for lack of marketability	0%

# Recent Accounting Pronouncements

In May 2014, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU")No. 2014-09, "Revenue from Contracts with Customers (Topic 606)," which amends the existing accounting standards for revenue recognition. ASU 2014-09 is based on principles that govern the recognition of revenue at an amount that the entity expects to be entitled to receive when products are transferred to customers. ASU 2014-09 will be effective for the Company in its year ended December 31, 2019, and for interim periods beginning with its first quarter of 2020. Early adoption is permitted. Subsequently, the FASB has issued the following standards related to ASU 2014-09: ASU No. 2016-08, Revenue from Contracts with Customers (Topic 606): Principal versus Agent Considerations; ASUNo. 2016-10, Revenue from Contracts with Customers (Topic 606): Interior (Topic 606): Narrow-Scope Improvements and Practical Expedients; and ASU 2016-20, Technical Correction and Improvements to Topic 606, Revenue from Contract with Customers. The Company must adopt ASU 2016-08, ASU 2016-10, ASU 2016-12 and ASU 2016-20 with ASU 2014-09 (collectively, the "new revenue standards"). The new revenue standards may be applied retrospectively to each prior period presented or prospectively with the cumulative effect recognized as of the date of adoption. The Company is currently evaluating the method and timing of its adoption and the impact of adopting the new revenue standards on its consolidated financial statements.

In November 2015, the FASB issued ASU No. 2015–17, "Income Taxes: Balance Sheet Classification of Deferred Taxes (Topic 740)," to require that deferred tax liabilities and assets be classified entirely as non-current. This ASU is effective for the Company in its year ended December 31, 2018 and for interim periods beginning with its first quarter of 2019. Early adoption is permitted, and the amended guidance may be applied prospectively to all deferred tax liabilities and assets or retrospectively to all periods presented. The Company is currently evaluating the impact of adopting this standard on its consolidated financial statements.

In February 2016, the FASB issued ASUNo. 2016-02, "Leases (Topic 842)." The amendments under this pronouncement will change the way all leases with duration of one year of more are treated. Under this guidance, lessees will be required to capitalize virtually all leases on the balance sheet as a right-of-use asset and an associated financing lease liability or capital lease liability. The right-of-use asset represents the lessee's right to use, or control the use of, a specified asset for the specified lease term. The lease liability represents the lessee's obligation to make lease payments arising from the lease, measured on a discounted basis. Based on certain characteristics, leases are classified as financing leases or operating leases. Financing lease liabilities, those that contain provisions similar to capitalized leases, are amortized in the same manner as capital leases are amortized under current accounting rules, as amortization expense and interest expense in the statement of operations. Operating lease liabilities are amortized on a straight-line basis over the life of the lease as lease expense in the statement of operations. This ASU is effective for the Company in its year ended December 31, 2020 and for interim periods beginning with its first quarter of 2021. The Company is currently evaluating the impact this standard will have on its policies and procedures pertaining to its existing and future lease arrangements, disclosure requirements and on its consolidated financial statements.

In March 2016, the FASB issued ASUNo. 2016-09, "Compensation—Stock Compensation (Topic 718): Improvements to Employee Share-Based Payment Accounting." The standard is intended to simplify several areas of accounting for share-based compensation arrangements, including the income tax impact, classification on the statement of cash flows and treatment of forfeitures. ASU 2016-09 is effective for the Company in its year ended December 31, 2018 and for interim periods beginning with its first quarter of 2019. Early adoption is permitted. The Company is currently evaluating the potential impact that this new standard will have on its consolidated financial statements.

In August 2016, the FASB issued ASU2016-15, "Statement of Cash Flows (Topic 230)," a consensus of the FASB's Emerging Issues Task Force," which provides guidance intended to reduce diversity in practice in how certain transactions are classified in the statement of cash flows. This ASU is effective for the Company in its year ended December 31, 2019 and for interim periods beginning with its first quarter of 2020. Early adoption is permitted. This adoption is not expected to have any significant impact on the Company's consolidated financial statements.

In May 2017, the FASB issued ASU No.2017-09, "Compensation—Stock Compensation (Topic 718): Scope of Modification Accounting," which amends the scope of modification accounting for share-based payment arrangements. Under the standard, an entity should not apply modification accounting if the fair value, vesting conditions, and classification of the awards are the same immediately before and after the modification. The current disclosure requirements in Topic 718 still applies regardless of whether an entity is required to apply modification accounting under the amendments in this update. The standard is effective for the Company beginning the first quarter of 2018, with early adoption permitted. The Company does not expect the adoption of ASU 2017-09 will have a material impact on its consolidated financial statements.

# NOTE 3. CONVERTIBLE NOTES PAYABLE AND RELATED TRANSACTIONS

### Acacia Note

In March 2017, the Company and Acacia Research Corporation ("Acacia") amended certain terms of the warrants that were included in the Investment Agreement that was entered into by the two companies on August 15, 2016 (refer to the Company's Prospectus for the full discussion of the Investment Agreement). Under the terms of the amendments, if the Company completed an IPO of its common stock with gross proceeds of at least \$15,000 (a "Qualifying IPO"), the exercise price for all warrants issued to Acacia would be the lower of \$13.6088 or the price of common stock issued in the IPO (see discussion of the IPO in Note 1). Also, the primary common stock purchase warrant issued to Acacia on August 15, 2016, as amended (the "Primary Warrant") would automatically be exercised upon the completion of a Qualifying IPO. As a result of these amendments, the fair value of the Primary Warrant decreased by \$3,118 and the credit associated with the reduction in fair value was recorded as a gain in Other Income (Expense) in the Company's condensed consolidated statement of operations for the first quarter of 2017.

Concurrent with the closing of the IPO on May 17, 2017, the convertible note payable to Acacia (the "Acacia Note"), which had a principal balance of \$20,000 and accumulated accrued interest of \$737, was automatically converted into 1,523,746 shares of the Company's common stock at a conversion price per share of \$13.6088. As a result of the conversion of the Acacia Note, the Company recorded a charge to earnings of \$4,347 representing the write-off of the unamortized balance of debt discounts associated with stock warrants and debt issuance costs which was included in Other Income (Expense) in the Company's condensed consolidated statement of operations for the second quarter of 2017.

Also, concurrent with the closing of the IPO, the Primary Warrant was automatically exercised by Acacia at an exercise price of \$13.6088 per share, which resulted in the issuance of 2,150,335 shares of the Company's common stock in exchange for a cash proceeds of \$29,263. Upon such exercise of the Primary Warrant, the Company issued to Acacia a five-year warrant to purchase 809,400 shares of the Company's common stock at an exercise price of \$13.6088 per share (the "10% Warrant"). Fifty percent of the shares under the 10% Warrant vested upon its issuance and the remaining half will vest in May 2018. The fair market value of the 10% Warrant is \$5,790, which was recorded as an expense in Other Income (Expense) in the Company's condensed consolidated statement of operations for the second quarter of 2017. Moreover, as a result of the exercise of the Primary Warrant, the balance of the Warrant Liability of \$3,996 was written-off resulting in a gain in Other Income (Expense) in the Company's condensed consolidated statement of operations for the second quarter of 2017.

# Bridge Loan

In March 2017, the Company entered into a Note Purchase Agreement with Acacia and Veritone LOC I, LLC ("VLOC") (collectively, the "Lenders"), which provided for a line of credit of up to \$8,000 (the "Bridge Loan"). The convertible promissory notes issued to the Lenders pursuant to the Bridge Loan accrued interest at the rate of eight percent (8%) per annum, compounded quarterly. The borrowings were due and payable on November 25, 2017, and the Company's obligations under this facility were secured by a security interest in substantially all of the assets of the Company, which was of equal priority to the security interests of

Acacia under the Acacia Note. The Company drew down the initial \$2,000 installment under the Bridge Loan upon the execution of the Note Purchase Agreement. In April 2017, the Company borrowed an additional \$2,000 under the Bridge Loan. In May 2017, the Lenders funded the remaining undrawn amounts of \$4,000 under the Bridge Loan, which increased the outstanding principal balance of the Bridge Loan to \$8,000.

Upon the execution of the Note Purchase Agreement, the Company issued an aggregate of 120,000 shares of the Company's common stock to the Lenders in accordance with the agreement. In addition, in connection with the funding of the \$8,000 principal amount of the Bridge Loan, the Company issued to the Lenders an aggregate of 180,000 shares of the Company's common stock and warrants to purchase an aggregate of 240,000 shares of common stock. Such warrants were automatically adjusted upon completion of the IPO to be exercisable to purchase an aggregate of 313,446 shares of the Company's common stock (which was equal to 1.5% of the fully diluted shares of common stock outstanding immediately following the closing of the IPO). The warrants have a term of ten years following the date of issuance and have an exercise price of \$13.6088 per share. As of June 30, 2017, no warrants issued in connection with the Bridge Loan had been exercised. The members of VLOC include entities controlled by Chad Steelberg and Ryan Steelberg, the Company's Chief Executive Officer and President, respectively, who own 50% of the VLOC's membership interests, and certain holders of the Company's redeemable convertible preferred stock.

Concurrent with the closing of the IPO on May 17, 2017, the \$8,000 principal balance of the Bridge Loan and accumulated accrued interest of \$45, were automatically converted into an aggregate of 590,717 shares of the Company's common stock at a conversion price per share of \$13.6088. As a result of the conversion of the Bridge Loan, the Company recorded a charge to earnings of \$2,915, representing the unamortized balance of the debt discounts associated with the common stock and warrants issued during the signing of the agreement and the debt issuance costs related to the March and April 2017 borrowings. The Company presented the write-off as an expense in Other Income (Expense) in the Company's condensed consolidated statement of operations in the second quarter of 2017. The Company also had recorded the issuance of common stock and warrants upon the funding of the remaining undrawn amounts of \$4,000 under the Bridge Loan (see share count discussion below). The Company recorded the issuance of common stock and warrants as a charge to earnings aggregating \$1,100 and \$1,770, respectively, and presented the issuances as an expense in Other Income (Expense) in the Company's condensed consolidated statement of operations in the second quarter of 2017 with a corresponding credit to Additional Paid-in Capital in the Company's condensed consolidated balance sheet

The following table represents a reconciliation of the principal amounts of the Acacia Note and the Bridge Loan (as described above) to the convertible notes payable included in the Company's condensed consolidated balance sheets as of:

	June 30, 2017	December 31, 2016
First Loan principal, August 15, 2016	\$ —	\$ 10,000
Second Loan principal, November 25, 2016	_	10,000
Debt discounts associated with stock warrants, net	_	(6,793)
Debt issuance costs, net	_	(105)
Accrued interest		286
Convertible note payable	<u>\$ —</u>	\$ 13,388

### NOTE 4. REDEEMABLE CONVERTIBLE PREFERRED STOCK

Concurrent with the closing of the IPO on May 17, 2017, the Series B Preferred Stock, which had a liquidation preference of \$18,668 and a net book value of \$18,463, was automatically converted into 2,309,135 shares of the Company's common stock, and the Series A Preferred Stock, which had a liquidation preference of \$8,598 and a net book value of \$6,272, was automatically converted into 2,922,798 shares of the Company's common stock. The differences between the liquidation preference and book value of the Series B and Series A Preferred Stock of \$205 and \$2,326, respectively, were recorded as a debit to accretion of redeemable convertible preferred stock in the Company's condensed statement of operations for the second quarter of 2017 and an increase to Additional Paid-in Capital in the Company's condensed consolidated balance sheet.

In the first quarter of 2017, the Company had recorded the amount of the beneficial conversion feature of the Series B and Series A Preferred Stock amounting to \$226 and \$328, respectively, as the respective conversion prices for such shares were less than the fair market value of the underlying preferred stock. The beneficial conversion feature resulted in the recording of a discount against the preferred stock and a corresponding credit to the Company's additional paid-in capital. The amortization of the discount for the Series B and Series A Preferred Stock during the three months ended June 30, 2017 were \$10 and \$14, respectively, and were presented as an addition to accretion of redeemable convertible preferred stock in the Company's condensed statement of operations for that period. For the six months ended June 30, 2017, the aggregate amortization of the discount for the Series B and Series A Preferred Stock were \$67 and \$96, respectively. As a result of the conversion of the Series B and Series A Preferred Stock, the Company recorded the write-off of the unamortized portion of the beneficial conversion feature of \$159 and \$232, respectively, as an addition to accretion of redeemable convertible preferred stock in the Company's condensed statement of operations in the second quarter of 2017.

In May 2017, the Board of Directors and the stockholders of the Company approved the decrease in the Company's authorized preferred stock from 11,500,000 shares to 1,000,000 shares (par value \$0.001 per share), effective following the closing of the IPO and the filing of the Company's amended and restated certificate of incorporation.

# NOTE 5. STOCKHOLDERS' EQUITY (DEFICIT)

### Reverse Split

In April 2017, the Company's Board of Directors and stockholders approved a0.6-for-1.0 reverse split of the Company's common stock, which was effected on April 20, 2017 (the "Reverse Split"). In connection with the reverse split, each share of the Company's issued and outstanding common stock was converted and combined into 0.6 shares of common stock. No fractional shares were issued in connection with the reverse split, and any fractional shares resulting from the Reverse Split will be cashed out. The Company has reflected the effect of the Reverse Split in these financial statements as if it had occurred at the beginning of the earliest period presented. As a result of the Reverse Split, the conversion price of each series of preferred stock was proportionally increased.

### Common Stock

In May 2017, the Board of Directors and the Company's stockholders approved the increase in the Company's authorized common stock to 75,000,000 shares from 38,500,000 shares (par value \$0.001 per share), effective following the closing of the IPO and the filing of the Company's amended and restated certificate of incorporation.

During the six months ended June 30, 2017, the Company issued 12,335,265 shares of common stock as follows:

- (a) 120,000 shares to the Bridge Loan Lenders upon execution of the Note Purchase Agreement in March 2017,
- (b) 90,000 shares in connection with the \$2,000 Bridge Loan borrowings in March and April 2017,
- (c) 2,500,000 shares for the IPO,
- (d) 2,309,135 shares for the conversion of the Series B preferred stock upon public offering,
- (e) 2,922,798 shares for the conversion of the Series A preferred stock upon public offering,
- (f) 2,150,335 shares for the exercise of the Primary Warrant upon public offering,
- (g) 1,523,746 shares for the conversion of the Acacia Note and accrued interest upon the IPO,
- (h) 590,717 shares for the conversion of the Bridge Loan and accrued interest upon the IPO, and
- (i) 90,000 shares in connection with the \$4,000 Bridge Loan borrowing in May 2017.

Also, the Company issued under its 2014 Plan Stock Options/Stock Issuance Plan (the "2014 Plan") 9,500 shares of restricted common stock to employees, 17,333 shares of restricted common stock to consultants for payment of services rendered, 10,121 shares of common stock to the Company's CEO as compensation for the second quarter of 2017 (see Note 7) and 1,580 shares upon the exercise of stock options (see Condensed Consolidated Statement of Stockholders Equity (Deficit) for more details).

# Common Stock Warrants

Upon signing of the Investment Agreement and in conjunction with the Company's First Loan with Acacia, the Company issued to Acacia a four-year warrant to purchase a number of shares of its common stock determined by dividing \$700,000 by an exercise price per share ranging from \$8.0833 to \$13.7323, with the actual exercise price to be determined by the type and/or valuation of its future equity financings. In conjunction with the Second Loan in November 2016, the Company issued to Acacia two additional four-year warrants, each to purchase a number of shares of the Company's common stock determined by dividing \$700,000 by an exercise price per share ranging from \$8.0833 to \$13.7323, with the actual exercise price to be determined by the type and/or valuation of its future equity financings). In March 2017, in connection with the amendment of the Primary Warrant, each of these warrants was amended to provide that the exercise prices thereof shall be equal to the lower of \$13.6088 or the initial public offering price per share. As a result, upon completion of the IPO on May 17, 2017, each of these three warrants became exercisable to purchase up to 51,437 shares of common stock, or an aggregate of 154,311 shares of common stock, at an exercise price per share of \$13.6088. In addition, upon the automatic exercise in full of the Primary Warrant, the Company issued to Acacia the 10% Warrant, which is a five-year warrant, to purchase 809,400 shares of the Company's common stock at an exercise price of \$13.6088 per share. Fifty percent of the shares under the 10% Warrant vested upon its issuance and the remaining fifty percent will vest in May 2018. The Company had also issued to the Bridge Loan Lenders in connection with the Bridge Loan warrants to purchase an aggregate of 313,446 shares of the Company had a total of 1,277,157 of warrants issued and outstanding.

# **Dispute Settlement**

On December 23, 2016, the Company entered into a settlement agreement and release relating to certain claims by a former employee, pursuant to which the Company paid to the former employee a lump sum cash payment of \$350 on January 4, 2017, which included a payment to the former employee to repurchase 7,500 shares of the Company's common stock in the amount of \$56, representing the fair value of such stock at that time. In addition, pursuant to the settlement agreement, Chad Steelberg, the Company's Chairman of the Board and Chief Executive Officer, purchased all of the former employee's membership interests in VIF I, LLC, a minority stockholder in the Company.

# Stock-Based Compensation

In 2014, the Company's Board of Directors and its stockholders adopted the 2014 Plan, which was amended in March 2015, October 2016 and April 2017. The 2014 Plan is administered by the compensation committee of the Board of Directors, which determines the recipients and the terms of the awards granted. The Plan provides that awards granted may be options, restricted stock or restricted stock option awards may be either incentive stock options or non-qualified options. Awards may be granted to eligible employees, directors and consultants under the 2014 Plan. In April 2017, the Company increased the number of shares reserved for issuance under the Plan by 900,000 shares to 4,450,585 shares.

During the second quarter of 2017, the Company granted the following equity awards to employees, directors and consultants under the 2014 Plan:

- A total of 2,089,638 time-vested stock options were granted pursuant to the Employment Agreements with Chad Steelberg and Ryan Steelberg, which have an
  exercise price of \$15.00 per share and a three-year vesting period;
- A total of 1,044,818 performance-based stock options were granted pursuant to the Employment Agreements with Chad Steelberg and Ryan Steelberg, which have an exercise price of \$15.00 per share. The performance-based stock option would vest upon the earlier of (a) the first date on which the market capitalization of the Company's common stock equals or exceeds \$400,000 over five consecutive business days, or (b) five years after the IPO; and
- A total of 666,702 stock options were granted to other employees and directors under the 2014 Plan, which have an exercise price of \$15.00 per share and a four-year vesting period.

The Company's Board of Directors has resolved not to make any further awards under the 2014 Plan following the completion of the Company's IPO. The 2014 Plan will continue to govern all outstanding awards granted thereunder.

In April 2017, the Company's Board of Directors and stockholders approved and adopted the 2017 Stock Incentive Plan (the "2017 Plan"), which became effective upon the execution of the underwriting agreement in connection with the Company's IPO. The 2017 Plan is administered by the compensation committee of the Board of Directors, which determines the recipients and the terms of the awards granted. Under the 2017 Plan, employees, non-employee directors, consultants and advisors may, at the discretion of the plan administrator, be granted options, stock appreciation rights, stock awards and restricted stock units. The Company has initially reserved 2,000,000 shares of its common stock for issuance under the 2017 Plan. The share reserve will increase automatically on the first trading day of January each calendar year, beginning with calendar year 2018, by an amount equal to 3% of the total number of shares of common stock outstanding on the last trading day in December of the immediately preceding calendar year, up to an annual maximum of 750,000 shares.

In the second quarter of 2017, under the 2017 Plan, the Company awarded to members of the Board of Directors 35,576 restricted stock units with a weighted average grant price of \$14.76. As of June 30, 2017, an aggregate of 1,964,424 shares of common stock were available for future grant under the 2017 Plan.

The Company recognizes compensation expense relating to awards granted under the 2014 and 2017 Plans ratably over the requisite service period, which is generally the vesting period. Expected forfeitures (currently at ten percent) are estimated based on historical and estimated future turnover of the Company's employees. The Company's stock-based compensation expense for the three and six months ended June 30, 2017, including expense for non-employees and the compensation of the Company's Chief Executive Officer totaled \$1,778 and \$1,903, respectively. The stock-based compensation expense recognized for the periods presented below was as follows:

		Three Months Ended June 30,		ths Ended e 30,
	2017	2016	2017	2016
Stock-based compensation expense by type of award:				
Restricted stock	\$ 375	\$ 28	\$ 451	\$ 48
Common stock awards	125	_	125	_
Stock options	1,278	1,453	1,327	1,474
Total	<u>\$ 1,778</u>	\$ 1,481	\$1,903	\$ 1,522
Stock-based compensation expense by operating expense grouping:	<u> </u>			
Sales and marketing	\$ 116	\$ 24	\$ 155	\$ 48
Research and development	92	6	104	12
General and administrative	1,570	1,451	1,644	1,462
Total	\$ 1,778	\$ 1,481	\$1,903	\$ 1,522

# Restricted Stock

Under the 2014 Plan, the Company has granted restricted stock that generally vests over four years from the date of the grant, unless the participant's service with the Company is terminated earlier. The fair value of the restricted stock grants was the estimated value per share of common stock at the date of grant determined by using both the option-pricing method and probability-weighted expected return method. Following the IPO, the fair value of the restricted stock granted is based on the closing market price of the Company's common stock at the grant date.

Restricted stock activity for the period presented was as follows:

		W	eighted
		Average Grant	
	Shares	Date l	Fair Value
Unvested at December 31, 2016	208,886	\$	4.80
Granted	52,433	\$	14.79
Forfeited	(18,100)	\$	10.03
Vested	(62,313)	\$	8.16
Unvested at June 30, 2017	180,906	\$	6.01

At June 30, 2017, total unrecognized compensation cost related to restricted stock was \$1,088, which is expected to be recognized over 2.7 years.

# Stock Options

Under the 2014 Plan, the Company has granted stock options at exercise prices equal to or greater than the fair value of the common stock on the grant date. These options expire ten years after the grant date and generally vest over a period of four years of continuous service following the vesting commencement date of such option (except for certain options granted to Chad Steelberg and Ryan Steelberg pursuant to their Employment Agreements), unless the optionee's continuous service with the Company is terminated earlier, with stock-based compensation expense recognized evenly over the requisite service period.

The fair value for each option granted was determined as of the grant date using the Black-Scholes option-pricing model. The Black-Scholes option-pricing model requires various assumptions, which are noted in the following table:

	Six Months Ended June 30, 2017
Expected terms (in years)	5.83 - 6.08
Expected volatility	65%
Risk-free interest rate	1.89% - 2.20%
Expected dividend yield	_

The expected term reflects the application of the simplified method. The simplified method defines the expected term as the average of the contractual term of the options and the vesting period for all tranches. The risk-free rate is based on the implied yield of U.S. Treasury notes as of the grant date with a remaining term approximately equal to the expected life of the award. Estimated volatility reflects historical volatility of the shares of publicly-traded peers of the Company until sufficient information regarding the volatility of the Company's shares becomes available.

The Company's stock option activity for the period presented was as follows:

			Weighted-Averag	ge
	Options	Exercise Price	Remaining Contractual Term	Aggregate Intrinsic Value
Outstanding at December 31, 2016	680,434	\$ 2.27	8.49 years	\$ 3,031
Options Granted	3,801,158	\$ 15.00		
Options Exercised	(1,580)	\$ 1.93		
Options Forfeited / Cancelled	(48,905)	\$ 2.88		
Outstanding at June 30, 2017	4,431,107	\$ 13.17	8.33 years	\$ 5,972
Exercisable at June 30, 2017	350,716			

The aggregate intrinsic value in the table above represents the difference between the fair value of the Company's common stock and the average option exercise price of in-the-money multiplied by the number of such options. At June 30, 2017, total unrecognized compensation expense related to stock options was \$33,114 and is expected to be recognized over a weighted average period of 2.3 years.

### Employee Stock Purchase Plan

In April 2017, the Company's Board of Directors and stockholders approved and adopted the employee stock purchase plan (the "ESPP"), which became effective upon the execution of the underwriting agreement in connection with the Company's IPO. The ESPP is administered by the compensation committee of the Board of Directors and is intended to qualify as an employee stock purchase plan under Section 423 of the Code. The purchase price for shares of the Company's common stock under the ESPP will be established by the plan administrator prior to the start of the offering period, but will not be less than 85% of the lower of the fair market value of the Company's common stock on (i) the date the eligible employee enters an offering period and (ii) the purchase date. Each purchase right granted to an employee will provide an employee with the right to purchase up to 1,000 shares of common stock on each purchase date within the offering period. The Company has initially reserved 1,000,000 shares of its common stock for issuance under the ESPP. The share reserve will increase automatically on the first trading day of January each calendar year, beginning with calendar year 2018, by an amount equal to 1% of the total number of shares of common stock outstanding on the last trading day in December of the immediately preceding calendar year, up to an annual maximum of 250,000 shares. As of June 30, 2017, no shares of common stock were purchased by the Company's employees under the ESPP.

# NOTE 6. RELATED PARTY TRANSACTIONS

In October 2014, the Company and Steel Ventures, LLC ("SVL"), an affiliated company whose shareholder has significant control over the Company, entered into an Intercompany Administrative Services Agreement (the "Service Agreement") effective October 1, 2014 for a two-year period, with two two-year renewal options. Pursuant to the Service Agreement, SVL agreed to make its executive management, professional, technical and clerical employees available to the Company to assist in the operation and administration of the Company's business. In addition, SVL agreed to make other services available to the Company through parties other than SVL's personnel. In consideration for the above, SVL invoiced the Company allocable costs based on a predefined allocation methodology. For the three and six months ended June 30, 2016, the Company incurred fees of \$392 and \$805, respectively, for services received under the Service Agreement. This agreement expired by its term in September 2016, and so no fees were incurred under such agreement for the three and six months ended June 30, 2017.

As fully discussed in the Prospectus, pursuant to the Investment Agreement, in August 2016, the Company entered into the Acacia Note, which provided for up to \$20,000 in borrowings through two \$10,000 advances, each bearing interest at the rate of 6.0% per annum. In August 2016, the Company borrowed the First Loan in an amount of \$10,000 that initially had a one-year term and, in November 2016, the Company borrowed the Second Loan, consisting of the remaining \$10,000 under the Acacia Note. The Second Loan also had a one-year term from the date of issuance. Upon the borrowing of the Second Loan, the maturity date of the First Loan was automatically extended to the maturity date of the Second Loan, with both loans becoming due and payable on November 25, 2017. The Acacia Note was amended and restated in March 2017 in connection with the Bridge Loan. Upon the completion of the IPO in May 2017, the outstanding \$20,000 principal and all accrued interest under the Acacia Note were converted into 1,523,746 shares of the Company's common stock at a conversion price per share of \$13.6088. Pursuant to the Investment Agreement, the Company issued to Acacia the five-year Primary Warrant to purchase up to a number of shares of the Company's common stock determined by dividing (i) \$50,000, less all outstanding principal and accrued interest under the Acacia Note, by (ii) an exercise price per share ranging from \$13.3028 to \$13.7323, with the actual exercise price per share to be determined by the amount of principal and accrued interest under the Acacia Note that are converted into the Company's common stock. Upon the completion of the IPO, the Primary Warrant was automatically exercised in full at an exercise price per share equal to \$13.6088. As such, the Company issued to Acacia an aggregate of 2,150,335 shares of the Company's common stock in exchange for cash proceeds of \$29,263.

In addition, in conjunction with the First Loan, the Company issued Acacia a four-year warrant to purchase a number of shares of its common stock determined by dividing \$700 by an exercise price per share ranging from \$8.0833 to \$13.7323, with the actual exercise price to be determined by the type and/or valuation of our future equity financings. In conjunction with the Second Loan in November 2016, the Company issued to Acacia two additional four-year warrants, each to purchase a number of shares of its common stock determined by dividing \$700 by an exercise price per share ranging from \$8.0833 to \$13.7323, with the actual exercise price to be determined by the type and/or valuation of our future equity financings). In March 2017, in connection with the amendment of the Primary Warrant, each of these warrants was amended to provide that the exercise prices thereof shall be equal to the lower of \$13.6088 or the initial public offering price per share. As a result, upon completion of the IPO, each of these three warrants became exercisable to purchase 51,437 shares of common stock, or an aggregate of 154,311 shares of common stock, at an exercise price per share of \$13.6088. Upon the exercise in full of the Primary Warrant in connection with the IPO, the Company issued to Acacia the 10% Warrant, with a term of five years, which provided for the issuance of 809,400 shares of our common stock at an exercise price of \$13.6088 per share, with fifty percent of the shares underlying the 10% Warrant vesting as of the issuance date of the 10% Warrant and the remaining fifty percent of the shares vesting on the first anniversary of the issuance date of the Company's Board of Directors.

As discussed in Note 3, the Company entered into the Note Purchase Agreement with Acacia and VLOC, which provided for an \$8,000 line of credit pursuant to Bridge Notes that accrue interest at the rate of eight percent (8%) per year, compounded quarterly, with Acacia and VLOC each purchasing equal amounts of such Bridge Notes. The Company borrowed the initial \$2,000 installment under the Bridge Loan in March 2017, and borrowed the second \$2,000 installment in April 2017. Prior to the completion of the IPO, the Lenders exercised their options to advance the remaining \$4,000 remaining available under the Bridge Loan. Upon the completion of the IPO, all of the \$8,000 principal and all accrued interest under the Bridge Notes were automatically converted into an aggregate of 590,717 shares of the Company's common stock at a conversion price of \$13.6088 per share. In connection with the Bridge Loan, the Company issued an aggregate of 120,000 shares of the Company's common stock to the Lenders upon the execution of the Note Purchase Agreement. In addition, upon the full funding of \$8,000 Bridge Loan, the Company had issued to the Lenders in the aggregate (a) 180,000 shares of the Company's common stock, and (b) fully vested ten year warrants to purchase an aggregate of 313,446 of shares of the Company's common stock. Such warrants have a ten-year term and have an exercise price per share equal to \$13.6088. Acacia's Chairman of the Board is a member of the Company's Board of Directors and Chad Steelberg and Ryan Steelberg, who are executive officers of the Company, collectively own fifty percent of VLOC's membership interests.

In March 2017, the Company entered into employment agreements with Chad Steelberg and Ryan Steelberg, who are executive officers of the Company (see Note 7 for the full discussions of the agreements).

The Company reimburses Chad Steelberg and Ryan Steelberg for the costs of their healthcare plans. During the three months ended June 30, 2017 and 2016, the Company expensed \$14 and \$15 for the cost of such plans, respectively. For the six months ended June 30, 2017 and 2016, the Company expensed \$29 and \$44 for the cost of such plans, respectively. As of June 30, 2017 and December 31, 2016, the Company had recorded an accrual of \$0 and \$73, respectively, related to these healthcare plans.

There were no other related party balances and transactions as of and for the six months ended June 30, 2017.

# NOTE 7. COMMITMENTS AND CONTINGENCIES

# Stockholders' Employment Agreements

In March 2017, the Company entered into three-year employment agreements with each of Chad Steelberg, the Company's Chief Executive Officer, and Ryan Steelberg, the Company's President. Under the agreement with Chad Steelberg, as of the end of each calendar quarter during the term of the agreement (following March 31, 2017) during which Chad Steelberg is still employed by the Company, the Company will issue to Mr. Steelberg a number of shares of its common stock calculated by dividing \$125 by the fair market value (as defined in the agreement) of the Company's common stock. The agreement with Ryan Steelberg provides that he shall receive an annual salary of \$350. During the three months ended June 30, 2017, Chad Steelberg received 10,121 shares of common stock valued at \$125.

The employment agreements also provided that the Company would issue time-based stock options to Chad Steelberg and Ryan Steelberg to allow each of them to purchase a number of shares of the Company's common stock equal to five percent of the fully diluted shares outstanding as of the closing of the Company's IPO. The exercise price per share of such time-based stock option is the IPO price of the Company's common stock (\$15), and such options would vest in monthly increments ratably over the three-year period following the closing date of the IPO (May 17, 2017). In addition, the employment agreements provided that the Company would issue to each of Chad Steelberg and Ryan Steelberg a performance-based stock option, which would allow each of them to purchase a number of shares of the Company's common stock equal to two and one-half percent of the fully diluted shares outstanding as of the closing of the Company's IPO. The performance-based stock option would vest upon the earlier of (a) the first date on which the market capitalization of the Company's common stock equals or exceeds \$400,000 over five consecutive business days, or (b) five years after the IPO. The exercise price of the performance-based awards is the initial public offering price of the Company's common stock (\$15.00). The vesting of the individual time-based and performance-based stock options would end if Chad Steelberg or Ryan Steelberg is no longer in the service of the Company, as defined in the Company's stock plan, subject to acceleration of vesting under certain circumstances as set forth in the option agreements. The stock options issuable under the Employment Agreements with Chad Steelberg and Ryan Steelberg were issued at the time of the pricing of the IPO. The total number of time-vested stock options awarded under these agreements was 2,089,638 and the total number of performance-based options awarded was 1,044,818.

# NOTE 8. SUBSEQUENT EVENTS

The Company evaluated subsequent events through August 7, 2017, the date these condensed consolidated financial statements were issued.

On July 24, 2017, the Company entered into an office building lease agreement for its Corporate Headquarters. The lease provides for the lease by the Company of approximately 37,875 square feet of office space in Costa Mesa, California. The lease has an initial term of 82 months and the Company has the option to extend the lease for two terms of five years each. Total minimum lease payments over the lease term is approximately \$9,465, net of abatements, of which approximately \$123 is expected to be paid over the next twelve months ending June 30, 2018.

There were no other material subsequent events that required recognition or additional disclosure in the accompanying condensed consolidated financial statements.

# Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion and analysis of our financial condition and results of operations together with the condensed consolidated financial statements and related notes that are included elsewhere in this Quarterly Report on Form 10-Q and our Prospectus included in the Registration Statement related to our initial public offering ("IPO"), which became effective on May 11, 2017, filed with the Securities and Exchange Commission ("SEC") pursuant to Rule 424(b) under the Securities Act of 1933, as amended. This discussion contains forward-looking statements based upon current expectations that involve risks and uncertainties. Our actual results may differ materially from those anticipated in these forward-looking statements as a result of various factors, including those set forth under "Risk Factors," set forth in Part II, Item 1A of this Quarterly Report on Form 10-Q. See "Forward-Looking Statements" above.

# Overview

Our company has two business lines: the Media Agency Business and the Artificial Intelligence ("AI") Platform Business.

# Media Agency Business

Since our inception, we have generated substantially all of our revenues from our media placement services performed under advertising contracts with our media clients. Our Media Agency Business includes media planning and strategy, media buying and placement, campaign messaging, clearance verification and attribution and custom analytics. We typically enter into agency contracts with our Media Agency customers that do not have a fixed term, but generally can be cancelled by us or the customer upon 30 to 90 days prior written notice without penalty.

In the past, our Media Agency Business has experienced volatility in its net revenues due to a number of factors, including (i) the timing of new large account wins; (ii) loss of customers who choose to replace our services by bringing their advertising placement in-house; (iii) customers who experience reductions in their advertising budgets due to issues with their own business; (iv) losses of customers who change providers from time to time based largely on pricing; and (v) the seasonality of the campaigns for certain large customers. Our Media Agency Business is also reliant on certain large key customers and has historically generated a significant portion of its net revenues from a few major customers. During the three and six months ended June 30, 2017, our ten largest customers by revenue collectively accounted for approximately 72% and 73% of our Media Agency net revenues, respectively.

# AI Platform Business

Our proprietary platform unlocks the power of AI cognitive computing so that unstructured audio and video data can be transformed and analyzed in conjunction with structured data in a seamless, automated manner to generate actionable intelligence for enterprises of all sizes. Our cloud-based, open platform integrates an ecosystem of a variety of best-of-breed cognitive engines, which can be orchestrated together, to reveal valuable multivariate insights from vast amounts of audio, video and structured data.

We are in the early stages of developing our AI Platform Business and are targeting industries that capture or use audio and video data, including in the media, politics, legal, law enforcement and other vertical markets. While we are in the process of rolling out our AI platform in a variety of markets, we have generated revenue primarily from our AI Platform Business in the media market.

We commenced commercial licensing of our service as a software ("SaaS") solutions in April 2015 with an initial focus on the media market to provide media owners and broadcasters with visibility on ad placements and the effectiveness of their media campaigns. We generally license access to our platform via a SaaS model directly to such media owners and broadcasters, and in the first six months of 2017, we signed license agreements with a leading national news and sports radio network and a major financial news television network. In the politics market, we have licensed our platform to political parties, consultants, candidates and special interest groups, who can use our platform to search and analyze large amounts of public and private media.

In the legal market, during the first quarter of 2017, we completed the integration of our platform with Relativity, kCura's industry leading eDiscovery software platform, enabling users to perform large scale analytics of audio and video discovery files using our AI platform within the Relativity environment. As of June 30, 2017, we had entered into 22 reseller agreements with providers of eDiscovery services and solutions. In the government market, during the second quarter of 2017, we signed a non-binding memorandum of understanding with The Safariland Group, a leading provider of body-worn cameras, accessories and software, regarding the parties' intent to enter into an agreement to integrate our platform with Safariland's product offerings to apply our AI technology to extract and process data from police body-worn camera footage, and we are currently negotiating the definitive agreement regarding such relationship.

The funding provided by the Acacia Investment (described below) in August 2016 and by our IPO in May 2017 has allowed us to expand our engineering team to continue the development and expansion of the capabilities of our AI platform, as well as to expand our sales and marketing team, both of which have enabled us to grow our AI Platform Business significantly. We expect to continue to invest significant resources and capital into developing our AI Platform Business, and therefore do not expect to achieve profitability in the foreseeable future.

### Acacia Investment

On August 15, 2016, we entered into an Investment Agreement with Acacia Research Corporation ("Acacia") that provided for Acacia to invest up to \$50 million in our Company, consisting of both debt and equity components. Pursuant to the Investment Agreement, on August 15, 2016, we entered into the Acacia Note, which provided for up to \$20 million in borrowings through two \$10 million advances, each bearing interest at the rate of 6.0% per annum. On August 15, 2016, we borrowed the First Loan of \$10 million that initially had a one-year term and, on November 25, 2016, we borrowed the Second Loan, consisting of the remaining \$10 million under the Acacia Note. The Second Loan also had a one-year term from the date of issuance. Upon the borrowing of the Second Loan, the maturity date of the First Loan was automatically extended to the maturity date of the Second Loan, with both loans becoming due and payable on November 25, 2017. The Acacia Note was secured by substantially all of our assets pursuant to a security agreement that we entered into with Acacia dated August 15, 2016, which was amended and restated in March 2017 in connection with our Bridge Loan (See "Bridge Loan Financing" below). Upon the completion of our IPO in May 2017, the outstanding \$20 million principal and all accrued interest under the Acacia Note was converted into 1,523,746 shares of our common stock at a conversion price per share of \$13.6088.

Pursuant to the Investment Agreement, we issued to Acacia a five-year warrant ("Primary Warrant") to purchase up to a number of shares of our common stock determined by dividing (i) \$50 million, less all outstanding principal and accrued interest under the Acacia Note, by (ii) an exercise price per share ranging from \$13.3028 to \$13.7323, with the actual exercise price per share to be determined by the amount of principal and accrued interest under the Acacia Note that are converted into our common stock. In March 2017, we amended the Primary Warrant to provide that it would be automatically exercised in full upon the closing of our IPO, and that the exercise price thereof would be equal to the lower of \$13.6088 or the IPO price per share. Upon the completion of our IPO, the Primary Warrant was automatically exercised in full at an exercise price per share of \$13.6088. As such, we issued to Acacia an aggregate of 2,510,335 shares of our common stock in exchange for cash proceeds of \$29.3 million.

In conjunction with the First Loan, we issued Acacia a four-year warrant to purchase a number of shares of our common stock determined by dividing \$0.7 million by an exercise price per share ranging from \$8.0833 to \$13.7323, with the actual exercise price to be determined by the type and/or valuation of our future equity financings. In conjunction with the Second Loan in November 2016, we issued to Acacia two additional four-year warrants, each to purchase a number of shares of our common stock determined by dividing \$0.7 million by an exercise price per share ranging from \$8.0833 to \$13.7323, with the actual exercise price to be determined by the type and/or valuation of our future equity financings. In March 2017, in connection with the amendment of the Primary Warrant, each of these warrants was amended to provide that the exercise prices thereof would be equal to the lower of \$13.6088 or the initial public offering price per share. As a result, upon completion of the IPO, each of these three warrants became exercisable to purchase 51,437 shares of common stock, or an aggregate of 154,311 shares of common stock, at an exercise price per share of \$13.6088.

Upon the exercise in full of the Primary Warrant in connection with the IPO, we issued to Acacia the 10% Warrant, with a term of five years, which provided for the issuance of 809,400 shares of our common stock at an exercise price of \$13.6088 per share, with fifty percent of the shares underlying the 10% Warrant vesting as of the issuance date of the 10% Warrant and the remaining fifty percent of the shares vesting on the first anniversary of the issuance date of the 10% Warrant.

# Bridge Loan Financing

In March 2017, we entered into a Note Purchase Agreement with Acacia and Veritone LOC I, LLC ("VLOC") (collectively, the "Lenders"), which provided for an \$8 million line of credit pursuant to secured convertible notes that accrue interest at the rate of eight percent (8%) per year, compounded quarterly (the "Bridge Notes"), with Acacia and VLOC each purchasing equal amounts of such Bridge Notes. The Bridge Notes were due and payable on November 25, 2017, and our obligations under such Bridge Notes were secured by a security interest in substantially all of our assets, which was of equal priority to the security interest of Acacia under the Acacia Note. We borrowed the initial \$2 million installment under the Bridge Loan in March 2017, and we borrowed the second \$2 million installment in April 2017. Prior to the completion of the IPO, the Lenders exercised their options to advance the remaining \$4 million remaining available under the Bridge Loan. Upon the completion of the IPO, all of the \$8 million principal and all accrued interest under the Bridge Notes were automatically converted into an aggregate of 590,717 shares of our common stock at a conversion price of \$13.6088 per share.

In connection with the Bridge Loan, we issued an aggregate of 120,000 shares of our common stock to the Lenders upon the execution of the Note Purchase Agreement. In addition, in connection with the funding of the \$8 million principal amount of the Bridge Loan, the Company issued to the Lenders an aggregate of 180,000 shares of our common stock and warrants to purchase an aggregate of 240,000 shares of our common stock. Such warrants were automatically adjusted upon completion of the IPO to be exercisable to purchase an aggregate of 313,446 shares of our common stock (representing 1.5% of our fully-diluted shares outstanding immediately following completion of the IPO) at an exercise price per share equal to \$13.6088, and have a ten-year life.

# Redeemable Convertible Preferred Stock

Upon completion of the IPO on May 17, 2017, our Series B Preferred Stock, which had a liquidation preference of \$18.7 million, was automatically converted into 2,309,135 shares of our common stock, and our Series A Preferred Stock, which had a liquidation preference of \$8.6 million, was automatically converted into 2,922,798 shares of our common stock.

# Initial Public Offering

In May 2017, we completed an underwritten IPO of 2,500,000 shares of our common stock at an IPO price per share to the public of \$15.00, pursuant to which we raised expected net proceeds of \$32.5 million, after deducting underwriting discounts and commissions and offering costs of approximately \$5.0 million. Upon completion of the IPO, all outstanding principal and accrued interest under the Acacia Note and the Bridge Loan were converted into shares of our common stock, the Primary Warrant was exercised in full by Acacia and all of our outstanding shares of Series A and Series B preferred stock were converted into shares of our common stock.

# **Factors Affecting our Performance**

We believe that the growth of our business and our future success are dependent upon many factors, including market acceptance of our product and market leadership, the success of our sales and marketing efforts, our expansion strategy, our investments for operational scale and our international growth. While each of these areas presents significant opportunities for us, they also pose important challenges that we must successfully address in order to sustain and increase the growth of our business and improve our results of operations. The investments that we make in these areas may not result in increased revenue or operating profit. Accordingly, these investments may delay or otherwise impair our ability to achieve profitability. The timing of our future profitability will depend upon many variables, including the success of our growth strategies and the timing and size of investments and expenditures that we choose to undertake, as well as market growth and other factors that are not within our control. We have not yet determined when we expect to achieve profitability.

**Product and Market Leadership.** We are committed to delivering market-leading products to continue to build and maintain credibility within the markets that we serve. We believe we must maintain and further develop our product and market position and strengthen our brand to drive future revenue growth. We intend to continue to invest in our engineering, product capabilities and sales and marketing activities to maintain and expand our position in the markets we serve. Our results of operations may fluctuate as we make these investments to drive increased customer adoption and usage of our platform.

To deliver a highly valued service offering and market-leading products to our customers, we need to add additional third-party cognitive engines to our platform. We measure the number of third-party cognitive engines that are active at the end of each period and establish goals for adding new cognitive engines each quarter. During the first half of 2017, we integrated 28 new third-party cognitive engines onto our platform.

Sales and Marketing. In order to acquire new customers in an efficient manner, we must maintain and expand our grassroots business development efforts and effectively generate additional sales to enterprises and customers across our targeted vertical markets.

Expansion Strategy. We are focused on driving greater awareness and adoption of our platform by enterprises and organizations across our vertical markets, as well as on expanding our existing customers' use of our products and platform. We believe that there is a significant opportunity to increase our penetration of these markets for cognitive analytics solutions and to drive additional sales to existing customers, and expect to invest in additional sales, marketing and customer service capabilities to support this growth.

In our media SaaS business, we are in the early stages of selling to large media companies. Our sales approach is to initially license our platform to these companies for use in one of their markets as a test. We believe that once these customers use our platform and understand its capabilities and value, many of them will choose to utilize our platform in their other markets. We expect to measure the success of this strategy by analyzing the number of new customers acquired and our revenues per customer.

Investments to Increase Scale. As our business grows and as we continue our platform optimization efforts, we expect to realize cost savings through improved economies of scale. In some cases, we may pass on these savings to our customers in the form of lower usage prices. In addition, such potential cost savings may be offset, partially or completely, by higher costs related to the release of new products and our expansion into new geographies. In addition, in some instances, we may acquire certain larger customers that we consider to be strategically important but that generate a lower gross margin. As a result, we expect our gross margins to fluctuate from period to period.

# Results of Operations (tables in \$000's, except percentages)

The following tables set forth items from our condensed consolidated statement of operations, as well as the period-over-period change, both in absolute dollars and on a percentage basis for the three and six month periods ended June 30, 2017 and 2016.

	Three Months I	Ended June 30, 2016	\$ Change	% Change
Net revenues	\$ 4,087	\$ 2,011	\$ 2,076	103.2%
Cost of revenues	336	299	37	12.4%
Gross profit	3,751	1,712	2,039	119.1%
Gross margin	91.8%	85.1%		22,72,0
Operating expenses:	) 110, V	00.17,0		
Sales and marketing	3,414	1,840	1,574	85.5%
Research and development	2,883	1,646	1,237	75.2%
General and administrative	5,302	4,754	548	11.5%
Total operating expenses	11,599	8,240	3,359	40.8%
Loss from operations	(7,848)	(6,528)	(1,320)	20.2%
Other income (expense), net	(13,746)	4	(13,750)	(343,750.0)%
Loss before provision for income taxes	(21,594)	(6,524)	(15,070)	231.0%
Provision for income taxes	1	1		_
Net loss	(21,595)	(6,525)	(15,070)	231.0%
Accretion of redeemable convertible preferred stock	(3,397)	(794)	(2,603)	327.8%
Net loss attributable to common stockholders	\$ (24,992)	\$ (7,319)	\$(17,673)	241.5%
	<del></del>			
	Six Months En		6 Cl	9/ Ch
Net revenues	2017	2016	\$ Change	% Change
Net revenues Cost of revenues	<b>2017</b> \$ 7,195	\$ 4,087	\$ 3,108	76.0%
Cost of revenues	2017 \$ 7,195 532	2016 \$ 4,087 619	\$ 3,108 (87)	76.0% (14.1)%
Cost of revenues Gross profit	2017 \$ 7,195 532 6,663	2016 \$ 4,087 619 3,468	\$ 3,108	76.0%
Cost of revenues Gross profit Gross margin	2017 \$ 7,195 532	2016 \$ 4,087 619	\$ 3,108 (87)	76.0% (14.1)%
Cost of revenues Gross profit Gross margin Operating expenses:	2017 \$ 7,195 532 6,663 92.6%	2016 \$ 4,087 619 3,468 84.9%	\$ 3,108 (87) 3,195	76.0% (14.1)% 92.1%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing	2017 \$ 7,195 532 6,663 92.6%	2016 \$ 4,087 619 3,468 84.9% 3,486	\$ 3,108 (87) 3,195	76.0% (14.1)% 92.1%
Cost of revenues Gross profit Gross margin Operating expenses:	2017 \$ 7,195 532 6,663 92.6%	2016 \$ 4,087 619 3,468 84.9%	\$ 3,108 (87) 3,195	76.0% (14.1)% 92.1%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing Research and development	2017 \$ 7,195	2016 \$ 4,087 619 3,468 84.9% 3,486 3,176	\$ 3,108 (87) 3,195	76.0% (14.1)% 92.1% 72.5% 93.5%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing Research and development General and administrative Total operating expenses	2017 \$ 7,195 532 6,663 92.6% 6,013 6,147 8,982	2016 \$ 4,087 619 3,468 84.9% 3,486 3,176 7,264	\$ 3,108 (87) 3,195 2,527 2,971 1,718	76.0% (14.1)% 92.1% 72.5% 93.5% 23.7%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing Research and development General and administrative	2017 \$ 7,195 532 6,663 92.6% 6,013 6,147 8,982 21,142	3,486 3,176 7,264 13,926	\$ 3,108 (87) 3,195 2,527 2,971 1,718 7,216	76.0% (14.1)% 92.1% 72.5% 93.5% 23.7% 51.8%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing Research and development General and administrative Total operating expenses Loss from operations Other income (expense), net	2017 \$ 7,195 532 6,663 92.6% 6,013 6,147 8,982 21,142 (14,479) (12,960)	3,486 3,176 7,264 13,926 (10,458) (27)	\$ 3,108 (87) 3,195 2,527 2,971 1,718 7,216 (4,021) (12,933)	76.0% (14.1)% 92.1% 72.5% 93.5% 23.7% 51.8% 38.4%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing Research and development General and administrative Total operating expenses Loss from operations	2017 \$ 7,195 532 6,663 92.6% 6,013 6,147 8,982 21,142 (14,479)	3,486 3,176 7,264 13,926 (10,458)	\$ 3,108 (87) 3,195 2,527 2,971 1,718 7,216 (4,021)	76.0% (14.1)% 92.1% 72.5% 93.5% 23.7% 51.8% 38.4% (47,900.0)%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing Research and development General and administrative Total operating expenses Loss from operations Other income (expense), net Loss before provision for income taxes	2017 \$ 7,195 532 6,663 92.6% 6,013 6,147 8,982 21,142 (14,479) (12,960) (27,439)	3,486 3,176 7,264 13,926 (10,458) (27) (10,485)	\$ 3,108 (87) 3,195 2,527 2,971 1,718 7,216 (4,021) (12,933) (16,954)	76.0% (14.1)% 92.1% 72.5% 93.5% 23.7% 51.8% 38.4% (47,900.0)% 161.7%
Cost of revenues Gross profit Gross margin Operating expenses: Sales and marketing Research and development General and administrative Total operating expenses Loss from operations Other income (expense), net Loss before provision for income taxes Provision for income taxes	2017 \$ 7,195 532 6,663 92.6% 6,013 6,147 8,982 21,142 (14,479) (12,960) (27,439) 3	3,486 3,176 7,264 13,926 (10,458) (27) (10,485) 3	\$ 3,108 (87) 3,195 2,527 2,971 1,718 7,216 (4,021) (12,933) (16,954)	76.0% (14.1)% 92.1% 72.5% 93.5% 23.7% 51.8% 38.4% (47,900.0)%

Three Months Ended June 30, 2017 Compared with the Three Months Ended June 30, 2016

Net Revenues

	Three Months Ended			
		June 30,		
	2017	2016	\$ Change	% Change
Media agency revenues, net	\$ 3,739	\$ 1,938	\$ 1,801	92.9%
AI Platform revenues	348	73	275	376.7%
Net revenues	<u>\$ 4,087</u>	\$ 2,011	\$ 2,076	103.2%

The year-over-year increase in net revenues was due primarily to the increase in our Media Agency revenue of \$1.8 million, or 93%, compared with the prior year quarter. Our media dollars placed per agency customer increased 39% quarter-over-quarter to \$0.7 million per agency customer in the second quarter of 2017 from \$0.5 million in the second quarter of 2016. In the second quarter of 2017, we added 16 net new Media Agency customers under master service agreements compared with 12 net new customers in the prior year quarter, and we had 45 active Media Agency customers at the end of the quarter, compared with 35 at the prior year quarter-end, an increase of 29%. Contributing to the year-over-year increase in net revenues was the increase in AI Platform revenues by \$0.3 million, or approximately 380%, from customers in the media market in second quarter of 2017 compared with the same quarter in the prior year. Our AI Platform Business added 114 accounts in the second quarter of 2017 and had 169 accounts at the end of the quarter. Also, in the second quarter of 2017, we ingested and processed 438,000 hours of video or audio content, compared with 178,000 hours in the same quarter of 2016, an increase of 146%.

# Cost of Revenue and Gross Profit

The cost of revenues increased by 12%, or less than \$0.1 million, due primarily to increased Media Agency production costs. Gross margin increased quarter-over-quarter due primarily to the operating leverage provided by our higher net revenue level.

# **Operating Expenses**

	Three Months Ended June 30,					
		2017		2016	\$ Change	% Change
Sales and marketing	\$	3,414	\$	1,840	\$ 1,574	85.5%
Research and development		2,883		1,646	1,237	75.2%
General and administrative		5,302		4,754	548	11.5%
Total operating expenses	\$	11,599	\$	8,240	\$ 3,359	40.8%
Percentage of net revenues:						
Sales and marketing		83.5%		91.5%		
Research and development		70.5%		81.8%		
General and administrative		129.7%		236.4%		
Total operating expenses		283.8%		409.7%		

Sales and marketing. The increase in sales and marketing expense in the second quarter of 2017 compared with the prior year quarter was due primarily to increased personnel and consulting costs resulting from the expansion of sales and marketing efforts in the legal, politics and government markets for our AI Platform Business. Also, our Media Agency costs increased due to variable compensation increases linked to the year-over-year net revenues increase and additional headcount to support the increased business volume.

**Research and development.** The increase in research and development expense in the second quarter of 2017 compared with the prior year quarter was attributable primarily to an increase in our personnel costs of \$1.1 million, resulting from a 147% net increase in headcount in the current year quarter, as we enhance our existing products and develop new products and other technical functionality.

General and administrative. The increase in general and administrative expense in the second quarter of 2017 compared with the prior year quarter was due primarily to an increase in our personnel costs of \$1.3 million, reflecting a 68% net increase in our headcount to support the expected growth of our business, offset by a reduction in settlement costs and legal fees which surged in the second quarter of 2016 due to a dispute with a dissenting shareholder.

In the second quarter of 2017, we recorded stock-based compensation expense of \$1.8 million, attributable primarily to a broad-based grant of equity awards to employees and directors that occurred in May 2017. Based on the equity awards that were outstanding and unvested as of June 30, 2017, we expect that our stock-based compensation expense will increase to approximately \$2.5 million per quarter in the second half of 2017.

In the second quarter of 2016, we recorded stock-based compensation expense of \$1.5 million, substantially all of which was attributable to the issuance of shares of our common stock to an entity beneficially owned by Chad Steelberg and Ryan Steelberg, in consideration of services rendered to us in the first four months of 2016. Such stock-based compensation expense did not recur in the 2017 period.

# Other Income (Expenses), Net

Other income (expense), net in the second quarter of 2017 was comprised primarily of the following:

- The net book value of debt discounts and issuance costs were written off when the related debt was converted into common stock in connection with the IPO, which resulted in expense of \$7.3 million;
- Upon the automatic exercise in full of the Primary Warrant in connection with the IPO, we issued to Acacia a five-year warrant to purchase up to 809,400 shares of our common stock at an exercise price of \$13.6088 per share (the "10% Warrant"). Fifty percent of the shares under the 10% Warrant vested upon its issuance and the remaining half will vest in May 2018. The fair market value of the 10% Warrant is \$5.8 million, which was recorded as an expense;
- Additional common stock and warrants were issued to the Bridge Loan lenders under the terms of the March 2017 agreements, and the fair value of these instruments is \$2.9 million, which was recorded as an expense;
- As the Primary Warrant was exercised in connection with the IPO, the carrying value of this warrant, which was reflected as a liability, was reversed which resulted in a gain of \$4.0 million; and
- Debt discount amortization, interest expense and debt issuance cost amortization from the beginning of the quarter through the date that the related debt instruments were converted to common stock resulted in expense of \$1.9 million.

# Six Months Ended June 30, 2017 Compared with the Six Months Ended June 30, 2016

# Net Revenues

	Six Mor			
	Ju	June 30,		
	2017	2016	\$ Change	% Change
Media agency revenues, net	\$6,638	\$ 3,974	\$ 2,664	67.0%
AI Platform revenues	557	113	444	392.9%
Net revenues	<u>\$7,195</u>	\$ 4,087	\$ 3,108	76.0%

The year-over-year increase in net revenues was due primarily to the increase in our Media Agency revenues of \$2.7 million, or 67%, compared with the prior year period. In the first half of 2017, we added 24 net new Media Agency customers under master service agreements compared with 16 net new customers in the prior year period and the increase in net revenues was driven primarily by revenues earned from these new customers. The customer count increase was the primary contributor to the media agency net revenue increase. Contributing to the year-over-year increase in net revenues was the increase in AI Platform revenues of \$0.4 million, or approximately 390%, primarily from customers in the Media Market. In the first half of 2017, our AI Platform Business added 146 accounts. Also, in the first half of 2017, we ingested and processed 805,000 hours of video or audio content, compared with 367,000 hours in the first half of 2016, an increase of 119%.

# Cost of Revenue and Gross Profit

The cost of revenues decreased by 14%, or approximately \$0.1 million, due primarily to the decrease in transcription costs per hour in our AI Platform Business, which were due, in part, to the higher volume of data being processed. This decrease was offset in part by the increased production costs in our Media Agency Business in the second quarter of 2017. Gross margin increased period-over-period due primarily to the operating leverage provided by our higher net revenue level.

# **Operating Expenses**

	Six Months En			
	2017	2016	\$ Change	% Change
Sales and marketing	\$ 6,013	\$ 3,486	\$ 2,527	72.5%
Research and development	6,147	3,176	2,971	93.5%
General and administrative	8,982	7,264	1,718	23.7%
Total operating expenses	\$ 21,142	\$ 13,926	\$ 7,216	51.8%
Percentage of net revenues:				
Sales and marketing	83.6%	85.3%		
Research and development	85.4%	77.7%		
General and administrative	124.8%	177.7%		
Total operating expenses	293.8%	340.7%		

Sales and marketing. The increase in sales and marketing expense in the first half of 2017 compared with the prior year period was due primarily to increased personnel and consulting costs resulting from the expansion of sales and marketing efforts in the legal, politics, and public-sector categories. Also, our Media Agency costs increased due to variable compensation increases linked to the year over year net revenues increase and additional headcount to support the increased business volume.

**Research and development.** The increase in research and development expense in the first half of 2017 compared with the prior year period was attributable primarily to an increase in our personnel costs of \$2.4 million, resulting from a 137% net increase in average month-end headcount in the current year period, as we enhance our existing products and develop new products and other technical functionality.

General and administrative. The increase in general and administrative expense in the first half of 2017 compared with the prior year period was due primarily to an increase in our personnel costs of \$2.2 million, reflecting a 72% net increase in average month-end headcount to support the expected growth of our business, offset by a reduction in settlement costs and legal fees.

# Other Income (Expenses), Net

Other income (expense), net for the first half of 2017 was comprised primarily of the following:

- The net book value of debt discounts and issuance costs were written off when the related debt was converted into common stock in connection with the IPO, which
  resulted in expense of \$7.3 million;
- Upon the automatic exercise in full of the Primary Warrant in connection with the IPO, we issued to Acacia the 10% Warrant. Fifty percent of the shares under the 10% Warrant vested upon its issuance and the remaining half will vest in May 2018. The fair market value of the 10% Warrant is \$5.8 million, which was recorded as an expense;
- Additional common stock and warrants were issued to the Bridge Loan lenders under the terms of the March 2017 agreements, and the fair value of these instruments is \$2.9 million, which was recorded as an expense;
- As the Primary Warrant was exercised in connection with the IPO, the carrying value of this warrant, which was reflected as a liability, was reversed which resulted in a gain of \$7.1 million; and,
- Debt discount amortization, interest expense and debt issuance cost amortization from the beginning of the year through the date that the related debt instruments were converted to common stock resulted in expense of \$4.2 million.

# Liquidity and Capital Resources

### Cash Flows

A summary of our net operating, investing and financing activities is shown in the table below.

	Six Month June	
	2017	2016
Cash used in operating activities	\$(16,477)	\$(15,612)
Cash used in investing activities	(30)	_
Cash provided by financing activities	70,341	
Net increase (decrease) in cash and cash equivalents	\$ 53,834	<u>\$(15,612</u> )

# Net Cash Used in Operating Activities

Our operating activities used cash of \$16.5 million in the first six months of 2017, compared with \$15.6 million in the first six months of 2016. Ouyear-to-date net loss increased to \$27.4 million in 2017 compared to a net loss of \$10.5 million in the same period in 2016. The first six months of 2017, included a net noncash charge of \$12.5 million related primarily to amortization of debt discounts and issuance costs, costs of stock warrants issued at the time of our IPO and write-off of debt discounts and issuance costs, offset in part by the change in fair value of warrant liability, none of which were incurred in 2016. Although the operating loss for the two year-to-date periods is still higher in 2017 after deducting the net noncash charges, in 2016, we had a larger amount of media payments to make than in 2017, which resulted in our cash used in operating activities to be similar in the two periods.

### Net Cash Provided by Financing Activities

Our financing activities provided cash of \$70.3 million in the first six months of 2017, due primarily to the \$33.2 million in net proceeds received from our IPO, the \$29.3 million in proceeds received from the exercise of the Primary Warrant by Acacia and \$8 million in borrowings under the Bridge Loan Agreement.

# Capital Resources

On August 15, 2016, we borrowed \$10 million in the First Loan under the Acacia Note and we borrowed the remaining \$10 million in the Second Loan on November 25, 2016. The Acacia Note accrued interest at the rate of 6.0% per annum and was due and payable on November 25, 2017. At December 31, 2016, the Acacia Note had a carrying value of \$13.4 million, net of unamortized deferred discounts and issuance costs of \$6.9 million. Upon the closing of our IPO in May 2017, the principal balance of \$20 million and the accumulated accrued interest of \$0.7 million under the Acacia Note were converted into shares of our common stock (see "Item 2. – Overview – Bridge Loan Financing."). Accordingly, no amounts remained outstanding under the Acacia Note at June 30, 2017.

In March 2017, we entered into the Note Purchase Agreement with Acacia and VLOC, which provided for an \$8 million Bridge Loan pursuant to the Bridge Notes, which accrued interest at the rate of eight percent (8%) per annum, compounded quarterly, and were due and payable on November 25, 2017 (see "Item 2. – Overview – Bridge Loan Financing."). In March 2017, we borrowed the first \$2 million installment under the Bridge Loan and, in April 2017, we borrowed an additional \$2 million under the Bridge Loan. In connection with our IPO, the Lenders elected to fund the \$4 million remaining undrawn under the Bridge Notes, and upon the closing of the IPO on May 17, 2017, all of the \$8 million principal amount under the Bridge Notes and the accumulated accrued interest of less than \$0.1 million were automatically converted into shares of our common stock at a conversion price per share of \$13.6088.

In May 2017, we completed our IPO, Acacia exercised the Primary Warrant and the Lenders funded the \$4 million remaining available under the Bridge Notes. As a result of these transactions, we received cash proceeds of approximately \$68 million. We believe that these cash proceeds will be sufficient to fund our cash requirements for the next 12 months and beyond. All outstanding principal and accrued interest under the Acacia Note and the Bridge Notes were also converted into common stock upon completion of the IPO in May 2017 (see "Item 2. – Overview – Acacia Investment" and "Overview – Bridge Loan Financing").

# Contractual Obligations

As discussed above, all of our outstanding principal and accumulated accrued interest under the Acacia Note and the Bridge Notes were converted into our shares of common stock. As of June 30, 2017, we have no material contractual obligations requiring disclosure.

On July 14, 2017, we entered into an office building lease agreement for our Corporate Headquarters. The lease provides for the lease by us of approximately 37,875 square feet of office space in Costa Mesa, California. The lease has an initial term of 82 months and we have the option to extend the lease for two terms of five years each. Total minimum lease payments over the lease term is \$9.5 million, net of abatements, of which approximately \$0.1 million is expected to be paid over the next twelve months ending June 30, 2018.

### Off-Balance Sheet Arrangements

We have not entered into any off-balance sheet arrangements and do not have any holdings in variable interest entities.

### Item 3. Quantitative and Qualitative Disclosures About Market Risk

Management believes there have been no material changes to our quantitative and qualitative disclosures about market risks during the six months ended June 30, 2017 compared with those discussed in the Prospectus (see Prospectus for discussion of our quantitative and qualitative disclosures about market risks).

# Item 4. Controls and Procedures

Evaluation of disclosure controls and procedures

Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, have evaluated the effectiveness of our disclosure controls and procedures as of the end of the period covered by this quarterly report. Disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended) are designed to ensure that information required to be disclosed by a company in the reports that it files or submits under the Exchange Act is (i) recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and (ii) accumulated and communicated to the company's management, including its principal executive officer and principal financial officer, as appropriate to allow timely decisions regarding required disclosure. Based on the evaluation of our disclosure controls and procedures, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective as of the end of the period covered by this quarterly report at the reasonable assurance level.

Changes in Internal Control over Financial Reporting

There were no changes in our internal control over financial reporting that occurred during the last fiscal quarter covered by this quarterly report that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

Limitations on Effectiveness of Controls and Procedures

Our disclosure controls and procedures and internal control over financial reporting are designed to provide reasonable assurance of achieving the desired control objectives. Our management recognizes that any control system, no matter how well designed and operated, is based upon certain judgments and assumptions and cannot provide absolute assurance that its objectives will be met. In addition, the design of disclosure controls and procedures must reflect the fact that there are resource constraints and that management is required to apply judgment in evaluating the benefits of possible controls and procedures relative to their costs. Similarly, an evaluation of controls cannot provide absolute assurance that misstatements due to error or fraud will not occur or that all control issues and instances of fraud, if any, have been detected.

# PART II. OTHER INFORMATION

# Item 1. Legal Proceedings

We are currently not a party to any material legal proceedings but may become involved in legal proceedings arising from our business activities. While management does not believe the ultimate disposition of these matters will have a material adverse impact on our consolidated results of operations, cash flows or financial position, litigation is inherently unpredictable, and depending on the nature and timing of these proceedings, an unfavorable resolution could materially affect our future consolidated results of operations, cash flows or financial position in a particular period.

# Item 1A. Risk Factors

You should carefully consider the risks described below, together with all of the other information included in this quarterly report on Form 10-Q, before making an investment decision. Our business, financial condition and results of operations could be materially and adversely affected by any of these risks or uncertainties. In that case, the trading price of our common stock could decline, and you may lose all or part of your investment.

# Risks Related to Our Business, Industry and Financial Condition

Most of our revenues are currently generated by our Media Agency Business, and our effort to expand our AI Platform business may not succeed.

In the first half of 2017 and fiscal years 2016 and 2015, most of our revenues were advertising-related revenues generated from media placement services performed under advertising contracts with our media clients. We typically receive a percentage of the total advertising placement by these customers with selected media sources. We did not commence licensing of our AI platform until April 2015, and SaaS licensing revenue from our AI platform was \$0.5 million in the six months ended June 30, 2017 and \$0.5 million and less than \$0.1 million in the years ended December 31, 2016 and 2015, respectively. In order for us to grow our business and achieve profitability, we must expand our revenue base by ramping up our SaaS licensing business and entering into additional licensing agreements. However, we are currently in the early stage of developing our SaaS licensing business, and there is no guarantee that we will succeed.

Many factors may adversely affect our ability to establish a viable and profitable SaaS licensing business, including but not limited to:

- Failure to add additional cognitive engines with sufficient levels of capability into our platform (or difficulties in integrating any such cognitive engines), or loss of access to such cognitive engines;
- Inability to expand the automation capabilities of our Conductor technology to other types of cognitive engines;
- Failure to add additional market-specific capabilities and analytics for each of our vertical markets;
- Failure to articulate the perceived benefits of our solution, or failure to persuade potential customers that such benefits justify the additional cost over single cognitive engine solutions;
- Introduction of competitive offerings by larger, better financed and more well-known companies;
- Inability to enter into satisfactory agreements relating to the integration of our platform with products of other companies to pursue particular vertical markets, such as our proposed agreement with the Safariland Group, Inc., or the failure of such relationships to achieve their anticipated benefits;
- · Failure to provide adequate customer support;
- Long sales cycles for customers in the government and law enforcement markets;
- · Failure to generate broad customer acceptance of or interest in our solutions;
- Increases in costs or lack of availability of certain cognitive engines;
- Challenges in operating our platform on secure government cloud platforms;
- Inability to continue to access public media for free;
- · Higher data storage and computing costs; and
- Difficulties in adding technical capabilities to our platform and ensuring future compatibility of additional third party providers.

If we fail to develop a successful SaaS licensing business, or if we are unable to ramp up our SaaS operations in a timely manner or at all, our business, results of operations and financial condition will suffer.

The artificial intelligence market is new and unproven, and it may decline or experience limited growth, which would adversely affect our ability to fully realize the potential of our platform.

The artificial intelligence market is relatively new and unproven and is subject to a number of risks and uncertainties. We believe that our future success will depend in large part on the growth of this market. The utilization of our platform by customers is still relatively new, and customers may not recognize the need for, or benefits of, our platform, which may prompt them to decide to adopt alternative products and services to satisfy their cognitive computing search and analytics requirements. In order to expand our business and extend our market position, we intend to focus our marketing and sales efforts on educating customers about the benefits and technological capabilities of our platform and the application of our platform to the specific needs of customers in different market verticals. Our ability to expand the market that our platform addresses depends upon a number of factors, including the cost, performance and perceived value of our platform. Market opportunity estimates are subject to significant uncertainty and are based on assumptions and estimates, including our internal analysis and industry experience. Assessing the market for our SaaS solutions in each of the vertical markets we are competing in, or are planning to compete in, is particularly difficult due to a number of factors, including limited available information and rapid evolution of the market. The market for our platform or AI cognitive computing in general may fail to grow significantly or be unable to meet the level of growth expected by us. As a result, we may experience significant reduction in demand for our products and services due to lack of customer acceptance, technological challenges, competing products and services, decreases in spending by current and prospective customers, weakening economic conditions and other causes. If our market does not experience significant growth, or if demand for our platform decreases, then our business, results of operations and financial condition will be adversely affected.

# We rely on third parties to develop cognitive engines for our platform and to create the related Application Program Interfaces (APIs).

A key element of our platform is the ability to incorporate and integrate cognitive engines developed by multiple third-party vendors into a single, open platform, and we plan to increase the number of third-party cognitive engines incorporated into our platform in order to enhance the performance and power of our platform. As we become increasingly dependent on third-party developers for new cognitive engines, we may encounter difficulties in identifying additional high-quality cognitive engines, entering into agreements for their inclusion in our ecosystem on acceptable terms or at all and/or in coordinating and integrating their technologies into our system. We may incur additional costs to modify and adjust existing functionalities of our platform to accommodate multiple classes of third-party cognitive engines, without the assurance that such costs can be recouped by the additional revenues generated by the new capabilities. As our platform becomes more complex due to the inclusion of various third-party cognitive engines, we may not be able to integrate them in a smooth or timely manner due to a number of factors, including incompatible software applications, lack of cooperation from developers, insufficient internal technical resources, and the inability to secure the necessary licenses or legal authorizations required. In addition, we currently use third-party providers to create the APIs to integrate such third party cognitive engines in order to make such services available through our platform. We plan to require such third party developers to create such APIs and will be dependent in part upon their ability to do so effectively and quickly. We may not have full control over the quality and performance of third-party providers, and therefore, any unexpected deficiencies or problems arising from these third-party providers may cause significant interruptions of our platform. The failure of third party developers to integrate their cognitive engines seamlessly into our platform and/or provide reliable,

Our competitors may acquire third party technologies, which could result in them blocking us from using the technology in our platform, offering it for free to the public or making it cost prohibitive for us to continue to incorporate their technologies in our platform.

Our success depends on our ability to attract and incorporate the leading cognitive engines into our platform. If any third party acquires a cognitive engine that is integral to our platform, they may preclude us from using it as a component of our platform or make it more expensive for us to utilize such cognitive engine. It is also possible that a third party acquirer of such technology could offer the cognitive engines and technologies to the public as a free add-on capability, in which case our customers would have less incentive to pay us for the use of our platform. If a key third party technology becomes unavailable to us or is impractical for us to continue to use, the functionality of our platform could be interrupted and our expenses could increase as we search for an alternative technology. As a result, our business, results of operations and financial condition could be adversely affected through the loss of customers, reputational harm and/or from increased operating costs.

# Our continuous access to public media may be restricted, disrupted or terminated, which would reduce the effectiveness of our platform.

The success of our platform for certain users depends substantially on our ability to continuously ingest and process large amounts of data available in the public media, and any interruption to our free access to such public media will adversely affect the performance and quality of our platform for such users. While we have not encountered any significant disruption of such access to

date, there is no guarantee that this trend will continue without costs. Public media sources may change their policies to restrict access or implement procedures to make it more difficult or costly for us to maintain access. Free access to certain public media has also been challenged in courts as a potential violation of laws. In a recent case, Fox News v. TVEyes, Fox News filed a lawsuit against a media-monitoring service company for alleged violation of copyright laws. The District Court held that TVEyes' indexing, viewing and archiving capabilities constituted fair use, but that its downloading, emailing and date-time search capabilities did not. Both parties have appealed the decision to the Court of Appeals for the Second Circuit. If the Court overturns the lower court decision as requested by Fox News, or other broadcasters pursue similar legal actions, our free access to some or all public media could be limited or eliminated entirely, which will severely reduce the effectiveness and capabilities of our platform and cause us to lose customers. If we no longer have free access to public media, our online media library and the capability and quality of our platform for some users will be significantly reduced. Furthermore, we may be forced to pay significant fees to public media sources in order to maintain access, which would adversely affect our financial condition and results of operations.

If we are not able to develop a strong brand for our platform and increase market awareness of our Company and our platform, then our business, results of operations and financial condition may be adversely affected.

We believe that the success of our platform will depend in part on our ability to develop a strong brand identity for our "Veritone", "Veritone Platform", "aiOS", "aiWARE" and other service marks and to increase the market awareness of our platform and its capabilities. The successful promotion of our brand will depend largely on our continued marketing efforts and our ability to offer high quality cognitive engines on our platform and ensure that the technology operates seamlessly for our customers. We also believe that it is important for us to be thought leaders in the AI-based cognitive computing market. Our brand promotion and thought leadership activities may not be successful or produce increased revenue. In addition, independent industry analysts often provide reviews of our platform and of competing products and services, which may significantly influence the perception of our platform in the marketplace. If these reviews are negative or not as positive as reviews of our competitors' products and services, then our brand may be harmed.

The promotion of our brand also requires us to make substantial expenditures, and we anticipate that these expenditures will increase as our industry becomes more competitive and as we seek to expand into new markets. These higher expenditures may not result in any increased revenue or incremental revenue that is sufficient to offset the higher expense levels. If we do not successfully maintain and enhance our brand, then our business may not grow, we may see our pricing power reduced relative to competitors and we may lose customers, all of which would adversely affect our business, results of operations and financial condition.

We expect that our brand and reputation may also be affected by customer reviews and reactions, including review and feedback through online social media channels. We must consistently provide high quality services to ensure that our customers have a positive experience using our platform. If customers complain about our services, if we do not handle customer complaints effectively or if we cannot generate positive reviews and commentaries on social media channels, then our brand and reputation may suffer, and our customers may lose confidence in us and reduce or cease their use of our platform.

# We may not be able to expand the capabilities of our proprietary Conductor technology to optimize our AI platform.

We recently enhanced the performance of our platform by adding our proprietary Conductor technology, which automates the selection of cognitive engines available on our platform within a class from the engines available on our platform. Our Conductor technology is designed to optimize data processing for performance, cost, and speed by choosing the best cognitive engine to deploy to generate the best results for each individual search. Our Conductor technology currently only works with transcription engines. While we are working on expanding our Conductor technology to other cognitive classes, we cannot guarantee that such expansion will be completed on a timely basis or at all. We may not be able to develop to effectively and quickly navigate and process multiple complex classes of cognitive engines, particularly those developed by third parties. Even if we are able to do so, we may not be able to develop Conductors that achieve the expected speed, performance and quality, which would have an adverse effect on our customer experience and satisfaction. In addition, we expect to incur significant costs in the development and deployment of our proprietary Conductors, and if we cannot achieve our expected performance goals and economic benefits, it will have an adverse effect on our financial condition and results of operations.

We currently generate significant revenue from a limited number of key customers and the loss of any of our key customers may harm our business, results of operations and financial results.

Our ten largest customers by revenue accounted for approximately 62.7% of our net revenues in the first half of 2017 and 72.2% and 81.0% in fiscal years 2016 and 2015, respectively. If any of our key customers decides not to renew its contract with us or renews on less favorable terms, or if any such customer decides to develop its own platform, our business, revenue and reputation could be materially and adversely affected.

For example, our two largest customers by revenue in 2015, LifeLock, Inc. and DraftKings, Inc., collectively accounted for approximately 42.7% of our net revenues in 2015, but only 10.5% of our net revenues in 2016. In September 2015, our agreement with LifeLock, Inc. was terminated, which contributed to the reduction in our net revenues in 2016 compared with the prior year. Furthermore, as a result of certain legal proceedings in which it was involved, DraftKings, Inc. reduced its marketing spend in 2016, and our net revenues related to our agreement with DraftKings, Inc. declined significantly. If we lose business with additional key customers, and are not able to gain additional customers or increase our revenue from other customers to offset the reduction of revenues from those key customers, our business, results of operations on a financial condition would be harmed.

Media Agency clients periodically review and change their advertising requirements and relationships. If we are unable to remain competitive or retain key clients, our business, results of operations and financial position may be adversely affected.

The media placement industry is highly competitive, and certain advertising clients periodically put their advertising, marketing and corporate communications business up for competitive review. Clients also review the cost/benefit of servicing all or a portion of their advertising and marketing needs in-house. We have won and lost accounts in the past as a result of these reviews. Because our Media Agency contracts generally can be cancelled by our customers upon 30- to 90- days prior written notice, clients can easily change media providers on short notice without any penalty. As a result, in order to retain existing clients and win new clients, we must continue to develop solutions that meet client needs, provide quality and effective client services, and achieve clients' return on investment requirements and pricing. In addition, our Media Agency Business is primarily engaged in the placement of endorsed media, and we may face increased competition in this business in the future from other advertising agencies that provide a more comprehensive range of advertising services to their customers. To be able to offer a broader range of services, we would need to add additional capabilities, such as television buying, and we may not be able to do so effectively. To the extent that we are not able to remain competitive or retain key clients, our revenue may be adversely affected, which could have a material adverse effect on our business, results of operations and financial position.

# Our Media Agency Business is dependent on growth in demand for endorsed media and the availability of sufficient media personalities to deliver such content.

Our Media Agency Business is primarily engaged in the placement of endorsed media, which depends on the availability of media personalities to deliver the endorsed media content. The endorsed media market is still at a relatively early stage of development, and its future growth is uncertain. Our ability to grow our sales in this business will be dependent in part upon the level of interest in endorsed media among advertisers, and upon the number of available media personalities and our ability to identify and engage an increasing number of such personalities on a cost effective basis. If demand for endorsed media fails to grow, or if we are unable to identify sufficient appropriate media personalities to deliver the endorsed media content, our ability to grow our Media Agency Business would be impacted materially.

Acquiring new clients and retaining existing Media Agency clients depends on our ability to avoid and manage conflicts of interest arising from other client relationships and attracting and retaining key personnel.

Our ability to acquire new Media Agency clients and to retain existing clients may, in some cases, be limited by clients' perceptions of, or policies concerning, conflicts of interest arising from other client relationships. If we are unable to manage these client relationships and avoid potential conflicts of interest, our business, results of operations and financial position may be adversely affected.

Our ability to acquire new Media Agency clients and to retain existing clients is dependent in large part upon our ability to attract and retain our key personnel in that business, who are an important aspect of our competitiveness. If we are unable to attract and retain key personnel, our ability to provide our services in the manner clients have come to expect may be adversely affected, which could harm our reputation and result in a loss of clients, which could have a material adverse effect on our business, results of operations and financial position.

# Our quarterly results may fluctuate significantly and period-to-period comparisons of our results may not be meaningful.

Our quarterly results, including the levels of our revenue, our operating expenses and other costs, and our operating margins, may fluctuate significantly in the future, and period-to-period comparisons of our results may not be meaningful. Accordingly, the results of any one quarter should not be relied upon as an indication of our future performance. In addition, our quarterly results may not fully reflect the underlying performance of our business.

Factors that may cause fluctuations in our quarterly results include, but are not limited to:

- the timing of new advertising program wins with our Media Agency customers;
- our ability to retain our existing customers and to expand our business with our existing customers;
- our ability to attract new customers, the type of customers we are able to attract, the size and needs of their businesses and the cost of acquiring these new customers:
- the timing and market acceptance of our SaaS solutions and other products introduced by us and our competitors;
- · variations in the timing of licensing revenues from our SaaS solutions as a result of trends impacting our target vertical markets;
- changes in our pricing policies or those of our competitors;
- the timing of our recognition of revenue and the mix of our revenues during the period;
- the amount and timing of operating expenses and other costs related to the maintenance and expansion of our business, infrastructure and operations;
- · the amount and timing of operating expenses and other costs associated with assessing or entering new vertical markets;
- the amount and timing of operating expenses and other costs related to the development or acquisition of businesses, services, technologies or intellectual property rights;
- the timing and impact of security breaches, service outages or other performance problems with our technology infrastructure and software solutions;
- the timing and costs associated with legal or regulatory actions;
- changes in the competitive dynamics of our industry, including consolidation among competitors, strategic partners or customers;
- loss of our executive officers or other key employees;
- · industry conditions and trends that are specific to the vertical markets in which we sell or intend to sell our SaaS solutions; and
- · general economic and market conditions.

Fluctuations in quarterly results may negatively impact the value of our common stock, regardless of whether they impact or reflect the overall performance of our business. If our quarterly results fall below the expectations of investors or any securities analysts who follow our stock, or below any guidance we may provide, the price of our common stock could decline substantially.

If we are not able to enhance or introduce new products that achieve market acceptance and keep pace with technological developments, our business, results of operations and financial condition could be harmed.

Our ability to attract new customers and increase revenue from existing customers depends in part on our ability to enhance and improve our platform, increase adoption and usage of our products and introduce new products and features, including products and services designed for a mobile user environment. The success of any enhancements or new products depends on several factors, including timely completion, adequate quality testing, actual performance quality, market-accepted pricing levels and overall market acceptance and demand. Enhancements and new products that we develop may not be introduced in a timely or cost-effective manner, may contain defects, may have interoperability difficulties with our platform or may not achieve the market acceptance necessary to generate significant revenue. If we are unable to successfully enhance our existing platform and capabilities to meet evolving customer requirements, increase adoption and usage of our platform, develop new products, or if our efforts to increase the usage of our products are more expensive than we expect, then our business, results of operations and financial condition could be harmed.

# The success of our business depends on our ability to expand into new vertical markets and attract new customers in a cost-effective manner.

In order to grow our business, we plan to drive greater awareness and adoption of our platform from enterprises across new vertical markets, including Legal, Government and Retail. We intend to increase our investment in sales and marketing, as well as in technological development, to meet evolving customer needs in these and other markets. There is no guarantee, however, that we will be successful in gaining new customers from any or all of these markets. We have limited experience in marketing and selling our products and services in these new markets, which may present unique and unexpected challenges and difficulties that are not present in our current business operations. For example, in order for us to offer products and services to certain government customers, we will be required to deploy our platform to operate in secure government cloud environments in order to enable our customers to maintain compliance with applicable regulations that govern the use, storage and transfer of certain government data. However, due to the secure nature of these government cloud environments, we may not be able to fully perform all functionalities and features of our platform or make available all of the third party cognitive engines within our non-government cloud platform ecosystem, which may limit or reduce the performance and quality of our services. Furthermore, we may incur additional costs to modify our current platform to conform to the cloud provider's requirements, and we may not be able to generate sufficient revenue to offset these costs. We will also be required to comply with certain regulations required by government customers, such as FedRAMP and CJIS, which will require us to incur costs, devote management time and modify our current platform and operations, and we may not be able to comply with those regulations effectively.

We use a variety of marketing channels to promote our products and platform, such as digital, print and social media advertising, email campaigns, industry events and public relations. If the costs of the marketing channels we use increase dramatically, then we may choose to use alternative and less expensive channels, which may not be as effective as the channels we currently use. As we add to or change the mix of our marketing strategies, we may need to expand into more expensive channels than those we are currently in, which could adversely affect our business, results of operations and financial condition. As part of our strategy to penetrate the new vertical markets, we will incur marketing expenses before we are able to recognize any revenue in such markets, and these expenses may not result in increased revenue or brand awareness. We have made in the past, and may make in the future, significant expenditures and investments in new marketing campaigns, and we cannot assure you that any such investments will lead to the cost-effective acquisition of additional customers. If we are unable to maintain effective marketing programs, then our ability to attract new customers or enter into new vertical markets could be adversely affected.

# Interruptions or performance problems associated with our technology and infrastructure may adversely affect our business and operating results.

Our continued growth depends in part on the ability of customers to access our platform at any time and within an acceptable amount of time. We have experienced, and may in the future experience, disruptions, outages and other performance problems due to a variety of factors, including infrastructure changes, introductions of new applications and functionality, software errors and defects, capacity constraints due to an increasing number of users accessing our platform simultaneously, or security related incidents. In addition, from time to time we may experience limited periods of server downtime due to server failure or other technical difficulties (as well as maintenance requirements). Because we also incorporate diverse software and hosted services from many third party vendors, we may encounter difficulties and delays in integrating and synthesizing these applications and programs, which may cause downtimes or other performance problems. It may become increasingly difficult to maintain and improve our performance, especially during peak usage times and as our platform becomes more complex and our user traffic increases. If our platform is unavailable or if our users are unable to access our platform within a reasonable amount of time or at all, our business would be adversely affected and our brand could be harmed. In the event of any of the factors described above, or certain other failures of our infrastructure, customer or consumer data may be permanently lost. To the extent that we do not effectively address capacity constraints, upgrade our systems as needed, and continually develop our technology and network architecture to accommodate actual and anticipated changes in technology, customers and consumers may cease to use our platform and our business and operating results may be adversely affected.

Our business depends on customers increasing their use of our services and/or platform, and we may experience loss of customers or decline in their use of our services and/or platform.

Our ability to grow and generate revenue depends, in part, on our ability to maintain and grow our relationships with existing customers and convince them to increase their usage of our services and/or platform. If our customers do not increase their use of our platform, then our revenue may not grow and our results of operations may be harmed. Our revenue model for advertising contracts is generally structured as a percentage of the total fees for the advertisement. If our customers reduce their spending on the placement of advertisements with media vendors, or if they decide to use other marketing or selling strategies, we will experience a decline in our revenue. In addition, many of our SaaS licensing contracts include a usage-based license fee that is based upon our customers' level of usage of our platform's cognitive engines. We cannot accurately predict customers' usage levels and the loss of customers or reductions in their usage levels of our platform may each have a negative impact on our business, results of operations and financial condition. If a significant number of customers cease using, or reduce their usage of, our platform, then we may be required to spend

significantly more on sales and marketing than we currently plan to spend in order to maintain or increase revenue from customers. Such additional sales and marketing expenditures could adversely affect our business, results of operations and financial condition. Most of our customers do not have long-term contractual financial commitments to us and, therefore, most of our customers may reduce or cease their use of our platform at any time without penalty or termination charges.

We rely upon Joyent, Amazon Web Services, Iron.io and Microsoft Azure to operate our platform, and any disruption of or interference with our use of such third party services would adversely affect our business operations.

Historically, we have used Joyent (a unit of Samsung) and Amazon Web Services (AWS) to host our platform and for our storage needs. We also utilize Iron.io (a unit of Oracle) for certain computing processes related to our services. Users of our platform need to be able to access our platform at any time, without interruption or degradation of performance. Joyent, Iron.io and AWS run their own platforms that we access, and we are, therefore, vulnerable to service interruptions at Joyent, Iron.io and AWS. We do not have control over the operations of Joyent, Iron.io or AWS, and we may experience interruptions, delays and outages in service and availability from time to time due to a variety of factors, including infrastructure changes, human or software errors, website hosting disruptions and capacity constraints. In addition, if our security, or that of Joyent, Iron.io or AWS, is compromised, our platform is unavailable to our customers, or our customers are unable to use our platform within a reasonable amount of time or at all, then our business, results of operations and financial condition could be adversely affected. In some instances, we may not be able to identify the cause or causes of these performance problems within a period of time acceptable to our customers.

Joyent, Iron.io and AWS provide us with hosting, computing and storage capacity pursuant to individual agreements with each of them that may be cancelled by providing 30 days' prior written notice, and in some cases, the agreements can be terminated immediately for cause without notice. Given the short term nature of our arrangements, we could experience interruptions on our platform and in our ability to make our platform available to customers, as well as delays and additional expenses in arranging alternative cloud infrastructure services.

While we have modified our platform to work on Microsoft's Azure secure government cloud, the secure nature of this secure government cloud limits certain features of our platform, which could impact a user's experience on our site and may make it harder to achieve broad market acceptance for our platform.

Any of the above circumstances or events may harm our reputation, cause customers to stop using our platform, impair our ability to increase revenue from existing customers, impair our ability to grow our customer base, subject us to financial penalties and liabilities under our service level agreements and otherwise harm our business, results of operations and financial condition.

The security of our platform, networks or computer systems may be breached, and any unauthorized access to our customer data will have an adverse effect on our business and reputation.

The use of our platform involves the storage, transmission and processing of our clients' private data as well as public media, and this private media may contain confidential and proprietary information of our clients or other personal or identifying information regarding our clients, their employees or other persons. Individuals or entities may attempt to penetrate our network or platform security, or that of our third party hosting and storage providers, and could gain access to our clients' private media, which could result in the destruction, disclosure or misappropriation of proprietary or confidential information of our clients' or their customers, employees and business partners. If any of our clients' private media is leaked, obtained by others or destroyed without authorization, it could harm our reputation, we could be exposed to civil and criminal liability, and we may lose our ability to access private media information, which will adversely affect the quality and performance of our platform.

In addition, our platform may be subject to computer malware, viruses and computer hacking, fraudulent use attempts and phishing attacks, all of which have become more prevalent in our industry. Though it is difficult to determine what, if any, harm may directly result from any specific interruption or attack, they may include the theft or destruction of data owned by us or our customers, and/or damage to our platform. Any failure to maintain the performance, reliability, security and availability of our products and technical infrastructure to the satisfaction of our customers may harm our reputation and our ability to retain existing customers and attract new users.

While we have implemented procedures and safeguards that are designed to prevent security breaches and cyber-attacks, there is no guarantee that they will be able to protect against all attempts to breach our systems, and we may not become aware in a timely manner of any such security breach. Unauthorized access to or security breaches of our platform, network or computer systems, or those of our technology service providers or third party cognitive engines, could result in the loss of business, reputational damage, regulatory investigations and orders, litigation, indemnity obligations, damages for contract breach, civil and criminal penalties for violation of applicable laws, regulations or contractual obligations, and significant costs, fees and other monetary payments for remediation. If customers believe that our platform does not provide adequate security for the storage of sensitive information or its transmission over the Internet, our business will be harmed. Customers' concerns about security or privacy may deter them from using our platform for activities that involve personal or other sensitive information.

# If we are not able to compete effectively, our business and operating results will be harmed.

While the market for audio and video search and analytics platforms is still in the early stages of development, we do face competition from various sources, including large, well-capitalized technology companies such as Google and IBM. These competitors may have better brand name recognition, greater financial and engineering resources and larger sales teams than we have. As a result, these competitors may be able to develop and introduce competing solutions that may have greater capabilities than ours or that are able to achieve greater customer acceptance, and they may be able to respond more quickly and effectively than we can to new or changing opportunities, technologies, standards or customer requirements. In addition, we may also compete with smaller third-party developers of cognitive engines, who may develop their own platforms that perform similar services as our platform. We expect that competition will increase and intensify as we continue to expand our serviceable markets and improve our platform and services. Increased competition is likely to result in pricing pressures, which could negatively impact our sales, profitability and market share.

# Privacy and data security laws and regulations could require us to make changes to our business, impose additional costs on us and reduce the demand for our software solutions.

Our business model contemplates that we will process both public media and our clients' private media. Our customers may store and/or transmit a significant amount of personal or identifying information through our platform. Privacy and data security have become significant issues in the United States and in other jurisdictions where we may offer our software solutions. The regulatory framework relating to privacy and data security issues worldwide is evolving rapidly and is likely to remain uncertain for the foreseeable future. Federal, state and foreign government bodies and agencies have in the past adopted, or may in the future adopt, laws and regulations regarding the collection, use, processing, storage and disclosure of personal or identifying information obtained from customers and other individuals. In addition to government regulation, privacy advocates and industry groups may propose various self-regulatory standards that may legally or contractually apply to our business. Because the interpretation and application of many privacy and data security laws, regulations and applicable industry standards are uncertain, it is possible that these laws, regulations and standards may be interpreted and applied in a manner inconsistent with our existing privacy and data management practices. As we expand into new jurisdictions or verticals, we will need to understand and comply with various new requirements applicable in those jurisdictions or verticals.

To the extent applicable to our business or the businesses of our customers, these laws, regulations and industry standards could have negative effects on our business, including by increasing our costs and operating expenses, and delaying or impeding our deployment of new core functionality and products. Compliance with these laws, regulations and industry standards requires significant management time and attention, and failure to comply could result in negative publicity, subject us to fines or penalties or result in demands that we modify or cease existing business practices. In addition, the costs of compliance with, and other burdens imposed by, such laws, regulations and industry standards may adversely affect our customers' ability or desire to collect, use, process and store personal information using our software solutions, which could reduce overall demand for them. Even the perception of privacy and data security concerns, whether or not valid, may inhibit market acceptance of our software solutions in certain verticals. Furthermore, privacy and data security concerns may cause our customers' clients, vendors, employees and other industry participants to resist providing the personal information necessary to allow our customers to use our applications effectively. Any of these outcomes could adversely affect our business and operating results.

# Failure to manage our growth effectively could increase our expenses, decrease our revenue and prevent us from implementing our business strategy.

Our ability to scale our business and achieve profitability requires substantial growth, which will put a strain on our business. To manage this and our anticipated future growth effectively, we must continue to maintain and enhance our platform and information technology infrastructure, as well as our financial and accounting systems and controls. We also must attract, train and retain a significant number of qualified software developers and engineers, technical and management personnel, sales and marketing personnel, customer support personnel and professional services personnel. Failure to effectively manage our rapid growth could lead us to over-invest or under-invest in development and operations, result in weaknesses in our platform, systems or controls, give rise to operational mistakes, losses, loss of productivity or business opportunities and result in loss of employees and reduced productivity of remaining employees. Our growth could require significant capital expenditures and might divert financial resources from other projects such as the development of new products and services. If our management is unable to effectively manage our growth, our expenses might increase more than expected, our revenue could decline or grow more slowly than expected, and we might be unable to implement our business strategy. The quality of our products and services might suffer, which could negatively affect our reputation and harm our ability to retain and attract customers.

We may pursue the acquisition of other companies, businesses or technologies, which could be expensive, divert our management's attention and/or fail to achieve the expected benefits.

As part of our expansion strategy, we may in the future seek to acquire businesses, services, technologies or intellectual property rights that we believe could complement, expand or enhance the features and functionality of our platform and our technical capabilities, broaden our service offerings or offer growth opportunities. The pursuit of potential acquisitions may divert the attention of management and cause us to incur various expenses in identifying, investigating and pursuing suitable acquisitions, whether or not such acquisitions are consummated. Acquisitions also could result in dilutive issuances of equity securities or the incurrence of debt, which could adversely affect our operating results and financial condition. In addition, we may experience difficulties in integrating the acquired personnel, operations and/or technologies successfully or effectively managing the combined business following the acquisition. We also may not achieve the anticipated benefits from the acquired business and may incur unanticipated costs and liabilities in connection with any such acquisitions.

#### Any failure to offer high-quality customer support may adversely affect our relationships with our customers.

Our ability to retain existing customers and attract new customers depends on our ability to maintain a consistently high level of customer service and technical support. Our customers depend on our service support team to assist them in utilizing our platform effectively and to help them to resolve issues quickly and to provide ongoing support. If we are unable to hire and train sufficient support resources or are otherwise unsuccessful in assisting our customers effectively, it could adversely affect our ability to retain existing customers and could prevent prospective customers from adopting our platform. We may be unable to respond quickly enough to accommodate short-term increases in demand for customer support. We also may be unable to modify the nature, scope and delivery of our customer support to compete with changes in the support services provided by our competitors. Increased demand for customer support, without corresponding revenue, could increase our costs and adversely affect our business, results of operations and financial condition. Our sales are highly dependent on our business reputation and on positive recommendations from customers. Any failure to maintain high-quality customer support, or a market perception that we do not maintain high-quality customer support, could adversely affect our reputation, business, results of operations and financial condition.

#### We plan to expand our international operations, which exposes us to significant risks.

We are planning to expand internationally to increase our revenue from customers outside of the United States as part of our growth strategy. We expect, in the future, to open foreign offices and hire employees to work at these offices in order to reach new customers and gain access to additional technical talent. Operating in international markets requires significant resources and management attention and will subject us to regulatory, economic and political risks in addition to those we already face in the United States. Because of our limited experience with international operations as well as developing and managing sales in international markets, our international expansion efforts may not be successful.

In addition, we will face risks in doing business internationally that could adversely affect our business, including, but not limited to:

- the difficulty of managing and staffing international operations and the increased operating, travel, infrastructure and legal compliance costs associated with numerous international locations;
- our ability to effectively price our products in competitive international markets;
- the need to adapt and localize our products for specific countries;
- the need to offer customer support in various languages;
- · difficulties in understanding and complying with U.S. laws, regulations and customs relating to U.S. companies operating in foreign jurisdictions;
- · difficulties in understanding and complying with local laws, regulations and customs in foreign jurisdictions, particularly in the areas of data privacy and personal privacy;
- difficulties with differing technical and environmental standards, data privacy and telecommunications regulations and certification requirements outside the United States, which could prevent customers from deploying our products or limit their usage;
- more limited protection for intellectual property rights in some countries; and
- political or social unrest or economic instability in a specific country or region in which we operate.

Our failure to manage any of these risks successfully could harm our international operations, and adversely affect our business, results of operations and financial condition

We may be sued by third parties for alleged infringement of their proprietary rights, which could adversely affect our business, results of operations and financial condition.

There has been considerable patent and other intellectual property development activity in the artificial intelligence industry, which has resulted in litigation based on allegations of infringement or other violations of intellectual property rights. Our future success depends, in part, on not infringing the intellectual property rights of others. In the future, we may receive claims from third parties, including our competitors, alleging that our platform and underlying technology infringe or violate such third party's intellectual property rights, and we may be found to be infringing upon such rights. We may be unaware of the intellectual property rights of others that may cover some or all of our technology. Any such claims or litigation could cause us to incur significant expenses and, if successfully asserted against us, could require that we pay substantial damages or ongoing royalty payments, prevent us from offering some portion of our platform, or require that we comply with other unfavorable terms. We may also be obligated to indemnify our customers or business partners in connection with any such litigation and to obtain licenses or modify our platform, which could further exhaust our resources. Patent infringement, trademark infringement, trade secret misappropriation and other intellectual property claims and proceedings brought against us, whether successful or not, could harm our brand, business, results of operations and financial condition. Litigation is inherently uncertain, and any judgment or injunctive relief entered against us or any adverse settlement could negatively affect our business, results of operations and financial condition. In addition, litigation can involve significant management time and attention and be expensive, regardless of the outcome. During the course of litigation, there may be announcements of the results of hearings and motions and other interim developments related to the litigation. If securities analysts or investors regard these announcements as negative, the

We could incur substantial costs in protecting or defending our intellectual property rights, and any failure to protect our intellectual property could adversely affect our business, results of operations and financial condition.

Our success depends, in part, on our ability to protect our brand and the proprietary methods and technologies that we develop under patent and other intellectual property laws of the United States and foreign jurisdictions so that we can prevent others from using our inventions and proprietary information. As of June 30, 2017, in the United States, we had nine issued patents, which expire between 2028 and 2031, and 30 patent applications pending for examination and one pending provisional application. As of such date, we also had 22 patent applications pending for examination in foreign jurisdictions, all of which are related to our U.S. patents and patent applications. There can be no assurance that we will be issued any additional patents or that any patents that have been issued or that may be issued in the future will provide significant protection for our intellectual property. In addition, as of June 30, 2017, we had five registered trademarks in the United States, and we have filed applications to register several additional marks. If we fail to protect our intellectual property rights adequately, our competitors might gain access to our technology and our business, results of operations and financial condition may be adversely affected.

We cannot assure you that the particular forms of intellectual property protection that we seek, or our business decisions about when to file patent applications and trademark applications, will be adequate to protect our business. We could be required to spend significant resources to monitor and protect our intellectual property rights. Litigation may be necessary in the future to enforce our intellectual property rights, determine the validity and scope of our proprietary rights or those of others, or defend against claims of infringement or invalidity. Such litigation could be costly, time-consuming and distracting to management, result in a diversion of significant resources, lead to the narrowing or invalidation of portions of our intellectual property and have an adverse effect on our business, results of operations and financial condition. Our efforts to enforce our intellectual property rights may be met with defenses, counterclaims and countersuits attacking the validity and enforceability of our intellectual property rights or alleging that we infringe the counterclaimant's own intellectual property. Any of our patents, copyrights, trademarks or other intellectual property rights could be challenged by others or invalidated through administrative process or litigation.

We also rely, in part, on confidentiality agreements with our business partners, employees, consultants, advisors, customers and others in our efforts to protect our proprietary technology, processes and methods. These agreements may not effectively prevent disclosure of our confidential information, and it may be possible for unauthorized parties to copy our software or other proprietary technology or information, or to develop similar software independently without our having an adequate remedy for unauthorized use or disclosure of our confidential information. In addition, others may independently discover our trade secrets and proprietary information, and in these cases we would not be able to assert any trade secret rights against those parties. Costly and time-consuming litigation could be necessary to enforce and determine the scope of our proprietary rights, and the failure to obtain or maintain trade secret protection could adversely affect our competitive business position.

In addition, the laws of some countries do not protect intellectual property and other proprietary rights to the same extent as the laws of the United States. To the extent we expand our international activities, our exposure to unauthorized copying, transfer and use of our proprietary technology or information may increase.

We cannot be certain that our means of protecting our intellectual property and proprietary rights will be adequate or that our competitors will not independently develop similar technology. If we fail to meaningfully protect our intellectual property and proprietary rights, our business, results of operations and financial condition could be adversely affected.

We depend on our executive officers and other key employees, and the loss of one or more of these employees or an inability to attract and retain highly skilled employees could adversely affect our business.

Our success depends largely upon the continued services of our Chief Executive Officer, Chad Steelberg, our President, Ryan Steelberg, and our other executive officers. We rely on our leadership team in the areas of strategy and implementation, research and development, operations, security, marketing, sales, support and general and administrative functions. We do not currently have any employment agreements with our executive officers that require them to continue to work for us for any specified period, and, therefore, they could terminate their employment with us at any time. The loss of Chad Steelberg or Ryan Steelberg, or one or more of the members of our management team, could adversely impact our business and operations and disrupt our relationships with our key customers.

# If we are unable to hire, retain and motivate qualified personnel, our business will suffer.

Our future success depends, in part, on our ability to continue to attract and retain highly skilled personnel. We believe that there is, and will continue to be, intense competition for highly skilled management, technical, sales and other personnel with experience in our industry. We must provide competitive compensation packages and a high-quality work environment to hire, retain and motivate employees. If we are unable to retain and motivate our existing employees and attract qualified personnel to fill key positions, we may be unable to manage our business effectively, including the development, marketing and sale of our products, which could adversely affect our business, results of operations and financial condition. To the extent we hire personnel from competitors, we also may be subject to allegations that they have been improperly solicited or that they have divulged proprietary or other confidential information.

Volatility in, or lack of performance of, our stock price may also affect our ability to attract and retain key personnel. Many of our key personnel are, or will soon be, vested in a substantial number of shares of common stock or stock options. Employees may be more likely to terminate their employment with us if the shares they own or the shares underlying their vested options have significantly appreciated in value relative to the original purchase prices of the shares or the exercise prices of the options, or, conversely, if the exercise prices of the options that they hold are significantly above the trading price of our common stock. If we are unable to retain our employees, our business, results of operations and financial condition could be adversely affected.

# We expect to require additional capital to support our business, and this capital might not be available on acceptable terms, if at all.

We intend to continue to make investments to support our business and may require additional funds. In particular, we expect to seek additional funds to develop new products and enhance our platform, expand our operations, including our sales and marketing organizations and our presence outside of the United States, improve our infrastructure or acquire complementary businesses, technologies, services, products and other assets. Accordingly, we expect to engage in equity or debt financings to secure additional funds. If we raise additional funds through future issuances of equity or convertible debt securities, our stockholders could suffer significant dilution, and any new equity securities we issue could have rights, preferences and privileges superior to those of holders of our common stock. Any debt financing that we may secure in the future could involve restrictive covenants relating to our capital raising activities and other financial and operational matters, which may make it more difficult for us to obtain additional capital and to pursue business opportunities. We may not be able to obtain additional financing on terms favorable to us, if at all. If we are unable to obtain adequate financing or financing on terms satisfactory to us when we require it, our ability to continue to support our business growth, scale our infrastructure, develop product enhancements and to respond to business challenges could be significantly impaired, and our business, results of operations and financial condition may be adversely affected.

#### Changes in laws and regulations related to the Internet or changes in the Internet infrastructure itself may diminish the demand for our products.

The future success of our business depends upon the continued use of the Internet as a primary medium for commerce, communications and business applications. Federal, state or foreign government bodies or agencies have in the past adopted, and may in the future adopt, laws or regulations affecting the use of the Internet as a commercial medium. Changes in these laws or regulations could require us to modify our products and platform in order to comply with these changes. In addition, government agencies or private organizations have imposed and may impose additional taxes, fees or other charges for accessing the Internet or commerce conducted via the Internet. These laws or charges could limit the growth of Internet-related commerce or communications generally, or result in reductions in the demand for Internet-based products and services such as our platform. In addition, the use of the Internet as a business tool could be adversely affected due to delays in the development or adoption of new standards and protocols to handle

increased demands of Internet activity, security, reliability, cost, ease-of-use, accessibility and quality of service. The performance of the Internet and its acceptance as a business tool has been adversely affected by "viruses", "worms" and similar malicious programs. If the use of the Internet is reduced as a result of these or other issues, then demand for our products could decline, which could adversely affect our business, results of operations and financial condition.

# We have a limited operating history, which makes it difficult to evaluate our current business and future prospects and increases the risk of your investment.

We were founded in 2014 and launched our platform in April 2015. As a result of our limited operating history, our ability to forecast our future results of operations is limited and subject to a number of uncertainties, including our ability to plan for future growth. We have encountered and will encounter risks and uncertainties frequently experienced by growing companies in rapidly changing industries, such as:

- market acceptance of our platform and new products;
- · reliability and scalability of our platform and services;
- adding new customers and new vertical markets;
- · retention of customers;
- the successful expansion of our business;
- · competition;
- our ability to control costs, particularly our operating expenses;
- network outages or security breaches and any associated expenses;
- executing acquisitions and integrating acquired businesses, technologies, services, products and other assets; and
- general economic and political conditions. If we do not address these risks successfully, our business, results of operations and financial condition may be adversely affected.

#### We have had a history of losses and we may be unable to achieve or sustain profitability.

We experienced net losses of \$27.4 million in the first half of 2017 and \$27.0 million and \$6.2 million in fiscal years 2016 and 2015, respectively. As of June 30, 2017 and December 31, 2016, we had an accumulated deficit of approximately \$77.1 million and \$45.2 million, respectively. We cannot predict if we will achieve profitability in the near future or at all. We expect to continue to expend substantial financial and other resources on, among other things:

- investments to expand and enhance our platform and technology infrastructure, make improvements to the scalability, availability and security of our platform, and develop new products;
- sales and marketing, including expanding our direct sales organization and marketing programs, and expanding our programs directed at increasing our brand awareness among current and new customers;
- · hiring additional employees;
- · expansion of our operations and infrastructure, both domestically and internationally; and
- general administration, including legal, accounting and other expenses related to being a public company.

These investments may not result in increased revenue or growth of our business. We cannot assure you that we will be able to generate net revenues sufficient to offset our expected cost increases and planned investments in our business and platform. As a result, we may incur significant losses for the foreseeable future, and may not be able to achieve and sustain profitability. If we fail to achieve and sustain profitability, then we may not be able to achieve our business plan, fund our business or continue as a going concern.

Our business is subject to the risks of earthquakes, fire, floods and other natural catastrophic events, and to interruption byman-made problems such as power disruptions, computer viruses, data security breaches or terrorism.

Our corporate headquarters are located in Southern California, a region known for seismic activity. A significant natural disaster, such as an earthquake, fire or a flood, occurring at our headquarters, at one of our other facilities or where a business partner is located could adversely affect our business, results of operations and financial condition. Further, if a natural disaster or man-made problem were to affect Joyent, Iron.io and/or AWS, our network service providers or Internet service providers, this could adversely

affect the ability of our customers to use our products and platform. In addition, natural disasters and acts of terrorism could cause disruptions in our or our customers' businesses, national economies or the world economy as a whole. We also rely on our network and third-party infrastructure and enterprise applications and internal technology systems for our engineering, sales and marketing and operations activities. Although we maintain incident management and disaster response plans, in the event of a major disruption caused by a natural disaster or man-made problem, we may be unable to continue our operations and may endure system interruptions, reputational harm, delays in our development activities, lengthy interruptions in service, breaches of data security and loss of critical data, any of which could adversely affect our business, results of operations and financial condition.

# Our directors, officers and principal stockholders have significant voting power and may take actions that may not be in the best interests of our other stockholders.

Our officers, directors and principal stockholders each holding more than 5% of our common stock, collectively control approximately 64% of our voting securities. If any of our officers, directors and principal stockholders purchases additional shares of common stock, the aggregate percentage of equity ownership of management may increase further. As a result, these stockholders, if they act together, will be able to control the management and affairs of our Company and most matters requiring stockholder approval, including the election of directors and approval of significant corporate transactions. This concentration of ownership may have the effect of delaying or preventing a change of control and might adversely affect the market price of our common stock. This concentration of ownership may not be in the best interests of our other stockholders. Due to such concentration of ownership, we may take actions with respect to our business that may conflict with the desire of other stockholders. Furthermore, pursuant to a voting agreement ("Voting Agreement"), Acacia and entities affiliated with our executive officers and directors will have the right to designate all nine directors on our Board. As a result, Acacia and our executive officers and directors will be able to exercise significant control over our business operations and on all matters requiring stockholder approval, including the election of directors, approval of significant corporate transactions and the definition of rights and privileges of all securities. Due to such controlling position, we may take actions with respect to our business that may conflict with the desire of other stockholders, including common stockholders.

# We expect that the price of our common stock will fluctuate substantially.

The market price of our common stock is likely to be highly volatile and may fluctuate substantially due to many factors, including, but not limited to:

- the volume and timing of our revenues and quarterly variations in our results of operations or those of others in our industry;
- announcement of new contracts with customers or termination of contracts with customers;
- the introduction of new services, content or features by us or others in our industry;
- disputes or other developments with respect to our or others' intellectual property rights;
- media exposure of our products or of those of others in our industry;
- · changes in governmental regulations;
- · changes in earnings estimates or recommendations by securities analysts; and
- · general market conditions and other factors, including factors unrelated to our operating performance or the operating performance of our competitors.

In recent years, the stock markets generally have experienced extreme price and volume fluctuations that have often been unrelated or disproportionate to the operating performance of the listed companies. Broad market and industry factors may significantly affect the market price of our common stock, regardless of our actual operating performance. These fluctuations may be even more pronounced in the trading market for our common stock shortly following our IPO.

In addition, in the past, class action litigation has often been instituted against companies whose securities have experienced periods of volatility in market price. Securities litigation brought against us following volatility in our stock price, regardless of the merit or ultimate results of such litigation, could result in substantial costs, which would hurt our financial condition and operating results and divert management's attention and resources from our business.

If securities or industry analysts do not publish or cease publishing research or reports about us, our business, our market or our competitors, or if such analysts adversely change their recommendations regarding our common stock, the market price and trading volume of our common stock could decline.

The trading market for our common stock will be influenced by the research and reports that securities or industry analysts may publish about us, our business, our market or our competitors. If few analysts commence coverage of us, the market price for our common stock could be negatively affected. If any of the analysts who may cover us adversely change their recommendations regarding our common stock or provide more favorable recommendations about our competitors, the market price of our common stock may decline. If any of the analysts who may cover us were to cease coverage of us or fail to publish reports on us regularly, visibility of our Company in the financial markets could decrease, which in turn could cause the market price or trading volume of our common stock to decline. These concerns may be exacerbated by the relatively small size of our common shares float, which is likely to result in limited trading volume for our common stock.

We have incurred and will continue to incur increased costs as a result of becoming a public company, including costs related to compliance with the Sarbanes-Oxley Act and other regulations.

As a public company, we have incurred and will continue to incur significant legal, accounting, insurance and other expenses that we had not incurred as a private company, including costs associated with public company reporting requirements. We also have incurred and will incur costs associated with compliance with the Sarbanes-Oxley Act and related rules implemented by the SEC. The expenses incurred by public companies generally for reporting and corporate governance purposes have been increasing. We expect these rules and regulations to increase our legal and financial compliance costs and to make some activities more time-consuming and costly, although we are currently unable to estimate these costs with any degree of certainty. In estimating these costs, we took into account expenses related to insurance, legal, accounting and compliance activities, as well as other expenses not currently incurred. These laws and regulations could also make it more difficult or costly for us to obtain certain types of insurance, including director and officer liability insurance, and we may be forced to accept reduced policy limits and coverage or incur substantially higher costs to obtain the same or similar coverage. These laws and regulations could also make it more difficult for us to attract and retain qualified persons to serve on our board of directors, our board committees or as our executive officers. Furthermore, if we are unable to satisfy our obligations as a public company, we could be subject to delisting of our common stock, fines, sanctions and other regulatory action and potentially civil litigation.

The Sarbanes-Oxley Act requires, among other things, that we maintain effective internal control over financial reporting and disclosure controls and procedures. In particular, we must perform system and process evaluation and testing of our internal control over financial reporting to allow management to report on the effectiveness of our internal control over financial reporting, as required by Section 404 of the Sarbanes-Oxley Act. In addition, we will be required to have our independent registered public accounting firm attest to the effectiveness of our internal control over financial reporting in the later of our second annual report on Form 10-K following the date on which we are no longer an emerging growth company unless we qualify as a "non-accelerated filer" as defined in Rule 12b-2 under the Securities Exchange Act of 1934 as amended, in which case we would be exempt from the auditor attestation requirements of Section 404 of The Sarbanes-Oxley Act. Our compliance with Section 404 of the Sarbanes-Oxley Act will require that we incur substantial accounting expense and expend significant management efforts. We currently do not have an internal audit group, and we will need to hire additional accounting and financial staff with appropriate public company experience and technical accounting knowledge. If we are not able to comply with the requirements of Section 404 in a timely manner, or if we or our independent registered public accounting firm identify deficiencies in our internal control over financial reporting that are deemed to be material weaknesses, the market price of our stock could decline and we could be subject to sanctions or investigations by NASDAQ, the SEC or other regulatory authorities, which would require additional financial and management resources.

Our ability to successfully implement our business plan and comply with Section 404 requires us to be able to prepare timely and accurate financial statements. We expect that we will need to continue to improve existing, and implement new operational and financial systems, procedures and controls to manage our business effectively. Any delay in the implementation of, or disruption in the transition to, new or enhanced systems, procedures or controls, may cause our operations to suffer and we may be unable to conclude that our internal control over financial reporting is effective and/or to obtain an unqualified report on internal control our financial reporting from our auditors as may be required under Section 404 of the Sarbanes-Oxley Act. This, in turn, could have an adverse impact on trading prices for our common stock, and could adversely affect our ability to access the capital markets.

We had identified a material weakness in our internal control over financial reporting for the years ended December 31, 2016 and 2015, and we may not be able to successfully maintain effective internal controls over financial reporting.

We identified control deficiencies in our financial reporting process that constituted a material weakness for the years ended December 31, 2015 and 2016. The material weakness related to the lack of competent accounting personnel with the appropriate level of knowledge, experience and training in GAAP and SEC reporting requirements with respect to equity transactions, resulting in several adjustments to the interim financial statements and also a restatement of our previously issued financial statements as of and for the years ended December 31, 2015 and 2016.

We have implemented certain measures to remediate this material weakness. For example, we hired a new Chief Financial Officer in October 2016, a new Corporate Controller in March 2017 and a new Senior Director of Financial Reporting in June 2017. We have also engaged outside consultants with requisite experience to assist us in the financial reporting process and utilized interim professionals to strengthen our accounting and financial reporting team. We believe that the actions we have taken remediated the material weakness.

However, there can be no assurance that we will not suffer from other material weaknesses in the future. If we fail to maintain effective internal control over financial reporting in the future, such failure could result in a material misstatement of our annual or quarterly financial statements that would not be prevented or detected on a timely basis and which could cause investors and other users to lose confidence in our financial statements, limit our ability to raise capital and have a negative effect on the trading price of our common stock. Additionally, failure to maintain effective internal control over financial reporting may also negatively impact our operating results and financial condition, impair our ability to timely file our periodic and other reports with the SEC, subject us to additional litigation and regulatory actions and cause us to incur substantial additional costs in future periods relating to the implementation of remedial measures.

# Substantial future sales of our common stock, or the perception in the public markets that these sales may occur, may depress our stock price.

Sales of a substantial amount of our common stock in the market, or the perception that these sales may occur, could adversely affect the market price of our common stock. Our outstanding shares include the shares of common stock we sold in the IPO, which may be resold immediately, and shares of common stock that will become available for sale 180 days (subject to extension in certain circumstances) under the terms of lock-up agreements entered into between the holders of those shares and the underwriters of the IPO. However, the underwriters can waive this restriction and allow these stockholders to sell their shares at any time after the offering. As these lockup restrictions end, the market price of the common stock could drop significantly if the holders of these restricted shares sell them or are perceived by the market as intending to sell them.

We have registered all common stock issuable under our 2014 Stock Option/Stock Issuance Plan and all common stock that we may issue under our 2017 Stock Incentive Plan and our 2017 Employee Stock Purchase Plan. As a result, these shares can be freely sold in the public market once vested and exercised, subject to the lock-up agreements referred to above. If a large number of these shares are sold in the public market, the sales could reduce our trading price.

In the future, we may also issue our securities if we need to raise additional capital or in connection with acquisitions. The number of shares of our common stock issued in connection with a financing or acquisition could constitute a material portion of our then-outstanding shares of our common stock.

We are an "emerging growth company," and we cannot be certain if the reduced reporting requirements applicable to emerging growth companies will make our common stock less attractive to investors.

We are an "emerging growth company," as defined in the JOBS Act. For as long as we continue to be an emerging growth company, we may take advantage of exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies, including not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. We could be an emerging growth company for up to five years, although circumstances could cause us to lose that status earlier, including if the market value of our common stock held by non-affiliates exceeds \$700 million as of any June 30 date before that time, in which case, we would no longer be an emerging growth company as of the following December 31. We cannot predict if investors will find our common stock less attractive because we may rely on these exemptions. If some investors find our common stock less attractive as a result, there may be a less active trading market for our common stock and our stock price may be more volatile.

We are a "controlled company" within the meaning of the NASDAQ rules and, as a result, are exempt from certain corporate governance requirements. Therefore, you may not have the same protections afforded to stockholders of companies that are subject to such requirements.

Acacia and entities affiliated with our Chief Executive Officer and our President (the Major Stockholders) will collectively beneficially own more than 50% of our voting power. Pursuant to the Voting Agreement, Acacia and the Major Stockholders (acting as a group) can designate and elect all nine directors on our Board of Directors. As a result, we will be considered a controlled company within the meaning of the corporate governance standards of NASDAQ. Under these rules, a controlled company may elect not to comply with certain of NASDAQ's corporate governance requirements, including:

- · A majority of our Board of Directors to consist of "independent directors" as defined by the applicable rules and regulations of NASDAQ;
- The compensation of our executive officers to be determined, or recommended to the Board of Directors for determination, by independent directors constituting a
  majority of the independent directors of the Board in a vote in which only independent directors participate or by a Compensation Committee comprised solely of
  independent directors; and
- That director nominees to be selected, or recommended to the Board of Directors for selection, by independent directors constituting a majority of the independent directors of the Board in a vote in which only independent directors participate or by a nomination committee comprised solely of independent directors.

Notwithstanding the availability of these exemptions, our current intent is to comply voluntarily with certain of NASDAQ's corporate governance requirements with respect to our Board. We cannot guarantee that a majority of our Board will consist of independent directors or that we will be able to comply with all of NASDAQ's other corporate governance requirements. As a controlled company, you will not have the same protections afforded to stockholders of companies that are subject to all of NASDAQ's corporate governance requirements.

#### We do not currently expect to pay any cash dividends.

The continued operation and expansion of our business will require substantial funding. Accordingly, we do not currently expect to pay any cash dividends on shares of our common stock. Any determination to pay dividends in the future will be at the discretion of our board of directors and will depend upon results of operations, financial condition, contractual restrictions, restrictions imposed by applicable law and other factors our board of directors deems relevant. Additionally, we expect these restrictions to continue in the future. Accordingly, if you purchase shares of our common stock, realization of a gain on your investment will depend on the appreciation of the price of our common stock, which may never occur. Investors seeking cash dividends in the foreseeable future should not purchase our common stock.

Our anti-takeover provisions could prevent or delay a change in control of our Company, even if such change in control would be beneficial to our stockholders.

Provisions of our amended and restated certificate of incorporation and amended and restated bylaws as well as provisions of Delaware law could discourage, delay or prevent a merger, acquisition or other change in control of our Company, even if such change in control would be beneficial to our stockholders. These include:

- authorizing the issuance of "blank check" preferred stock that could be issued by our board of directors to increase the number of outstanding shares and thwart a takeover attempt;
- a provision for a classified board of directors so that not all members of our board of directors are elected at one time;
- the removal of directors only for cause;
- no provision for the use of cumulative voting for the election of directors;
- limiting the ability of stockholders to call special meetings;
- · requiring all stockholder actions to be taken at a meeting of our stockholders (i.e. no provision for stockholder action by written consent); and
- establishing advance notice requirements for nominations for election to the board of directors or for proposing matters that can be acted upon by stockholders at stockholder meetings.

In addition, the Delaware General Corporation Law prohibits us, except under specified circumstances, from engaging in any mergers, significant sales of stock or assets or business combinations with any stockholder or group of stockholders who owns at least 15% of our common stock.

Our amended and restated certificate of incorporation designates the Court of Chancery of the State of Delaware as the sole and exclusive forum for certain types of actions and proceedings that may be initiated by our stockholders, which could limit our stockholders' ability to obtain a favorable judicial forum for disputes with us or our directors, officers or other employees.

Our amended and restated certificate of incorporation provides that, unless we consent in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for:

- any derivative action or proceeding brought on our behalf;
- · any action asserting a claim of breach of a fiduciary duty owed by any of our directors, officers or other employees to us or to our stockholders;
- any action asserting a claim against us arising pursuant to any provision of the Delaware General Corporation Law, our certificate of incorporation or our bylaws; or
- any action asserting a claim against us governed by the internal affairs doctrine.

Any person or entity purchasing or otherwise acquiring any interest in shares of our capital stock shall be deemed to have notice of and consented to this provision of our certificate of incorporation. This choice-of-forum provision may limit a stockholder's ability to bring a claim in a judicial forum that it finds favorable or convenient for disputes with us or our directors, officers or other employees, which may discourage such lawsuits against us and our directors, officers and other employees. Alternatively, if a court were to find these provisions of our certificate of incorporation inapplicable to, or unenforceable in respect of, one or more of the specified types of actions or proceedings, we may incur additional costs associated with resolving such matters in other jurisdictions, which could adversely affect our business, financial condition or results of operations.

#### Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

#### Recent Sales of Unregistered Equity Securities

Shares and Warrants Issued in Connection with the Primary Warrant and Bridge Loan

In connection with the execution of the Bridge Loan, we issued to the Lenders (a) a total of 300,000 shares of common stock, and (b) fully vestedten-year warrants to purchase an aggregate of 240,000 shares of our common stock, which automatically adjusted upon completion of the IPO to be exercisable to purchase an aggregate of 313,446 shares of our common stock (which was equal to 1.5% of the fully diluted shares of common stock outstanding immediately following the closing of the IPO) at an exercise price of \$13.6088 per share.

Concurrent with the closing of our IPO, we issued 2,150,335 shares of our common stock to Acacia pursuant to the automatic exercise of the Primary Warrant in exchange for cash proceeds of \$29.3 million. In addition, upon exercise of the Primary Warrant, we issued to Acacia a five-year warrant to purchase 809,400 shares of our common stock at an exercise price of \$13.6088 per share.

No underwriters were involved in the foregoing issuances of securities. The securities described above were issued to sophisticated investors in reliance upon the exemption from the registration requirements of the Securities Act, as set forth in Section 4(a)(2) under the Securities Act, relative to transactions by an issuer not involving any public offering, to the extent an exemption from such registration was required.

Shares and Awards Issued under Employee Stock Plans

In the first six months of 2017, we issued a total of 52,433 shares of restricted common stock to employees and consultants, 10,121 shares of common stock to our CEO as compensation for the second quarter of 2017, and 1,580 shares upon the exercise of stock options.

In May 2017, we granted a total of 3,801,158 stock options to officers and employees, and a total of 35,576 restricted stock units to directors.

The issuances of stock options, restricted shares and restricted stock units, and the shares of common stock issuable upon the exercise of the options described above, were issued pursuant to written compensatory plans or arrangements with our employees, directors and consultants, in reliance on the exemption provided by Rule 701 promulgated under the Securities Act, or pursuant to Section 4(a)(2) under the Securities Act, relative to transactions by an issuer not involving any public offering, to the extent an exemption from such registration was required.

# Use of Proceeds

On May 17, 2017, we completed our IPO, pursuant to which we issued and sold an aggregate of 2,500,000 shares of our common stock at a price to the public of \$15.00 per share pursuant to a registration statement on Form S-1 (File No. 333-216726), which was declared effective by the SEC on May 11, 2017. We received expected net proceeds of approximately \$32.5 million, after deducting discounts and commissions of approximately \$3.2 million and other offering expenses of approximately \$1.8 million.

No payments for such expenses were made directly or indirectly to any of our officers and directors, or persons owning 10% or more of any class of our equity securities, or to any of our affiliates. Wunderlich Securities, Inc. acted as the sole book-running manager of the offering. Craig-Hallum Capital Group, LLC and Northland Securities, Inc. acted as co-managers of the offering.

There has been no material change in the use of proceeds from our initial public offering from that described in our Prospectus. As described in our Prospectus, we intend to use the proceeds for working capital or other general corporate purposes, including funding our growth strategies.

# Item 3. Defaults Upon Senior Securities

None

# Item 4. Mine Safety Disclosures

Not applicable

# Item 5. Other Information

None

101.PRE

XBRL Taxonomy Extension Presentation Linkbase Document

Item 6.	Exhibits
3.1	Third Amended and Restated Certificate of Incorporation of Veritone, Inc. (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form8-K filed with the Securities and Exchange Commission on May 23, 2017).
3.2	Amended and Restated Bylaws of Veritone, Inc. (incorporated by reference to Exhibit 3.2 to the Company's Current Report on Form8-K filed with the Securities and Exchange Commission on May 23, 2017).
10.1	Lease Agreement dated July 14, 2017, between the Company and PRII/MCC South Coast Property Owner, LLC.
31.1	Certification of Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS	XBRL Instance Document
101.SCH	XBRL Taxonomy Extension Schema Document
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document
101.LAB	XBRL Taxonomy Extension Label Linkbase Document

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Veritone, Inc.

August 8, 2017

By /s/ Chad Steelberg

Chad Steelberg

Chief Executive Officer and Chairman of the Board

(Principal Executive Officer)

August 8, 2017

By /s/ Peter F. Collins

Peter F. Collins

Executive Vice President and Chief Financial Officer

(Principal Financial and Accounting Officer)

# LEASE AGREEMENT

By and Between

# PR II/MCC SOUTH COAST PROPERTY OWNER, LLC, a Delaware limited liability company

("Landlord")

 $\quad \text{and} \quad$ 

VERITONE, INC., a Delaware corporation

("Tenant")

July 14, 2017 575 Anton Boulevard Costa Mesa, California

# TABLE OF CONTENTS

	Page
ARTICLE 1 PREMISES; COMMON AREAS	4
ARTICLE 2 TERM AND CONDITION OF PREMISES	5
ARTICLE 3 USE, NUISANCE, OR HAZARD	6
ARTICLE 4 RENT	7
ARTICLE 5 RENT ADJUSTMENT	8
ARTICLE 6 SERVICES TO BE PROVIDED BY LANDLORD	19
ARTICLE 7 REPAIRS AND MAINTENANCE BY LANDLORD	24
ARTICLE 8 REPAIRS AND CARE OF PREMISES BY TENANT	25
ARTICLE 9 TENANT'S EQUIPMENT AND INSTALLATIONS	27
ARTICLE 10 FORCE MAJEURE	28
ARTICLE 11 CONSTRUCTION, MECHANICS' AND MATERIALMAN'S LIENS	28
ARTICLE 12 ARBITRATION	29
ARTICLE 13 INSURANCE	29
ARTICLE 14 QUIET ENJOYMENT	31
ARTICLE 15 ALTERATIONS	32
ARTICLE 16 FURNITURE, FIXTURES, AND PERSONAL PROPERTY	34
ARTICLE 17 PERSONAL PROPERTY AND OTHER TAXES	35
ARTICLE 18 ASSIGNMENT AND SUBLETTING	36
ARTICLE 19 DAMAGE OR DESTRUCTION	41
ARTICLE 20 CONDEMNATION	44
ARTICLE 21 HOLD HARMLESS	45
ARTICLE 22 DEFAULT BY TENANT	46
ARTICLE 23 LIEN FOR RENT	50
ARTICLE 24 INTENTIONALLY OMITTED	50
ARTICLE 25 ATTORNEYS' FEES	50
ARTICLE 26 NON-WAIVER	51
ARTICLE 27 RULES AND REGULATIONS	51
ARTICLE 28 ASSIGNMENT BY LANDLORD; RIGHT TO LEASE	52
ARTICLE 29 LIABILITY OF LANDLORD	52
ARTICLE 30 SUBORDINATION AND ATTORNMENT	52

ARTICLE 31 HOLDING OVER	55
ARTICLE 32 SIGNS	56
ARTICLE 33 HAZARDOUS SUBSTANCES	57
ARTICLE 34 COMPLIANCE WITH LAWS AND OTHER REGULATIONS	60
ARTICLE 35 SEVERABILITY	62
ARTICLE 36 NOTICES	62
ARTICLE 37 OBLIGATIONS OF, SUCCESSORS, PLURALITY, GENDER	62
ARTICLE 38 ENTIRE AGREEMENT	62
ARTICLE 39 CAPTIONS	63
ARTICLE 40 CHANGES	63
ARTICLE 41 AUTHORITY	63
ARTICLE 42 BROKERAGE	64
ARTICLE 43 EXHIBITS	64
ARTICLE 44 APPURTENANCES	64
ARTICLE 45 PREJUDGMENT REMEDY, REDEMPTION, AND COUNTERCLAIM	65
ARTICLE 46 RECORDING	65
ARTICLE 47 MORTGAGEE PROTECTION	65
ARTICLE 48 OTHER LANDLORD CONSTRUCTION	65
ARTICLE 49 PARKING	67
ARTICLE 50 ELECTRICAL CAPACITY	69
ARTICLE 51 OPTIONS TO EXTEND LEASE	69
ARTICLE 52 TELECOMMUNICATIONS LINES AND EQUIPMENT	74
ARTICLE 53 ERISA	74
ARTICLE 54 TENANT'S RIGHT OF FIRST OFFER	75
ARTICLE 55 TEMPORARY SPACE	78
ARTICLE 56 24 HOUR FITNESS CLUB	80
ARTICLE 57 TENANT'S ROOFTOP AND EXTERIOR ANTENNA RIGHTS	80
ARTICLE 58 PATIO AND BALCONY AREAS	83
ARTICLE 59 TENANT SECURITY SYSTEMS	84
ARTICLE 60 LANDLORD'S REPRESENTATIONS AND WARRANTIES	86
ARTICLE 61 HELISTOP USE	86
ARTICLE 62 LETTER OF CREDIT	87
ARTICLE 63 ADDITIONAL IMPROVEMENTS	91

# LEASE AGREEMENT

THIS LEASE AGREEMENT, (this "Lease") is made and entered into as of July 14, 2017, by and between PR II/MCC SOUTH COAST PROPERTY OWNER, LLC, a Delaware limited liability company ("Landlord"), and Tenant identified in the Basic Lease Information below.

# BASIC LEASE INFORMATION

Tenant: VERITONE, INC., a Delaware corporation.

Premises: Suite 100 on the first (1st) floor ("Suite 100") and Suite 200 on the second (2nd) floor ("Suite 200") of the Building, containing approximately 37,875 square feet of rentable area, outlined in Exhibit B to this Lease. (The interior design and layout of the Premises shown on Exhibit B is preliminary.) The Premises includes the Patio Area, as defined in Article 58, however, the Patio Area is excluded from the rentable square footage of the Premises.

Building: The Building commonly known as 575 Anton Boulevard, Costa Mesa, California. The rentable area of the Building is 253,737 square feet.

#### Base Rent:

Period		
(In Months)	Moi	thly Base Rent
1 - 10	_	Abated*
11 – 12	\$	80,000.00**
13 - 18	\$	82,500.00**
19 - 24	\$	124,987.50
25 - 36	\$	128,737.13
37 - 48	\$	132,599.24
49 - 60	\$	136,577.22
61 - 72	\$	140,674.53
73 - 82	\$	144,894.77

As an inducement to Tenant entering into this Lease, so long as no Event of Default shall have occurred under this Lease, Base Rent in the amount of \$121,200.00 per month shall be abated for the first ten (10) months commencing as of the Commencement Date, or if the Commencement Date is other than the first day of a calendar month, commencing as of the first day of the first full calendar month of the Term. The amount of Base Rent set forth in the foregoing table for that period reflects that rent abatement. During such abatement period, Tenant shall still be responsible for the payment of all of its other monetary obligations under the Lease.

As an inducement to Tenant entering into this Lease, during the eleventh (11th) through eighteenth (18th) months after the Commencement Date, for purposes of calculating Base Rent only, the Premises shall be deemed to contain only 25,000 square feet of rentable area. The amount of Base Rent set forth in the foregoing table for that period reflects that deemed square footage. The amount equal to the difference between Base Rent payable based on those deemed square footages and the Base Rent that would be payable if the actual square footage of the Premises were used is referred to herein as the "Rental Abatement." In the event of an Event of Default by Tenant under the terms of the Lease that results in early termination pursuant to the provisions of <u>Article 22</u> of the Lease, then as part of the recovery set forth in <u>Article 22</u> of the Lease, Landlord shall be entitled to the recovery of the Rental Abatement.

Letter of Credit Required Amount: \$606,000.00.

Rent Payable Upon Execution: \$80,000.00.

Tenant's Building Percentage 14.93%.

Base Year: Calendar year 2018.

Commencement Date: The earlier of (i) the date Tenant commences business operations in the Premises, or (ii) the date of Substantial Completion (as defined in the Tenant Work Letter attached hereto as <a href="Exhibit C">Exhibit C</a>) of the Premises, which Commencement Date is anticipated to be January 1, 2018 (the "Anticipated Commencement Date"). The Commencement Date shall be confirmed by the parties by the execution of the Commencement Date Memorandum attached hereto as <a href="Exhibit F">Exhibit F</a> as provided in Section <a href="2.1">2.1</a>.

Expiration Date: The date that is the day prior to the date that iseighty-two (82) months after the Commencement Date. If the Expiration Date falls on a day other than the last day of the calendar month, then, the Expiration Date shall be extended to the last day of the calendar month in which the day that the Term of this Lease would otherwise end but for this proviso occurs, and the Term of this Lease shall be extended accordingly.

#### Landlord's Address:

c/o The Prudential Insurance Company of America 4 Embarcadero Center, 27th Floor San Francisco, California 94111 Attn: PRISA II Asset Manager

With a copy by the same method to:

c/o The Prudential Insurance Company of America 7 Giralda Farms Madison, New Jersey 07940 Attention: James Marinello With a copy by the same method to:

McCarthy Cook & Co. 575 Anton Boulevard, Suite 350 Costa Mesa, California 92626 Attention: Brian M. Harnetiaux

Address for rental payment:

PRII/MCC South Coast Property PO Box 51731 Los Angeles, CA 90051-0060

# Tenant's Address:

575 Anton Boulevard, Suite 900 Costa Mesa, CA 92626 Attention: General Counsel

(If on or after the Commencement Date to the

Premises)

Attention: General Counsel

# With a copy to:

Much Shelist 191 North Wacker Drive, Suite 1800 Chicago, IL 60606 Attention: Glenn Taxman

Landlord's Broker: McCarthy Cook & Co.

Tenant's Broker: Jones Lang LaSalle and Newmark Grubb Knight Frank

Minimum Parking Allocation: One hundred thirteen (113) parking passes, which is based on three (3) parking passes per 1,000 rentable square feet of area in the Premises.

Maximum Parking Allocation: Two hundred sixty-five (265) parking passes, which is based on seven (7) parking passes per 1,000 rentable square feet of area in the Premises.

The Basic Lease Information is incorporated into and made part of this Lease. Each reference in this Lease to any Basic Lease Information shall mean the applicable information set forth in the Basic Lease Information, except that in the event of any conflict between an item in the Basic Lease Information and this Lease, this Lease shall control. Additional defined terms used in the Basic Lease Information shall have the meanings given those terms in this Lease.

# ARTICLE 1 PREMISES; COMMON AREAS

- 1.1 Subject to all of the terms and conditions hereinafter set forth, Landlord hereby leases to Tenant and Tenant hereby leases from Landlord the Premises. The property shown on Exhibit A to this Lease and all improvements thereon and appurtenances on that land thereto, including, but not limited to, the Building, access roadways, and all other related areas, shall be collectively hereinafter referred to as the "Project." The parties hereto hereby acknowledge that the purpose of Exhibit A and Exhibit B is to show the approximate location of the Premises in the Building and the general layout of the Project and such Exhibits are not meant to constitute an agreement, representation or warranty as to the construction of the Premises, the Building or the Project, the precise area of the Premises, the Building or the Project, the specific location of the Building, "Common Areas," as that term is defined in Section 1.3, below, or the elements thereof or of the access ways to the Premises, or the Project, or the identity or existence of any other tenant or occupant of the Project.
- 1.2 For purposes of this Lease, (1) "rentable area" and "usable area" shall be calculated pursuant to the Office Buildings: Standard Methods of Measurement (ANSI/BOMA Z65.1, 2010) (the "BOMA Standards"); (2) "rentable square feet" and "rentable footage" shall have the same meaning as the term "rentable area;" and (3) "usable square feet" and "usable square footage" shall have the same meaning as the term "usable area." Notwithstanding anything to the contrary in this Lease, the recital of the rentable area herein above set forth is for descriptive purposes only. Tenant shall have no right to terminate this Lease or receive any adjustment or rebate of any Base Rent or Additional Rent (as hereinafter defined) payable hereunder if said recital is incorrect. Tenant has inspected the Premises and is fully familiar with the scope and size thereof and agrees to pay the full Base Rent and Additional Rent set forth herein in consideration for the use and occupancy of said space, regardless of the actual number of square feet contained therein.
- 1.3 Tenant shall have the non-exclusive right to use in common with other tenants in the Project, and subject to the rules and regulations referred to in Article 27 of this Lease, those portions of the Project which are provided, from time to time, for use in common by Landlord, Tenant and any other tenants of the Project (such areas, together with such other portions of the Project designated by Landlord, in its discretion, including certain areas designated for the exclusive use of certain tenants, or to be shared by Landlord and certain tenants, are collectively referred to herein as the "Common Areas"). The manner in which the Common Areas are maintained and operated shall be at the reasonable discretion of Landlord and the use thereof shall be subject to such rules, regulations and restrictions as Landlord may make from time to time. Landlord reserves the right to close temporarily, make alterations or additions to, or change the location of elements of the Project and the Common Areas; provided that in the event of any such Common Area closure, change, alteration or modification, except in emergency situations as reasonably determined by Landlord, Landlord shall exercise commercially reasonable efforts to perform the same in a manner that is reasonably designed to minimize interference with Tenant's access to and permitted use of the Premises consistent with Comparable Buildings (as defined below). Subject to "Applicable Laws," as that term is defined in Section 5.1(a) of this Lease, except when and where Tenant's right of access is specifically excluded in this Lease, and except in the event of an emergency, Tenant shall have the right of

access to the Premises, the Building, and the parking facilities servicing the Building twenty-four (24) hours per day, seven (7) days per week during the "Term," as that term is defined in Section 2.1, below.

# ARTICLE 2 TERM AND CONDITION OF PREMISES

- 2.1 The term of this Lease (the "Term") shall commence on the Commencement Date and end on the Expiration Date, unless sooner terminated (the "Termination Date") as hereinafter provided. The Commencement Date of this Lease and the obligation of Tenant to pay Base Rent, Additional Rent and all other charges hereunder shall not be delayed or postponed by reason of any delay by Tenant in performing changes or alteration in the Premises not required to be performed by Landlord. In the event the Term shall commence on a day other than the first day of a month, then the Base Rent shall be immediately paid for such partial month prorated in accordance with Section 4.4 below. If Landlord does not deliver possession of the Premises to Tenant on or before the Anticipated Commencement Date, Landlord shall not be subject to any liability nor shall the validity of this Lease nor the obligations of Tenant hereunder be affected. In the event that the Commencement Date is a date which is other than the Anticipated Commencement Date, within a reasonable period of time after the date Tenant takes possession of the Premises Landlord shall deliver to Tenant a Commencement Date Memorandum in the form attached hereto as Exhibit F, setting forth the Commencement Date and the Expiration Date, and Tenant shall execute and return such Commencement Date Memorandum to Landlord within five (5) business days after Tenant's receipt thereof. In the event that the Commencement Date does not occur by the Outside Commencement Date (as defined below), then Tenant shall not be obligated to pay the Temporary Space Base Rent (as defined below) for the balance of the Temporary Space Term (as defined below). The term "Outside Commencement Date" initially means March 1, 2018, but shall be extended by one day for every one day in delay in Substantial Completion of the Tenant Improvements caused by (i) Tenant Delays (as defined in Exhibit C), (ii) inability to obtain or delays in obtaining necessary permits despite Landlord's commercially reasonable efforts to diligently obtain such necessary permits, and/or (iii) any other one or more Force Majeure Events (as defined in Article 10). In the event that the Commencement Date does not occur by the Out Date (as defined below), then Tenant shall be entitled by notice in writing to Landlord within ten (10) days thereafter to cancel this Lease, in which event the parties shall be discharged from all obligations hereunder; provided further, however, that if such written notice of Tenant is not received by Landlord within such ten (10)-day period, Tenant's right to cancel this Lease hereunder shall terminate and be of no further force or effect. The term "Out Date" initially means August 31, 2018, but shall be extended by one day for every one day in delay in substantial completion of Landlord's Work caused by (i) Tenant Delays and/or (ii) any other one or more Force Majeure Events.
- 2.2 Except as expressly set forth in this Lease and in the Tenant Work Letter attached hereto as Exhibit C (the "Tenant Work Letter"), Landlord shall not be obligated to provide or pay for any improvement, remodeling or refurbishment work or services related to the improvement, remodeling or refurbishment of the Premises, and Tenant shall accept the Premises in its "As Is" condition on the Commencement Date.

- 2.3 Except as herein expressly set forth, neither Landlord nor Landlord's agents have made any representations or promises with respect to the condition of the Building, the Premises, the land upon which the Building is constructed, or any other matter or thing affecting or related to the Building or the Premises, and no rights, easements or licenses are acquired by Tenant by implication or otherwise except as expressly set forth in this Lease.
- 2.4 Notwithstanding Section 2.3 above, Landlord warrants that the roof, structural components of the Building, HVAC system, electrical and plumbing systems, elevator, parking lot or site lighting (the "Covered Items"), other than those constructed by Tenant, shall be in good operating condition on the date possession of the Premises is delivered to Tenant. If a non-compliance with such warranty exists as of the delivery of possession, or if one of such Covered Items should malfunction or fail within one hundred twenty (120) days after the delivery of possession to Tenant, Landlord's sole obligation with respect to such matter, promptly after receipt of written notice from Tenant setting forth in reasonable detail the nature and extent of such non-compliance, malfunction or failure, rectify the same at Landlord's expense. If Tenant does not give Landlord the required notice within one hundred twenty (120) days after the delivery of possession to Tenant, Landlord shall have no obligation with respect to that warranty other than obligations regarding the Covered Items set forth elsewhere in this Lease. Nothing in this Section 2.4 shall limit Landlord's obligation with respect to correction of latent defects in the Premises as provided in the Tenant Work Letter.

# ARTICLE 3 USE, NUISANCE, OR HAZARD

- 3.1 The Premises shall be used and occupied by Tenant solely for general office purposes and any other lawful purposes (other than retail uses) consistent with a first-class office building and for no other purposes without the prior written consent of Landlord.
- 3.2 Tenant shall not use, occupy, or permit the use or occupancy of the Premises for any purpose which Landlord, in its reasonable discretion, deems to be illegal, immoral, or dangerous; permit any public or private nuisance; do or permit any act or thing which may unreasonably disturb the quiet enjoyment of any other tenant of the Project; keep any substance or carry on or permit any operation which might introduce offensive odors or conditions into other portions of the Project, use any apparatus which might make undue noise or set up vibrations in or about the Project; permit anything to be done which would increase the premiums paid by Landlord for special causes of loss form property insurance on the Project or its contents or cause a cancellation of any insurance policy covering the Project or any part thereof or any of its contents; or permit anything to be done which is prohibited by or which shall in any way conflict with any law, statute, ordinance, or governmental rule, regulation or covenants, conditions and restrictions affecting the Project, including without limitation the CC&R's (as defined below) now or hereinafter in force. Should Tenant do any of the foregoing without the prior written consent of Landlord, and the same is not cured within fifteen (15) days after written notice from Landlord (which 15 day period shall be subject to extension if the nature of the breach is such that it is not possible to cure the same within such 15 day period so long as the Tenant commences the cure of such breach within such 15 day period and diligently prosecutes the same to completion) it shall constitute an Event of Default (as hereinafter defined) and shall enable Landlord to resort to any of its remedies hereunder.

3.3 The ownership, operation, maintenance and use of the Project shall be subject to certain conditions and restrictions contained in an instrument ("CC&R's") recorded or to be recorded against title to the Project. Tenant agrees that regardless of when those CC&R's are so recorded, this Lease and all provisions hereof shall be subject and subordinate thereto and Tenant shall comply therewith; provided, however, that except as required by Applicable Laws (as defined below), Tenant's obligation to comply with CC&R's recorded after the date of this Lease shall be subject to Tenant's prior consent, which will not be withheld unless the same would materially adversely affect Tenant's rights under this Lease and/or materially increase Tenant's obligations under this Lease. The CC&R's affecting the Project as of the date of this Lease are listed on Exhibit H attached hereto. True and correct copies of the documents listed on Exhibit H have been delivered to Tenant. Accordingly, as a consequence of that subordination, during any period in which the entire Project is not owned by Landlord, (a) the portion of Operating Expenses and Taxes (each as defined below) for the Common Areas shall be allocated among the owners of the Project as provided in the CC&R's, and (b) the CC&R's shall govern the maintenance and insuring of the portions of the Project not owned by Landlord. Tenant shall, promptly upon request of Landlord, sign all documents reasonably required to carry out the foregoing into effect.

#### ARTICLE 4 RENT

- 4.1 Tenant hereby agrees to pay Landlord the Base Rent. For purposes of Rent adjustment under the Lease, the number of months is measured from the first day of the calendar month in which the Commencement Date falls. Each monthly installment (the "Monthly Rent") shall be payable by check or by money order or by wire transfer or by Federal Reserve Automated Clearing House (ACH) deposit to an account as directed by Landlord by written notice to Tenant on or before the first day of each calendar month. Landlord agrees to accept payment by Federal Reserve Automated Clearing House (ACH) deposit only so long as such system is available for Landlord's use. In addition to the Base Rent, Tenant also agrees to pay Tenant's Share of increases in Operating Expenses and Tenant's Tax Share of increases in Taxes over the Base Year of 2018 (each as hereinafter defined), and any and all other sums of money as shall become due and payable by Tenant as set forth in this Lease, all of which shall constitute additional rent under this Lease (the "Additional Rent"). Landlord expressly reserves the right to apply any payment received to Base Rent or any other items of Rent that are not paid by Tenant. The Base Rent, the Monthly Rent and the Additional Rent are sometimes hereinafter collectively called "Rent" and shall be paid when due in lawful money of the United States without demand, deduction, abatement, or offset to the addresses for the rental payment set forth in the Basic Lease Information, or as Landlord may designate from time to time.
- 4.2 In the event any Monthly or Additional Rent or other amount payable by Tenant hereunder is not paid within five (5) days after written notice from Landlord that such payment was not received when due, Tenant shall pay to Landlord a late charge (the "Late Charge"), as Additional Rent, in an amount of five percent (5%) of the amount of such late payment. Failure to pay any Late Charge shall be deemed a Monetary Default (as hereinafter defined). Provision for the Late Charge shall be in addition to all other rights and remedies available to Landlord hereunder, at law or in equity, and shall not be construed as liquidated damages or limiting Landlord's remedies in any manner. Failure to charge or collect such Late

Charge in connection with any one (1) or more such late payments shall not constitute a waiver of Landlord's right to charge and collect such Late Charges in connection with any other similar or like late payments.

- 4.3 Simultaneously with the execution hereof, Tenant shall deliver to Landlord the Rent Payable Upon Execution as payment of the first installment of Monthly Rent due hereunder.
- 4.4 If the Term commences on a date other than the first day of a calendar month or expires or terminates on a date other than the last day of a calendar month, the Rent for any such partial month shall be prorated to the actual number of days in such partial month.
- 4.5 All Rents and any other amount payable by Tenant to Landlord hereunder, if not paid within thirty (30) days after its due date, shall bear interest from the date due until paid at a rate equal to the prime commercial rate established from time to time by Bank of America, plus four percent (4%) per annum; but not in excess of the maximum legal rate permitted by law. Failure to charge or collect such interest in connection with any one (1) or more delinquent payments shall not constitute a waiver of Landlord's right to charge and collect such interest in connection with any other or similar or like delinquent payments.
- 4.6 If Tenant fails to make when due four (4) consecutive payments of Monthly Rent in any calendar year or makes three (3) consecutive payments of Monthly Rent in any calendar year which are returned to Landlord by Tenant's financial institution for insufficient funds, Landlord may require, by giving written notice to Tenant, that all future payments of Rent shall be made in cashier's check or by money order. The foregoing is in addition to any other remedy of Landlord hereunder, at law or in equity.
- 4.7 No Rent or other payment in respect of the Premises shall be based in any way upon net income or profits from the Premises. Tenant may not enter into or permit any sublease or license or other agreement in connection with the Premises which provides for a rental or other payment based on net income or profit.

#### ARTICLE 5 RENT ADJUSTMENT

#### 5.1 Definitions.

- (a) "Operating Expenses", as said term is used herein, shall mean all expenses, costs, and disbursements of every kind and nature which Landlord shall pay or become obligated to pay because of or in connection with the ownership, operation, management, security, repair, restoration, replacement, or maintenance of the Project, or any portion thereof. Operating Expenses shall be computed in accordance with generally accepted real estate practices, consistently applied, and shall include, but not be limited to, the items as listed below:
  - (i) Wages, salaries, other compensation and any and all taxes, insurance and benefits of, the Building manager and of all other persons engaged in the operation, maintenance and security of the Project; provided that if such

other persons are employed by Landlord they are at or below the grade of senior Building manager or chief Building engineer;

- (ii) Payments under any equipment rental agreements or management agreements, including without limitation the cost of any actual or charged management fee and all expenses for the Project management office including rent, office supplies, and materials therefor;
- (iii) Costs of all supplies, equipment, materials, and tools and amortization (including interest at the rate of eight percent (8%) per annum on the unamortized cost) of the cost of acquiring or the rental expense of personal property used in the maintenance, operation and repair of the Project, or any portion thereof:
- (iv) All costs incurred in connection with the operation, maintenance, and repair of the Project including without limitation, the following: (A) the cost of operation, repair, maintenance and replacement of all systems and equipment and components thereof of the Project; (B) the cost of janitorial, alarm, security and other services, replacement of wall and floor coverings, ceiling tiles and fixtures in common areas, maintenance and replacement of curbs and walkways, repair to roofs and re-roofing; (C) the cost of licenses, certificates, permits and inspections and the cost of contesting any governmental enactments which are reasonably anticipated by Landlord to increase Operating Expenses, and the cost incurred in connection with a transportation system management program or similar program; (D) the cost of landscaping, decorative lighting, and relamping, the cost of maintaining fountains, sculptures, bridges; and (E) costs, fees, charges or assessments imposed by, or resulting from any mandate imposed on Landlord by, any federal, state or local government for fire and police protection, trash removal, community services, or other services which do not constitute "Taxes" as that term is defined below. With respect to any element of Operating Expenses that includes replacement, if any such replacements are of a materially higher quality than the specifications for the replaced item [determined as if such replaced item were new and in good condition], and if the cost of installing such upgrades materially exceeds the cost of replacing the item with another like quality, then the excess cost of such upgrade replacement item shall not be included in Operating Expenses.
- (v) The cost of supplying all utilities, the cost of operating, maintaining, repairing, replacing, renovating and managing the utility systems, mechanical systems, sanitary, storm drainage systems, communication systems and escalator and elevator systems, and the cost of supplies, tools, and equipment and maintenance and service contracts in connection therewith.
- (vi) Costs and expenses of complying with, or participating in, conservation, recycling, sustainability, energy efficiency, waste reduction or other programs or practices implemented or enacted from time to time at the Building, including, without limitation, in connection with any LEED (Leadership in

Energy and Environmental Design) rating or compliance system or program, including that currently coordinated through the U.S. Green Building council or Energy Star rating and/or compliance system or program (collectively, "Conservation Costs"); provided that if the Conservation Cost is a capital improvement, the cost thereof shall be amortized as provided in clause (viii), below;

(vii) The cost of all insurance carried by Landlord in connection with the Project as reasonably determined by Landlord, including without limitation commercial general liability insurance, physical damage insurance covering damage or other loss caused by fire, earthquake, flood or other water damage, explosion, vandalism and malicious mischief, theft or other casualty, rental interruption insurance and such insurance as may be required by any lessor under any present or future ground or underlying lease of the Building or Project or any holder of a mortgage, deed of trust or other encumbrance now or hereafter in force against the Building or Project or any portion thereof, and any deductibles payable thereunder; including, without limitation, Landlord's cost of any self insurance deductible or retention;

(viii) Capital improvements made to or capital assets acquired for the Project, or any portion thereof, after the Commencement Date that (1) are intended to reduce Operating Expenses or (2) are necessary for the health, safety and/or security of the Project, its occupants and visitors and are deemed advisable in the reasonable judgment of Landlord or (3) are Conservation Costs or (4) are required under any and all applicable laws, statutes, codes, ordinances, orders, rules, regulations, conditions of approval and requirements of all federal, state, county, municipal and governmental authorities and all administrative or judicial orders or decrees and all permits, licenses, approvals and other entitlements issued by governmental entities, and rules of common law, relating to or affecting the Project, the Premises or the Building or the use or operation thereof, whether now existing or hereafter enacted, including, without limitation, the Americans with Disabilities Act of 1990, 42 USC 12111 et seq. (the "ADA") as the same may be amended from time to time, all Environmental Requirements (as hereinafter defined), and any CC&R's, or any corporation, committee or association formed in connection therewith, or any supplement thereto recorded in any official or public records with respect to the Project or any portion thereof (collectively, "Applicable Laws") (except for capital repairs, replacements or other improvements to remedy a condition existing prior to the Commencement Date that was not in compliance with then-current Applicable Laws in their form existing as of the Commencement Date and as interpreted pursuant to the then-current interpretation of such Applicable Laws by the applicable governmental authority as of the Commencement Date), which capital costs, or an allocable portion thereof, shall be amortized over the useful life of the improvements as reasonably determined by Landlord consistent with prudent property management practices for Comparable Buildings, together with interest on the unamortized ba

- (ix) fees, charges and other costs, including management fees (or amounts in lieu thereof), consulting fees, legal fees and accounting fees, of all contractors, engineers, consultants and other persons engaged by Landlord or otherwise incurred by or charged by Landlord in connection with the management, operation, maintenance and repair of the Buildings and the Project; and
- (x) payments, fees or charges under the CC&R's and any easement, license, operating agreement, declaration, restricted covenant, or instrument pertaining to the sharing of costs by the Project, or any portion thereof.

Expressly excluded from Operating Expenses are the following items:

- (xi) Repairs and restoration paid for by the proceeds of any insurance policies or amounts otherwise reimbursed to Landlord or paid by any other source (other than by tenants paying their share of Operating Expenses);
  - (xii) Principal, interest, and other costs directly related to financing the Project or ground lease rental or depreciation;
- (xiii) The cost of special services to tenants (including Tenant) for which a special charge is made and the cost of any services which are not offered to Tenant or for which Tenant is charged directly or are exclusively provided to another tenant or occupant of the Building;
- (xiv) The costs of repair of casualty damage or for restoration following condemnation to the extent covered by insurance proceeds or condemnation awards;
- (xv) The costs of any capital expenditures except as expressly permitted to be included in Operating Expenses as provided under clauses (vii), and (viii) above;
- (xvi) Advertising and leasing commissions; costs, including permit, license and inspection costs and supervision fees, incurred with respect to the installation of tenant improvements within the Project or incurred in renovating or otherwise improving, decorating, painting or redecorating vacant space within the Project or promotional or other costs in order to market space to potential tenants;
- (xvii) The legal fees and related expenses and legal costs incurred by Landlord (together with any damages awarded against Landlord) due to the bad faith violation by Landlord or any tenant of the terms and conditions of any lease of space in the Project;
- (xviii) Costs incurred: (x) to comply with Applicable Laws with respect to any Hazardous Materials (as defined below) which were in existence in, on, under or about the Project (or any portion thereof) prior to the

Commencement Date, and were of such a nature that a federal, state or municipal governmental or quasi-governmental authority, if it had then had knowledge of the presence of such Hazardous Materials, in the state, and under the conditions that they then existed in, on, under or about the Project, would have then required the removal, remediation or other action with respect thereto; and/or (y) with respect to Hazardous Materials which are disposed of or otherwise introduced into, on, under or about the Project after the date hereof by Landlord or Landlord's agents or employees and are of such a nature, at time of disposition or introduction, that a federal, state or municipal governmental or quasi-governmental authority, if it had then had knowledge of the presence of such Hazardous Materials, in the state, and under the conditions, that they then existed in, on, under or about the Project, would have then required the removal, remediation or other action with respect thereto; provided, however, Operating Expenses shall include costs incurred in connection with the clean-up, remediation, monitoring, management and administration of (and defense of claims related to) the presence of (1) Hazardous Materials used by Landlord (provided such use is not negligent and is in compliance with Applicable Laws) in connection with the operation, repair and maintenance of the Project to perform Landlord's obligations under this Lease (such as, without limitation, fuel oil for generators, cleaning solvents, and lubricants) and which are customarily found or used in Comparable Buildings and (2) Hazardous Materials created, released or placed in the Premises, Building or the Project by Tenant (or Tenant's affiliates or their tenants, contractors, employees or agents) prior to or after the Commencement Date;

- (xix) The attorneys' fees in connection with the negotiation and preparation of letters, deal memos, letters of intent, leases, subleases and/or assignments, space planning costs, and other costs and expenses incurred in connection with lease, sublease and/or assignment negotiations and transactions with present or prospective tenants or other occupants of the Project;
  - (xx) The expenses in connection with services or other benefits which are not available to Tenant;
- (xxi) The overhead and profit paid to Landlord or to subsidiaries or affiliates of Landlord for goods and/or services in the Project to the extent the same exceeds the costs of such goods and/or services rendered by qualified, unaffiliated third parties on a competitive basis;
  - (xxii) The costs arising from Landlord's charitable or political contributions;
  - (xxiii) The costs (other than ordinary maintenance and insurance) for sculpture, paintings and other objects of art;
  - (xxiv) The interest and penalties resulting from Landlord's failure to pay any items of Operating Expense when due;

(xxv) The Landlord's general corporate overhead and general and administrative expenses, costs of entertainment, dining, automobiles or travel for Landlord's employees, and costs associated with the operation of the business of the partnership or entity which constitutes Landlord as the same are distinguished from the costs of the operation of the Project, including partnership accounting and legal matters, costs of defending any lawsuits with any mortgagee, costs of selling, syndicating, financing, mortgaging or hypothecating any of Landlord's interest in the Project, costs of any disputes between Landlord and its employees (if any) not engaged in the operation of the Project, disputes of Landlord with management, or outside fees paid in connection with disputes with other Project tenants or occupants (except to the extent such dispute is based on Landlord's good faith efforts to benefit Tenant or meet Landlord's obligations under this Lease);

- (xxvi) The costs arising from the gross negligence or willful misconduct of Landlord;
- (xxvii) The management office rental to the extent such rental exceeds the fair market rental for such space;
- (xxviii) The costs of correction of latent defects in the Project to the extent covered by warranties;
- (xxix) Costs and expenses incurred in connection with leasing space in the Building, such as leasing commissions, tenant allowances, space planning fees and advertising and promotional expenses and legal fees for the preparation of leases;
  - (xxx) Depreciation or amortization of any improvements except as specifically set forth in this Lease;
- (xxxi) Any late fees, fines, penalties and interest on past due amounts incurred by Landlord due to Landlord's violation of any Applicable Law, rule or regulation;
  - (xxxii) Costs incurred for relocating tenants within the Building;
- (xxxiii) Costs or expenses (including fines, penalties and legal fees) incurred due to the violation by Landlord, its employees, agents and/or contractors or any other tenant in the Building, of any terms and conditions of this Lease or of the leases of other tenants in the Building;
- (xxxiv) Contributions to Operating Expense reserves; provided, however, the maintenance of such reserves shall not prohibit Landlord from passing through to Tenant (as an Operating Expense) items includable in Operating Expenses pursuant to the Lease once such items have been purchased from such reserve or once the expenses covered by such reserve have been incurred;

(xxxv) Costs or fees relating to the title insurance or defense of Landlord's title to or interest in the Project, or any part thereof;

(xxxvi) management fees in excess of an amount equal to three percent (3%) of all gross receipts for the Project (as fully grossed up for a one hundred percent (100%) occupancy level); and

(xxxvii) Costs (including capital repairs, replacements or other improvements) to remedy a condition existing prior to the Commencement Date that was not in compliance with the then current Applicable Laws in their form existing as of the Commencement Date and as interpreted pursuant to the then-current interpretation of such Applicable Laws by the applicable government authority as of the Commencement Date;

(b) "Taxes" shall mean all real property taxes, ad valorem taxes, personal property taxes, and all other taxes, assessments, embellishments, use and occupancy taxes, transit taxes, water, sewer and pure water charges not included in Section 5.1.(a)(v) above, excises, levies, license fees or taxes, and all other similar charges, levies, penalties, or taxes, if any, which are levied, assessed, or imposed, by any Federal, State, county, or municipal authority, whether by taxing districts or authorities presently in existence or by others subsequently created, upon, or due and payable in connection with, or a lien upon, all or any portion of the Project, or facilities used in connection therewith, and rentals or receipts therefrom and all taxes of whatsoever nature that are imposed in substitution for or in lieu of any of the taxes, assessments, or other charges included in its definition of Taxes, and any costs and expenses of contesting the validity of same. Taxes shall include, without limitation: (i) Any tax on the rent, right to rent or other income from the Project, or any portion thereof, or as against the business of leasing the Project, or any portion thereof; (ii) Any assessment, tax, fee, levy or charge in addition to, or in substitution, partially or totally, of any assessment, tax, fee, levy or charge previously included within the definition of real property tax, it being acknowledged by Tenant and Landlord that Proposition 13 was adopted by the voters of the State of California in the June 1978 election ("Proposition 13") and that assessments, taxes, fees, levies and charges may be imposed by governmental agencies for such services as fire protection, street, sidewalk and road maintenance, refuse removal and for other governmental services formerly provided without charge to property owners or occupants, and, in further recognition of the decrease in the level and quality of governmental services and amenities as a result of Proposition 13, Taxes shall also include any governmental or private assessments or the Project's contribution towards a governmental or private cost-sharing agreement for the purpose of augmenting or improving the quality of services and amenities normally provided by governmental agencies; (iii) Any assessment, tax, fee, levy, or charge allocable to or measured by the area of the Premises, the tenant improvements in the Premises, or the Rent payable hereunder, including, without limitation, any business or gross income tax or excise tax with respect to the receipt of such rent, or upon or with respect to the possession, leasing, operating, management, maintenance, alteration, repair, use or occupancy by Tenant of the Premises, or any portion thereof; (iv) Any assessment, tax, fee, levy or charge, upon this transaction or any document to which Tenant is a party, creating or transferring an

interest or an estate in the Premises; (v) All of the real estate taxes and assessments imposed upon or with respect to the Buildings and all of the real estate taxes and assessments imposed on the land and improvements comprising the Project, and (vi) assessments attributable to the Project by any governmental or quasi-governmental agency that Landlord is required to pay. For purposes of this Lease, Taxes shall be calculated as if the tenant improvements in the Buildings were fully constructed and the Project, the Buildings, and all tenant improvements in the Buildings were fully assessed for real estate tax purposes, and accordingly, Taxes shall be deemed to be increased appropriately. Notwithstanding anything to the contrary contained in this Section 5.1(b), there shall be excluded from Taxes (i) all excess profits taxes, franchise taxes, gift taxes, capital stock taxes, inheritance and succession taxes, estate taxes, federal and state net income taxes, and other taxes to the extent applicable to Landlord's net income (as opposed to rents, receipts or income attributable to operations at the Project), (ii) any items included as Operating Expenses, and (iii) any items paid by Tenant under Section 17.1 of this Lease.

- (c) "Lease Year" shall mean the twelve (12) month period commencing January 1st and ending December 31st.
- (d) "Tenant's Building Percentage" shall mean Tenant's percentage of the entire Building as determined by dividing the rentable area of the Premises by the total rentable area of the Building using the BOMA Standards. If there is a change in the total Building rentable area as a result of an addition to the Building, partial destruction, modification or similar cause, which event causes a reduction or increase on a permanent basis, Landlord shall cause adjustments in the computations as shall be necessary to provide for any such changes.
  - (e) "Market Area" shall mean Costa Mesa, California (the "City").
  - (f) "Comparable Buildings" shall mean comparable Class "A" office use buildings in the Market Area.
- 5.2 Notwithstanding anything to the contrary in this Article 5, Tenant shall not be obligated to pay any Operating Expenses prior to the first (\$\frac{1}{2}\$t) anniversary of the Commencement Date. In the event that the Operating Expenses of Landlord's operation of the Project during any Lease Year of the Term shall exceed the actual Operating Expenses for the Project for the Base Year, Tenant shall pay to Landlord, as Additional Rent, Tenant's Share (as hereinafter defined) of the difference between the Operating Expenses for a particular Lease Year and the Base Year. "Tenant's Share" shall be determined by multiplying any such difference between Operating Expenses for any Lease Year and the Base Year or pro rata portion thereof, respectively, by Tenant's Building Percentage. Landlord shall, in advance of each Lease Year, estimate what Tenant's Share will be for such Lease Year based, in part, on Landlord's operating budget for such Lease Year, and Tenant shall pay Tenant's Share as so estimated each month (the "Monthly Escalation Payments"). The Monthly Escalation Payments shall be due and payable at the same time and in the same manner as the Monthly Rent. Base Year Operating Expenses shall not include market-wide labor-rate increases due to extraordinary circumstances, including, but not limited to, boycotts and strikes, and utility rate increases due to extraordinary

circumstances including, but not limited to, conservation surcharges, boycotts, embargoes or other shortages or amortized costs relating to capital improvements or any one time Conservation Costs. Landlord may elect to segregate Operating Expenses into two or more subcategories. For example, Landlord may segregate insurance costs, electrical costs and/or other utility costs from other Operating Expenses. If Landlord elects to make that segregation, (a) Tenant's Share of Operating Expenses shall be determined separately for each such subcategory by making separate calculations of the increase in the cost of each subcategory of Operating Expense from the Base Year to the applicable calendar year, and (b) Tenant shall pay as Tenant's Share of Operating Expenses Tenant's Share of the increase in each such subcategory.

5.3 Landlord shall, within one hundred eighty (180) days after the end of each Lease Year other than the Base Year, or as soon thereafter as reasonably possible, provide Tenant with a written statement of the actual Operating Expenses incurred during such Lease Year for the Project and such statement shall set forth Tenant's Share of such Operating Expenses. Tenant shall pay Landlord, as Additional Rent, the difference between Tenant's Share of Operating Expenses and the amount of Monthly Escalation Payments made by Tenant attributable to said Lease Year, such payment to be made within thirty (30) days of the date of Tenant's receipt of said statement (except as provided in Section 5.4 below); similarly, Tenant shall receive a credit if Tenant's Share is less than the amount of Monthly Escalation Payments collected by Landlord during said Lease Year, such credit to be applied to future Monthly Escalation Payments to become due hereunder. If utilities, janitorial services or any other components of Operating Expenses increase during any Lease Year, Landlord may revise Monthly Escalation Payments due during such Lease Year by giving Tenant written notice to that effect; and thereafter, Tenant shall pay, in each of the remaining months of such Lease Year, a sum equal to the amount of the revised difference in Operating Expenses multiplied by Tenant's Building Percentage divided by the number of months remaining in such Lease Year.

5.4 Within ninety (90) days following Tenant's receipt of the Operating Expense statement or Taxes statement, Tenant may give Landlord notice (the "Review Notice") stating that Tenant elects to review Landlord's calculation of the amount of Operating Expenses or Taxes payable by Tenant for the Lease Year to which such statement applies and identifying with reasonable specificity the records of Landlord reasonably relating to such matters that Tenant desires to review. Tenant may not deliver more than one (1) Review Notice with respect to any Lease Year. If Tenant fails to give Landlord such a Review Notice within that ninety (90) day period, Tenant shall be deemed to have approved the applicable statement. If Tenant timely gives the Review Notice, Tenant shall be entitled to conduct or require an audit to be conducted, provided that (a) not more than one (1) such audit may be conducted during any Lease Year of the Term, (b) the records for each Lease Year may be audited only once, (c) such audit is commenced within one hundred fifty (150) days following Tenant's receipt of the applicable statement, and (d) such audit is completed and a copy thereof is delivered to Landlord within two hundred ten (210) days following Tenant's receipt of the applicable statement. Tenant's auditor must be a member of a nationally or regionally recognized accounting firm and must not charge a fee based on the amount that the accountant is able to save Tenant by the inspection. As a condition precedent to any inspection by Tenant's accountant, such accountant shall provide Landlord with its written agreement that such accountant shall maintain in strict confidence any and all information obtained in connection with the review, and except as may be required by a court of competent jurisdiction, shall not disclose such information to any person or entity other

than to the management personnel of Tenant. An overcharge of Operating Expenses or Taxes by Landlord shall not entitle Tenant to terminate this Lease. No subtenant shall have the right to audit. Any assignee's audit right will be limited to the period after the effective date of the assignment. No audit shall be permitted if an Event of Default by Tenant has occurred and is continuing under this Lease, including without limitation any failure by Tenant to pay any amount due under this Article 5. If Landlord responds to any such audit with an explanation of any issues raised in the audit, such issues shall be deemed resolved unless Tenant responds to Landlord with further written objections within thirty (30) days after receipt of Landlord's response to the audit. In no event shall payment of Rent ever be contingent upon the performance of such audit. For purposes of any audit, Tenant or Tenant's duly authorized representative, at Tenant's sole cost and expense, shall have the right, upon fifteen (15) days' written notice to Landlord, to inspect Landlord's books and records pertaining to Operating Expenses and Taxes at the offices of Landlord or Landlord's managing agent during ordinary business hours, provided that such audit must be conducted so as not to interfere with Landlord's business operations and must be reasonable as to scope and time. If actual Operating Expenses or Taxes are finally determined (by agreement of the parties, arbitration or a court of competent jurisdiction) to have been overstated or understated by Landlord for any calendar year, then the parties shall within thirty (30) days thereafter make such adjustment payment or refund as is applicable, and if actual Operating Expenses and Taxes are finally determined (by agreement of the parties, arbitration or a court of competent jurisdiction) to have been overstated by Landlord for any calendar year by in excess of seven percent (7%), then Landlord shall pay the reasonable cost of Tenant's audit, not to exceed \$5,000.00.

5.5 If the Expiration Date or Termination Date under this Lease does not fall on December 31 of any given year, then: (i) the period commencing on the January 1 immediately preceding said Expiration Date or Termination Date and continuing through, to and, including said Expiration Date or Termination Date shall be referred to in this Lease as the "Last Partial Year" and (ii) Tenant's Share of Operating Expenses for the Last Partial Year shall be, calculated by proportionately reducing the Base Year Operating Expenses to reflect the number of calendar months in said Last Partial Year (the "Adjusted Base Operating Expenses"). The Adjusted Base Operating Expenses shall then be compared with the actual Operating Expenses for said Last Partial Year to determine the amount of any increases or decreases in the actual Operating Expenses for said Last Partial Year over the Adjusted Base Operating Expenses. Tenant shall pay its Tenant's Share of any such increases within thirty (30) days following the receipt of a final statement or, as applicable, Landlord shall refund any overpayment concurrently with delivery of such final statement.

5.6 If the occupancy of the Building during any part of any Lease Year is less than ninety-five percent (95%), Landlord shall make an appropriate adjustment of the variable components of Operating Expenses for that Lease Year, as reasonably determined by Landlord using sound accounting and management principles, to determine the amount of Operating Expenses that would have been incurred had the Building been ninety-five percent (95%) occupied. This amount shall be considered to have been the amount of Operating Expenses for that Lease Year. For purposes of this Section 5.6, "variable components" include only those component expenses that are affected by variations in occupancy levels.

- 5.7 In the event that the Taxes during any Lease Year of the Term shall exceed the Taxes for the Project for the Base Year, Tenant shall pay to Landlord, as Additional Rent, "Tenant's Tax Share" (as hereinafter defined) of the difference between the Taxes for a particular Lease Year and the Base Year. "Tenant's Tax Share" shall be determined by multiplying any such difference between Taxes for any Lease Year and the Base Year or pro rata portion thereof, respectively, by Tenant's Building Percentage. Landlord shall, in advance of each Lease Year, estimate what Tenant's Tax Share will be for such Lease Year and Tenant shall pay Tenant's Tax Share as so estimated each month (the "Monthly Tax Payments"). The Monthly Tax Payments shall be due and payable at the same time and in the same manner as the Monthly Rent.
- 5.8 Landlord shall, within one hundred eighty (180) days after the end of each Lease Year other than the Base Year, or as soon thereafter as reasonably possible, provide Tenant with a written statement of the actual Taxes incurred during such Lease Year for the Project and such statement shall set forth Tenant's Tax Share of such Taxes. Tenant shall pay Landlord, as Additional Rent, the difference between Tenant's Tax Share of any increases in Taxes and the amount of Monthly Tax Payments made by Tenant attributable to said Lease Year, such payment to be made within thirty (30) days of the date of Tenant's receipt of said statement; similarly, Tenant shall receive a credit if Tenant's Tax Share is less than the amount of Monthly Tax Payments collected by Landlord during said Lease Year, such credit to be applied to future Monthly Tax Payments to become due hereunder. If Taxes increase during any Lease Year, Landlord may revise Monthly Tax Payments due during such Lease Year by giving Tenant written notice to that effect; and, thereafter, Tenant shall pay, in each of the remaining months of such Lease Year, a sum equal to the amount of revised difference in Taxes multiplied by Tenant's Tax Percentage divided by the number of months remaining in such Lease Year. Despite any other provision of this Article 5. Landlord may adjust Operating Expenses and/or Taxes and submit a corrected statement to account for Taxes or other government public-sector charges (including utility charges) that are for that given year but that were first billed to Landlord after the date that is ten (10) business days before the date on which the statement was furnished.
- 5.9 Tenant's Tax Share of Taxes for the Last Partial Year, if any, shall be calculated by proportionately reducing the Base Year Taxes to reflect the number of calendar months in said Last Partial Year (the "Adjusted Base Taxes"). The Adjusted Base Taxes shall then be compared with the actual Taxes for said Last Partial Year to determine the amount of any increases or decreases in the actual Taxes for such Last Partial Year over the Adjusted Base Taxes. Tenant shall pay its Tenant's Tax Share of any such increases within thirty (30) days following the receipt of a final statement or, as applicable, Landlord shall refund any overpayment concurrently with delivery of such final statement.
- 5.10 If the Taxes for the Base Year or the Taxes for any Lease Year are changed as a result of protest, appeal or other action taken by a taxing authority, the Taxes as so changed shall be deemed the Taxes for such Lease Year or Base Year (as applicable). Any reasonable expenses incurred by Landlord in attempting to protest, reduce or minimize Taxes shall be included in Taxes in the Lease Year in which those expenses are paid. Landlord shall have the exclusive right to conduct such contests, protests and appeals of the Taxes as Landlord shall determine is appropriate in Landlord's sole discretion. Tenant shall be entitled to receive its

proportionate share of any refunds received by Landlord with respect to any Taxes paid by Tenant. This provision shall survive the Expiration Date or Termination Date of the Lease for a period of two (2) years.

- 5.11 Tenant's obligation with respect to Additional Rent and the payment of Tenant's Share of Operating Expenses and Tenant's Tax Share of Taxes shall survive the Expiration Date or Termination Date of this Lease.
- 5.12 The parties acknowledge that the Building is a part of a multi-building project and that the costs and expenses incurred in connection with the Project <u>Ye.</u>, the Operating Expenses and Taxes) should be shared between the tenants of the Building and the tenants of the other buildings in the Project. Accordingly, as set forth in <u>Section 5.1</u> above, the Operating Expenses and Taxes are determined annually for the Project as a whole, and a portion of the Operating Expenses and Taxes, which portion shall be determined by Landlord on an equitable basis, shall be allocated to the tenants of the Building (as opposed to the tenants of any other buildings in the Project) and such portion shall be the Operating Expenses and Taxes for purposes of this Lease. Such portion of the Operating Expenses and Taxes altributable to the Building shall include all the Operating Expenses and Taxes attributable to the Project as a whole. Landlord shall have the right, from time to time, to equitably allocate some or all of the Operating Expenses and Taxes for the Project among different portions or occupants of the Project (the "Cost Pools"), in Landlord's reasonable discretion. Such Cost Pools may include, but shall not be limited to, the office space tenants of a building of the Project or of the Project, and the retail space tenants of a building of the Project or of the Project. The Operating Expenses and Taxes within each such Cost Pool shall be allocated and charged to the tenants within such Cost Pool in an equitable manner.
- 5.13 Notwithstanding anything to the contrary set forth in this Lease, Taxes for the Base Year and any subsequent Lease Year during the Term shall be calculated assuming the Project is assessed at its taxable value as determined under Section 51(a)(1) (but excluding Section 51(a)(2)) of the California Revenue and Taxation Code ("Taxable Value") without taking into account any decreases in real estate taxes obtained in connection with Proposition 8 or an assessed value lower than the Taxable Value, and, therefore, the Taxes payable under this Lease may be greater than those actually incurred by Landlord. Tax refunds under Proposition 8 or otherwise shall not be deducted from Taxes nor refunded to Tenant, but rather shall be the sole property of Landlord.

#### ARTICLE 6 SERVICES TO BE PROVIDED BY LANDLORD

- 6.1 Subject to Articles 5 and 10 herein, and provided Tenant is not in default under this Lease, Landlord agrees to furnish or cause to be furnished to the Premises the following utilities and services, subject to the conditions and in accordance with the standards set forth herein:
- 6.1.1 Subject to limitations imposed by all governmental rules, regulations and guidelines applicable thereto, Landlord shall provide heating, ventilation and air

conditioning ("HVAC") when necessary for normal comfort for normal office use in the Premises from 8:00 A.M. to 6:00 P.M. Monday through Friday (collectively, the "Building Hours"), except for the date of observation of New Year's Day, President's Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Day and, at Landlord's discretion, other locally and nationally recognized holidays observed by landlords of Comparable Buildings (collectively, the "Holidays"). If Tenant desires HVAC service on Saturdays from 9:00 A.M. to 1:00 P.M., Tenant shall give Landlord such prior notice, if any, as Landlord shall from time to time establish as appropriate, which shall not exceed 24 hours' notice, of Tenant's desired use in order to supply such HVAC, and Landlord shall supply such HVAC to Tenant during those hours on Saturday without charge other than including the cost thereof in Operating Expenses for the Building.

- 6.1.2 Landlord shall provide electrical wiring and facilities for connection to Tenant's lighting fixtures and incidental use equipment for normal office use, provided that (i) the connected electrical load of the incidental use equipment does not exceed an average of five (5.0) watts per rentable square foot of the Premises during the Building Hours, calculated on a monthly basis, and the electricity so furnished for incidental use equipment will be at a nominal one hundred twenty (120) volts and no electrical circuit for the supply of such incidental use equipment will require a current capacity exceeding twenty (20) amperes, and (ii) the connected electrical load of Tenant's lighting fixtures does not exceed an average of one (1.0) watt per rentable square foot of the Premises during the Building Hours, calculated on a monthly basis, and the electricity so furnished for Tenant's lighting will be at a nominal one hundred twenty (120) volts. Tenant will design Tenant's electrical system serving any equipment producing nonlinear electrical loads to accommodate such nonlinear electrical loads, including, but not limited to, oversizing neutral conductors, derating transformers and/or providing power-line filters. Engineering plans shall include a calculation of Tenant's fully connected electrical design load with and without demand factors and shall indicate the number of watts of unmetered and submetered loads. Tenant shall bear the cost of replacement of lamps, starters and ballasts for non-Building standard lighting fixtures within the Premises. To the extent that Tenant is not billed directly by a public utility, Tenant shall pay, within thirty (30) days of Landlord's demand, for all electricity used by Tenant in the Premises. The charge shall be at the rates charged for such services by the local public utility. Alternatively, Landlord may elect to include electricity costs in Operating Expenses.
- 6.1.3 Landlord shall, as part of Operating Expenses, provide city water from the regular Building outlets for drinking, kitchen, lavatory and toilet purposes in the Building Common Areas and the Premises.
- 6.1.4 Landlord shall, as part of Operating Expenses, provide nonexclusive, non-attended automatic passenger elevator service during the Building Hours, shall have one elevator available at all other times, including on the Holidays, except in the event of emergency.
- 6.1.5 Landlord shall provide customary weekday janitorial services to the Premises five (5) days per week, Sunday through Thursday, except the date of observation of the Holidays, in and about the Premises and customary occasional window washing services as determined by Landlord.

6.1.6 Subject to Landlord's rules, regulations, and restrictions and the terms of this Lease, Landlord shall permit Tenant to utilize the existing Building risers, raceways, shafts and conduit to the extent (i) such use does not exceed Tenant's Building Percentage of the space made available to tenants in such facilities, as determined by Landlord in Landlord's sole and absolute discretion, and (ii) Tenant's requirements are consistent with the requirements of a typical general office user. Tenant may only use vendors selected by Landlord to provide services to Tenant through the use of the Leased Project Building risers, raceways, shafts and conduit.

6.1.7 Landlord shall contract for on-site access control services for the Project (but not individually for Tenant or the Premises) of such scope and type as Landlord may determine in its sole discretion; provided that such access control services will include a lobby attendant on Monday through Friday, excluding Holidays, from 7:00 A.M. to 11:00 P.M. and one roving security guard on the Project at all times, subject to Applicable Laws. Landlord shall not be liable in any manner to Tenant or Tenant's employees, agents, subtenants, contractors, licensees, or invitees for any acts (including criminal acts) of others, or for any direct, indirect, or consequential damages, or any injury or damage to, or interference with, Tenant's business, including, but not limited to, loss of profits, loss of rents or other revenues, loss of business opportunity, loss of goodwill or loss of use, or other loss or damage, bodily injury or death, related to any malfunction, circumvention or other failure of any security services which Landlord provides or for the failure of any security services to prevent bodily injury, death, or property damage, or loss, or to apprehend any person suspected of causing such injury, death, damage or loss Landlord shall provide each of Tenant's employees with an initial card key at no cost.

6.2 Except as otherwise expressly provided by this Lease, Tenant shall not, without Landlord's prior written consent, use heat-generating machines, machines other than normal fractional horsepower office machines, or equipment or lighting other than Building standard lights in the Premises, which (in any such case) materially increases the water, electricity, heat or air conditioning normally furnished for the Premises by Landlord pursuant to the terms of Section 6.1 of this Lease. If Tenant uses water, electricity, heat or air conditioning in excess of that supplied by Landlord pursuant to Section 6.1 of this Lease, Tenant shall pay to Landlord, within thirty (30) days after billing, the cost of such excess consumption, the cost of the installation, operation, and maintenance of equipment which is installed in order to supply such excess consumption, and (based on reasonable estimates provided by qualified contractors who do not provide maintenance and repair service with respect to the particular system or equipment in question in the Project) the cost of the increased wear and tear on existing equipment caused by such excess consumption (without any markup or fees); and Landlord may install devices (including without limitation Quadlogic revenue grade meters (or comparable devices) and related data transmission and collection systems and software) to separately meter (or sub-meter) any increased use and in such event Tenant shall pay the increased cost directly to Landlord, on demand, at the rates charged by the public utility company furnishing the same, including the cost of such additional metering (or sub-metering) devices without any markup or fees. Notwithstanding the foregoing, the installation, operation, maintenance and repair of Temperature Balance Equipment (as defined in Section 6.7) shall be governed by the terms of Section 6.7. Tenant's use of electricity shall never exceed the capacity of the feeders to the Project or the risers or wiring installation. If Tenant desires to use heat, ventilatio

conditioning during hours other than those for which Landlord is obligated to supply such utilities pursuant to the terms of Section 6.1 of this Lease, Tenant shall give Landlord such prior notice, if any, as Landlord shall from time to time establish as appropriate, not to exceed 24 hours' notice as to after-hours HVAC, of Tenant's desired use in order to supply such utilities, and Landlord shall supply such utilities to Tenant at Landlord's "Actual Cost" (as defined below) (which shall be treated as Additional Rent) for a minimum one-hour period. Reference in this Lease to Landlord's "Actual Cost" will apply to cost charged by Landlord to provide HVAC service to Tenant hereunder and shall be calculated by Landlord in good faith to reflect the actual costs paid or incurred by Landlord to provide HVAC service, including the cost of utilities provided by the utility provider, costs of maintenance and repair and (based on reasonable estimates provided by qualified contractors who provide maintenance and repair service with respect to the particular system or equipment in question) costs of wear and tear and depreciation of systems and equipment, as applicable, and a five percent (5%) administrative fee, but excluding profits, it being agreed that if the amount of actual costs paid or incurred by Landlord cannot be reasonably and readily ascertained, "Actual Cost" hereunder shall be the amount reasonably estimated by Landlord. As of the date of this Lease, Landlord's Actual Cost for after hours HVAC is \$60 per hour. During the Term, Tenant shall be entitled to twenty (20) hours per month of after hours HVAC at no cost. Landlord shall have the exclusive right, but not the obligation, to provide any additional services which may be required by Tenant, including, without limitation, locksmithing, lamp replacement, additional janitorial service, and additional repairs and maintenance. If Tenant requests any such additional services, then Tenant shall pay to Landlord, within thirty (30) days after billing, the cost of

6.3 Landlord shall not be liable for any loss or damage arising or alleged to arise in connection with the failure, stoppage, or interruption of any such services; nor shall the same be construed as an eviction of Tenant, work an abatement of Rent, entitle Tenant to any reduction in Rent, or relieve Tenant from the operation of any covenant or condition herein contained; it being further agreed that Landlord reserves the right to discontinue temporarily such services or any of them at such times as may be necessary by reason of repair or capital improvements performed within the Project, accident, unavailability of employees, repairs, alterations or improvements, or whenever by reason of strikes, lockouts, riots, acts of God, or any other happening or occurrence beyond the reasonable control of Landlord. In the event of any such failure, stoppage or interruption of services, Landlord shall use reasonable diligence to have the same restored. Without limiting Section 6.9, neither diminution nor shutting off of light or air or both, nor any other effect on the Project by any structure erected or condition now or hereafter existing on lands adjacent to the Project, shall affect this Lease, abate Rent, or otherwise impose any liability on Landlord. Furthermore, Landlord shall not be liable under any circumstances for a loss of, or injury to, property or for injury to, or interference with, Tenant's business, including, without limitation, loss of profits, however occurring, through or in connection with or incidental to a failure to furnish any of the services or utilities as set forth in this <u>Article 6</u>. Except in cases of emergency, Landlord shall give Tenant reasonable prior notice of any delay or diminution in any service and shall use reasonable efforts to minimize any inconvenience to Tenant where such delay or diminution is the result of Landlord's voluntary actions.

- 6.4 Landlord shall have the right to reduce heating, cooling, or lighting within the Premises and in the public area in the Building as required by any mandatory fuel or energy-saving program.
- 6.5 Unless otherwise provided by Landlord, Tenant shall separately arrange with the applicable local public authorities or utilities, as the case may be, for the furnishing of and payment of all telephone and facsimile services as may be required by Tenant in the use of the Premises. Tenant shall directly pay for such telephone and facsimile services as may be required by Tenant in the use of the Premises, including the establishment and connection thereof, at the rates charged for such services by said authority or utility; and the failure of Tenant to obtain or to continue to receive such services for any reason whatsoever shall not relieve Tenant of any of its obligations under this Lease.
- 6.6 At all times during the Term Landlord shall have the right to select the utility company or companies that shall provide electric, telecommunication and/or other utility services to the Premises and, subject to all Applicable Laws, Landlord shall have the right at any time and from time to time during the Term to either (a) contract for services from electric, telecommunication and/or other utility service provider(s) other than the provider with which Landlord has a contract as of the date of this Lease (the "Current Provider"), or (b) continue to contract for services from the Current Provider. The cost of such utility services and any energy management and procurements services in connection therewith shall be Operating Expenses.
- 6.7 If the temperature otherwise maintained in any portion of the Premises by the heating, air conditioning or ventilation system ("HVAC") is affected in any material respect as a result of (i) the type or quantity of any lights, machines or equipment (including without limitation typical office equipment) used by Tenant in the Premises, (ii) the occupancy of such portion of the Premises by more than one person per one hundred twenty (120) square feet of rentable area therein, (iii) an electrical load for lighting or power in excess of the limits specified in Section 6.1.2 above, or (iv) any rearrangement of partitioning or other improvements, then at Tenant's sole cost, Landlord may install any HVAC equipment, or modify any existing HVAC equipment (including the standard HVAC equipment) Landlord deems necessary to restore the temperature balance (such new HVAC equipment or modifications to existing HVAC equipment termed herein "Temperature Balance Equipment"). Tenant agrees to cooperate with Landlord and to abide by the regulations and requirements which Landlord may reasonably prescribe for the proper functioning and protection of the HVAC system. Landlord makes no representation to Tenant regarding the adequacy or fitness of the HVAC equipment in the Building to maintain temperatures that may be required for, or because of, any computer or communications rooms, machine rooms, conference rooms or other areas of high concentration of personnel of electrical usage, any other uses other than or in excess of the fractional horsepower normally required for office equipment or occupancy of the Premises that exceeds a density greater than one (1) person per one hundred fifty (150) rentable square feet of space in the Premises, and Landlord shall have no liability for loss or damage suffered by Tenant or others in connection therewith. Tenant shall pay to Landlord on demand the cost of installing, operating, maintaining and repairing any Temperature Balance Equipment.
- 6.8 Landlord acknowledges and agrees that Tenant's intended and permitted uses may include the use of servers, server rooms, workstations, and similar or accompanying

equipment, including, but not limited to, UPS machines, and that such uses are hereby permitted so long as (i) Tenant pays any excess electrical charges attributable to same, (ii) the installation, construction and maintenance thereof is performed by Tenant at Tenant's sole cost and expense and in compliance with all applicable laws, (iii) the Building and Building systems have sufficient capacity and space to accommodate those facilities, (iv) the area occupied by servers and server rooms does not exceed two percent (2%) of the rentable area of the Premises, (v) the area occupied by servers and server rooms are served by adequate supplemental HVAC systems installed by Tenant in accordance with this Lease at Tenant's sole cost and expense, and (vi) the server room and supplemental HVAC systems are separately metered at Landlord's cost and Tenant pays the cost of all utilities serving those supplemental HVAC systems and the server room.

6.9 Notwithstanding anything to the contrary in Section 6.2 or elsewhere in this Lease, if (a) Landlord fails to provide Tenant with the HVAC service, the water service, the electrical service or elevator service described in Section 6.1, or Landlord enters the Premises and such entry interferes with Tenant's reasonable use of the Premises, or work or damage to other portions of the Building interfere with Tenant's access to or from or the use of the Premises, (b) such failure or Landlord's entry is not due to any one or more Force Majeure Events or to an event covered by Article 19, (c) Tenant has given Landlord reasonably prompt written notice of such failure or that such entry by Landlord is unreasonably interfering with Tenant's use of the Premises and (d) as a result of such failure or entry all or any part of the Premises are rendered untenantable (and, as a result, all or such part of the Premises are not used by Tenant during the applicable period) for more than five (5) consecutive business days, then Tenant shall be entitled to an abatement of Rent proportional to the extent to which the Premises are thereby rendered unusable by Tenant, commencing with the later of (i) the sixth business day during which such untenantability continues or (ii) the sixth business day after Landlord receives such notice from Tenant, until the Premises (or part thereof affected) are again usable or until Tenant again uses the Premises (or part thereof rendered unusable) in its business, whichever first occurs. If, but for the untenantability having been caused by one or more Force Majeure Events, Tenant would have received rental abatement under this Section 6.9, and the rental loss insurance carried by Landlord, the premiums for which are included in Operating Expenses, covers rent loss for such abatement, then Tenant shall receive rent abatement to the extent Landlord actually receives proceeds from rent loss insurance attributable to that rent abatement. The foregoing rental abatement shall be Tenant's exclusive remedy

# ARTICLE 7 REPAIRS AND MAINTENANCE BY LANDLORD

7.1 Landlord shall provide for the cleaning, maintenance and repair of the Base Building, the Common Areas of the Project and the Parking Facilities in keeping with the standard for Comparable Buildings as part of Operating Expenses. Unless otherwise expressly stipulated herein, Landlord shall not be required to make any improvements or repairs of any kind or character to the Premises during the Term, except such repairs as may be required to the

Base Building (as defined below) and other structural elements and equipment of the Project, and subject to Section 13.4, below, such additional maintenance as may be necessary because of the damage caused by persons other than Tenant, its agents, employees, licensees, or invitees. As used in this Lease, the "Base Building" shall include the structural portions of the Building, and the public restrooms, elevators, exit stairwells and the systems and equipment located in the internal core of the Building on the floor or floors on which the Premises are located. The term "Base Building," as used in this Lease, shall not be deemed to have the same meaning as the term "Base, Shell and Core," as the same is defined in Section 1 of the Tenant Work Letter.

7.2 Landlord or Landlord's officers, agents, and representatives (subject to any security regulations imposed by any governmental authority) shall have the right to enter all parts of the Premises at all reasonable hours upon reasonable prior notice to Tenant (other than in an emergency) to inspect, clean, make repairs, alterations, and additions to the Project or the Premises which it may deem necessary or desirable, to make repairs to adjoining spaces, to cure any defaults of Tenant hereunder that Landlord elects to cure pursuant to Section 22.5, below, to post notices of non-responsibility, to show the Premises to prospective tenants (during the final nine (9) months of the Term or at any time after the occurrence of an Event of Default that remains uncured), mortgagees or purchasers of the Building, or to provide any service which it is obligated or elects to furnish to Tenant; and Tenant shall not be entitled to any abatement or reduction of Rent by reason thereof. Landlord shall have the right to enter the Premises at any time and by any means in the case of an emergency. Tenant shall have the right to have one or more of its employees accompany Landlord or Landlord's officers, agents, and representatives during their entry into the Premises pursuant to this Lease; provided that Tenant's failure to make a Tenant employee available at the time of Landlord's entry into the Premises shall not limit Landlord's officers, agents, representatives' right to enter the Premises.

7.3 Except as otherwise expressly provided in this Lease, Landlord shall not be liable for any failure to make any repairs or to perform any maintenance and there shall be no abatement of rent and no liability of Landlord by reason of any injury to or interference with Tenant's business arising from the making of any repairs, alterations or improvements in or to any portion of the Project, Building or the Premises or in or to fixtures, and equipment therein. Tenant hereby waives all rights it would otherwise have under California Civil Code Sections 1932(1) and 1942(a), or any similar law, statute or ordinance now or hereafter in effect, to make repairs at Landlord's expense, to deduct repair costs from Rent and/or terminate this Lease as the result of any failure by Landlord to maintain or repair.

# ARTICLE 8 REPAIRS AND CARE OF PREMISES BY TENANT

8.1 If the Building, the Project, or any portion thereof, including but not limited to, the elevators, boilers, engines, pipes, and other apparatus, or members of elements of the Building (or any of them) used for the purpose of climate control of the Building or operating of the elevators, or of the water pipes, drainage pipes, electric lighting, or other equipment of the Building or the roof or outside walls of the Building and also the Premises improvements, including but not limited to, the carpet, wall coverings, doors, and woodwork, become damaged or are destroyed through the negligence, carelessness, or misuse of Tenant, its servants, agents, employees, or anyone permitted by Tenant to be in the Building, or through it or them, then the

reasonable cost of the necessary repairs, replacements, or alterations shall be borne by Tenant who shall pay the same to Landlord as Additional Rent within thirty (30) days after written demand, subject to Section 13.4 below. Landlord shall have the exclusive right, but not the obligation, to make any repairs necessitated by such damage.

8.2 Subject to Section 13.4 below, Tenant agrees, at its sole cost and expense, to repair or replace any damage or injury done to the Project, or any part thereof, caused by Tenant, Tenant's agents, employees, licensees, or invitees which Landlord elects not to repair. Tenant shall not injure the Project or the Premises and, at Tenant's sole cost and expense, shall maintain the Premises, including without limitation all improvements, fixtures and furnishings therein, and the floor or floors on which the Premises are located, in good order, repair and condition at all times during the Term. If Tenant fails to keep such elements of the Premises in such good order, condition, and repair as required hereunder to the satisfaction of Landlord, Landlord may restore the Premises to such good order and condition and make such repairs without liability to Tenant for any loss or damage that may accrue to Tenant's property or business by reason thereof, and within thirty (30) days after completion thereof, Tenant shall pay to Landlord, as Additional Rent, upon written demand, the cost of restoring the Premises to such good order and condition and of the making of such repairs, plus an additional charge of five percent (5%) thereof. Tenant shall leave the Premises at the end of each business day in a reasonably tidy condition for the purpose of allowing the performance of Landlord's cleaning services. Upon the Expiration Date or the Termination Date, Tenant shall surrender and deliver up the Premises to Landlord in the same condition in which it existed at the Commencement Date, excepting only ordinary wear and tear and damage arising from any cause not required to be repaired by Tenant. Upon the Expiration Date or the Termination Date, Landlord shall have the right to re-enter and take possession of the Premises.

8.3 In the event of an emergency situation, Landlord shall, upon notice from Tenant, immediately commence any necessary repairs and pursue the same to completion with reasonable diligence if the work to correct the emergency situation involves elements of the Building that are Landlord's obligation to repair and maintain under Article 7. If Landlord fails to commence such repairs within twenty-four (24) hours of actual receipt of such notice, Tenant shall have the right to perform such repairs to correct the emergency situation under the limited circumstances where all of the following conditions exist (the "Self Help Conditions"): (a) Landlord has not commenced such repairs within twenty-four (24) hours of actual receipt of such notice or, if later, by the date Tenant attempts to commence such repairs; (b) at the time Tenant intends to commence such repairs, Tenant provides Landlord with written notice of the fact that Tenant intends to commence such repairs on Landlord's behalf; (c) the action required to correct the emergency situation does not require Tenant to have access to the premises of any other tenant or occupant of the Building; and (d) the action required to correct the emergency situation does not involve the performance of work with respect to the roof of the Building. If all of the Self Help Conditions exist and Tenant elects to cure the default of Landlord, any such curative work must be performed by contractors that have previously performed similar work for Landlord in the Building, unless such contractors are unwilling or unable to perform, or timely perform, such work (or unless Landlord has not provided Tenant with a list of such contractors), in which event Tenant may utilize the services of any other qualified contractor which normally and regularly performs similar work in Comparable Buildings. In addition, if Landlord has a warranty or service and maintenance agreement for a particular item that requires service, repair

or replacement as part of the curative work (and Tenant is aware of such warranty or contract), Tenant shall endeavor to contract with the issuer of such warranty or contract for the performance of the curative work so as not to adversely affect such warranty or contract. In attempting to correct the emergency situation, Tenant shall perform only so much work as is reasonably necessary to correct the emergency situation. All such work shall be diligently pursued to completion and shall be conducted in a manner which does not interfere with the normal operations of the Building and other tenants of the Building. If the work to correct the emergency situation involves elements of the Building that are Landlord's obligation to repair and maintain under Article 7, then upon completion of the curative work by Tenant and receipt of paid invoices, Landlord shall reimburse Tenant for all expenses reasonably incurred by Tenant in correcting the emergency situation, net of any amounts that would have been payable by Tenant if Landlord had performed the work. For purposes hereof, an "emergency situation" shall mean a situation which immediately threatens substantial damage or injury to the Premises or to persons in the Premises. If Landlord fails to repay Tenant for such costs within thirty (30) days after Landlord's receipt of the paid invoices, Tenant shall be entitled to deliver notice (the "Obligation Payment Notice") thereof to Landlord and to any mortgage or trust deed holder of the Building whose identity and address have been previously provided to Tenant. If Landlord still fails to fulfill any such obligation within twenty (20) business days after Landlord's receipt of the Obligation Payment Notice from Tenant and if Landlord fails to deliver notice to Tenant within such twenty (20) business day period explaining Landlord's good faith reasons that the amounts described in Tenant's Obligation Payment Notice are not due and payable by Landlord ("Obligation Payment Refusal Notice"), Tenant shall be entitled to offset that amount (not to exceed the amount set forth in the Obligation Payment Notice) plus simple interest at the prime rate plus two percent (2%), but in no event in excess of the maximum interest rate permitted by law (the "Agreed Rate"), from the date Landlord's original payment was due to be paid until the date of the offset therefor, against Tenant's next obligations to pay Rent. However, if Tenant is in default under the Lease beyond applicable notice and cure periods at the time that such offset would otherwise be applicable. Tenant shall not be entitled to such offset until such default is cured. If Landlord delivers an Obligation Payment Refusal Notice, and if Landlord and Tenant are not able to agree on the amounts to be so paid by Landlord, if any, within ten (10) business days after Tenant's receipt of an Obligation Payment Refusal Notice, Tenant may submit such dispute to arbitration in accordance with Article 12 of this Lease. If Tenant prevails in any such arbitration, Tenant shall be entitled to offset such arbitration award against Tenant's next obligations to pay Rent and the award shall include simple interest at the Agreed Rate calculated from the date such amount was due to Tenant until the date of Tenant's application of such amounts as a credit against Rent. In no event shall the total amounts offset under this Section 8.3 in any calendar month exceed fifty percent (50%) of the Base Rent payable in that calendar month.

# ARTICLE 9 TENANT'S EQUIPMENT AND INSTALLATIONS

9.1 Except for desk or table-mounted typewriters, adding machines, office calculators, dictation equipment, personal computers, and other similar office equipment consistent with first-class general office use in Comparable Buildings or as otherwise provided in Article 6, Tenant shall not install within the Premises any fixtures, equipment, facilities, or other improvements without the specific written consent of Landlord which consent shall not be

unreasonably withheld, conditioned or delayed, subject to Article 15, below. Tenant shall not, without the specific written consent of Landlord (which consent shall not be unreasonably withheld, conditioned, or delayed), install or maintain any apparatus or device within the Premises which shall increase the usage of electrical power or water for the Premises to an amount greater than would be normally required for general office use for space of comparable size in the Market Area; and if any such apparatus or device is so installed, Tenant agrees to furnish Landlord a written agreement to pay for any additional costs of utilities as the result of said installation.

## ARTICLE 10 FORCE MAJEURE

10.1 It is understood and agreed that with respect to any service or other obligation to be furnished or obligations to be performed by either party that in no event shall either party be liable for failure to furnish or perform the same when prevented from doing so by strike, lockout, breakdown, accident, supply, or inability by the exercise of reasonable diligence to obtain supplies, parts, or employees necessary to furnish such service or meet such obligation; or because of war or other emergency; or for any cause beyond the reasonable control with the party obligated for such performance; or for any cause due to any act or omission of the other party or its agents, employees, licensees, invitees, or any persons claiming by, through, or under the other party; or because of the failure of any public utility to furnish services; or because of order or regulation of any federal, state, county or municipal authority (collectively, "Force Majeure Events"). Nothing in this Section 10.1 shall limit or otherwise modify or waive Tenant's obligation to pay Base Rent and Additional Rent as and when due pursuant to the terms of this Lease.

# ARTICLE 11 CONSTRUCTION, MECHANICS' AND MATERIALMAN'S LIENS

- 11.1 Tenant shall not suffer or permit any construction, mechanics' or materialman's lien to be filed against the Premises or any portion of the Project by reason of work, labor services, or materials supplied or claimed to have been supplied to Tenant. Nothing herein contained shall be deemed or construed in any way as constituting the consent or request of Landlord, expressed or implied, by inference or otherwise, for any contractor, subcontractor, laborer, or materialman to perform any labor or to furnish any materials or to make any specific improvement, alteration, or repair of or to the Premises or any portion of the Project; nor of giving Tenant any right, power, or authority to contract for, or permit the rendering of, any services or the furnishing of any materials that could give rise to the filing of any construction, mechanics' or materialman's lien against the Premises or any portion of the Project.
- 11.2 If any such construction, mechanics' or materialman's lien shall at any time be filed against the Premises or any portion of the Project as the result of any act or omission of Tenant, Tenant covenants that it shall, within twenty (20) days after Tenant has notice of the claim for lien, procure the discharge thereof by payment or by giving security or in such other manner as is or may be required or permitted by law or which shall otherwise satisfy Landlord. If Tenant fails to take such action, Landlord, in addition to any other right or remedy it may have, may take such action as may be reasonably necessary to protect its interests. Any

amounts paid by Landlord in connection with such action, all other expenses of Landlord incurred in connection therewith, including reasonable attorneys' fees, court costs, and other necessary disbursements shall be repaid by Tenant to Landlord within twenty (20) days after demand.

#### ARTICLE 12 ARBITRATION

12.1 In the event that a dispute arises under Section 5.3 above, the same shall be submitted to arbitration in accordance with the provisions of applicable state law, if any, as from time to time amended. Arbitration proceedings, including the selection of an arbitrator, shall be conducted pursuant to the rules, regulations, and procedures from time to time in effect as promulgated by the American Arbitration Association (the "Association"). Prior written notice of application by either party for arbitration shall be given to the other at least ten (10) days before submission of the application to the said Association's office in the city wherein the Building is situated (or the nearest other city having an Association office). The arbitrator shall hear the parties and their evidence. The decision of the arbitrator may be entered in the appropriate court of law; and the parties consent to the jurisdiction of such court and further agree that any process or notice of motion or other application to the court or a judge thereof may be served outside the state wherein the Building is situated by registered mail or by personal service, provided a reasonable time for appearance is allowed. The costs and expenses of each arbitration hereunder and their apportionment between the parties shall be determined by the arbitrator in his or her award or decision, subject to the penultimate sentence of this section. No arbitrable dispute shall be deemed to have arisen under this Lease (a) prior to the expiration of the period of twenty (20) days after the date of the giving of written notice by the party asserting the existence of the dispute, together with a description thereof sufficient for an understanding thereof, and (b) where Tenant disputes the amount of a Tenant payment required hereunder (e.g., Operating Expense excess under Section 5.3 hereof), prior to Tenant paying in full the amount billed by Landlord, including the disputed amount. The prevailing party in such arbitration shall be reimbursed for its expenses, in

#### ARTICLE 13 INSURANCE

13.1 Landlord shall maintain, as a part of Operating Expenses, special causes of loss form property insurance on the Project (excluding, at Landlord's option, the property required to be insured by Tenant pursuant to Section 13.2(e), below) in an amount equal to the full replacement cost of the Project, subject to such deductibles as Landlord may determine. Landlord shall not be obligated to insure, and shall not assume any liability of risk of loss for, any of Tenant's furniture, equipment, machinery, goods, supplies, improvements or alterations upon the Premises. Such insurance shall be maintained with an insurance company selected, and in amounts desired, by Landlord or Landlord's mortgagee, and payment for losses thereunder shall be made solely to Landlord subject to the rights of the holder of any mortgage or deed of trust which may now or hereafter encumber the Project. Additionally Landlord may maintain such additional insurance, including, without limitation, earthquake insurance, flood insurance,

liability insurance and/or rent insurance, as Landlord may in its sole discretion elect. The cost of all such additional insurance shall also be part of the Operating Expenses. Any or all of Landlord's insurance may be provided by blanket coverage maintained by Landlord or any affiliate of Landlord under its insurance program for its portfolio of properties or by Landlord or any affiliate of Landlord's program of self-insurance, and in such event Operating Expenses shall include the portion of the reasonable cost of blanket insurance or self-insurance that is allocated to the Project.

13.2 Tenant, at its own expense, shall maintain with insurers authorized to do business in the State of California and which are ratedA- or better and have a financial size category of at least VIII in the most recent Best's Key Rating Guide, or any successor thereto (or if there is none, an organization having a national reputation), (a) commercial general liability insurance, including Broad Form Property Damage and Contractual Liability with the following minimum limits: General Aggregate \$3,000,000.00; Products/Completed Operations Aggregate \$2,000,000.00; Each Occurrence \$2,000,000.00; Personal and Advertising Injury \$1,000,000.00; Medical Payments \$5,000.00 per person, (b) Umbrella/Excess Liability on a following form basis with the following minimum limits: General Aggregate \$10,000,000.00; Each Occurrence \$10,000,000.00; (c) Workers' Compensation with statutory limits; (d) Employer's Liability insurance with the following limits: Bodily injury by disease per person \$1,000,000.00; Bodily injury by accident policy limit \$1,000,000.00; Bodily injury by disease policy limit \$1,000,000.00; (e) property insurance on special causes of loss insurance form covering (i) all office furniture, business and trade fixtures, office equipment, free-standing cabinet work, movable partitions, merchandise and all other items of Tenant's property on the Premises installed by, for, or at the expense of Tenant, (ii) the "Tenant Improvements," as that term is defined in Section 2.1 of the Tenant Work Letter, and any other improvements which exist in the Premises as of the Commencement Date (excluding the Base Building) (the "Original Improvements"), and (iii) all other improvements, alterations and additions to the Premises (such insurance shall be for the full replacement cost value (subject to reasonable deductible amounts) new without deduction for depreciation of the covered items and in amounts that meet any co-insurance clauses of the policies of insurance and shall include coverage for damage or other loss caused by fire or other peril including, but not limited to, vandalism and malicious mischief, theft, water damage of any type, including sprinkler leakage, bursting or stoppage of pipes, and explosion); and (f) business auto liability insurance having a combined single limit of not less than One Million Dollars (\$1,000,000.00) per occurrence and insuring Tenant against liability for claims arising out of ownership, maintenance or use of any owned, hired or non-owned automobiles. At all times during the Term, such insurance shall be maintained, and Tenant shall cause a current and valid certificate of such policies to be deposited with Landlord. If Tenant fails to have a current and valid certificate of such policies on deposit with Landlord at all times during the Term and such failure is not cured within three (3) business days following Tenant's receipt of notice thereof from Landlord, Landlord shall have the right, but not the obligation, to obtain such an insurance policy, and Tenant shall be obligated to pay Landlord the amount of the premiums applicable to such insurance within ten (10) days after Tenant's receipt of Landlord's request for payment thereof. Said policy of liability insurance shall name Landlord, Landlord's affiliates and subsidiaries designated by Landlord, and Landlord's managing agent as additional insureds and Tenant as the insured and shall be noncancellable with respect to Landlord except after thirty (30) days' written notice from the insurer to Landlord.

- 13.3 Tenant shall adjust not more often than once during the initial Term and any Extension Period under Article 51 the amount of coverage established in Section 13.2 hereof to such amount as in Landlord's reasonable opinion, adequately protects Landlord's interest; provided the same is consistent with the amount of coverage customarily required of comparable tenants in the Project and in Comparable Buildings. In addition, Landlord shall not require such adjustment prior to the third anniversary of the Commencement Date.
- 13.4 Notwithstanding anything in this Lease to the contrary, Landlord and Tenant each hereby waives any and all rights of recovery, claim, action, or cause of action against the other, its agents, employees, licensees, or invitees for any loss or damage to or at the Premises or the Project or any personal property of such party therein or thereon by reason of fire, the elements, or any other cause which would be insured against under the terms of (i) special causes of loss form property insurance or (ii) the liability insurance referred to in Section 13.2. to the extent of such insurance, regardless of cause or origin, including omission of the other party hereto, its agents, employees, licensees, or invitees. Landlord and Tenant covenant that no insurer shall hold any right of subrogation against either of such parties with respect thereto. This waiver shall be ineffective against any insurer of Landlord or Tenant to the extent that such waiver is prohibited by the laws and insurance regulations of the State of California. The parties hereto agree that any and all such insurance policies required to be carried by either shall be endorsed with a subrogation clause, substantially as follows: "This insurance shall not be invalidated should the insured waive, in writing prior to a loss, any and all right of recovery against any party for loss occurring to the property described therein, "and shall provide that such party's insurer waives any right of recovery against the other party in connection with any such loss or damage.
- 13.5 In the event Tenant's occupancy or conduct of business in or on the Premises, whether or not Landlord has consented to the same, results in any increase in premiums for the insurance carried from time to time by Landlord with respect to the Building, Tenant shall pay any such increase in premiums as Rent within thirty (30) days after bills for such additional premiums shall be rendered by Landlord. In determining whether increased premiums are a result of Tenant's use or occupancy of the Premises, a schedule issued by the organization computing the insurance rate on the Building showing the various components of such rate, shall be conclusive evidence of the several items and charges which make up such rate. Tenant shall promptly comply with all reasonable requirements of the insurance authority or of any insurer now or hereafter in effect relating to the Premises.

### ARTICLE 14 QUIET ENJOYMENT

14.1 Provided Tenant is not in default under this Lease after the expiration of any period for cure in the performance of all its obligations under this Lease, including, but not limited to, the payment of Rent and all other sums due hereunder, Tenant shall peaceably and quietly hold and enjoy the Premises for the Term, without hindrance by Landlord, subject to the provisions and conditions set forth in this Lease.

## ARTICLE 15 ALTERATIONS

15.1 Tenant agrees that it shall not make or allow to be made any alterations, physical additions, or improvements in or to the Premises without first obtaining the written consent of Landlord in each instance. As used herein, the term "Minor Alteration" refers to an alteration that (a) does not affect the outside appearance of the Building and is not visible from the Common Areas, (b) is non-structural and does not impair the strength or structural integrity of the Building, and (c) does not affect the mechanical, electrical, HVAC or other systems of the Building. Landlord agrees not to unreasonably withhold its consent to any Minor Alteration. Landlord's consent to any other alteration may be conditioned, given, or withheld in Landlord's sole discretion. Notwithstanding the foregoing, Landlord consents to any repainting, recarpeting, or other purely cosmetic changes or upgrades to the Premises, so long as (i) the aggregate cost of such work is less than \$50,000.00 in any twelve-month period, (ii) such work constitutes a Minor Alteration (iii) no building permit is required in connection therewith, and (iv) such work conforms to the then existing Building standards. At the time of said request, Tenant shall submit to Landlord plans and specifications of the proposed alterations, additions, or improvements; and Landlord shall have a period of not less than thirty (30) days therefrom in which to review and approve or disapprove said plans; provided that if Landlord determines in good faith that Landlord requires a third party to assist in reviewing such plans and specifications, Landlord shall instead have a period of not less than sixty (60) days in which to review and approve or disapprove said plans. Tenant shall pay to Landlord upon demand the reasonable cost and expense of Landlord in (A) reviewing said plans and specifications by third party consultants, and (B) inspecting the alterations, additions, or improvements to determine whether the same are being performed in accordance with the approved plans and specifications and all laws and requirements of public authorities, including, without limitation, the fees of any architect or engineer employed by Landlord for such purpose. In any instance where Landlord grants such consent, and permits Tenant to use its own contractors, laborers, materialmen, and others furnishing labor or materials for Tenant's construction (collectively, "Tenant's Contractors"), Landlord's consent shall be deemed conditioned upon each of Tenant's Contractors (1) working in harmony and not interfering with any laborer utilized by Landlord, Landlord's contractors, laborers, or materialmen; and (2) furnishing Landlord with evidence of acceptable liability insurance, worker's compensation coverage and if required by Landlord, completion bonding, and if at any time such entry by one or more persons furnishing labor or materials for Tenant's work shall cause such disharmony or interference, the consent granted by Landlord to Tenant may be withdrawn immediately upon written notice from Landlord to Tenant. Landlord agrees not to require completion bonding for any Alterations costing less than \$250,000.00 so long as the initially named Tenant or any Permitted Transferee is the Tenant under this Lease. If Tenant is using Tenant's Contractors for Tenant's construction, the contract with such Tenant's Contractor(s) shall provide for a guaranteed maximum price or a stipulated sum as the contract amount and shall be fully executed and delivered by Tenant and Tenant's Contractor(s) prior to the commencement of construction. Tenant, at its expense, shall obtain all necessary governmental permits and certificates for the commencement and prosecution of alterations, additions, or improvements and for final approval thereof upon completion, and shall cause any alterations, additions, or improvements to be performed in compliance therewith and with all Applicable Laws (including without limitation, California Energy Code, Title 24) and all requirements of public authorities and with all applicable requirements of insurance bodies. All

alterations, additions, or improvements shall be diligently performed in a good and workmanlike manner, using new materials and equipment at least equal in quality and class to be better than (a) the original installations of the Building, or (b) the then standards for the Comparable Building. Upon the completion of work and upon request by Landlord, Tenant shall provide Landlord copies of all waivers or releases of lien from each of Tenant's Contractors. No alterations, modifications, or additions to the Project or the Premises shall be removed by Tenant either during the Term or upon the Expiration Date or the Termination Date without the express written approval of Landlord. Tenant shall not be entitled to any reimbursement or compensation resulting from its payment of the cost of constructing all or any portion of said improvements or modifications thereto unless otherwise expressly agreed by Landlord in writing.

15.2 Upon submission of any plans for Landlord's approval, Tenant may request prior to the installation of specific Required Removal Items (as defined below) in the Premises as an Alteration or as part of any Tenant Improvements performed pursuant to the Work Letter, that Landlord agree not to require Tenant to remove such Required Removal Items upon expiration or termination of the Lease or agree to permit Tenant to remove any item it may otherwise not be permitted to remove under the terms of this Lease. Such consent, which may be granted or denied in Landlord's sole discretion, must be granted in writing in order to be binding against Landlord and shall be granted or denied concurrently with Landlord's approval of the applicable plan. If so required by Landlord, Tenant, prior to the expiration of the Term or termination of this Lease, shall, at Tenant's sole cost and expense, (i) remove any or all Required Removal Items, (ii) restore the Premises to the condition existing prior to the installation of such Required Removal Items, and (iii) repair all damage to the Premises or Project caused by the removal of such Required Removal Items. Tenant shall use a contractor reasonably approved by Landlord for such removal and repair. If Tenant fails to remove, restore and repair under this Section, then Landlord may remove such Required Removal Items and perform such restoration and repair, and Tenant shall reimburse Landlord for costs and expenses incurred by Landlord in performing such removal, restoration and repair. As used in this Section 15.2, "Required Removal Items" means (a) with respect to the Tenant Improvements performed pursuant to the Work Letter anyNon-Standard Improvements (such as fish tanks or interior climbing walls) which are so unusual or atypical that they are not reasonably suited for use by any successor occupant of the Premises for general office purposes. Specifically, and without limiting the generality of the foregoing, any interior Building Standard Stairs (as defined in Exhibit C-1) inst

15.3 Landlord's approval of Tenant's plans for work shall create no responsibility or liability on the part of Landlord for their completeness, design sufficiency, or compliance with all laws, rules, and regulations of governmental agencies or authorities, including, but not limited to, the Americans with Disabilities Act. Landlord may, at its option, at Tenant's expense, require that Landlord's contractors be engaged for any work upon the integrated Building mechanical or electrical systems or other Building or leasehold improvements.

- 15.4 At least five (5) days prior to the commencement of any work permitted to be done by persons requested by Tenant on the Premises, Tenant shall notify Landlord of the proposed work and the names and addresses of Tenant's Contractors. During any such work on the Premises, Landlord, or its representatives, shall have the right to go upon and inspect the Premises at all reasonable times, and shall have the right to post and keep posted thereon building permits and notices of non-responsibility or to take any further action which Landlord may deem to be proper for the protection of Landlord's interest in the Premises.
- 15.5 During such times as Tenant is performing work or having work or services performed in or to the Premises, Tenant shall require its contractors, and their subcontractors of all tiers, to obtain and maintain commercial general liability, automobile, workers compensation, employer's liability, builder's risk, and equipment/property insurance in such amounts and on such terms as are customarily required of such contractors and subcontractors on similar projects. The amounts and terms of all such insurance are subject to Landlord's written approval, which approval shall not be unreasonably withheld. The commercial general liability and auto insurance carried by Tenant's contractors and their subcontractors of all tiers pursuant to this section shall name Landlord, Landlord's managing agent, and such other persons as Landlord may reasonably request from time to time as additional insureds with respect to liability arising out of or related to their work or services (collectively, "Additional Insureds"). Such insurance shall provide primary coverage without contribution from any other insurance carried by or for the benefit of Landlord, Landlord's managing agent, or other Additional Insureds. Such insurance shall also waive any right of subrogation against each Additional Insured. Tenant shall obtain and submit to Landlord, prior to the earlier of (i) the entry onto the Premises by such contractors or subcontractors or (ii) commencement of the work or services, certificates of insurance evidencing compliance with the requirements of this section. All of such alterations shall be insured by Tenant pursuant to Article 13 of this Lease immediately upon completion thereof.
  - 15.6 Tenant's initial improvement of the Premises shall be governed by Exhibit C and not the provisions of this Article 15 (other than Section 15.2 and 15.5).
  - 15.7 Landlord is obligated to perform the initial build-out of the Premises in accordance with the Tenant Work Letter.

# ARTICLE 16 FURNITURE, FIXTURES, AND PERSONAL PROPERTY

- 16.1 Tenant, at its sole cost and expense, may remove its trade fixtures, office supplies and moveable office furniture and equipment not attached to the Project or Premises provided:
  - (a) Such removal is made prior to the Expiration Date or the Termination Date;
  - (b) No Event of Default exists under this Lease at the time of such removal; and
  - (c) Tenant promptly repairs all damage caused by such removal.

16.2 If Tenant does not remove its trade fixtures, office supplies, and moveable furniture and equipment as herein above provided prior to the Expiration Date or the Termination Date (unless prior arrangements have been made with Landlord and Landlord has agreed in writing to permit Tenant to leave such items in the Premises for an agreed period), then, in addition to its other remedies, at law or in equity, Landlord shall have the right to have such items removed and stored at Tenant's sole cost and expense and all damage to the Project or the Premises resulting from said removal shall be repaired at the cost of Tenant; Landlord may elect that such items automatically become the property of Landlord upon the Expiration Date or the Termination Date, and Tenant shall not have any further rights with respect thereto or reimbursement therefor subject to the provisions of Applicable Law. All other property in the Premises, any alterations, or additions to the Premises (including wall-to-wall carpeting, paneling, wall covering, specially constructed or built-in cabinetry or bookcases), and any other article attached or affixed to the floor, wall, or ceiling of the Premises shall become the property of Landlord and shall remain upon and be surrendered with the Premises as a part thereof at the Expiration or Termination Date regardless of who paid therefor; and Tenant hereby waives all rights to any payment or compensation therefor.

16.3 All the furnishings, fixtures, equipment, effects, and property of every kind, nature, and description of Tenant and of all persons claiming by, through, or under Tenant which, during the continuance of this Lease or any occupancy of the Premises by Tenant or anyone claiming under Tenant, may be on the Premises or elsewhere in the Project shall be at the sole risk and hazard of Tenant, and if the whole or any part thereof shall be destroyed or damaged by fire, water, or otherwise, or by the leakage or bursting of water pipes, steam pipes, or other pipes, by theft, or from any other cause, no part of said loss or damage is to be charged to or be borne by Landlord unless due to the gross negligence or willful misconduct of Landlord or its employees, agents or contractors.

# ARTICLE 17 PERSONAL PROPERTY AND OTHER TAXES

17.1 During the Term hereof, Tenant shall pay, prior to delinquency, all business and other taxes, charges, notes, duties, and assessments levied, and rates or fees imposed, charged, or assessed against or in respect of Tenant's occupancy of the Premises or in respect of the personal property, trade fixtures, furnishings, equipment, and all other personal and other property of Tenant contained in the Project (including without limitation taxes and assessments attributable to the cost or value of any leasehold improvements made in or to the Premises by or for Tenant (to the extent that the assessed value of those leasehold improvements exceeds the assessed value of standard office improvements in other space in the Project regardless of whether title to those improvements is vested in Tenant or Landlord)), and shall hold Landlord harmless from and against all payment of such taxes, charges, notes, duties, assessments, rates, and fees, and against all loss, costs, charges, notes, duties, assessments, rates, and fees, and any and all such taxes. Tenant shall cause said fixtures, furnishings, equipment, and other personal property to be assessed and billed separately from the real and personal property of Landlord. In the event any or all of Tenant's fixtures, furnishings, equipment, and other personal property shall be assessed and taxed with Landlord's real property, Tenant shall pay to Landlord Tenant's share of such taxes within ten (10) days after delivery to Tenant by Landlord of a statement in writing setting forth the amount of such taxes applicable to Tenant's

property. In addition, Tenant shall be liable for and shall pay ten (10) days before delinquency any (i) rent tax, gross receipts tax, or sales tax, service tax, transfer tax or value added tax, or any other applicable tax on the rent or services herein or otherwise respecting this Lease; or (ii) taxes assessed upon or with respect to the possession, leasing, operation, management, maintenance, alteration, repair, use or occupancy by Tenant of the Premises or any portion of the Project. If any of such taxes are billed to Landlord or included in bills to Landlord for taxes, then Tenant shall pay to Landlord all such amounts within thirty (30) days after receipt of Landlord's invoice therefor. If applicable law prohibits Tenant from reimbursing Landlord for any such taxes, but Landlord may lawfully increase the Base Rent to account for Landlord's payment of such taxes, the Base Rent payable to Landlord shall be increased to net to Landlord the same return without reimbursement of such Imposition as would have been received by Landlord with reimbursement of such taxes.

#### ARTICLE 18 ASSIGNMENT AND SUBLETTING

18.1 Except as set forth in Section 18.8 below, Tenant shall not, without the prior written consent of Landlord, which consent shall not be unreasonably withheld, conditioned or delayed beyond the express time periods provided for in this Article 18 (except that Landlord shall in no event be obligated to consent to an encumbrance of this Lease or any transfer by operation of law): (a) assign, convey, mortgage or otherwise transfer this Lease or any interest hereunder, or sublease the Premises, or any part thereof, whether voluntarily or by operation of law; or (b) permit the use of the Premises or any part thereof by any person other than Tenant and its employees. Any such transfer, sublease or use described in the preceding sentence (a "Transfer") occurring without the prior written consent of Landlord shall, at Landlord's option, be void and of no effect. Landlord's consent to any Transfer shall not constitute a waiver of Landlord's right to withhold its consent to any future Transfer. Landlord may require as a condition to its consent to any assignment of this Lease that the assignee execute an instrument in which such assignee assumes the remaining obligations of Tenant hereunder; provided that the acceptance of any assignment of this Lease by the applicable assignee shall automatically constitute the assumption by such assignee of all of the remaining obligations of Tenant that accrue following such assignment. The voluntary or other surrender of this Lease by Tenant or a mutual cancellation hereof shall not work a merger and shall, at the option of Landlord, terminate all or any existing sublease or may, at the option of Landlord, operate as an assignment to Landlord of Tenant's interest in any or all such subleases.

18.2 For purposes of this Lease, the term "Transfer" shall also include (i) if a Tenant is a partnership or limited liability company, the withdrawal or change, voluntary, involuntary or by operation of law, of fifty percent (50%) or more of the partners, members or managers thereof, or transfer of twenty-five percent (25%) or more of partnership or membership interests therein within a twelve (12) month period, or the dissolution of the partnership or the limited liability company without immediate reconstitution thereof, and (ii) if Tenant is a corporation whose stock is not publicly held and not traded through an exchange or over the counter or any other form of entity, (A) the dissolution, merger, consolidation or other reorganization of Tenant, the sale or other transfer of more than an aggregate of fifty percent (50%) of the voting shares or other interests of or in Tenant (other than to immediate family members by reason of gift or death), within a twelve (12) month period, or (B) the sale,

mortgage, hypothecation or pledge of more than an aggregate of fifty percent (50%) of the value of the unencumbered assets of Tenant within a twelve (12) month period.

18.3 If Tenant desires the consent of Landlord to a Transfer, Tenant shall submit to Landlord, at least thirty (30) days prior to the proposed effective date of the Transfer, a written notice (the "Transfer Notice") which includes (a) the name of the proposed sublessee or assignee, (b) the nature of the proposed sublessee's or assignee's business, (c) the terms and provisions of the proposed sublease or assignment, and (d) current financial statements and information on the proposed sublessee or assignee. Upon receipt of the Transfer Notice, Landlord may request additional information concerning the Transfer or the proposed sublessee or assignee (the "Additional Information"). Subject to Landlord's rights under Section 18.6, Landlord shall not unreasonably withhold its consent to any assignment or sublease (excluding an encumbrance or transfer by operation of law), which consent or lack thereof shall be provided within thirty (30) days of receipt of Tenant's Transfer Notice; provided, however, Tenant hereby agrees that it shall be a reasonable basis for Landlord to withhold its consent if Landlord has not received the Additional Information requested by Landlord. Without limiting any other reasonable basis for Landlord to withhold its consent to the proposed Transfer, Landlord and Tenant agree that for purposes of this Lease and any Applicable Law, Landlord shall not be deemed to have unreasonably withheld its consent if, in the judgment of Landlord: (i) the transferee is of a character or engaged in a business which is not in keeping with the standards or criteria used by Landlord in leasing the Project, or the general character or quality of the Project; (ii) the financial condition of the transferee is such that it may not be able to perform its obligations in connection with this Lease (or otherwise does not satisfy Landlord's standards for financial standing with respect to tenants under direct leases of comparable economic scope); (iii) the transferee, or any person or entity which directly or indirectly controls, is controlled by, or is under common control with, the transferee, is a tenant of or negotiating for space in the Project occupies space in the Project or has negotiated with Landlord within the preceding ninety (90) days (or is currently negotiating with Landlord) to lease space in the Project and Landlord then has space available in the Building meeting the requirements of the Transferee with respect to size, location and term (provided, however, that existing tenants occupying space on the same floor as the part of the Premises proposed to be subleased, or an immediately adjacent floor, shall be excluded from the foregoing provision unless Landlord has other space then available for lease on such floor or a floor immediately adjoining a floor on which such existing tenant's premises are located); (iv) the transferee has the power of eminent domain, is a governmental agency or an agency or subdivision of a foreign government; (v) an Event of Default by Tenant has occurred and is uncured at the time Tenant delivers the Transfer Notice to Landlord; (vi) in the judgment of Landlord, such a Transfer would violate any term, condition, covenant, or agreement of Landlord involving the Project or any other tenant's lease within it or would give an occupant of the Project a right to cancel or modify its lease; (vii) [intentionally omitted]; (viii) in Landlord's judgment, the use of the Premises by the proposed transferee would not be comparable to that of Tenant, would result in more density of occupants per square foot of the Premises over the density of Tenant's use prior to the proposed Transfer, would increase the burden on elevators or other Building systems or equipment over the burden thereon prior to the proposed Transfer, would require increased services by Landlord or would require any alterations to the Project to comply with applicable laws; (ix) the transferee intends to use the space for purposes which are not permitted under this Lease; (x) the terms of the proposed Transfer would allow the transferee (other than a Permitted Transferee who is an assignee of

Tenant's entire interest in this Lease) to exercise a right of renewal, right of expansion, right of first offer, or other similar right held by Tenant (or will allow the transferee (other than a Permitted Transferee who is an assignee of Tenant's entire interest in this Lease) to occupy space leased by Tenant pursuant to any such right); (xi) the proposed Transfer would result in more than three subleases per each full floor of the Premises being in effect at any one time during the Term; or (xii) any ground lessor or mortgagee whose consent to such Transfer is required fails to consent thereto. Tenant hereby waives any right to terminate the Lease and/or recover damages as remedies for Landlord wrongfully withholding its consent to any Transfer and agrees that Tenant's sole and exclusive remedy therefor shall be to seek specific performance of Landlord's obligation to consent to such Transfer.

18.4 Landlord and Tenant agree that, in the event of any approved assignment or subletting, the rights of any such assignee or sublessee of Tenant herein shall be subject to all of the terms, conditions, and provisions of this Lease, including, without limitation, restriction on use, assignment, and subletting and the covenant to pay Rent. During any period in which an Event of Default exists and remains uncured, Landlord may collect the rent owing by the assignee or sublessee directly from such assignee or sublessee and apply the amount so collected to the Rent herein reserved. No such consent to or recognition of any such assignment or subletting shall constitute a release of Tenant or any guarantor of Tenant's performance hereunder from further performance by Tenant or such guarantor of covenants undertaken to be performed by Tenant herein. Tenant and any such guarantor shall remain liable and responsible for all Rent and other obligations herein imposed upon Tenant, and Landlord may condition its consent to any Transfer upon the receipt of a written reaffirmation from each such guarantor in a form acceptable to Landlord (which shall not be construed to imply that the occurrence of a Transfer without such a reaffirmation would operate to release any guarantor). Consent by Landlord to a particular assignment, sublease, or other transaction shall not be deemed a consent to any other or subsequent transaction. In any case where Tenant desires to assign, sublease or enter into any related or similar transaction, whether or not Landlord consents to such assignment, sublease, or other transaction, Tenant shall pay any reasonable attorneys' fees incurred by Landlord in connection with such assignment, sublease or other transaction, including, without limitation, fees incurred in reviewing documents relating to, or evidencing, said assignment, sublease, or other transaction; provided that those costs shall not exceed \$2,500.00 with respect to any single Transfer so long as Tenant and the proposed transferee execute

18.5 Tenant shall be bound and obligated to pay Landlord a portion of any sums or economic consideration payable to Tenant by any sublessee, assignee, licensee, or other transferee, within ten (10) days following receipt thereof by Tenant from such sublessee, assignee, licensee, or other transferee, as the case might be, as follows:

(a) In the case of an assignment, fifty percent (50%) of any sums or other economic consideration received by Tenant as a result of such assignment shall be paid to Landlord, after first deducting reasonable costs incurred by Tenant in connection with the Transfer for the following: (i) broker fees, (ii) legal fees incurred in connection

with the negotiation and documentation of the Transfer, (iii) accounting fees incurred in connection with the negotiation and documentation of the Transfer, (iv) costs of tenant improvements constructed by Tenant (with Landlord's prior consent) in connection with the Transfer (without reimbursement out of any allowance provided by Landlord), (v) reasonable tenant improvement allowances granted by Tenant (without reimbursement out of any allowance provided by Landlord), and (vi) reasonable rent abatements or concessions granted by Tenant (collective, the "Transaction Costs"). Economic consideration, however, shall exclude any consideration received by Tenant pursuant to a sale, merger or consolidation of Tenant with another entity.

- (b) In the case of a subletting, fifty percent (50%) of any sums or economic consideration received by Tenant as a result of such subletting shall be paid to Landlord after first deducting (i) the Rent due hereunder prorated to reflect only Rent allocable to the sublet portion of the Premises, (ii) the Transaction Costs amortized over the term of the Sublease.
- (c) Tenant shall provide Landlord with a detailed statement setting forth any sums or economic consideration Tenant either has or will derive from such Transfer, the deductions permitted under (a) and (b) of this Section 18.5, and the calculation of the amounts due Landlord under this Section 18.5. In addition, Landlord or its representative shall have the right at all reasonable times to audit the books and records of Tenant with respect to the calculation of the Transfer profits. If such inspection reveals that the amount paid to Landlord was incorrect, then within ten (10) days of Tenant's receipt of the results of such audit, Tenant shall pay Landlord the deficiency and the cost of Landlord's audit.
- 18.6 If this Lease is assigned to any person or entity pursuant to the provisions of the Bankruptcy Code, 11 U.S.C. Section 101 et seq. or any successor or substitute therefor (the "Bankruptcy Code"), any and all monies or other consideration payable or otherwise to be delivered in connection with such assignment shall be paid or delivered to Landlord, shall be and remain the exclusive property of Landlord, and shall not constitute property of Tenant or of the estate of Tenant within the meaning of the Bankruptcy Code. Any such monies or other consideration not paid or delivered to Landlord shall be held in trust for the benefit of Landlord and shall be promptly paid or delivered to Landlord. Any person or entity to whom this Lease is so assigned shall be deemed, without further act or deed, to have assumed all of the remaining obligations arising under this Lease as of the date of such assignment. Any such assignee shall, upon demand therefor, execute and deliver to Landlord an instrument confirming such assumption.
  - 18.7 Landlord shall have the following option with respect to any assignment of this Lease or a Triggering Sublease (as defined below) proposed by Tenant:
  - (a) Notwithstanding any other provision of this Article, Landlord has the option, by written notice to Tenant (the "Recapture Notice") within thirty (30) days after receiving any Transfer Notice to recapture the space covered by the proposed sublease or the entire Premises in the case of an assignment (the "Subject Space") by terminating this Lease for the Subject Space or taking an assignment or a sublease of the

Subject Space from Tenant. A timely Recapture Notice terminates this Lease or creates an assignment or a sublease for the Subject Space for the same term as the proposed Transfer, effective as of the date specified in the Transfer Notice. After such termination, Landlord may (but shall not be obligated to) enter into a lease with the party to the sublease or assignment proposed by Tenant. As used herein, "Triggering Subletting" means subleasing of seventy-five percent (75%) or more of the Premises, either in a single transaction or, in the aggregate, following a series of transactions, for a term or terms expiring during the last six (6) months of the Term.

(b) To determine the new Base Rent under this Lease in the event Landlord recaptures the Subject Space without terminating this Lease, the original Base Rent under the Lease shall be multiplied by a fraction, the numerator of which is the rentable square feet of the Premises retained by Tenant after Landlord's recapture and the denominator of which is the total rentable square feet in the Premises before Landlord's recapture. The Additional Rent, to the extent that it is calculated on the basis of the rentable square feet within the Premises, shall be reduced to reflect Tenant's proportionate share based on the rentable square feet of the Premises retained by Tenant after Landlord's recapture. This Lease as so amended shall continue thereafter in full force and effect. Either party may require a written confirmation of the amendments to this Lease necessitated by Landlord's recapture of the Subject Space. If Landlord recaptures the Subject Space, Landlord shall, at Landlord's sole expense, construct any partitions required to segregate the Subject Space from the remaining Premises retained by Tenant. Tenant shall, however, pay for painting, covering or otherwise decorating the surfaces of the partitions facing the remaining Premises retained by Tenant.

18.8 Notwithstanding anything to the contrary in this Article 18. Tenant may Transfer all or part of its interest in this Lease or all or part of the Premises (a "Permitted Transfere") to the following types of entities (a "Permitted Transferee") without the written consent of Landlord: (a) any parent, subsidiary or affiliate corporation which Controls (as defined below), is Controlled by or is under common Control with Tenant (collectively, an "Affiliate"); (b) any corporation, limited partnership, limited liability company or other business entity in which or with which Tenant, an Affiliate of Tenant, or their respective corporate successors or assigns, is merged or consolidated, in accordance with applicable statutory provisions governing merger and consolidation of business entities, so long as in both cases (a) and (b), (i) Tenant's obligations hereunder are assumed by the Permitted Transferee; and (ii) the Permitted Transfers the Net Worth Threshold as of the effective date of the Permitted Transfer; or (c) any corporation, limited partnership, limited liability partnership, limited liability company or other business entity which acquires all or substantially all of Tenant's assets and/or ownership interests, if the Transferee satisfies the Net Worth Threshold as of the effective date of the Transfer. Tenant shall remain liable for the performance of all of the obligations of Tenant hereunder, or if Tenant no longer exists because of a merger, consolidation, or acquisition, the surviving or acquiring entity shall expressly assume in writing, the obligations of Tenant hereunder. Additionally, the Permitted Transferee shall comply with all of the terms and conditions of this Lease, whether accruing prior to and/or from and after the consummation of the Transfer. No later than ten (10) days prior to the effective date of any Permitted Transfer, Tenant agrees to furnish Landlord with (1) copies of the instrument effecting any of the

foregoing Transfers, (2) documentation establishing Tenant's satisfaction of the requirements set forth above applicable to any such Transfer, and (3) evidence of insurance as required under this Lease with respect to the Permitted Transferee. The occurrence of a Permitted Transfer shall not waive Landlord's rights as to any subsequent Transfers. As used herein, the term "Net Worth Threshold" shall mean the proposed Permitted Transferee has a tangible net worth equal to or greater than (x) that of Tenant immediately prior to such transaction, and (y) that of the originally named Tenant as of December 31 of the year prior to the Commencement Date (determined in accordance with generally accepted accounting principles consistently applied and excluding from the determination of total assets all assets which would be classified as intangible assets under generally accepted accounting principles, including, without limitation, goodwill, licenses, tradenarks, trade names, copyrights and franchises), and as evidenced by financial statements audited by a certified public accounting firm reasonably acceptable to Landlord. The term "Control" shall mean the possession of the power to direct or cause the direction of the management and policy of such corporation, partnership, limited liability company or other entity, whether through the ownership of voting securities, by statute or by contract, and whether directly or indirectly through Affiliates. The provisions of Sections 18.3, 18.5 and 18.7 shall not apply to any Permitted Transfer.

### ARTICLE 19 DAMAGE OR DESTRUCTION

19.1 If the Premises or Building should be damaged or destroyed by fire or other casualty, Tenant shall give immediate written notice to Landlord. If the Premises or any common areas of the Building or Project serving or providing access to the Premises shall be damaged by fire or other casualty, Landlord shall promptly and diligently, subject to reasonable delays for insurance adjustment or other matters beyond Landlord's reasonable control, and subject to all other terms of this Article 19, restore the base, shell, and core of the Premises and such common areas. Such restoration shall be to substantially the same condition of the base, shell, and core of the Premises and common areas prior to the casualty, except for modifications required by zoning and building codes and other laws or by the holder of a mortgage on the Project, or the lessor of a ground or underlying lease with respect to the Project and/or the Building, or any other modifications to the common areas deemed desirable by Landlord, provided access to the Premises and any common restrooms serving the Premises shall not be materially impaired. Landlord shall not be liable for any inconvenience or annoyance to Tenant or its visitors, or injury to Tenant's business resulting in any way from such damage or the repair thereof; provided however, that if such fire or other casualty shall have damaged the Premises or common areas necessary to Tenant's occupancy, Landlord shall allow Tenant a proportionate abatement of Base Rent and Tenant's Share of Operating Expenses and Tenant's Tax Share of Taxes, during the time and to the extent the Premises are unfit for occupancy for the purposes permitted under this Lease, and not occupied by Tenant as a result thereof; provided, however, if the damage or destruction was caused by the negligence or willful misconduct of Tenant or Tenant's employees, contractors, licensees or invitees, such abatement shall occur only to the extent rental abatement insurance proceeds are received by Landlord (or would have been r

19.2 Within sixty (60) days following the date of discovery of the damage, Landlord shall deliver to Tenant a written estimate from Landlord's contractor of the time needed to rebuild and/or restore the Premises and/or the Building (the "Restoration Notice"). Notwithstanding the terms of Section 19.1 of this Lease, Landlord may elect not to rebuild and/or restore the Premises, the Building and/or any other portion of the Real Property and instead terminate this Lease by notifying Tenant in writing of such termination within sixty (60) days after the date of Landlord's discovery of such damage (the "Damage Discovery Date"), such notice to include a termination date giving Tenant ninety (90) days to vacate the Premises, but Landlord may so elect only if the Building shall be damaged by fire or other casualty or cause, whether or not the Premises are affected, and one or more of the following conditions is present: (i) repairs cannot reasonably be completed within two hundred seventy (270) days of the Damage Discovery Date (when such repairs are made without the payment of overtime or other premiums); (ii) the holder of any mortgage on the Project or ground or underlying lessor with respect to the Project and/or the Building shall require that the insurance proceeds or any portion thereof be used to retire the mortgage debt, or shall terminate the ground or underlying lease, as the case may be; or (iii) the damage is not fully covered (except for deductible amounts in the case of a casualty other than earthquake or flood, and except for deductible amounts under earthquake or flood insurance that are less than 10% of the replacement cost of the restoration of the damage) by Landlord's insurance policies. In addition, if the Premises or the Building is destroyed or damaged to any substantial extent during the last twelve (12) months of the Term, then notwithstanding anything contained in this Article 19, Landlord shall have the option to terminate his Lease by giving written notice to Tenant of the exercise of such

19.3 If there is an occurrence of any damage to the Premises that does not result in the termination of this Lease pursuant to this Article 19, then upon notice (the "Landlord Repair Notice") to Tenant from Landlord, Tenant shall assign to Landlord (or to any party designated by Landlord) all insurance proceeds payable to Tenant under Tenant's insurance required under Sections 13.2(e)(ii) and (iii) above with respect to any improvements in the Premises required to be insured by Tenant hereunder (excluding proceeds for Tenant's Property), and Landlord shall repair any injury or damage to the Tenant Improvements, alterations and the Original Improvements installed in the Premises and shall return such Tenant Improvements, alterations and Original Improvements to their original condition; provided that if the cost of such repair by Landlord exceeds the sum of (A) amount of insurance proceeds received by Landlord from Tenant's insurance carrier, as assigned by Tenant, plus (B) any insurance proceeds received by Landlord with respect to such Tenant Improvements, alterations and Original Improvements (it being acknowledged and agreed that Tenant's insurance as to the Tenant Improvements, Alterations and Original Improvements is primary in nature and Landlord's insurance, if any, with respect to same is secondary in nature), the cost of such repairs shall be paid by Tenant to Landlord prior to Landlord's commencement of repair of the damage. In the event that Landlord does not deliver the Landlord Repair Notice within forty-five (45) days following the Damage Discovery Date, Tenant shall, at its sole cost and expense, repair any

injury or damage to the Tenant Improvements, alterations, and the Original Improvements installed in the Premises and shall return such Tenant Improvements, alterations, and Original Improvements to their original condition. Whether or not Landlord delivers a Landlord Repair Notice, prior to the commencement of construction, Tenant shall submit to Landlord, for Landlord's review and approval, all plans, specifications and working drawings relating thereto, and Landlord shall select the contractors to perform such improvement work.

19.4 If (i) Landlord does not elect to terminate this Lease pursuant to Landlord's termination right as provided hereinabove, (ii) the damage constitutes a Tenant Damage Event (as defined below, and either (a) the repairs cannot, in the reasonable opinion of Landlord's contractor, be completed within two hundred seventy (270) days after the Damage Discovery Date, or (b) the damage occurs during the last twelve (12) months of the Term and will reasonably require in excess of ninety (90) days to repair, Tenant may elect, no earlier than sixty (60) days after the Damage Discovery Date and not later than ninety (90) days after the Damage Discovery Date (or within thirty (30)) days after receipt of the Restoration Notice, if later), to terminate this Lease by written notice to Landlord effective as of the date specified in the notice, which date shall not be less than thirty (30) days nor more than sixty (60) days after the date such notice is given by Tenant (prior to which Tenant shall be entitled to an abatement of Rent as provided in Section 19.1). In the event of a Tenant Damage Event, and if the Lease does not terminate pursuant to the other provisions of this Article 19, then if Landlord fails to substantially complete the repair of such damage comprising a Tenant Damage Event on or before the date (the "Outside Restoration Completion Date") which is three (3) months after the date estimated for completion of such repair by Landlord's contractor in the Restoration Notice, then Tenant shall have the option, exercisable by written notice to Landlord within thirty (30) days after the Outside Restoration Completion Date, to terminate this Lease ("Tenant's Second Termination Option"). The Outside Restoration Completion Date shall be extended by delays in the completion of the repair of the damage comprising a Tenant Damage Event to the extent caused by Force Majeure Events (other than the casualty that caused the damage) for up to ninety (90) days or by Tenant, its agents, employees or contractors. If Tenant exercises Tenant's Second Termination Option, the Lease shall terminate as of a date specified in Tenant's termination notice which is not less than thirty (30) days nor more than sixty (60) days after Tenant's notice to Landlord of the exercise of Tenant's Second Termination Option. As used herein, a "Tenant Damage Event" shall mean damage to all or any part of the Premises or any Common Areas necessary to Tenant's occupancy of the Premises by fire or other casualty, which damage (A) is not the result of the willful misconduct of Tenant or any of the Tenant Parties, and (B) substantially interferes with Tenant's use of or access to the Premises.

19.5 In the event this Lease is terminated in accordance with the terms of this <u>Article 19</u>, Tenant shall assign to Landlord (or to any party designated by Landlord) all insurance proceeds payable to Tenant under Tenant's insurance required under Sections 13.2(e)(ii) and (iii).

19.6 The provisions of this Lease, including this <u>Article 19</u>, constitute an express agreement between Landlord and Tenant with respect to damage to, or destruction of, all or any portion of the Premises or the Project, and any statute or regulation of the State of California, including without limitation Sections 1932(2) and 1933(4) of the California Civil Code, with respect to any rights or obligations concerning damage or destruction in the absence of an express agreement between the parties (and any other statute or regulation now or hereafter

in effect with respect to such rights or obligations), shall have no application to this Lease or to any damage or destruction to all or any portion of the Premises or the Project.

#### ARTICLE 20 CONDEMNATION

- 20.1 If all of the Premises is condemned by eminent domain, inversely condemned or sold under threat of condemnation for any public or quasi-public use or purpose ("Condemned"), this Lease shall terminate as of the earlier of the date the condemning authority takes title to or possession of the Premises, and Rent shall be adjusted to the date of termination.
- 20.2 If any portion of the Premises or Building is condemned and such partial condemnation materially impairs Tenant's ability to use the Premises for Tenant's business, Landlord shall have the option in Landlord's sole and absolute discretion of either (i) relocating Tenant to comparable space within the Project at Landlord's sole cost, which space shall be acceptable to Tenant in its sole discretion or (ii) terminate this Lease as of the earlier of the date title vests in the condemning authority or as of the date an order of immediate possession is issued and Rent shall be adjusted to the date of termination. If such partial condemnation does not materially impair Tenant's ability to use the Premises for the business of Tenant, Landlord shall promptly restore the Premises to the extent of any condemnation proceeds recovered by Landlord, excluding the portion thereof lost in such condemnation, and this Lease shall continue in full force and effect. In the event the Premises are not substantially restored within 270 days after the taking, then Tenant may elect to terminate this Lease within 30 days after the expiration of such 270 day period. If any part of the Premises shall be taken, and this Lease shall not be terminated as provided in this Article 20, the Rent shall be proportionately abated.
- 20.3 If there shall be taken by exercise of the power of eminent domain, or by conveyance in lieu, during the Term, more than twenty-five percent (25%) of the Rentable Area of the Premises such that it objectively prevents the conduct of Tenant's business in the Premises and Landlord has not provided replacement space acceptable to Tenant in its sole discretion, Tenant may terminate this Lease by giving Landlord notice to that effect within thirty (30) days after the date of the taking or transfer in lieu thereof.
- 20.4 If the Premises are wholly or partially condemned, Landlord shall be entitled to the entire award paid for such condemnation, and Tenant waives any claim to any part of the award from Landlord or the condemning authority; provided, however, Tenant shall have the right to recover from the condemning authority such compensation as may be separately awarded to Tenant in connection with costs in removing Tenant's merchandise, furniture, fixtures, leasehold improvements and equipment to a new location. No condemnation of any kind shall be construed to constitute an actual or constructive eviction of Tenant or a breach of any express or implied covenant of quiet enjoyment. Tenant hereby waives the effect of Sections 1265.120 and 1265.130 of the California Code of Civil Procedure.
- 20.5 In the event of a temporary condemnation not extending beyond the Term, this Lease shall remain in effect, Tenant shall continue to pay Rent and Tenant shall receive any award made for such condemnation except damages to any of Landlord's property. If a temporary condemnation is for a period which extends beyond the Term, this Lease shall

terminate as of the date of initial occupancy by the condemning authority and any such award shall be distributed in accordance with the preceding section. If a temporary condemnation remains in effect at the expiration or earlier termination of this Lease, Tenant shall pay Landlord the reasonable cost of performing any obligations required of Tenant with respect to the surrender of the Premises.

#### ARTICLE 21 HOLD HARMLESS

- 21.1 Tenant agrees to defend, with counsel approved by Landlord, all actions against Landlord, any member, partner, trustee, stockholder, officer, director, employee, or beneficiary of Landlord (collectively, "Landlord Parties"), holders of mortgages secured by the Premises or the Project and any other party having an interest therein (collectively with Landlord Parties, the "Indemnified Parties") with respect to, and to pay, protect, indemnify, and save harmless, to the extent permitted by law, all Indemnified Parties from and against, any and all liabilities, losses, damages, costs, expenses (including reasonable attorneys' fees and expenses), causes of action, suits, claims, demands, or judgments of any nature to which any Indemnified Party is subject because of its estate or interest in the Premises or the Project arising from (a) injury to or death of any person, or damage to or loss of property on the Premises, the Project, on adjoining sidewalks, streets or ways, or, in any of the foregoing cases, connected with the use, condition, or occupancy of the Premises, the Wall (as defined below), the Project sidewalks streets, or ways, except to the extent, if any, caused by the gross negligence or willful misconduct of Landlord or its employees, contractors or agents, (b) any violation of this Lease by or attributable to Tenant, or (c) subject to Section 13.4, any act, fault, omission, or other misconduct of Tenant or its agents, contractors, licensees, sublessees, or invitees. Tenant agrees to use and occupy the Premises, the Wall and other facilities of the Project at its own risk, and hereby releases the Indemnified Parties from any and all claims for any damage or injury to the fullest extent permitted by law.
- 21.2 Tenant agrees that Landlord shall not be responsible or liable to Tenant, its agents, employees, or invitees for fatal omon-fatal bodily injury or property damage occasioned by the acts or omissions of any other tenant, or such other tenant's agents, employees, licensees, or invitees, of the Project. Landlord shall not be liable to Tenant for losses to property due to theft or burglary, or damages from criminal acts, done by any persons on the Project other than Landlord or its employees or agents.
- 21.3 Subject to Section 13.4, Landlord shall indemnify and hold harmless Tenant and its agents, directors, officers, shareholders, partners, members, employees and invitees, from all liability for any loss of or damage or injury to any person (including death resulting therefrom) or property occurring in, on or about the Project excluding the Premises, to the extent caused by the negligence or willful misconduct of Landlord or Landlord's employees, agents, or contractors and not insured (or required to be insured) by Tenant under this Lease (provided, however, that Landlord's indemnity shall, in no event, extend to loss of profits, loss of business or other consequential damages incurred by Tenant).

## ARTICLE 22 DEFAULT BY TENANT

- 22.1 The term "Event of Default" refers to the occurrence of any one (1) or more of the following:
- (a) Failure of Tenant to pay when due any sum required to be paid hereunder within five (5) days of receipt of written notice from Landlord (the "Monetary Default"); provided that Tenant acknowledges that any such written notice provided hereunder shall be in lieu of, and not in addition to, any notice to pay rent or quit pursuant to any applicable statutes;
- (b) Failure of Tenant, after thirty (30) days written notice thereof, to perform any of Tenant's obligations, covenants, or agreements except a Monetary Default, provided that if the cure of any such failure is not reasonably susceptible of performance within such thirty (30) day period, then an Event of Default of Tenant shall not be deemed to have occurred so long as Tenant has promptly commenced and thereafter diligently prosecutes such cure to completion;
- (c) Tenant, or any guarantor of Tenant's obligations under this Lease (the "Guarantor"), admits in writing that it cannot meet its obligations as they become due; or is declared insolvent according to any law; or assignment of Tenant's or Guarantor's property is made for the benefit of creditors; or a receiver or trustee is appointed for Tenant or Guarantor or its property; or the interest of Tenant or Guarantor under this Lease is levied on under execution or other legal process; or any petition is filed by or against Tenant or Guarantor to declare Tenant bankrupt or to delay, reduce, or modify Tenant's debts or obligations; or any petition filed or other action taken to reorganize or modify Tenant's or Guarantor's capital structure if Tenant is a corporation or other entity. Any such levy, execution, legal process, or petition filed against Tenant or Guarantor shall not constitute a breach of this Lease provided Tenant or Guarantor shall vigorously contest the same by appropriate proceedings and shall remove or vacate the same within ninety (90) days from the date of its creation, service, or filing;
  - (d) The abandonment (as defined in California Civil Code section 1951.3) of the Premises by Tenant;
- (e) The discovery by Landlord that any financial statement given by Tenant or any of its assignees, successors-in-interest, or Guarantors was actually known by Tenant to be materially false when given; or
- (f) If Tenant or any Guarantor shall die, cease to exist as a corporation or partnership, or be otherwise dissolved or liquidated or become insolvent, or shall make a transfer in fraud of creditors.
- 22.2 In the event of any Event of Default by Tenant, Landlord, at its option, may pursue one or more of the following remedies without notice or demand in addition to all other rights and remedies provided for at law or in equity:

(a) Landlord may continue this Lease in full force and effect, and this Lease shall continue in full force and effect as long as Landlord does not terminate Tenant's right to possession, and Landlord shall have the right to collect Rent when due. Landlord may enter the Premises and relet it, or any part of it, to third parties for Tenant's account, provided that any Rent in excess of the Rent due hereunder shall be payable to Landlord. Tenant shall be liable immediately to Landlord for all costs Landlord incurs in reletting the Premises, including, without limitation, brokers' commissions, expenses of cleaning and redecorating the Premises required by the reletting and like costs. Reletting may be for a period shorter or longer than the remaining Term of this Lease. Tenant shall pay to Landlord the Rent and other sums due under this Lease on the dates the Rent is due, less the Rent and other sums Landlord receives from any reletting. No act by Landlord allowed by this Section 22.2(a) shall terminate this Lease unless Landlord notifies Tenant in writing that Landlord elects to terminate this Lease.

"The lessor has the remedy described in Civil Code Section 1951.4 (lessor may continue the lease in effect after lessee's breach and abandonment and recover rent as it becomes due, if lessee has the right to sublet or assign subject only to reasonable limitations)."

(b) Landlord may terminate Tenant's right to possession of the Premises at any time by giving written notice to that effect. No act by Landlord other than giving written notice to Tenant shall terminate this Lease. Acts of maintenance, efforts to relet the Premises or the appointment of a receiver on Landlord's initiative to protect Landlord's interest under this Lease shall not constitute a termination of Tenant's right to possession. On termination, Landlord shall have the right to remove all personal property of Tenant and store it at Tenant's cost and to recover from Tenant as damages: (i) the worth at the time of award of unpaid Rent and other sums due and payable which had been earned at the time of termination; plus (ii) the worth at the time of award of the amount by which the unpaid Rent and other sums due and payable which would have been payable after termination until the time of award exceeds the amount of the Rent loss that Tenant proves could have been reasonably avoided; plus (iii) the worth at the time of award of the amount by which the unpaid Rent and other sums due and payable for the balance of the Term after the time of award exceeds the amount of the Rent loss that Tenant proves could be reasonably avoided; plus (iv) any other amount necessary to compensate Landlord for all the detriment proximately caused by Tenant's failure to perform Tenant's obligations under this Lease, or which, in the ordinary course of things, would be likely to result therefrom, including, without limitation, any costs or expenses incurred by Landlord: (A) in retaking possession of the Premises, including reasonable attorneys' fees and costs therefor; (B) maintaining or preserving the Premises for reletting to a new tenant, including repairs or alterations to the Premises for the reletting; (C) leasing commissions; (D) any other costs necessary or appropriate to relet the Premises; and (E) at Landlord's election, such other amounts in addition to or in lieu of the foregoing as may be permitted from time to time b

The "worth at the time of award" of the amounts referred to in Sections 22.2(b)(i) and 22.2(b)(ii) shall be calculated by allowing interest at the lesser of ten percent (10%) per annum or the

maximum rate permitted by law, on the unpaid Rent and other sums due and payable from the termination date through the date of award. The "worth at the time of award" of the amount referred to in Section 22.2(b)(iii) shall be calculated by discounting the amount at the discount rate of the Federal Reserve Bank of San Francisco at the time of award, plus one percent (1%). Tenant waives redemption or relief from forfeiture under California Code of Civil Procedure Sections 1174 and 1179, or under any other present or future law, if Tenant is evicted or Landlord takes possession of the Premises by reason of any Event of Default by Tenant.

- 22.3 If Landlord shall exercise any one or more remedies hereunder granted or otherwise available, it shall not be deemed to be an acceptance or surrender of the Premises by Tenant whether by agreement or by operation of law; it is understood that such surrender can be effected only by the written agreement of Landlord and Tenant. No alteration of security devices and no removal or other exercise of dominion by Landlord over the property of Tenant or others in the Premises shall be deemed unauthorized or constitute a conversion, Tenant hereby consenting to the aforesaid exercise of dominion over Tenant's property within the Premises after any Event of Default.
- 22.4 Each right and remedy provided for in this Lease shall be cumulative and shall be in addition to every other right or remedy provided for in this Lease or now or hereafter existing at law or in equity or by statute or otherwise, including, but not limited to, suits for injunctive relief and specific performance. The exercise or beginning of the exercise by Landlord of any one or more of the rights or remedies provided for in this Lease or now or hereafter existing at law or in equity, or by statute or otherwise shall not preclude the simultaneous or later exercise by Landlord for any or all other rights or remedies provided for in this Lease or now or hereafter existing at or in equity or by statute or otherwise. All such rights and remedies shall be considered cumulative and non-exclusive. All costs incurred by Landlord in connection with collecting any Rent or other amounts and damages owing by Tenant pursuant to the provisions of this Lease, or to enforce any provision of this Lease, including reasonable attorneys' fees from the date such matter is turned over to an attorney, whether or not one or more actions are commenced by Landlord, shall also be recoverable by Landlord from Tenant. If any notice and grace period required under subparagraphs 22.1(a) or (b) was not previously given, a notice to pay rent or quit, or to perform or quit, as the case may be, given to Tenant under any statute authorizing the forfeiture of leases for unlawful detainer shall also constitute the applicable notice for grace period purposes required by subparagraphs 22.1(a) or (b). In such case, the applicable grace period under subparagraphs 22.1(a) or (b) and under the unlawful detainer statute shall run concurrently after the one such statutory notice, and the failure of Tenant to cure the default within the greater of the two (2) such grace periods shall constitute both an unlawful detainer and an Event of Default entitling Landlord to the remedies provided for in this Lease and/or by said statute.
- 22.5 If Tenant should fail to make any payment or cure any default hereunder within the time herein permitted and such failure constitutes an Event of Default (except in the case where if Landlord in good faith believes that action prior to the expiration of any cure period under Section 22.1 is necessary to prevent damage to persons or property, in which case Landlord may act without waiting for such cure period to expire), Landlord, without being under any obligation to do so and without thereby waiving such default, may make such payment and/or remedy such default for the account of Tenant (and enter the Premises for such purpose),

and thereupon, Tenant shall be obligated and hereby agrees to pay Landlord, upon demand, all reasonable costs, expenses, and disbursements, plus ten percent (10%) overhead cost incurred by Landlord in connection therewith.

- 22.6 [Intentionally Omitted.]
- 22.7 Nothing contained in this Article 22 shall limit or prejudice the right of Landlord to prove and obtain as damages in any bankruptcy, insolvency, receivership, reorganization, or dissolution proceeding, an amount equal to the maximum allowed by any statute or rule of law governing such a proceeding and in effect at the time when such damages are to be proved, whether or not such amount be greater, equal, or less than the amounts recoverable, either as damages or Rent, referred to in any of the preceding provisions of this Article 22. Notwithstanding anything contained in this Article to the contrary, any such proceeding or action involving bankruptcy, insolvency, reorganization, arrangement, assignment for the benefit of creditors, or appointment of a receiver or trustee, as set forth above, shall be considered to be an Event of Default only when such proceeding, action, or remedy shall be taken or brought by or against the then holder of the leasehold estate under this Lease.
- 22.8 Landlord is entitled to accept, receive, in check or money order, and deposit any payment made by Tenant for any reason or purpose or in any amount whatsoever, and apply them at Landlord's option to any obligation of Tenant, and such amounts shall not constitute payment of any amount owed, except that to which Landlord has applied them. No endorsement or statement on any check or letter of Tenant shall be deemed an accord and satisfaction or recognized for any purpose whatsoever. The acceptance of any such check or payment shall be without prejudice to Landlord's rights to recover any and all amounts owed by Tenant hereunder and shall not be deemed to cure any other default nor prejudice Landlord's rights to pursue any other available remedy. Landlord's acceptance of partial payment of Rent does not constitute a waiver of any rights, including without limitation any right Landlord may have to recover possession of the Premises.
- 22.9 In the event that Tenant's right of possession of the Premises is terminated prior to the end of the initial Term by reason of an Event of Default by Tenant, then immediately upon such termination, an amount shall be due and payable by Tenant to Landlord equal to the unamortized portion as of that date (which amortization shall be based on an interest rate of eight percent (8%) per annum and amortized over the initial Term of the Lease) of the sum of (a) the cost of Landlord's Work (if any), (b) the Allowance (if any), (c) the value of any free Base Rent (i.e., the Base Rent stated in this Lease to be abated as an inducement to Tenant's entering into this Lease) enjoyed as of that date by Tenant, and (d) the amount of all commissions paid by Landlord in order to procure this Lease.
- 22.10 Tenant waives the right to terminate this Lease on Landlord's default under this Lease. Tenant's sole remedy on Landlord's default is an action for damages or injunctive or declaratory relief. The foregoing limitation shall not apply if, as determined in a final judgment by a court having jurisdiction, Landlord's default is so material as to render the Premises unfit for occupancy or prevent Tenant's use of the Premises. Landlord's failure to perform any of its obligations under this Lease shall constitute a default by Landlord under this Lease if the failure continues for thirty (30) days after written notice of the failure from Tenant to

Landlord. If the required performance cannot be completed within thirty (30) days, Landlord's failure to perform shall constitute a default under the Lease unless Landlord promptly commences to cure the failure and diligently and continuously prosecutes such cure to completion as soon as reasonably possible. All obligations of each party hereunder shall be construed as covenants, not conditions.

#### ARTICLE 23 LIEN FOR RENT

23.1 To secure the payment of all Rent due and to become due hereunder and the faithful performance of all the other covenants of this Lease required to be performed by Tenant, Tenant hereby gives to Landlord an express contract lien ("Lien") on and first security interest in and to all property, equipment, machinery, trade fixtures, chattels, and merchandise ("Tenant's Property") which may be placed in the Premises, and also upon all proceeds of any insurance which may accrue to Tenant by reason of damage to or destruction of any such property, and agrees that this Lease shall constitute a security agreement with respect thereto. All exemption laws are hereby waived by Tenant. This Lien is given in addition to any statutory liens and shall be cumulative thereto. Tenant authorizes Landlord to file UCC-1 Financing Statements referencing this Security Agreement in a form satisfactory to Landlord, and to file originals of such statements with the Secretary of State and the clerk(s) of the county(ies) where (a) the Premises are located, and (b) Tenant maintains its principal business office or residence, or wherever else such statements would ordinarily be filed to protect creditor's rights under California law. In addition to all other rights of Landlord under this Lease, upon Tenant's default, Landlord shall have all of the remedies of a secured party with respect to said property, equipment, machinery, trade fixtures, chattels, and merchandise. Landlord's lien shall be subordinate and inferior to any bank financing or purchase money financing procured by Tenant's Property and Landlord agrees to enter into a commercially reasonable agreement confirming that subordination upon Tenant's request.

# ARTICLE 24 INTENTIONALLY OMITTED

#### ARTICLE 25 ATTORNEYS' FEES

25.1 All costs and expenses, including reasonable attorneys' fees (whether or not legal proceedings are instituted), involved in collecting rents, enforcing the obligations of Tenant, or protecting the rights or interests of Landlord under this Lease, whether or not an action is filed, including without limitation the cost and expense of instituting and prosecuting legal proceedings or recovering possession of the Premises after default by Tenant or upon expiration or sooner termination of this Lease, shall be due and payable by Tenant on demand, as Additional Rent. In addition, and notwithstanding the foregoing, if either party hereto shall file any action or bring any proceeding against the other party arising out of this Lease or for the declaration of any rights hereunder, the prevailing party in such action shall be entitled to recover from the other party all costs and expenses, including reasonable attorneys' fees incurred by the prevailing party, as determined by the trier of fact in such legal proceeding. For purposes of this provision, the terms "attorneys' fees" or "attorneys' fees and costs," or "costs and expenses" shall

mean the fees and expenses of legal counsel (including external counsel and in-house counsel) of the parties hereto, which include printing, photocopying, duplicating, mail, overnight mail, messenger, court filing fees, costs of discovery, and fees billed for law clerks, paralegals, investigators and other persons not admitted to the bar for performing services under the supervision and direction of an attorney. For purposes of determining in-house counsel fees, the same shall be considered as those fees normally applicable to a partner in a law firm with like experience in such field. In addition, the prevailing party shall be entitled to recover reasonable attorneys' fees and costs incurred in enforcing any judgment arising from a suit or proceeding under this Lease, including without limitation post-judgment motions, contempt proceedings, garnishment, levy and debtor and third party examinations, discovery and bankruptcy litigation, without regard to schedule or rule of court purporting to restrict such award. This post-judgment award of attorneys' fees and costs provision shall be severable from any other provision of this Lease and shall survive any judgment/award on such suit or arbitration and is not to be deemed merged into the judgment/award or terminated with the Lease.

### ARTICLE 26 NON-WAIVER

26.1 Neither acceptance of any payment by Landlord from Tenant nor, failure by Landlord to complain of any action, non-action, or default of Tenant shall constitute a waiver of any of Landlord's rights hereunder. Time is of the essence with respect to the performance of every obligation of each party under this Lease in which time of performance is a factor. Waiver by either party of any right or remedy arising in connection with any default of the other party shall not constitute a waiver of such right or remedy or any other right or remedy arising in connection with either a subsequent default of the same obligation or any other default. No right or remedy of either party hereunder or covenant, duty, or obligation of any party hereunder shall be deemed waived by the other party unless such waiver is in writing, signed by the other party or the other party's duly authorized agent.

# ARTICLE 27 RULES AND REGULATIONS

27.1 Such reasonable rules and regulations applying to all lessees in the Project for the safety, care, and cleanliness of the Project and the preservation of good order thereon are hereby made a part hereof as Exhibit D. and Tenant agrees to comply with all such rules and regulations. Landlord shall have the right at all times to change such rules and regulations or to amend them in any reasonable and non-discriminatory manner as may be deemed advisable by Landlord, all of which changes and amendments shall be sent by Landlord to Tenant in writing and shall be thereafter carried out and observed by Tenant. In the event that Tenant notifies Landlord of any non-compliance with rules and regulations by any other lessee in the Project which adversely affects Tenant's use and enjoyment of the Premises, Landlord shall use its reasonable efforts to cause such other tenant to comply with such rules and regulations; provided that Landlord shall have no obligation to commence or prosecute any action or proceeding against such other lessee and Landlord shall not have any liability to Tenant for any failure of any other lessees of the Project to comply with such rules and regulations. Landlord agrees not to enforce the Rules and Regulations in a manner that discriminates against Tenant.

### ARTICLE 28 ASSIGNMENT BY LANDLORD; RIGHT TO LEASE

28.1 Landlord shall have the right to transfer or assign, in whole or in part, all its rights and obligations hereunder and in the Premises and the Project. In such event, no liability or obligation shall accrue or be charged to Landlord with respect to the period from and after such transfer or assignment and assumption of Landlord's obligations by the transferee or assignee; provided that (a) any successor pursuant to a voluntary transfer (but not as part of an involuntary transfer resulting from a foreclosure or deed in lieu thereof) shall have assumed Landlord's obligations under this Lease, and (b) Landlord's obligation to return any Letter of Credit held on behalf of Tenant shall not be released until that Letter of Credit is turned over to that successor Landlord or returned to Tenant.

28.2 Landlord reserves the absolute right to effect such other tenancies in the Project as Landlord in the exercise of its sole business judgment shall determine to best promote the interests of the Buildings or Project. Tenant does not rely on the fact, nor does Landlord represent, that any specific tenant or type or number of tenants shall, during the Lease Term, occupy any space in the Buildings or Project.

## ARTICLE 29 LIABILITY OF LANDLORD

29.1 It is expressly understood and agreed that the obligations of Landlord under this Lease shall be binding upon Landlord and its successors and assigns and any future owner of the Project only with respect to events occurring during its and their respective ownership of the Project. In addition, Tenant agrees to look solely to Landlord's interest in the Project for recovery of any judgment against Landlord arising in connection with this Lease, it being agreed that neither Landlord nor any successor or assign of Landlord nor any future owner of the Project, nor any partner, shareholder, member, or officer of any of the foregoing shall ever be personally liable for any such judgment. The limitations of liability contained in this Section 29.1 shall inure to the benefit of Landlord's and the Landlord Parties' present and future partners, beneficiaries, officers, directors, trustees, shareholders, agents and employees, and their respective partners, heirs, successors and assigns. Under no circumstances shall any present or future partner of Landlord (if Landlord is a partnership), or trustee or beneficiary (if Landlord or any partner of Landlord a trust), have any liability for the performance of Landlord's obligations under this Lease. Notwithstanding any contrary provision herein, neither Landlord nor the Landlord Parties shall be liable under any circumstances for any indirect or consequential damages or any injury or damage to, or interference with, Tenant's business, including but not limited to, loss of profits, loss of rents or other revenues, loss of business opportunity, loss of goodwill or loss of use, in each case, however occurring.

# ARTICLE 30 SUBORDINATION AND ATTORNMENT

30.1 This Lease, and all rights of Tenant hereunder, are, and shall continue to be, subject and subordinate in all respects to: (a) all ground leases, overriding leases and underlying leases of the Land and/or the Building now or hereafter existing; (b) all mortgages

that may now or hereafter affect the Land, the Building and/or any of such leases, whether or not such mortgages shall also cover other land and/or buildings; (c) each and every advance made or hereafter to be made under such mortgages; (d) all renewals, modifications, replacements and extensions of such leases and such mortgages; and (e) all spreaders and consolidations of such mortgages. The leases to which this Lease is, at the time referred to, subject and subordinate pursuant to this Article are herein sometimes called "superior leases," the mortgages to which this lease is, at the time referred to, subject and subordinate are herein sometimes called "superior mortgages," the lessor of a superior lease or its successor in interest at the time referred to is sometimes herein called a "lessor" and the mortgagee under a superior mortgage or its successor in interest at the time referred to is sometimes herein called a "mortgagee." Tenant agrees, with respect to any of the foregoing documents, that no documentation other than this Lease shall be required to evidence such subordination. If any holder of a mortgage shall elect for this Lease to be superior to the lien of its mortgage and shall give written notice thereof to Tenant, then this Lease shall automatically be deemed prior to such mortgage whether this Lease is dated earlier or later than the date of said mortgage or the date of recording thereof. Tenant agrees to execute such reasonable documents as may be further required to evidence such subordination or to make this Lease prior to the lien of any mortgage or deed of trust, as the case may be, and Tenant's failure to do so within ten (10) days after written demand shall, if Landlord so elects, constitute an Event of Default. Tenant hereby attorns to all successor owners of the Building, whether or not such ownership is acquired as a result of a sale through foreclosure or otherwise.

30.2 If the lessor of a superior lease or the mortgagee of a superior mortgage shall succeed to the rights of Landlord under this Lease, whether through possession or foreclosure action or delivery of a new lease or deed, then, at the request of such party so succeeding to Landlord's rights (hereinafter sometimes called a "successor landlord"), and upon such successor landlord's written agreement to accept Tenant's attornment, Tenant shall attorn to and recognize such successor landlord as Tenant's landlord under this Lease, and shall promptly execute and deliver any instrument such successor landlord may reasonably request to evidence such attornment. Upon such attornment this Lease shall continue in full force and effect as, or as if it were, a direct lease between such successor landlord and Tenant upon all of the terms, conditions and covenants as are set forth in this Lease and shall be applicable after such attornment, except such successor landlord shall not be (a) liable in any way to Tenant for any act or omission, neglect or default on the part of Landlord under this Lease unless it is of a continuing nature and continues after successor landlord is given written notice thereof and has a reasonable opportunity to cure the same after taking possession of the Property, (b) responsible for any monies owing by or on deposit with Landlord to the credit of Tenant unless such money has been delivered to and received by the successor landlord, (c) subject to any defenses, counterclaims, abatements or offsets that theretofore accrued to Tenant against Landlord except to the extent the basis of such counterclaim, abatement or offset is of a continuing nature and continues after successor landlord is given written thereof and has a reasonable opportunity to cure the same after taking possession of the Property, (d) bound by any amendments, terminations or modifications of this Lease subsequent to such superior lease or superior mortgage, or by any previous prepayment of fixed rent for more than one (1) month (ex

performance of any work to be done by the Landlord under this Lease to render the Premises ready for occupancy by the Tenant, or (g) required to remove any person occupying the Premises or any part thereof.

- 30.3 This Lease may not be modified or amended so as to reduce the Base Rent and/or Additional Rent, shorten the Term, or otherwise materially affect the rights of Landlord hereunder, or be canceled or surrendered, without the prior written consent in each instance of the ground lessors and of any superior mortgages whose mortgages shall require such consent. Any such modification, agreement, cancellation or surrender made without such prior written consent shall be null and void.
- 30.4 Tenant agrees if this Lease expires or is terminated or canceled for any reason or by any means whatsoever by reason of a default under a ground lease or mortgage, and the ground lessor or mortgage so elects by written notice to Tenant, this Lease shall automatically be reinstated for the balance of the term that would have remained but for such termination, expiration or cancellation, at the same rental, and upon the same agreements, covenants, conditions, restrictions and provisions herein contained, with the same force and effect as if no such termination, expiration or cancellation had taken place. Tenant covenants to execute and deliver any instrument required to confirm the validity of the foregoing.
- 30.5 Tenant shall, at such time or times as Landlord may request, upon not less than fifteen (15) days' prior written request by Landlord, sign and deliver to Landlord an estoppel certificate, which shall be substantially in the form of Exhibit E, attached hereto (or such other commercially reasonable form as may be required by any prospective mortgagee or purchaser of the Project, or any portion thereof), indicating therein any exceptions thereto that may exist at that time, and shall also contain such other information and agreements as may be reasonably requested, it being intended that any such statement delivered pursuant to this Article may be relied upon by Landlord and by any prospective purchaser of all or any portion of the Project, or a holder or prospective holder of any mortgage encumbering the Project, or any portion thereof. Tenant's failure to deliver such statement within five (5) days after Landlord's second written request therefor shall constitute an Event of Default (as that term is defined elsewhere in this Lease) and shall conclusively be deemed to be an admission by Tenant of the matters set forth in the request for an estoppel certificate.
- 30.6 Tenant shall deliver to Landlord prior to the execution of this Lease and thereafter at any time upon Landlord's request, Tenant's current audited financial statements, including a balance sheet and profit and loss statement for the most recent prior year (collectively, the "Statements"), which Statements shall accurately and completely reflect the financial condition of Tenant. Landlord shall have the right to deliver the same to any proposed purchaser of the Building or the Project, and to any encumbrancer of all or any portion of the Building or the Project. Landlord agrees not to request copies of financial statements more often than once in every twelve-month period, unless required in connection with a proposed sale or financing. Notwithstanding the foregoing, if (i) Tenant is required to file reports under the Securities Exchange Act of 1934, as amended, (ii) Tenant is current in its reporting obligations thereunder, and (iii) the reports required by such act are available to the public, including Landlord, then Tenant shall not be obligated to provide Landlord with financial statements pursuant to this Section 30.6.

30.7 Tenant acknowledges that Landlord is relying on the Statements previously delivered by Tenant to Landlord in its determination to enter into this Lease, and Tenant represents to Landlord, which representation shall be deemed made on the date of this Lease and again on the Commencement Date, that no material change in the financial condition of Tenant, as reflected in the Statements, has occurred since the date Tenant delivered the Statements to Landlord. The Statements are represented and warranted by Tenant to be correct and to accurately and fully reflect Tenant's true financial condition as of the date of submission of any Statements to Landlord.

30.8 Concurrently with Tenant's execution of this Lease, Tenant shall execute a non-disturbance, subordination and attornment agreement with the lender ("Lender") set forth on the Subordination, Non-Disturbance & Attornment Agreement ("SNDA") in the form attached to this Lease as Exhibit K and return the same to Landlord for execution by Lender. Landlord agrees to use commercially reasonable efforts to deliver to Tenant from the Lender and any future mortgagee or beneficiary a written subordination and non-disturbance agreement in recordable form acceptable to such mortgagee or beneficiary in its reasonable discretion providing that so long as Tenant performs all of the terms of this Lease, Tenant's possession under this Lease shall not be disturbed and Tenant shall not be joined by the holder of any mortgage or deed of trust in any action or proceeding to foreclose thereunder, except where such is necessary for jurisdictional or procedural reasons. Landlord and Tenant shall each pay its own costs incurred in obtaining that subordination and non-disturbance agreement. Landlord's failure to obtain anon-disturbance, subordination and attornment agreement for Tenant shall have no effect on the rights, obligations and liabilities of Landlord and Tenant or be considered to be a default by Landlord hereunder.

## ARTICLE 31 HOLDING OVER

31.1 In the event Tenant, or any party claiming under Tenant, retains possession of the Premises after the Expiration Date or Termination Date, such possession shall be that of a tenant at sufferance and an unlawful detainer. No tenancy or interest shall result from such possession, and such parties shall be subject to immediate eviction and removal. Tenant or any such party shall pay Landlord, as Base Rent for the period of such holdover, a monthly amount equal to the Holdover Percentage (as defined below) of (a) the Base Rent for the last period prior to the date of such termination plus (b) Additional Rent attributable to Operating Expenses and Taxes as provided in Article 5 of this Lease during the time of holdover, together with all other Additional Rent and other amounts payable pursuant to the terms of this Lease. As used herein, the term "Holdover Percentage" means one hundred twenty-five percent (125%) with respect to the first three months of the holdover and one hundred fifty percent (150%) with respect to any holding over in excess of three months. Such tenancy at sufferance shall be subject to every other applicable term, covenant and agreement contained herein. Tenant shall also be liable for any and all damages sustained by Landlord as a result of such holdover after the date that is the later of (a) the date that is thirty (30) days after the expiration or earlier termination of the Term of this Lease, or (b) the date that is thirty (30) days after Landlord has notified Tenant that Landlord has executed a letter of intent or lease with another tenant for all or any portion of the Premises. Tenant shall vacate the Premises and deliver same to Landlord immediately upon Tenant's receipt of notice from Landlord to so

vacate. The Rent during such holdover period shall be payable to Landlord on demand. Landlord's acceptance of Rent if and after Tenant holds over shall not convert Tenant's tenancy at sufferance to any other form of tenancy or result in a renewal or extension of the Term of this Lease, unless otherwise specified by notice from Landlord to Tenant.

### ARTICLE 32 SIGNS

- 32.1 No sign, symbol, or identifying marks shall be put upon the Project, Building, in the halls, elevators, staircases, entrances, parking areas, or upon the doors or walls, without the prior written approval of Landlord in its sole discretion. Should such approval ever be granted, all signs or lettering shall conform in all respects to the sign and/or lettering criteria established by Landlord and comply with all Applicable Laws. Landlord, at Landlord's sole cost and expense, reserves the right to change the door plaques as Landlord deems reasonably desirable.
  - 32.2 Landlord, at Tenant's sole cost and expense, shall provide Tenant with Building standard lobby and suite signage.
- 32.3 Tenant may, at Tenant's sole cost and expense, install (a) one "eyebrow" sign at the main entrance to the Building facing Anton Boulevard, (b) one "eyebrow" sign either above Tenant's south entrance to the Building or at the top of the proposed archway (the "Archway") near Tenant's south entrance to the Building as shown on the Exhibit B-3 attached hereto, as elected by Tenant (the "Eyebrow Sign"), and (c) one line of signage at the top of the Project's new parking structure facing the 405 freeway, each identifying Tenant's name and logo (collectively, "Tenant's Signage"). Notwithstanding the foregoing, if Landlord is not able to obtain approvals for the construction of the Archway or otherwise determines not to proceed with the construction of the Archway, the Eyebrow Sign shall instead be located above Tenant's south entrance to the Building. The graphics, materials, color, design, lettering, size, location and specifications of Tenant's Signage shall be subject to the approval of Landlord and all applicable governmental authorities. The costs of the actual signs comprising Tenant's Signage and the installation, design, construction, and any and all other costs associated with Tenant's Signage, including, without limitation, utility charges and hook-up fees, permits, and maintenance and repairs, shall be the sole responsibility of Tenant. Should Tenant's Signage require repairs and/or maintenance, as determined in Landlord's reasonable judgment, Landlord shall have the right to provide notice thereof to Tenant and Tenant (except as set forth above) shall cause such repairs and/or maintenance to be performed within fifteen (15) business days after receipt of such notice from Landlord, at Tenant's sole cost and expense; provided, however, if such repairs and/or maintenance are reasonably expected to require longer than fifteen (15) business days to perform, Tenant shall commence such repairs and/or maintenance within such fifteen (15) business day period and shall diligently prosecute such repairs and maintenance to completion. Should Tenant fail to perform, commence, or diligently prosecute such repairs and/or maintenance within the applicable periods described in the immediately preceding sentence, Landlord shall, upon the delivery of an additional five (5) business days' prior written notice, have the right to cause such work to be performed and to charge Tenant as Additional Rent for the cost of such work. At the expiration or earlier termination of this Lease or termination of Tenant's sign rights as provided below, Landlord shall, at Tenant's sole cost and

expense, cause the Tenant's Signage to be removed and the area of the Building and/or parking structure affected by Tenant's Signage to be restored to the condition existing prior to the installation of Tenant's Signage. The right to Tenant's Signage is personal Original Tenant and any Permitted Transferee that is an assignee of Original Tenant's entire interest in this Lease. To the extent Tenant desires to change the name and/or logo set forth on Tenant's Signage, such name and/or logo shall not have a name which relates to an entity which is of a character or reputation, or is associated with a political faction or orientation, which is inconsistent with the quality of the Project, or which would otherwise reasonably offend a landlord of the Comparable Buildings. All of Tenant's rights to install and maintain Tenant's Signage on the Building and/or parking structure in accordance with this Section 32.3 shall permanently terminate upon notice from Landlord following (a) a Monetary Default under this Lease and/or (b) the date upon which Tenant and its Permitted Transferees cease to occupy collectively at least 25,000 rentable square feet within the Building (the "Minimum Occupancy Requirement for Signage"); provided that the Minimum Occupancy Requirement for Signage shall not apply to the Eyebrow Sign so long as Tenant and/or Permitted Transferees occupy all of the portion of the Premises located on the first (1st) floor of the Building.

# ARTICLE 33 HAZARDOUS SUBSTANCES

- 33.1 Except for Hazardous Material (as defined below) contained in products used by Tenant for ordinary cleaning and office purposes in quantities not violative of applicable Environmental Requirements, Tenant shall not permit or cause any party to bring any Hazardous Material upon the Premises and/or the Project or transport, store, use, generate, manufacture, dispose, or release any Hazardous Material on or from the Premises and/or the Project without Landlord's prior written consent. Tenant, at its sole cost and expense, shall operate its business in the Premises in strict compliance with all Environmental Requirements (as defined below) and all requirements of this Lease. Tenant shall complete and certify to disclosure statements as requested by Landlord from time to time relating to Tenant's transportation, storage, use, generation, manufacture, or release of Hazardous Materials on the Premises, and Tenant shall promptly deliver to Landlord a copy of any notice of violation relating to the Premises or the Project of any Environmental Requirement. Without limiting the generality of the foregoing, Tenant shall, at such intervals as Landlord may require, provide to Landlord or its designated consultant a list of all Hazardous Materials used by Tenant in the Premises. Tenant shall reimburse Landlord within thirty (30) days after demand for the actual costs and fees charged by Landlord's consultant to review the list of chemicals provided by the Tenant.
- 33.2 The term "Environmental Requirements" means all applicable present and future statutes, regulations, ordinances, rules, codes, judgments, permits, authorizations, orders, policies or other similar requirements of any governmental authority, agency or court regulating or relating to health, safety, or environmental conditions on, under, or about the Premises or the environment, including without limitation, the following: the Comprehensive Environmental Response, Compensation and Liability Act; the Resource Conservation and Recovery Act; the Clean Air Act; the Clean Water Act; the Toxic Substances Control Act and all state and local counterparts thereto; all applicable California requirements, including, but not limited to, Sections 25115, 25117, 25122.7, 25140, 25249.8, 25281, 25316 and 25501 of the California Health and Safety Code and Title 22 of the California Code of Regulations, Division 4.5,

Chapter 11, and any policies or rules promulgated thereunder as well as any County or City ordinances that may operate independent of, or in conjunction with, the State programs, and any common or civil law obligations including, without limitation, nuisance or trespass, and any other requirements of <a href="Article 3">Article 3</a> of this Lease. The term "Hazardous Materials" means and includes any substance, material, waste, pollutant, or contaminant that is or could be regulated under any Environmental Requirement or that may adversely affect human health or the environment, including, without limitation, any solid or hazardous waste, hazardous substance, asbestos, petroleum (including crude oil or any fraction thereof, natural gas, synthetic gas, polychlorinated biphenyls (PCBs), and radioactive material). For purposes of Environmental Requirements, to the extent authorized by law, Tenant is and shall be deemed to be the responsible party, including without limitation, the "owner" and "operator" of Tenant's "facility" and the "owner" of all Hazardous Materials brought on the Premises by Tenant, its agents, employees, contractors or invitees, and the wastes, by-products, or residues generated, resulting, or produced therefrom.

33.3 Tenant, at its sole cost and expense, shall remove all Hazardous Materials stored, disposed of or otherwise released by Tenant, its assignees, subtenants, agents, employees, contractors or invitees onto or from the Premises, in a manner and to a level satisfactory to Landlord in its sole discretion, but in no event to a level and in a manner less than that which complies with all Environmental Requirements and does not limit any future uses of the Premises or require the recording of any deed restriction or notice regarding the Premises. Tenant shall perform such work at any time during the Term of the Lease upon written request by Landlord or, in the absence of a specific request by Landlord, before Tenant's right to possession of the Premises terminates or expires. If Tenant fails to perform such work within the time period specified by Landlord or before Tenant's right to possession terminates or expires (whichever is earlier), Landlord may at its discretion, and without waiving any other remedy available under this Lease or at law or equity (including without limitation an action to compel Tenant to perform such work), perform such work at Tenant's cost. Tenant shall pay all costs incurred by Landlord in performing such work within ten (10) days after Landlord's request therefor. Such work performed by Landlord is on behalf of Tenant and Tenant remains the owner, generator, operator, transporter, and/or arranger of the Hazardous Materials for purposes of Environmental Requirements. Tenant agrees not to enter into any agreement with any person, including without limitation any governmental authority, regarding the removal of Hazardous Materials that have been disposed of or otherwise released onto or from the Premises without the written approval of Landlord.

33.4 Tenant shall indemnify, defend, and hold Landlord harmless from and against any and all losses (including, without limitation, diminution in value of the Premises or the Project and loss of rental income from the Project), claims, demands, actions, suits, damages (including, without limitation, punitive damages), expenses (including, without limitation, remediation, removal, repair, corrective action, or cleanup expenses), and costs (including, without limitation, actual attorneys' fees, consultant fees or expert fees and including, without limitation, removal or management of any asbestos brought into the Premises or disturbed in breach of the requirements of this <a href="https://doi.org/10.1001/NTML

regardless of whether Tenant had knowledge of such noncompliance. The obligations of Tenant under this Article 33 shall survive any termination of this Lease.

- 33.5 Landlord shall have access to, and a right to perform inspections and tests of, the Premises to determine Tenant's compliance with Environmental Requirements, its obligations under this Article 33, or the environmental condition of the Premises. Access shall be granted to Landlord upon Landlord's prior written notice to Tenant and at such times so as to minimize, so far as may be reasonable under the circumstances, any disturbance to Tenant's operations. Such inspections and tests shall be conducted at Landlord's expense, unless such inspections or tests reveal that Tenant has not complied with any Environmental Requirement, in which case Tenant shall reimburse Landlord for the reasonable cost of such inspection and tests. Landlord's receipt of or satisfaction with any environmental assessment in no way waives any rights that Landlord holds against Tenant. Tenant shall promptly notify Landlord of any communication or report that Tenant makes to any governmental authority regarding any possible violation of Environmental Requirements or release or threat of release of any Hazardous Materials onto or from the Premises. Tenant shall, within five (5) days of receipt thereof, provide Landlord with a copy of any documents or correspondence received from any governmental agency or other party relating to a possible violation of Environmental Requirements or claim or liability associated with the release or threat of release of any Hazardous Materials onto or from the Premises.
- 33.6 In addition to all other rights and remedies available to Landlord under this Lease or otherwise, Landlord may, in the event of a breach of the requirements of this <u>Article 33</u> that is not cured within thirty (30) days following written notice of such breach by Landlord, require Tenant to provide financial assurance (such as insurance, escrow of funds or third party guarantee) in an amount and form satisfactory to Landlord. The requirements of this <u>Article 33</u> are in addition to and not in lieu of any other provision in the Lease.
- 33.7 Tenant acknowledges receipt of a copy of (i) that certain Phase I Environmental Site Assessment dated December 10, 2015, prepared by Cardno ATC and (ii) that certain Phase I Site Assessment and Preliminary Indoor Air Quality Investigation dated August 25, 2005 prepared by LFR Levine-Fricke (collectively, "Hazardous Substance Reports"). Landlord acknowledges to Tenant that: (i) Landlord has not authorized any other studies for hazardous or toxic materials at the Premises, Project or Building other than the Hazardous Substance Reports; and (ii) Landlord does not know of any surveys for toxic or Hazardous Materials at the Premises or the Building other than the Hazardous Substance Reports. Notwithstanding the preceding sentence, unless Tenant obtains a reliance letter from the author(s) thereof, Tenant shall not rely on and Tenant hereby represents to Landlord that it has not relied on the Hazardous Substance Reports.
- 33.8 Tenant shall not be liable to Landlord under this Lease with respect to any Hazardous Materials existing on the Premises or the Project prior to the date Tenant takes possession of the Premises (the "Pre-Existing Hazardous Materials") or any Hazardous Materials placed on the Premises or the Project by Landlord or any other person other than Tenant, Tenant's employees, agents, contractors, invitees, assignees, or sublessees except to the extent that any hazard posed by such Pre-Existing Hazardous Materials is exacerbated by, or the cost to clean up, remove or remediate such Pre-Existing Hazardous Materials is increased as a result of,

the acts, omissions negligence or willful misconduct of Tenant or any of Tenant's employees, agents, contractors, invitees, assignees, or sublessees.

# ARTICLE 34 COMPLIANCE WITH LAWS AND OTHER REGULATIONS

34.1 Tenant, as its sole cost and expense, shall promptly comply with all laws, statutes, ordinances, and governmental rules, regulations, or requirements now in force or which may hereafter become in force, of federal, state, county, and municipal authorities, including without limitation the Americans with Disabilities Act and the California Energy Code, Title 24, with the requirements of any board of fire underwriters or other similar body now or hereafter constituted, and with any occupancy certificate issued pursuant to any law by any public officer or officers, which impose, any duty upon Landlord or Tenant, insofar as any thereof relate to or affect the condition, use, alteration, or occupancy of the Premises. Landlord's approval of Tenant's plans for any improvements shall create no responsibility or liability on the part of Landlord for their completeness, design sufficiency, or compliance with all laws, rules, and regulations of governmental agencies or authorities, including, but not limited to, the Americans with Disabilities Act.

34.2 As an inducement to Landlord to enter into this Lease, Tenant hereby represents and warrants that: (i) Tenant is not, nor is it owned or controlled directly or indirectly by, any person, group, entity or nation named on any list issued by the Office of Foreign Assets Control of the United States Department of the Treasury pursuant to Executive Order 13224 or any similar list or any law, order, rule or regulation or any Executive Order of the President of the United States as a terrorist, "Specially Designated National and Blocked Person" or other banned or blocked person (any such person, group, entity or nation being hereinafter referred to as a "Prohibited Person"); (ii) Tenant is not (nor is it owned or controlled, directly or indirectly, by any person, group, entity or nation which is) acting directly or indirectly for or on behalf of any Prohibited Person; and (iii) neither Tenant (nor any person, group, entity or nation which owns or controls Tenant, directly or indirectly) has conducted or will conduct business or has engaged or will engage in any transaction or dealing with any Prohibited Person, including without limitation any assignment of this Lease or any subletting of all or any portion of the Premises or the making or receiving of any contribution of funds, goods or services to or for the benefit of a Prohibited Person. Tenant covenants and agrees (a) to comply with all requirements of law relating to money laundering, anti-terrorism, trade embargos and economic sanctions, now or hereafter in effect, (b) to immediately notify Landlord in writing if any of the representations, warranties or covenants set forth in this Section 34.2 are no longer true or have been breached or if Tenant has a reasonable basis to believe that they may no longer be true or have been breached, (c) not to use funds from any Prohibited Person to make any payment due to Landlord under the Lease and (d) at the request of Landlord, to provide such information as may be requested by Landlord to determine Tenant's complia

34.3 Pursuant to California Civil Code Section 1938, Tenant is hereby notified that, as of the date hereof, the Project has not undergone an inspection by a "Certified Access Specialist" and except to the extent expressly set forth in this Lease, Landlord shall have no liability or responsibility to make any repairs or modifications to the Premises or the Project in order to comply with accessibility standards. The following disclosure is hereby made pursuant to applicable California law: "A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction-related accessibility standards within the premises." Tenant acknowledges that Landlord has made no representation regarding compliance of the Premises or the Project with accessibility standards. Any CASp inspection shall be conducted in compliance with reasonable rules in effect at the Building with regard to such inspections and shall be subject to Landlord's prior written consent.

34.4 Landlord shall, as part of Operating Expenses, subject to the terms and exclusions set forth in Article 5, comply with all laws, statutes, ordinances, and governmental rules, regulations, or requirements that impose any obligation with respect to the Common Area, the structural elements of the Building or to the Building systems other than systems installed by Tenant (which shall be the sole responsibility of Tenant), but only to the extent (i) the same are applicable to Landlord and the Building, (ii) Landlord is required by the applicable governmental authority to take such action, (iii) such action is not the result of a "Trigger Event" (as defined below), and (iv) Landlord's failure to comply therewith would prohibit Tenant from obtaining or maintaining a certificate of occupancy (or its legal equivalent) for the Premises, or would unreasonably and materially affect the safety of Tenant's employees or create a significant health hazard for Tenant's employees or otherwise materially and adversely affect Tenant's use or occupancy of the Premises. As used herein, the term "Trigger Event" means one or more of the following events or circumstances:

- (a) Tenant's particular use of the Premises (other than normal office uses);
- (b) The manner of conduct of Tenant's business or operation of its installations, equipment or other property outside those of normal office use;
- (c) The performance of any alterations or the installation of any Tenant systems; or
- (d) The breach of any of Tenant's obligations under this Lease.

### ARTICLE 35 SEVERABILITY

35.1 This Lease shall be construed in accordance with the laws of the State of California. If any clause or provision of this Lease is illegal, invalid, or unenforceable under present or future laws effective during the Term, then it is the intention of the parties hereto that the remainder of this Lease shall not be affected thereby. It is also the intention of both parties that in lieu of each clause or provision that is illegal, or unenforceable, there is added as a part of this Lease a clause or provision as similar in terms to such illegal, invalid, or unenforceable clause or provision as may be possible and still be legal, valid, and enforceable.

### ARTICLE 36 NOTICES

36.1 All notices, demands, designations, approvals or other communications (collectively, "Notices") given or required to be given by either party to the other hereunder or by law shall be in writing, shall be (i) sent by United States certified or registered mail, postage prepaid, return receipt requested ("Mail"), (ii) transmitted by telecopy, if such telecopy is promptly followed by a Notice sent by Mail, (iii) delivered by a nationally recognized overnight courier, or (iv) delivered personally. Any Notice shall be sent, transmitted, or delivered, as the case may be, to Tenant at the appropriate address set forth in the Basic Lease Information, or to such other place as Tenant may from time to time designate in a Notice to Landlord, or to Landlord at the addresses set forth in the Basic Lease Information, or to such other places as Landlord may from time to time designate in a Notice to Tenant. Any Notice will be deemed given (A) three (3) days after the date it is posted if sent by Mail, (B) the date the telecopy is transmitted, (C) the date the overnight courier delivery is made, or (D) the date personal delivery is made. Any Notice given by an attorney on behalf of Landlord or by Landlord's managing agent shall be considered as given by Landlord and shall be fully effective.

# ARTICLE 37 OBLIGATIONS OF, SUCCESSORS, PLURALITY, GENDER

37.1 Landlord and Tenant agree that all the provisions hereof are to be construed as covenants and agreements as though the words imparting such covenants were used in each paragraph hereof, and that, except as restricted by the provisions hereof, shall bind and inure to the benefit of the parties hereto, their respective successors, and assigns. If the rights of Tenant hereunder are owned by two or more parties, or two or more parties are designated herein as Tenant, then all such parties shall be jointly and severally liable for the obligations of Tenant hereunder. Whenever the singular or plural number, masculine or feminine or neuter gender is used herein, it shall equally include the other.

### ARTICLE 38 ENTIRE AGREEMENT

38.1 This Lease and any attached addenda or exhibits constitute the entire agreement between Landlord and Tenant. No prior or contemporaneous written or oral leases or

representations shall be binding. This Lease shall not be amended, changed, or extended except by written instrument signed by Landlord and Tenant.

38.2 THE SUBMISSION OF THIS LEASE BY LANDLORD, ITS AGENT OR REPRESENTATIVE FOR EXAMINATION OR EXECUTION BY TENANT DOES NOT CONSTITUTE AN OPTION OR OFFER TO LEASE THE PREMISES UPON THE TERMS AND CONDITIONS CONTAINED HEREIN OR A RESERVATION OF THE PREMISES IN FAVOR OF TENANT, IT BEING INTENDED HEREBY THAT THIS LEASE SHALL ONLY BECOME EFFECTIVE UPON THE EXECUTION HEREOF BY LANDLORD AND DELIVERY OF A FULLY EXECUTED LEASE TO TENANT.

### ARTICLE 39 CAPTIONS

39.1 Paragraph captions are for Landlord's and Tenant's convenience only, and neither limit nor amplify the provisions of this Lease.

### ARTICLE 40 CHANGES

- 40.1 If, in connection with obtaining financing or refinancing for the Building and/or the Land, a banking, insurance, or other lender shall request reasonable modifications to this Lease as a condition to such financing or refinancing, Tenant shall not unreasonably withhold, delay, or defer its consent thereto, provided such modifications do not increase the obligations, or decrease the rights, of Tenant hereunder or decrease the obligations or increase the rights of Landlord hereunder. In no event shall the following requested modification of this Lease (or modifications comparable thereto) be deemed to adversely affect the leasehold interest hereby created or decrease the rights of Tenant hereunder:
  - (a) give notice of any default by Landlord under this Lease to such lender and/or permit the curing of such defaults by such lender, together with the granting of such additional time for such curing as may be reasonably required for such lender to obtain possession of the Building; and/or
    - (b) obtain such lender's consent for any amendment, termination or modification of this Lease or for any assignment or sublease.

#### ARTICLE 41 AUTHORITY

41.1 All rights and remedies of Landlord under this Lease, or those which may be provided by law, may be exercised by Landlord in its own name individually, or in its name by its agent, and all legal proceedings for the enforcement of any such rights or remedies, including distress for Rent, unlawful detainer, and any other legal or equitable proceedings may be commenced and prosecuted to final judgment and be executed by Landlord in its own name individually or in its name by its agent. Landlord and Tenant each represent to the other that each has full power and authority to execute this Lease and to make and perform the agreements herein contained, and Tenant expressly stipulates that any rights or remedies available to

Landlord, either by the provisions of this Lease or otherwise, may be enforced by Landlord in its own name individually or in its name by its agent or principal.

### ARTICLE 42 BROKERAGE

42.1 Tenant represents and warrants to Landlord that it has dealt only with Tenant's Broker and Landlord's Broker, in negotiation of this Lease. Landlord shall make payment of the brokerage fee due the Landlord's Broker pursuant to and in accordance with a separate agreement between Landlord and Landlord's Broker. Landlord's Broker shall pay a portion of its commission to Tenant's Broker pursuant to a separate agreement between Landlord's Broker and Tenant's Broker. Except for amounts owing to Landlord's Broker and Tenant's Broker, each party hereby agrees to indemnify and hold the other party harmless of and from any and all damages, losses, costs, or expenses (including, without limitation, all attorneys' fees and disbursements) by reason of any claim of or liability to any other broker or other person claiming through the indemnifying party and arising out of or in connection with the negotiation, execution, and delivery of this Lease. Additionally, except as may be otherwise expressly agreed upon by Landlord in writing, Tenant acknowledges and agrees that Landlord and/or Landlord's agent shall have no obligation for payment of any brokerage fee or similar compensation to any person with whom Tenanth has dealt or may in the future deal with respect to leasing of any additional or expansion space in the Building or renewals or extensions of this Lease. Tenant hereby irrevocably recognizes Tenant's Broker as Tenant's sole and exclusive broker with respect to any lease by Tenant of all or any portion of the Offer Space. Landlord shall have no obligation to pay any commission or other fee to any broker other than Tenant's Broker with respect to any lease by Tenant of all or any portion of the Offer Space.

### ARTICLE 43 EXHIBITS

43.1 Exhibits A through J are attached hereto and incorporated herein for all purposes and are hereby acknowledged by both parties to this Lease.

### ARTICLE 44 APPURTENANCES

44.1 The Premises include the right of ingress and egress thereto and therefrom; however, Landlord reserves the right to make changes and alterations to the Building, fixtures and equipment thereof, in the street entrances, doors, halls, corridors, lobbies, passages, elevators, escalators, stairways, toilets and other parts thereof which Landlord may deem necessary or desirable; provided that Tenant at all times has a means of access to the Premises (subject to a temporary interruption due to Force Majeure Events or necessary maintenance that cannot reasonably be performed without such interruption of access). In exercising its rights under this Section 44.1, Landlord shall make commercially reasonable efforts to minimize the disruption to Tenant's business operations during standard business hours. Neither this Lease nor any use by Tenant of the Building or any passage, door, tunnel, concourse, plaza or any other area connecting the garages or other buildings with the Building, shall give Tenant any right or easement of such use and the use thereof may, without notice to Tenant, be regulated or

discontinued at any time and from time to time by Landlord without liability of any kind to Tenant and without affecting the obligations of Tenant under this Lease.

# ARTICLE 45 PREJUDGMENT REMEDY, REDEMPTION, AND COUNTERCLAIM

45.1 Tenant, for itself and for all persons claiming through or under it, hereby expressly waives any and all rights which are, or in the future may be, conferred upon Tenant by any present or future law to redeem the Premises, or to any new trial in any action for ejection under any provisions of law, after reentry thereupon, or upon any part thereof, by Landlord, or after any warrant to dispossess or judgment in ejection. If Landlord shall acquire possession of the Premises by summary proceedings, or in any other lawful manner without judicial proceedings, it shall be deemed a reentry within the meaning of that word as used in this Lease. In the event that Landlord commences any summary proceedings or action for nonpayment of Rent or other charges provided for in this Lease, Tenant shall not interpose any counterclaim of any nature or description in any such proceeding or action.

### ARTICLE 46 RECORDING

46.1 Tenant shall not record this Lease but will, at the request of Landlord, execute a memorandum or notice thereof in recordable form satisfactory to both Landlord and Tenant specifying the date of commencement and expiration of the Term of this Lease and other information required by statute. Either Landlord or Tenant may then record said memorandum or notice of lease at the cost of the recording party.

# ARTICLE 47 MORTGAGEE PROTECTION

47.1 Tenant agrees to give any mortgagees and/or trust deed holders having a lien on Landlord's interest in the Project, by registered mail, a copy of any notice of default served upon Landlord, provided that prior to such notice Tenant has been notified, in writing of the address of such mortgagees and/or trust deed holders. Tenant further agrees that if Landlord shall have failed to cure such default within the time provided for in this Lease, then such mortgagees and/or trust deed holders shall have an additional thirty (30) days within which to cure such default or if such default cannot be cured within that time, then such additional time as may be necessary to cure such default (including but not limited to commencement of foreclosure proceedings, if necessary to effect such cure) in which event this Lease shall not be terminated while such remedies are being so diligently pursued. Nothing contained herein shall obligate such lessor or mortgagee to remedy such act or omission.

# ARTICLE 48 OTHER LANDLORD CONSTRUCTION

48.1 Tenant acknowledges that portions of the Project may be under construction following Tenant's occupancy of the Premises, and that such construction may result in levels of noise, dust, odor, obstruction of access, etc. which are in excess of that present in a fully constructed project. Tenant hereby waives any and all rent offsets or claims of

constructive eviction which may arise in connection with such construction. If any excavation or construction is made adjacent to, upon or within the Building, or any part thereof, Tenant shall afford to any and all persons causing or authorized to cause such excavation or construction license to enter upon the Premises for the purpose of doing such work as such persons shall deem necessary to preserve the Building or any portion thereof from injury or damage and to support the same by proper foundations, braces and supports, without any claim for damages or indemnity or abatement of Rent (subject to the express provisions of this Lease), or of a constructive or actual eviction of Tenant.

48.2 It is specifically understood and agreed that Landlord has made no representation or warranty to Tenant and has no obligation and has made no promises to alter, remodel, improve, renovate, repair or decorate the Premises, the Building, or any part thereof and that no representations respecting the condition of the Project, Premises or the Building, or the areas in the vicinity of the Project have been made by Landlord to Tenant except as specifically set forth herein or in the Tenant Work Letter. However, Tenant hereby acknowledges that Landlord is currently renovating or may during the Term of this Lease renovate, improve, alter, or modify (collectively, the "Renovations") the Project, the Building and/or the Premises, including without limitation the Parking Facilities (as defined below), the Common Areas, systems and equipment, roof, and structural portions of the same, which Renovations may include, without limitation, (i) installing sprinklers in the Building Common Areas and tenant spaces, (ii) modifying the Common Areas and tenant spaces to comply with applicable laws and regulations, including regulations relating to the physically disabled, seismic conditions, and building safety and security, (iii) installing new floor covering, lighting, and wall coverings in the Building Common Areas, and (iv) creating additional parking areas or occupied space within the Project, and in connection with any Renovations, Landlord may, among other things, erect scaffolding or other necessary structures in the Building, limit or eliminate access to portions of the Project, including portions of the Common Areas, or perform work in the Building, which work may create noise, dust or leave debris in the Building. Similarly, other properties in the vicinity of the Project may undergo substantial construction or renovation during the Term of this Lease (the "Area Renovations"), which may cause substantial disturbance to traffic and parking, and may cause dust, noise and vibrations which may affect the Project. Tenant hereby agrees that such Renovations or Area Renovations and Landlord's actions in connection with such Renovations or Area Renovations shall in no way constitute a constructive eviction of Tenant nor entitle Tenant to any abatement of Rent. Landlord shall have no responsibility or for any reason be liable to Tenant for any direct or indirect injury to or interference with Tenant's business arising from the Renovations or Area Renovations, nor shall Tenant be entitled to any compensation or damages from Landlord for loss of the use of the whole or any part of the Premises or of Tenant's personal property or improvements resulting from the Renovations or Area Renovations or Landlord's actions in connection with such Renovations or Area Renovations, or for any inconvenience or annoyance occasioned by such Renovations or Area Renovations or Landlord's actions in connection with such Renovations or Area Renovations.

48.3 In exercising its rights under this <u>Article 48</u>, Landlord shall make commercially reasonable efforts to minimize the disruption to Tenant's business operations during standard business hours.

48.4 As of the date of this Lease, Landlord has constructed or is constructing on the Project the following amenities: Food truck runway, outdoor social lounges, on-site café, tenant conference center, executive lounge, relaxation lounge & showers, event lawn, recreation gaming area, outdoor work & social areas, modernized entries, new lobbies and campus promenade. Landlord reserves the right to modify and/or eliminate any such amenities, and to add and/or eliminate different amenities in the Project, from time to time during the Term in its sole discretion. The design and location of such amenities, whether existing, under construction or constructed in the future, shall be determined by Landlord in its sole discretion.

### ARTICLE 49 PARKING

49.1 Commencing upon the date Tenant shall commence the conduct of business in the Premises or any portion thereof, Landlord shall provide Tenant, on an unassigned, non-exclusive and unlabeled basis, the Maximum Parking Allocation of parking passes in the parking structures and surface parking facilities of the Project (collectively, the "Parking Facilities"). Landlord does not guarantee the availability of spaces at all times against the actions of other tenants of the Project. Without limiting the foregoing, in no event shall this Lease be void or voidable, nor shall Landlord be liable to Tenant for any loss or damage, nor shall there be any abatement of Rent hereunder (other than the charge paid hereunder for any parking space no longer made available), by reason of any reduction in Tenant's parking rights hereunder by reason of strikes, lock-outs, labor disputes, shortages of material or labor, fire, flood or other casualty, acts of God or any other cause beyond the reasonable control of Landlord. Tenant shall pay Landlord or the operator of the Parking Facilities, as directed by Landlord, the rate or charge in effect from time to time for parking passes in the Parking Facilities in accordance with the terms and provisions set forth in Section 49.3 below; provided, however, if less than the Minimum Parking Allocation of parking passes are used, then Tenant shall nonetheless pay such rate for the Minimum Parking Allocation of parking passes are used. If the charge for parking is not paid when due, and such failure continues for thirty (30) days after written notice to Tenant of such failure, then in addition to any other remedies afforded Landlord under this Lease by reason of nonpayment of Rent, Landlord may terminate Tenant's rights under this Article 49 with respect to the applicable parking spaces. Further, if at any time Tenant releases to Landlord any parking pass(es) provided for in this Article 49 in excess of the Minimum Parking Allocation, then Tenant's right under this Article 49 to use such relea

49.2 A parking pass does not entitle the holder thereof to park in any particular parking space in the parking facilities. Landlord may issue parking permits, install a gate system, utilize valet parking and impose any other system as Landlord deems necessary for the use of the parking area. Tenant agrees that it and its employees and invitees shall not park their automobiles in any parking spaces designated as handicapped parking (unless Tenant's employee or invitee qualifies for such use), loading area, visitor parking or as may be otherwise reserved or

allocated, as the same may be reallocated or redesignated from time to time (the "Excluded Parking Areas"). Landlord shall not be responsible for any damage to or theft of any vehicle in the parking area, and shall not be required to keep parking spaces clear of unauthorized vehicles or to otherwise supervise the use of the parking area. Notwithstanding the foregoing, in the event that Tenant notifies Landlord of use of parking spaces by unauthorized vehicles which adversely affect Tenant's use and enjoyment of the parking rights granted to Tenant under this Article 49, Landlord shall use its reasonable efforts to cause such other users to cease such unauthorized use (including, without limitation, having unauthorized vehicles parked in preferred or personalized spaces allocated to Tenant towed; provided that such unauthorized vehicle continues to park in preferred or personalized spaces allocated to Tenant after two (2) prior written notices from Landlord); provided that Landlord shall have no obligation to commence or prosecute any action or proceeding against, or revoke parking rights of, such other user or hire additional personnel or install additional equipment, and Landlord shall not have any liability to Tenant for any unauthorized use. Landlord reserves the right to change any existing or future parking area, roads, or driveways, or increase or decrease the size thereof and make any repairs or alterations it deems necessary to the parking area, roads and driveways and Landlord agrees to use commercially reasonable efforts to minimize any interference with Tenant's parking in the course of such repairs or alterations. Tenant shall supply Landlord with an identification roster listing, for each parking pass, the name of the employee and the make, color and registration number of the vehicle to which such parking pass has been assigned, and shall provide a revised roster to Landlord monthly indicating changes thereto. Tenant shall cause each such individual to execute the standard waiver form, if any, for garage users used in the applicable portion of the Parking Facilities. Tenant shall be responsible for causing any employee of Tenant or other person using parking spaces allocated to Tenant to comply with all such parking rules and regulations. Landlord may delegate its responsibilities hereunder to a parking operator in which case such parking operator shall have all the rights of control attributed hereby to the Landlord. The parking passes provided to Tenant pursuant to this Article 49 are provided to Tenant solely for use by Tenant's own personnel and such passes may not be transferred, assigned, subleased or otherwise alienated by Tenant without Landlord's prior approval. In no event shall Tenant seek or collect any payment from any assignee or subtenant for the transfer of any such parking rights. No deductions from the monthly parking charge shall be made for days on which the Tenant does not use any of the parking passes entitled to be used by Tenant. Tenant shall be responsible for the full amount of any taxes imposed by any governmental authority in connection with the renting of such parking passes by Tenant or the use of such parking facilities by Tenant.

49.3 During the initial Term, parking charges payable by Tenant shall be as follows: (a) unreserved parking passes shall be \$0.00 per pass per month from the Commencement Date through the end of the fourteenth (14th) month after the Commencement Date and \$60.00 per pass per month thereafter through the expiration of the initial Term; (b) preferred parking passes shall be \$100.00 per pass per month from the Commencement Date through the expiration of the initial Term; and (c) personalized parking passes shall be \$150.00 per pass per month from the Commencement Date through the expiration of the initial Term.

49.4 If at any time or from time to time on or after the Commencement Date Landlord determines that an Adverse Parking Condition (as defined below) exists and Tenant is leasing more than one hundred thirteen (113) parking passes out of Tenant's Maximum Parking Allocation (the number of such excess over one hundred thirteen (113) parking passes being

referred to herein as the "Excess Parking Pass Amount"), then Landlord may require that the users of the Excess Parking Pass Amount of Tenant's parking passes park on the roof of a parking structure within the Project designated by Landlord. If despite such relocation, Landlord determines that an Adverse Parking Conditions continues to exist, then Landlord may also require that the users of Tenant's parking passes use valet assist parking (at no charge to Tenant) or overflow parking locations designated by Landlord. As used in this paragraph, "Adverse Parking Condition" means there is insufficient parking for Tenant's employees or other tenants of the Project within the unreserved areas of the Parking Facilities.

49.5 Notwithstanding anything to the contrary in this Lease, during the initial Term, Tenant may convert up to forty-five (45) of its unreserved parking passes to preferred parking passes and/or personalized parking passes, six (6) of which shall be VIP "walk-in" spaces located directly adjacent to the west side of the ground floor of the Building. The rates for such preferred parking passes and/or personalized parking passes shall be as provided in this Lease. The location of any specific reserved parking spaces shall be on the ground floor or the second floor of the parking structure, as determined by Landlord in its sole discretion; provided that Landlord shall locate at least fifteen (15) of such reserved spaces on the ground floor of the parking structure.

### ARTICLE 50 ELECTRICAL CAPACITY

Tenant covenants and agrees that, to its knowledge, at all times, its use of electric energy shall never exceed the capacity of the existing feeders to the Building or the risers of wiring installation. Excluding the initial build-out of the Premises, any additional riser or risers to supply Tenant's electrical requirements upon written request of Tenant shall be installed by Landlord at the sole cost and expense of Tenant, if, in Landlord's sole judgment, the same are necessary and will not cause or create a dangerous or hazardous condition or entail excess or unreasonable alterations, repairs or expense or interfere with or disrupt other tenants or occupants. In addition to the installation of such riser or risers, Landlord will also, at the sole cost and expense of Tenant, install all other equipment proper and necessary in connection therewith subject to the aforesaid terms and conditions.

### ARTICLE 51 OPTIONS TO EXTEND LEASE

- 51.1 First Extension Option. Tenant shall have the option to extend this Lease (the "First Extension Option") for one additional term of five (5) years (the "First Extension Period"), upon the terms and conditions hereinafter set forth:
  - (a) If the First Extension Option is exercised, then the Base Rent per annum for such First Extension Period (the "First Option Rent") shall be the Fair Market Rental Value (as defined hereinafter) for the Premises as of the commencement of the First Extension Option for such First Extension Period.
    - (b) The First Extension Option must be exercised by Tenant, if at all, only at the time and in the manner provided in this subsection 51.1(b).

- (i) If Tenant wishes to exercise the First Extension Option, Tenant must, on or before the date occurring nine (9) months before the expiration of the initial Lease Term (but not before the date that is twelve (12) months before the expiration of the Initial Lease Term), exercise the First Extension Option by delivering written notice (the "First Exercise Notice") to Landlord. If Tenant timely and properly exercises its First Extension Option, the Lease Term shall be extended for the First Extension Period upon all of the terms and conditions set forth in the Lease, as amended, except that the rent for the First Extension Period shall be as provided in subsection 51.1(a) and Tenant shall have no further options to extend the Lease Term except for the Second Extension Option.
  - (ii) If Tenant fails to deliver a timely First Exercise Notice, Tenant shall be considered to have elected not to exercise the First Extension Option.
- (c) It is understood and agreed that the First Extension Option hereby granted is personal to Original Tenant and is not transferable except to a Permitted Transferee in connection with an assignment of Tenant's entire interest in this Lease. In the event of any assignment or subletting of the Premises or any part thereof (other than to a Permitted Transferee), the First Extension Option shall automatically terminate and shall thereafter be null and void.
- (d) Tenant's exercise of the First Extension Option shall, if Landlord so elects in its absolute discretion, be ineffective in the event that (i) an Event of Default by Tenant remains uncured at the time of delivery of the First Exercise Notice or at the commencement of the First Extension Period or (ii) Tenant shall have reduced the size of the Premises below 23,993 rentable square feet by agreement with Landlord or pursuant to an express right in this Lease.
- 51.2 Second Extension Option. Tenant shall have the option to extend this Lease (the "Second Extension Option") for one additional term of five (5) years (the "Second Extension Period"), upon the terms and conditions hereinafter set forth:
  - (a) If the Second Extension Option is exercised, then the Base Rent per annum for such Second Extension Period (the "Second Option Rent") shall be the Fair Market Rental Value for the Premises as of the commencement of the Second Extension Option for such Second Extension Period.
    - (b) The Second Extension Option must be exercised by Tenant, if at all, only at the time and in the manner provided in thisubsection 51.2(b).
    - (i) If Tenant wishes to exercise the Second Extension Option, Tenant must, on or before the date occurring nine (9) months before the expiration of the First Extension Period (but not before the date that is twelve (12) months before the expiration of the First Extension Period), exercise the Second Extension Option by delivering written notice (the "Second Exercise Notice") to Landlord. If Tenant timely and properly exercises its Second Extension Option,

the Lease Term shall be extended for the Second Extension Period upon all of the terms and conditions set forth in the Lease, as amended, except that the rent for the Second Extension Period shall be as provided in <u>subsection 51.2(a)</u> and Tenant shall have no further options to extend the Lease Term.

- (ii) If Tenant fails to deliver a timely Second Exercise Notice, Tenant shall be considered to have elected not to exercise the Second Extension Option.
- (c) It is understood and agreed that the Second Extension Option hereby granted is personal to Original Tenant and is not transferable except to a Permitted Transferee in connection with an assignment of Tenant's entire interest in this Lease. In the event of any assignment or subletting of the Premises or any part thereof (other than to a Permitted Transferee), the Second Extension Option shall automatically terminate and shall thereafter be null and void.
- (d) Tenant's exercise of the Second Extension Option shall, if Landlord so elects in its absolute discretion, be ineffective in the event that (i) an Event of Default by Tenant remains uncured at the time of delivery of the Second Exercise Notice or at the commencement of the Second Extension Period or (ii) Tenant shall have reduced the size of the Premises below 23,993 rentable square feet by agreement with Landlord or pursuant to an express right in this Lease.
- (e) The Second Extension Option shall terminate and shall thereafter be null and void in the event Tenant does not exercise the First Extension Option or for any reason Tenant's exercise of the First Extension Option is ineffective.
- 51.3 Fair Market Rental Value. The provisions of this Section shall apply in any instance in which this Lease provides that the Fair Market Rental Value is to apply.
  - (a) "Fair Market Rental Value" means the annual amount per square foot that a willing tenant would pay and a willing landlord would accept in arm's length negotiations, without any additional inducements, for a lease of the applicable space on the applicable terms and conditions for the applicable period of time (other than Base Year, which shall be the calendar year in which the first day of the applicable Extension Period falls). Fair Market Rental Value shall be determined considering the most recent new direct leases (and market renewals and extensions, if applicable) in the Building and in Comparable Buildings owned or managed by Landlord in the Market Area.
    - (b) In determining the rental rate of comparable space, the parties shall include all escalations and take into consideration the following concessions:
      - (i) Rental abatement concessions, if any, being granted to tenants in connection with the comparable space;
      - (ii) Tenant improvements or allowances provided or to be provided for the comparable space, taking into account the value of the existing

improvements in the Premises, based on the age, quality, and layout of the improvements;

- (iii) Brokerage commissions paid in comparable transactions;
- (iv) The amount of parking rent, if any, per parking permit paid;
- (v) Operating expense and tax protection granted such as a base year or expense stop; and
- (vi) Lease takeover payments.
- (c) If in determining the Fair Market Rental Value the parties determine that the economic terms of leases of comparable space include a tenant improvement allowance, Landlord may, at Landlord's sole option, elect to do the following:
  - (i) Grant some or all of the value of the tenant improvement allowance as an allowance for the refurbishment of the Premises; and
  - (ii) Reduce the Base Rent component of the Fair Market Rental Value to be an effective rental rate that takes into consideration the total dollar value of that portion of the tenant improvement allowance that Landlord has elected not to grant to Tenant (in which case that portion of the tenant improvement allowance evidenced in the effective rental rate shall not be granted to Tenant).
  - 51.4 Determination of Fair Market Rental Value. The determination of Fair Market Rental Value shall be as provided in this Section 51.4.
- (a) <u>Negotiated Agreement</u>. Landlord and Tenant shall diligently attempt in good faith to agree on the Fair Market Rental Value on or before the date (the "Outside Agreement Date") that is five (5) months prior to the date upon which the applicable Extension Period is to commence.
- (b) <u>Parties' Separate Determinations</u>. If Landlord and Tenant fail to reach agreement on or before the Outside Agreement Date, Landlord and Tenant shall each make a separate determination of the Fair Market Rental Value and notify the other party of this determination within fifteen (15) business days after the Outside Agreement Date.
  - (i) <u>Two Determinations</u>. If each party makes a timely determination of the Fair Market Rental Value, those determinations shall be submitted to arbitration in accordance with subsection (c).
    - (ii) One Determination. If Landlord or Tenant fails to make a determination of the Fair Market Rental Value within thefifteen-day period, that

failure shall be conclusively considered to be that party's approval of the Fair Market Rental Value submitted within the fifteen-day period by the other party.

- (c) <u>Arbitration</u>. If both parties make timely individual determinations of the Fair Market Rental Value under subsection (b), the Fair Market Rental Value shall be determined by arbitration under this subsection (c).
  - (i) <u>Scope of Arbitration</u>. The determination of the arbitrators shall be limited to the sole issue of whether Landlord's or Tenant's submitted Fair Market Rental Value is the closest to the actual Fair Market Rental Value as determined by the arbitrators, taking into account the requirements of <u>Section 51.3</u>.
  - (ii) Qualifications of Arbitrator(s). The arbitrators must be licensed real estate brokers who have been active in the leasing of commercial multi-story properties in the Market Area over the five-year period ending on the date of their appointment as arbitrator(s) and who as an individual has not personally represented either party or its affiliates during such period.
  - (iii) <u>Parties' Appointment of Arbitrators</u>. Within fifteen (15) days after the Outside Agreement Date, Landlord and Tenant shall each appoint one arbitrator and notify the other party of the arbitrator's name and business address.
  - (iv) Appointment of Third Arbitrator. If each party timely appoints an arbitrator, the two (2) arbitrators shall, within ten (10) days after the appointment of the second arbitrator, agree on and appoint a third arbitrator (who shall be qualified under the same criteria set forth above for qualification of the initial two (2) arbitrators) and provide notice to Landlord and Tenant of the arbitrator's name and business address. The parties shall not directly or indirectly consult with the third arbitrator regarding his or her opinion of the Fair Market Rental Value prior to such appointment.
  - (v) <u>Arbitrators' Decision</u>. Within thirty (30) days after the appointment of the third arbitrator, the three (3) arbitrators shall decide whether the parties will use Landlord's or Tenant's submitted Fair Market Rental Value and shall notify Landlord and Tenant of their decision. The decision of the majority the three (3) arbitrators shall be binding on Landlord and Tenant.
  - (vi) If Only One Arbitrator is Appointed. If either Landlord or Tenant fails to appoint an arbitrator within fifteen (15) days after the Outside Agreement Date, the arbitrator timely appointed by one of them shall reach a decision and notify Landlord and Tenant of that decision within thirty (30) days after the arbitrator's appointment. The arbitrator's decision shall be binding on Landlord and Tenant.
    - (vii) If Only Two Arbitrators Are Appointed. If each party appoints an arbitrator in a timely manner, but the two (2) arbitrators fail to agree

on and appoint a third arbitrator within the required period, the arbitrators shall be dismissed without delay and the issue of Fair Market Rental Value shall be submitted to binding arbitration under the real estate arbitration rules of JAMS, subject to the provisions of this section.

(viii) If No Arbitrator Is Appointed. If Landlord and Tenant each fail to appoint an arbitrator in a timely manner, the matter to be decided shall be submitted without delay to binding arbitration under the real estate arbitration rules of JAMS subject the provisions of this Section 51.4(c).

(ix) Cost of Arbitration. Each party shall pay its own arbitrator. The other costs of the arbitration and the third arbitrator shall be split between the parties.

# ARTICLE 52 TELECOMMUNICATIONS LINES AND EQUIPMENT

52.1 Tenant may install, maintain, replace, remove or use any electrical, communications or computer wires and cables (collectively, the "Lines") at the Project in or serving solely the Premises, provided that (i) Tenant shall obtain Landlord's prior written consent, which consent shall not be unreasonably withheld, use an experienced and qualified contractor approved in writing by Landlord, and comply with all of the other provisions of <a href="Articles 8">Articles 8</a> and <a href="Articles 8">15</a> of this Lease, (ii) an acceptable number of spare Lines and space for additional Lines shall be maintained for existing and future occupants of the Project, as determined in Landlord's reasonable opinion, (iii) the Lines therefor (including riser cables) shall be appropriately insulated to prevent excessive electromagnetic fields or radiation, and shall be surrounded by a protective conduit reasonably acceptable to Landlord, (iv) any new or existing Lines servicing the Premises shall comply with all applicable governmental laws and regulations, (v) as a condition to permitting the installation of new Lines, Landlord may require that Tenant remove existing Lines located in or serving the Premises and repair any damage in connection with such removal, and (vi) Tenant shall pay all costs in connection therewith. Landlord reserves the right to require that Tenant remove any Lines located in or serving the Premises which are installed in violation of these provisions, or which are at any time in violation of any laws or represent a dangerous or potentially dangerous condition. Landlord further reserves the right to require that Tenant remove any earlier termination of this Lease.

### ARTICLE 53 ERISA

53.1 It is understood that from time to time during the Lease Term, Landlord may be subject to the provisions of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") and as a result may be prohibited by law from engaging in certain transactions. Tenant represents and warrants to the best of its knowledge after due inquiry that at the time this Lease is entered into and at any time thereafter when its terms are amended or modified, neither Tenant nor its affiliates (within the meaning of part VI(c) of Department of Labor Prohibited Transaction Class Exemption 84-14 ("PTE 84-14", as amended), has or will

have the authority to appoint or terminate The Prudential Insurance Company of America ("Prudential") as an investment manager to any employee benefit plan then holding a ten percent (10%) or greater interest in the Prudential separate account PRISA II, nor the authority to negotiate the terms of any management agreement between Prudential and any such employee pension benefit plan for its investment in PRISA II. Further, Tenant is not "related" to Prudential within the meaning of part VI(h) of PTE 84-14.

# ARTICLE 54 TENANT'S RIGHT OF FIRST OFFER

- 54.1 As used herein, "Offer Space" means up to 23,038 square feet of rentable space located on the third (3<sup>rd</sup>) floor of the Building that is contiguous to the Premises (including without limitation the space currently occupied by McCarthy Cook on the date of this Lease and known as Suite 350 ("Suite 350") and the Balcony adjoining the Suite 350). If, at any time during the Term, any portion of the Offer Space become Available (as defined below), Landlord shall give Tenant a written notice (the "Availability Notice") identifying the particular Offer Space (the "Specific Offer Space") that is Available (as defined below), and give Tenant the first opportunity to lease such Specific Offer Space prior to leasing the space to any other tenant or potential tenant. As used herein, "Available" means that the space (i) is not part of the Premises, (ii) is not then subject to any rights of a tenant to renew their lease or expand their premises as set forth in their lease, and (iv) Landlord has received an bona fide offer from a third party to lease the applicable space.
- 54.2 Landlord agrees that, subject to the other terms and conditions of this Article 54, Suite 350 shall be Available on (and Landlord shall give Tenant an Availability Notice with respect to Suite 350 that provides for delivery of Suite 350 on or about) the date (the "Suite 350 Availability Date") that is the earlier of (a) the date that is twenty-four (24) months following the Commencement Date and the (b) date that is nine (9) months after the date upon which Tenant gives Landlord written notice that Tenant desires to lease Suite 350.
- 54.3 The location and configuration of the Specific Offer Space shall be determined by Landlord in its reasonable discretion; provided that Landlord shall not designate Specific Offer Space that would result in any space not included in the Specific Offer Space being not Configured For Leasing (as defined below). For purposes of this Lease, "Configured For Leasing" means the applicable space must have convenient access to the central corridor on the applicable floor and must have a size and configuration that complies with all applicable building codes and other laws and is such that Landlord judges, in its reasonable discretion, that Landlord will be able to lease such space to a third party. The Availability Notice shall:
  - (a) Describe the particular Specific Offer Space (including rentable area, usable area and location);
  - (b) Include an attached floor plan identifying such space;
  - (c) State the date (the "Specific Offer Space Delivery Date") the space will be available for delivery to Tenant;

- (d) (i) if the Specific Offer Space Delivery Date is on or before the second (2nd) anniversary of the Commencement Date, state the Base Rent, which shall be equal to, and adjust at the same times and by the same percentage as, the amount of Base Rent due for the Premises on a per rentable square foot basis (with the exception that the Base Rent abatement period granted to Tenant with respect to the Premises shall be prorated with respect to the Specific Offer Space), and the Allowance shall be as provided in Section 54.6(e) or (ii) if the Specific Offer Space Delivery Date is after the second (2nd) anniversary of the Commencement Date, state Landlord's determination of the Fair Market Rental Value of the Specific Offer Space;
- (e) Specify the increase in the Letter of Credit Required Amount (and any "burn-off" of that increased amount) that will apply to reflect the addition of the Specific Offer Space to the Premises; and
- (f) If the Specific Offer Space Delivery Date is after the second (2nd) anniversary of the Commencement Date, specify the length of the term of the leasing of the Specific Offer Space that will be available (the "Specific Offer Space Term").
- 54.4 If Tenant wishes to exercise Tenant's rights set forth in this Article 54 with respect to the Specific Offer Space, then within ten (10) business days of delivery of the Availability Notice to Tenant, Tenant shall deliver irrevocable notice to Landlord (the "First Offer Exercise Notice") offering to lease the Specific Offer Space on the terms and conditions as may be specified by Landlord in the Availability Notice. If the Specific Offer Space Delivery Date is after the second (2nd) anniversary of the Commencement Date, and Tenant, concurrently with Tenant's exercise of the first offer right, notifies Landlord that it does not accept Landlord's determination of Fair Market Rental Value set forth in the Availability Notice, the Fair Market Rental Value of the Specific Offer Space shall be determined in accordance with the procedures set forth in Article 51 (except that the Base Year shall be the calendar year in which the Specific Offer Space Delivery Date falls and the Outside Agreement Date shall be the date that is thirty (30) days after Tenant delivers the First Offer Exercise Notice). Otherwise, the Fair Market Rental Value shall be as set forth in the Availability Notice.
- 54.5 In the event Tenant fails to give a First Offer Exercise Notice in response to any Availability Notice, Tenant shall have no further rights to receive an Availability Notice and Tenant's rights under this Article 54 shall terminate with respect to the Specific Offer Space identified in that Availability Notice and Landlord shall be free to lease such Specific Offer Space to anyone on any terms at any time during the Term, without any obligation to provide Tenant with any further right to lease that space; provided, however, that if the net present value of the economic terms of Landlord's proposed lease to such third party are more than ten percent (10%) more favorable to the third party (determined using an eight percent (8%) discount rate) than the net present value of the economic terms proposed by Landlord in the Availability Notice (determined using an eight percent (8%) discount rate), then before entering into such third party lease Landlord shall notify Tenant of such materially more favorable economic terms and Tenant shall have the right to lease the Specific Offer Space upon such materially more favorable economic terms by delivering written notice thereof to Landlord within ten (10) business days after Tenant's receipt of Landlord's notice; and provided further, however, if Landlord does not enter into a lease or leases pertaining to the entire Specific Offer Space identified by Landlord in

such Availability Notice within nine (9) months after the date Landlord first delivered such Availability Notice to Tenant, then Tenant's rights under this Article 54 to receive an Availability Notice shall again apply with respect to the unleased portion of such Specific Offer Space that is Available. In the event in the event Landlord leases such Specific Offer Space to someone other than Tenant as provided in this Section 54.5 (each, a "Superior Lease"), then following the expiration or earlier termination of the Superior Lease including without limitation any renewal or extension of such Superior Lease, whether or not such renewal or extension is pursuant to an express written provision in such Superior Lease, and regardless of whether any such renewal or extension is consummated pursuant to a lease amendment or a new lease, Landlord shall again be obligated to comply with the terms of this Article 54 with respect to such space.

54.6 If Tenant timely and validly gives the First Offer Exercise Notice, then beginning on the Specific Offer Space Delivery Date and continuing (i) if the Specific Offer Space Delivery Date is on or before the fourth (4th) anniversary of the Commencement Date, for the balance of the Term (including any extensions), or (ii) if the Specific Offer Space Delivery Date is after the fourth (4th) anniversary of the Commencement Date, for the Specific Offer Space Term:

- (a) The Specific Offer Space shall be part of the Premises under this Lease (so that the term "Premises" in this Lease shall refer to the space in the Premises immediately before the Specific Offer Space Delivery Date plus the Specific Offer Space);
  - (b) Tenant's Building Percentage shall be adjusted to reflect the increased rentable area of the Premises;
  - (c) Base Rent for the Specific Offer Space shall be as specified in the Availability Notice, subject to Section 54.4;
- (d) The amount of the Letter of Credit Tenant must provide (and any "burn-off" of that increased amount) shall be increased by the amounts specified in the Availability Notice;
- (e) Tenant's lease of the Specific Offer Space shall be on the same terms and conditions as affect the original Premises from time to time, except as otherwise provided in this section. Tenant's obligation to pay Rent with respect to the Specific Offer Space shall begin on the Specific Offer Space Delivery Date. The Specific Offer Space shall be leased to Tenant in its "as-is" condition and Landlord shall not be required to construct improvements in, or contribute any tenant improvement allowance for, the Specific Offer Space; provided that (1) except with respect to Suite 350, if the Specific Offer Space Delivery Date is on or before the second (2nd) anniversary of the Commencement Date, Landlord shall construct improvements in the Specific Offer Space generally in accordance with the terms of Exhibit C, except that (i) the Allowance for the Specific Offer Space shall be equal to the Allowance for the Premises on a per rentable square foot basis, and (ii) the Allowance for the Specific Offer Space and the amount of the unused Allowance for the Specific Offer Space that may be used as a credit against Base Rent will be prorated by multiplying each of those amounts by a fraction, (i) the

numerator of which is the number of months remaining in the Term after the Specific Offer Space Delivery Date, and (ii) the denominator of which is eighty-two (82); (2) except with respect to Suite 350, if the Specific Offer Space Delivery Date is after the second (2nd) anniversary of the Commencement Date, Landlord shall provide free rent and a tenant improvement allowance to the extent (if any) included in the Fair Market Rental Value; and (3) with respect to Suite 350, the tenant improvement allowance shall be \$15.00 per square foot of rentable area and, in determining Fair Market Rental Value, tenant improvements and tenant improvement allowances shall be ignored and Landlord shall provide free rent to the extent (if any) included in the Fair Market Rental Value; Tenant's construction of any improvements in the Specific Offer Space shall comply with the terms of this Lease concerning alterations; and

- (f) If requested by Landlord, Landlord and Tenant shall confirm in writing the addition of the Specific Offer Space to the Premises on the terms and conditions set forth in this section, but Tenant's failure to execute or deliver such written confirmation shall not affect the enforceability of the First Offer Exercise Notice.
- 54.7 Tenant's rights and Landlord's obligations under this <u>Article 54</u> are expressly subject to and conditioned upon there not existing an Event of Default by Tenant under this Lease, either at the time of delivery of the First Offer Exercise Notice or at the time the Specific Offer Space is to be added to the Premises.
- 54.8 It is understood and agreed that Tenant's rights under this Article 54 are personal to Original Tenant and not transferable except to a Permitted Transferee that is an assignee of Original Tenant's entire interest in this Lease. In the event of any assignment of this Lease except to a Permitted Transferee that is an assignee of Original Tenant's entire interest in this Lease, this expansion right shall automatically terminate and shall thereafter be null and void.

### ARTICLE 55 TEMPORARY SPACE

- 55.1 Prior to the execution of this Lease, Landlord and Tenant entered into a Temporary Use License Agreement (the "Temporary License Agreement") with respect to approximately 20,558 rentable square feet of space comprising the entire ninth (9th) floor of the Building as shown on Exhibit B-1 of this Lease (the "Temporary Space"), as it may be increased by mutual agreement of the parties in their respective sole discretion.
- 55.2 Effective on the date (the "Temporary Space Commencement Date") that is the latest of the full and final execution of this Lease by Landlord and Tenant, delivery of the Letter of Credit and all prepaid rental, if any, required under this Lease, delivery of all initial certificates of insurance required by this Lease (which certificates of insurance shall specifically cover both the Temporary Space during the Temporary Space Term, as hereinafter defined, and the Premises) the Temporary License Agreement shall terminate and Tenant's occupancy and use of the Temporary Space shall be governed by the terms of this Article 55. Commencing on the Temporary Space Commencement Date, and ending on the date that is thirty (30) days after the Commencement Date of this Lease (such period being referred to herein as the "Temporary

Space Term"), Landlord shall allow Tenant to use the Temporary Space for the uses permitted by this Lease. During the Temporary Space Term, the Temporary Space shall be deemed the "Premises" for purposes of Section 13 and Article 21 of the Lease. Such Temporary Space shall be accepted by Tenant in its "as-is" condition and configuration, it being agreed that Landlord shall be under no obligation to perform any work in the Temporary Space or to incur any costs in connection with Tenant's move in, move out or occupancy of the Temporary Space. Tenant acknowledges that it shall be entitled to use and occupy the Temporary Space at its sole cost, expense and risk. Tenant shall not construct any improvements or make any alterations of any type to the Temporary Space without the prior written consent of Landlord. All costs in connection with making the Temporary Space ready for occupancy by Tenant shall be the sole responsibility of Tenant. Tenant shall have no right to sublease or assign the Temporary Space.

- 55.3 The Temporary Space shall be subject to all the terms and conditions of the Lease except as expressly modified herein, provided that the Base Rent payable for the Temporary Space during the Temporary Space Term shall equal \$24,669.60 per month (the "Temporary Space Base Rent"). For clarity, the Temporary Space Base Rent represents a payment of \$1.20 per square foot per month that is the equivalent of Landlord's estimate of the monthly Operating Expenses and Taxes. The Temporary Space Base Rent shall be payable on or before the commencement of the Temporary Space Term and on the first day of each month thereafter during the Temporary Space Term. Any prorated amount of the fee paid by Tenant under the Temporary License Agreement with respect to the period after the termination of the Temporary License Agreement shall be credited against Tenant's obligation to pay Temporary Space Base Rent under this Lease. Tenant shall not be entitled to receive any allowances, abatement or other financial concession in connection with the Temporary Space which was granted with respect to the Premises unless such concessions are expressly provided for herein with respect to the Temporary Space, and the Temporary Space shall not be subject to any renewal or expansion rights of Tenant under the Lease.
- 55.4 On or before termination of the Temporary Space Term, Tenant shall vacate the Temporary Space and deliver the same to Landlord in the same condition that the Temporary Space was delivered to Tenant, ordinary wear and tear excepted and damage by casualty excepted. At the expiration or earlier termination of the Temporary Space Term, Tenant shall remove all debris, all items of Tenant's personalty, and any trade fixtures of Tenant from the Temporary Space. Tenant shall be fully liable for all damage Tenant or Tenant's agents, employees, contractors, or subcontractors cause to the Temporary Space.
- 55.5 Tenant shall have no right to hold over or otherwise occupy the Temporary Space at any time following the expiration or earlier termination of the Temporary Space Term, and in the event of such holdover, Landlord shall immediately be entitled to institute dispossessory proceedings to recover possession of the Temporary Space, without first providing notice thereof to Tenant. In the event of holding over by Tenant after expiration or termination of the Temporary Space Term without the written authorization of Landlord, Tenant shall pay, for such holding over, \$98,678.40 per month for each month or partial month of holdover, plus all consequential damages that Landlord incurs as a result of the Tenant's hold over after the date that is the later of (a) the date that is thirty (30) days after the expiration or earlier termination of the Temporary Space Term, or (b) the date that is thirty (30) days after Landlord has notified Tenant that Landlord has executed a letter of intent or lease with another

tenant for all or any portion of the Temporary Space. During any such holdover, Tenant's occupancy of the Temporary Space shall be deemed that of a tenant at sufferance, and in no event, either during the Temporary Space Term or during any holdover by Tenant, shall Tenant be determined to be a tenant-at-will under applicable law. While Tenant is occupying the Temporary Space, Landlord or Landlord's authorized agents shall be entitled to enter the Temporary Space, upon reasonable notice, to display the Temporary Space to prospective tenants.

- 55.6 Landlord hereby leases to Tenant, and Tenant hereby hires from Landlord, the Personal Property listed on Exhibit G hereto which is located in the Temporary Space (the "Included Personal Property"). Tenant shall:
  - (a) accept the Included Personal Property in its "as is" condition as of the date hereof, as the same may be affected by reasonable wear and tear after the date hereof;
  - (b) insure the Included Personal Property against loss or damage by fire or other casualty (and all of the provisions of this Lease applicable to insurance required to be carried by Tenant shall be applicable thereto); and
  - (c) surrender the Included Personal Property to Landlord in the Temporary Space upon the expiration or sooner termination of the Temporary Space Term in the same condition as at the commencement of this Lease, as the same may be affected by reasonable wear and tear or damage by fire or other casualty; provided, however, that if the Included Personal Property shall have been damaged by fire or other casualty and not repaired or replaced then upon such expiration or sooner termination Tenant shall pay to Landlord the full replacement cost thereof.

### ARTICLE 56 24 HOUR FITNESS CLUB

56.1 During the Term, Landlord shall provide Tenant with twelve (12) free memberships for Tenant's employees to the 24 Hour Fitness Club located at the Project so long as that 24 Hour Fitness Club is in operation in the Project. Tenant's use of the memberships and the 24 Hour Fitness Club shall be subject to the rules, regulations and requirements of 24 Hour Fitness, and shall be at the sole risk and hazard of Tenant, without representation or warranty of any kind from Landlord.

# ARTICLE 57 TENANT'S ROOFTOP AND EXTERIOR ANTENNA RIGHTS

57.1 During the Term and subject to the terms of this Article 57, Tenant may install (a) on the roof of the Building telecommunications antennae, microwave dishes or other communication equipment, as necessary for the operation of Tenant's business within the Premises, including any cabling or wiring necessary to connect this rooftop equipment to the Premises (collectively, the "Rooftop Equipment"), and (b) on the exterior of the Building,

wireless network antennae, as necessary for the operation of Tenant's business within the Premises, including any cabling or wiring necessary to connect this wireless network antennae to the Premises (collectively, the "Wireless Equipment"). The Rooftop Equipment and the Wireless Equipment are sometimes referred to herein collectively as the "Exterior Equipment". If Tenant wishes to install any Exterior Equipment, Tenant shall first notify Landlord in writing, which notice shall fully describe the Exterior Equipment, including, without limitation, its purpose, weight, size and desired location on the Building and its intended method of connection to the Premises. All of Tenant's Rooftop Equipment must be located within a total aggregate area not to exceed 4 square feet, at locations reasonably approved by Landlord prior to any installation. Landlord hereby consents to the installation of Rooftop Equipment consisting of one (1) roof-mounted antennae and/or satellite dish (collectively, the "Initial Rooftop Equipment"). Landlord also reserves the right to restrict the number and size of dishes, antennae and other Rooftop Equipment in addition to the Initial Rooftop Equipment installed on the roof of the Building in its sole discretion. The location of Tenant's Exterior Equipment must be located within a total aggregate area of not to exceed four (4) square feet, at locations approved by Landlord in its sole discretion prior to any installation. Landlord hereby consents to the installation of Wireless Equipment consisting of antennae co-located with each of Landlord's wireless network antennae on the exterior of the Building (collectively, the "Initial Wireless Equipment"). Landlord reserves the right to restrict the number and size of wireless network antennae in addition to the Initial Wireless Equipment installed on the Building in its sole discretion.

57.2 Tenant will be solely responsible, at Tenant's sole expense, for the installation, maintenance, repair and removal of the Exterior Equipment, and Tenant shall at all times maintain the Exterior Equipment in good condition and repair. Landlord agrees that the named Tenant hereunder shall not pay any rental charge for Tenant's use of the rooftop or Building exterior pursuant to the terms of this <a href="Article 57">Article 57</a> for the Initial Rooftop Equipment and the Initial Wireless Equipment, provided, however, if any successor to the named Tenant wishes to utilize rooftop space or if Tenant seeks to use rooftop space for Rooftop Equipment in addition to the Initial Rooftop Equipment or Building exterior for Wireless Equipment in addition to the Initial Wireless Equipment, Landlord reserves the right to impose a charge for such use, which shall be consistent with market rates.

57.3 The installation of the Exterior Equipment shall constitute an alteration and shall be performed in accordance with and subject to the provisions of Article 15 of this Lease. Tenant shall comply with all applicable laws, rules and regulations relating to the installation, maintenance and operation of Exterior Equipment at the Building (including, without limitation, all construction rules and regulations) and will pay all costs and expenses relating to such Exterior Equipment, including the cost of obtaining and maintaining any necessary permits or approvals for the installation, operation and maintenance thereof in compliance with applicable laws, rules and regulations. The installation, operation and maintenance of the Exterior Equipment at the Building shall not adversely affect the structure or operating systems of the Building or the business operations of any other tenant or occupant at the Building. For purposes of determining Landlord's and Tenant's respective rights and obligations with respect to the use of the roof, the portion of the roof affected by the Rooftop Equipment shall be deemed to be a portion of the Premises (provided that such portion shall not be measured for purposes of determining the area of the Premises); consequently, all of the provisions of this Lease respecting Tenant's obligations hereunder shall apply to the installation,

use and maintenance of the Exterior Equipment, including without limitation provisions relating to compliance with requirements as to insurance, indemnity, repairs and maintenance. Tenant shall be solely responsible for the repair, maintenance, replacement and removal of the Wireless Equipment. Tenant's Wireless Equipment must be collated with Landlord's existing campus wireless network system and antennae. Tenant may install cabling and wiring through the Building interior conduits, risers, and pathways of the Building in accordance with <a href="https://example.com/repairs/https://

57.4 Tenant's right to install and maintain Exterior Equipment is non-exclusive and is subject to termination or revocation as set forth herein, including pursuant to Section 22.2(b) of this Lease. Landlord shall be entitled to all revenue from use of the roof other than revenue from the Exterior Equipment installed by Tenant. Subject to the terms set forth below in this Section 57.4, Landlord at its election may require the relocation, reconfiguration or removal of the Exterior Equipment, if (a) in Landlord's reasonable judgment the Exterior Equipment is interfering with the use of the rooftop for the helipad (if any) or other Building operations (including without limitation maintenance, repairs and replacements of the roof), (b) in Landlord's reasonable judgment the Exterior Equipment is interfering with the business operations of other tenants or occupants of the Building, (c) in Landlord's reasonable judgment the Exterior Equipment is causing damage to the Building or (d) Tenant otherwise fails to comply with the terms of this Article 57. If relocation or reconfiguration becomes necessary due to interference difficulties, Landlord and Tenant will reasonably cooperate in good faith to agree upon an alternative location or configuration that will permit the operation of the Exterior Equipment for Tenant's business at the Premises without interfering with other operations at the Building or communications uses of other tenants or occupants. If removal is required due to any breach or default by Tenant under the terms of this Article 57, Tenant shall remove the Exterior Equipment upon thirty (30) days' written notice from Landlord. Any relocation, removal or reconfiguration of the Exterior Equipment as provided above shall be at Tenant's sole cost and expense. In addition to the other rights of relocation and removal as set forth herein, Landlord reserves the right to require relocation of Tenant's Exterior Equipment at any time at its election at Landlord's cost (but not more frequently than once per year) so long as Tenant is able to continue operating its Exterior Equipment in substantially the same manner as it was operated prior to its relocation. In connection with any relocation of Tenant's Exterior Equipment at the request of or required by Landlord (other than in the case of a default by Tenant hereunder), Landlord shall provide Tenant with at least thirty (30) days' prior written notice of the required relocation and will conduct the relocation in a commercially reasonable manner and in such a way that will, to the extent reasonably possible, prevent interference with the normal operation of Tenant's Exterior Equipment. In connection with any relocation, Landlord further agrees to work with Tenant in good faith to relocate Tenant's Exterior Equipment to a location that will permit its normal operation for Tenant's business operations. Landlord acknowledges that relocation of Tenant's Exterior Equipment may be disruptive to Tenant's business and, without limiting its rights to require such removal, confirms that it will not exercise its rights hereunder in a bad faith manner or for the purpose of harassing or causing a hardship to Tenant.

57.5 If Tenant fails to comply with the terms of this <u>Article 57</u> within thirty (30) days following written notice by Landlord (or such longer period as may be reasonably required to comply so long as Tenant is diligently attempting to comply), Landlord may take such action as may be necessary to comply with these requirements. In such event.

Tenant agrees to reimburse Landlord for all costs incurred by Landlord to effect any such maintenance, removal or other compliance subject to the terms of this Article 57, including interest on all such amounts, accruing from the date which is ten (10) days after the date of Landlord's demand until the date paid in full by Tenant, with all such amounts being Additional Rent under this Lease.

57.6 Tenant agrees to indemnify Landlord, its partners, agents, officers, directors, employees and representatives from and against any and all liability, expense, loss or damage of any kind or nature from any suits, claims or demands, including reasonable attorneys' fees, arising out of Tenant's installation, operation, maintenance, repair, relocation or removal of the Exterior Equipment, except to the extent any such liability, expense, loss or damage results from the gross negligence or intentional misconduct of Landlord or its agents, partners, officers, directors, employees, contractors or representatives. At the expiration or earlier termination of the Lease, Tenant may and, upon request by Landlord, shall remove all of the Exterior Equipment, including any wiring or cabling relating thereto, at Tenant's sole cost and expense and will repair at Tenant's cost any damage resulting from such removal. If Landlord does not require such removal, any Exterior Equipment remaining at the Building after the expiration or earlier termination of this Lease which is not removed by Tenant shall be deemed abandoned and shall become the property of Landlord.

57.7 Subject to compliance with the construction rules for the Building and Landlord's reasonable and nondiscriminatory rules and regulations regarding access to the roof and, upon receipt of Landlord's prior written consent to such activity (which shall not be unreasonably withheld, conditioned or delayed), Tenant and its representatives shall have access to and the right to go upon the roof of the Building, on a seven (7) day per week, twenty-four (24) hour basis, to exercise its rights and perform its obligations under this <a href="Article 57">Article 57</a>. Tenant acknowledges that, except in the case of an emergency or when a Building engineer is not made available to Tenant in sufficient time to allow Tenant to avoid or minimize interruption of use of the Rooftop Equipment, advance notice is required and a Building engineer must accompany all persons gaining access to the rooftop. Tenant may install Exterior Equipment at the Building only in connection with its business operations at the Premises, and may not lease or license any rights or equipment to third parties or allow the use of any rooftop equipment or wireless network by any party other than Tenant. Tenant acknowledges that Landlord has made no representation or warranty as to Tenant's ability to operate Exterior Equipment at the Building and Tenant acknowledges that helicopters, other equipment installations and other structures and activities at or around the Building may result in interference with Tenant's Exterior Equipment. Except as set forth in this <a href="Article 57">Article 57</a>, Landlord shall have no obligation to prevent, minimize or in any way limit or control any existing or future interference with Tenant's Exterior Equipment.

# ARTICLE 58 PATIO AND BALCONY AREAS

58.1 Tenant shall have the exclusive use of the balcony adjacent to Suite 200 as indicated on Exhibit B-2 to this Lease (the "Balcony Area") and one patio on the West side of Suite 100 and two patios on the South side of Suite 100 as indicated on Exhibit B-3 to this Lease (collectively, the "Patio Area"). The Balcony Area and the Patio Area are sometimes referred to

herein collectively as the "Patio and Balcony Areas." Tenant shall be solely responsible for the cleaning and maintenance of the Patio and Balcony Areas and shall maintain the Patio and Balcony Areas in a clean and neat condition at all times consistent with the balance of the Common Areas as determined by Landlord. Tenant shall be responsible for all furniture, fixtures and equipment ("FF&E") installed in the Patio and Balcony Areas and all of Tenant's FF&E selections shall be subject to Landlord's prior written approval, which shall not be unreasonably withheld. Landlord shall have no liability with respect to any loss, damage or theft of Tenant's FF&E unless caused by Landlord's gross negligence or willful misconduct. Tenant shall not have any right to assign or sublet all or any portion of the Patio and Balcony Areas separate from the Premises. The Patio and Balcony Areas shall be deemed part of the Premises; provided that it shall not be included in the calculation of the rentable square footage of the Premises. In addition to the improvements described in the Construction Drawings (as defined in Exhibit C to this Lease), Landlord shall also construct, at Landlord's sole expense, (a) with respect to the Balcony Area: full-height sliding glass doors, hardscape flooring installation, base building hardscape barriers and other landscape and privacy screening elements, and (b) with respect to the Patio Area: full-height sliding glass doors on the west and south sides of the Premises, hardscape flooring installation, lush landscape privacy screening elements, an exclusive ground floor exterior entrance off the parking structure, and a security perimeter surrounding the first floor patio, each in accordance with Landlord's Building standards for such improvements. Any security system used on Tenant's security perimeter must be installed by Tenant in accordance with this Lease at Tenant's sole cost and expense (the "Perimeter Security System"). Subject to Article 57, Tenant shall not enclose, modify, alter or improve, or install any telecommunications, electrical or other systems (other than the Perimeter Security System) in or on, the Patio and Balcony Areas; provided that notwithstanding the foregoing, Tenant may, at Tenant's sole cost and expense, install electrical service to the Patio and Balcony Areas subject to Landlord's prior written approval, which shall not be unreasonably withheld. Tenant acknowledges that as of the date of this Lease, additional levels of parking are being added to the existing parking structures adjacent to the Premises and that the construction on the parking structures must be coordinated with the construction of the Patio Area. Accordingly, delays affecting the construction of the parking structure may affect the completion of the Patio Area improvements. In the event the Patio Area improvements are not completed by March 31, 2018, for any reason other than Tenant Delays, Tenant as Tenant's sole and exclusive remedy therefor shall receive a daily rent credit in the amount of \$250.00 per day until the Patio Area is usable by Tenant.

## ARTICLE 59 TENANT SECURITY SYSTEMS

59.1 Subject to compliance with the terms of Article 15, as applicable, Tenant shall have the right to install as an Alteration, at Tenant's sole cost and expense, a surveillance camera system (the "Security Camera System") inside of the Premises together with a security camera on the exterior of the Building in a location approved by Landlord, which shall not be unreasonably withheld; provided that (a) any such Security Camera System installed by Tenant complies with all Applicable Laws, (b) the plans and specifications for any such Security Camera System shall be subject to Landlord's prior written approval, which consent shall not be unreasonably withheld, and (c) any such Security Camera System shall be compatible with the Building's structure and systems. In no event shall Landlord have any liability or responsibility for the operation, maintenance or monitoring of any Security Camera System installed by Tenant

and any additional costs incurred for any such monitoring, operation or maintenance and for any modification to the Building necessary as a result of the installation or operation of such Security Camera System shall be paid by Tenant. On the expiration or earlier termination of this Lease, Tenant shall remove the Security Camera System, the Perimeter Security System and all related wiring and systems and restore the Premises and portions of the Building affected by the Security Camera System and the Perimeter Security System to the condition existing prior to the installation of that Security Camera System and the Perimeter Security System at Tenant's sole cost and expense.

59.2 At all times during the Term, Tenant may provide all security operations and equipment which Tenant may from time to time determine to be necessary or appropriate to serve the particular needs of Tenant in connection with its use of the Premises, including without limitation the Security Camera System ("Tenant's Security"). All costs and expenses with respect to Tenant's Security (including, but not limited to, insurance for tort liability and uninsured liability arising out of the acts and omissions of Tenant's Security personnel) shall be borne by Tenant. The standards of quality and performance and the level of service for Tenant's Security shall be established by Tenant in its reasonable discretion, giving weight to the degree of anticipated risk and the effect of such Security functions on the other tenants in the Building, and Tenant shall consult with Landlord from time to time and use all reasonable efforts to resolve to the reasonable satisfaction of Landlord any objections to Tenant's Security. Tenant's right to exercise reasonable discretion shall not however, limit or otherwise affect Tenant's obligations to Landlord pursuant to any other provision of this Lease. All costs related to the risks inherent with respect to Tenant's Security (including but not limited to insurance for tort liability and uninsured liability arising out of the acts or omissions of Tenant's Security personnel) shall be solely the responsibility of Tenant. Landlord and its agents and employees shall have no liability or obligation with respect to such costs, and Tenant hereby indemnifies and holds harmless Landlord and its agents and employees therefrom, except to the extent caused by the gross negligence or willful misconduct of Landlord or its agents or employees. Tenant hereby further indemnifies Landlord and its agents and employees against and holds Landlord and its agents and employees harmless from and against any and all loss, cost, liability, damage and expense, including without limitation, penalties, fines and actual attorneys' fees and costs, which Landlord or its agents or employees may incur arising out of or in connection with the provision by Tenant of all or any portion of Tenant's Security, except to the extent caused by the gross negligence or willful misconduct of Landlord or its agents or employees. Tenant shall comply with all applicable laws, rules and regulations in connection with its provision of security for the Premises, including, without limitation, the carrying and use of firearms by Tenant's Security personnel and the personnel of Tenant's agents and independent contractors retained by Tenant. Tenant hereby indemnifies Landlord and its agents and employees against and holds Landlord and its agents and employees harmless from and against any and all loss, cost, liability, damage and expense, including without limitation, penalties, fines and actual attorneys' fees and costs, which Landlord or its agents or employees may incur arising out of or in connection with the carrying and/or use of firearms by Tenant's Security personnel or the personnel of Tenant's agents or independent contractors.

# ARTICLE 60 LANDLORD'S REPRESENTATIONS AND WARRANTIES

- 60.1 Landlord, in order to induce Tenant to enter into this Lease, hereby represents and warrants that:
  - (a) Landlord is duly organized and validly existing under the laws of the State of Delaware and has full power and authority to enter into this Lease;
- (b) (i) Landlord is the sole fee simple owner of the Building and Project, and (ii) to Landlord's knowledge, none of the CC&R's prohibits or requires consent to the exercise of any of Tenant's material rights under this Lease;
- (c) To Landlord's knowledge, Landlord is not a party to any agreement or litigation the terms of which could reasonably be expected to adversely affect the ability of Landlord to perform its obligations under this Lease or which would constitute a default on the part of Landlord under this Lease, or otherwise adversely affect Tenant's rights or entitlements under this Lease; and
- (d) Landlord has no actual knowledge that the Building is not zoned to permit the use of the Building and Premises for general office purposes. For purposes of this Lease, "Landlord's knowledge" shall be deemed to mean and limited to the current actual knowledge of Thomas McCarthy, Edward Cook, Brian Harnetiaux, Michael Coppin or Megan Allen at the time of execution of this Lease and not any implied, imputed, or constructive knowledge of said individual or of Landlord or any parties related to or comprising Landlord and without any independent investigation or inquiry having been made or any implied duty to investigate or make any inquiries; it being understood and agreed that such individual shall have no personal liability in any manner whatsoever hereunder or otherwise related to the transactions contemplated hereby.

### ARTICLE 61 HELISTOP USE

61.1 Landlord and Tenant shall enter into a helipad access agreement substantially in the form of Exhibit J, granting Tenant non-exclusive access to the helipad located at the 555 Anton Blvd. building in the Project ("Helipad Access Agreement"). The form and substance of the Helipad Access Agreement shall be subject to modification to meet the requirements of any applicable governmental authority having jurisdiction over the use of the helipad. Any termination of this Lease terminates all rights under this Article 61 and the Helipad Access Agreement. Tenant's rights under this Article 61 are personal to the Original Tenant and may be exercised only by the Original Tenant while occupying at least one contiguous, full floor of the Premises. Any assignment or subletting by Tenant of this Lease or of all or a portion of the Premises (such that Tenant and its Permitted Transferees cease to occupy collectively at least 23,993 rentable square feet within the Building) (even if such assignment or subletting does not require the consent of Landlord or is approved by Landlord) terminates Tenant's rights under this Article 61, unless Landlord consents to the contrary in writing at the time of such subletting or assignment, or unless such assignment or subletting is to a Permitted Transferee under this

Lease. Tenant's rights to use the Helipad are subject to rights of any tenants of the Project existing on the Date of this Lease.

### ARTICLE 62 LETTER OF CREDIT

62.1 Letter of Credit. Tenant agrees to provide, at Tenant's sole cost and expense, a Letter of Credit (as defined below) in the Letter of Credit Required Amount (as defined below) as additional security for the faithful performance and observance by Tenant of all of the provisions of this Lease, on the terms and conditions set forth below. The use, application or retention of the Letter of Credit, or any portion thereof, by Landlord shall not prevent Landlord from exercising any other right or remedy provided by this Lease or by law, it being intended that Landlord shall not first be required to proceed against the Letter of Credit and the Letter of Credit shall not operate as a limitation on any recovery to which Landlord may otherwise be entitled. As used herein the term Letter of Credit Required Amount initially means \$606,000.00. Subject to the remaining terms of this Article 62, and provided the Reduction Conditions (as defined below) have been satisfied at the particular reduction effective date, Tenant shall have the right to reduce the Required Amount so that the new Required Amount shall be \$484,800.00 effective as of the first day of the nineteenth (19th) month of the Term. \$363,600.00 effective as of the first day of the thirty-first (31st) month of the Term, \$242,400.00 effective as of the first day of the forty-third (43d) month of the Term, \$151,500.00 effective as of the first day of the fifty-fifth (55th) month of the Term, and \$121,200.00 effective as of the first day of the seventy-third (73d) month of the Term. If Tenant is not entitled to reduce the Required Amount as of a particular reduction effective date due to the failure of one or more of the Reduction Conditions, then as and when the Reduction Conditions are satisfied in full, Tenant shall be entitled to reduce the Required Amount by the amount of the reduction Tenant would have been entitled to had all of the Reduction Conditions been satisfied on such reduction effective date. If Tenant is entitled to a reduction in the Required Amount, Tenant shall provide Landlord with written notice requesting that the Required Amount be reduced as provided above (the Reduction Notice). If Tenant provides Landlord with a Reduction Notice, and Tenant is entitled to reduce the Required Amount as provided herein, the reduction shall be effectuated by Tenant replacing the Letter of Credit then being held by Landlord with a new Letter of Credit in the new Required Amount or amending the then-existing Letter of Credit to that new Required Amount. The term Reduction Conditions means no Event of Default shall have occurred and be continuing under this Lease.

62.2 <u>Delivery of Letter of Credit</u> (a) Tenant shall cause a Letter of Credit, in the amount of the Letter of Credit Required Amount to be issued by the L/C Bank (as defined below) in favor of Landlord, and its successors, assigns and transferees; (b) Tenant will cause the Letter of Credit to remain in full force and effect during the entire Term (or be replaced by a new Letter of Credit issued by the L/C Bank not less than thirty (30) days prior to the expiration date of the prior Letter of Credit) and thereafter until thirty (30) days after expiration or earlier termination of the Lease; and (c) the initial Letter of Credit will be delivered to Landlord upon the execution and delivery of this Lease by Tenant. So long as no Event of Default then exists, Landlord shall return the Letter of Credit to Tenant within thirty (30) days after the Expiration Date. The specific requirements for the Letter of Credit and the rights of Landlord to make draws thereon will be as set forth in this <u>Article 62</u>. All of Tenant's rights and all of Landlord's

obligations under this Lease are strictly contingent on Tenant's delivering and thereafter causing the Letter of Credit to remain in full force and effect during the entire Term.

62.3 Draws on the Letter of Credit. Immediately upon, and at any time or from time to time after, the occurrence of any one or more Draw Events (as defined below), Landlord will have the unconditional right to draw on the Letter of Credit in accordance with this Article 62. Landlord shall provide written notice of any such draw on the Letter of Credit to Tenant on the date of such draw, unless Landlord is prevented from giving notice by application of the bankruptcy code's automatic stay, in which case no such notice shall be required. Upon the payment to Landlord of the Draw Proceeds, Landlord will hold the Draw Proceeds in its own name and for its own account, without liability for interest, to use and apply any and all of the Draw Proceeds only (a) to cure any Event of Default by Tenant; (b) to pay any other sum to which Landlord becomes obligated by reason of Tenant's failure to carry out its obligations under this Lease; or (c) to compensate Landlord for any monetary loss or damage which Landlord suffers thereby that is finally determined by a court to have been proximately caused by Tenant's failure to carry out its obligations under this Lease. In addition, if the Draw Event is the failure of Tenant to renew the Letter of Credit as required hereunder, then Landlord shall be entitled to draw the entire Letter of Credit as a cash security deposit, held as a pledge under the California Uniform Commercial Code to secure Tenant's obligations under this Lease. Landlord will return the amounts so drawn, less any amounts applied as provided herein, upon delivery by Tenant of a replacement Letter of Credit. Among other things, it is expressly understood that the Draw Proceeds will not be considered an advance payment of Base Rent or Additional Rent or a measure of Landlord's damages resulting from any Event of Default hereunder (past, present or future). Further, immediately upon the occurrence and during the continuance of any one or more Draw Events, Landlord may, from time to time and without prejudice to any other remedy, use the Draw Proceeds (whether from a contemporaneous or prior draw on the Letter of Credit) to the extent necessary to (a) cure any Event of Default by Tenant; (b) to pay any other sum to which Landlord becomes obligated by reason of Tenant's failure to carry out its obligations under this Lease; or (c) compensate Landlord for any monetary loss or damage which Landlord suffers thereby that is finally determined by a court to have been proximately caused by Tenant's failure to carry out its obligations under this Lease. Any delays in Landlord's draw on the Letter of Credit or in Landlord's use of the Draw Proceeds as provided in this Article 62 will not constitute a waiver by Landlord of any of its rights hereunder with respect to the Letter of Credit or the Draw Proceeds. Following any such application of the Draw Proceeds, Tenant will either pay to Landlord within five (5) business days following Landlord's demand therefor the cash amount so applied in order to restore the Draw Proceeds to the full amount thereof immediately prior to such application or cause the Letter of Credit to be replenished to its full amount thereunder. Failure to either pay that cash amount or cause the Letter of Credit to be replenished to its full amount thereunder within five (5) business days after written demand shall constitute an Event of Default without the right to any further notice or cure period. Landlord will not be liable for any indirect, consequential, special or punitive damages incurred by Tenant arising from a claim that Landlord violated the bankruptcy code's automatic stay in connection with any draw by Landlord of any Draw Proceeds, Landlord's liability (if any) under such circumstances being limited to the reimbursement of direct costs as and to the extent expressly provided in this Section 62.3. Nothing in this Lease or in the Letter of Credit will confer upon Tenant any property rights or interests in any Draw Proceeds; provided, however, that upon the expiration or earlier termination of this Lease, and so long as there then exist no Draw Events or Events of

Default hereunder, Landlord agrees to return any remaining unapplied balance of the Draw Proceeds then held by Landlord, and the Letter of Credit itself (if and to the extent not previously drawn in full) to Tenant. Landlord may draw on the Letter of Credit and/or apply any Security Deposit in any order.

## 62.4 Applicable Definitions.

Draw Event means each of the following events:

- (a) the occurrence of any one or more of the following which shall have also been preceded, simultaneously accompanied, or succeeded by an Event of Default under this Lease regardless of the absence of any notice of default which might otherwise be required with respect to an Event of Default if the giving of notice to Tenant about such breach by Tenant is stayed or barred due to one of the following events: (i) Tenant's filing of a petition under any chapter of the Bankruptcy Code, or under any federal, state or foreign bankruptcy or insolvency statute now existing or hereafter enacted, or Tenant's making a general assignment or general arrangement for the benefit of creditors, (ii) the filing of an involuntary petition under any chapter of the Bankruptcy Code, or under any federal, state or foreign bankruptcy or insolvency statute now existing or hereafter enacted, or the filing of a petition for adjudication of bankruptcy or for reorganization or rearrangement, by or against Tenant and such filing not being dismissed within sixty (60) days, (iii) the entry of an order for relief under any chapter of the Bankruptcy Code, or under any federal, state or foreign bankruptcy or insolvency statute now existing or hereafter enacted), for Tenant is defined in the Bankruptcy Code (or of an equivalent thereto under any federal, state or foreign bankruptcy or insolvency statute now existing or hereafter enacted), for Tenant, or the appointment of a trustee or receiver to take possession of substantially all of Tenant's assets located at the Premises or of Tenant's interest in this Lease and possession not being restored to Tenant within sixty (60) days, or (v) the subjection of all or substantially all of Tenant's assets located at the Premises or of Tenant's interest in this Lease to attachment, execution or other judicial seizure and such subjection not being discharged within sixty (60) days;
- (b) the failure of Tenant, not less than thirty (30) days prior to the stated expiration date of the Letter of Credit then in effect, to cause an extension, renewal or replacement issuance of the Letter of Credit, to be effected, which extension, renewal or replacement issuance will be made by the L/C Bank, will otherwise meet all of the requirements of the initial Letter of Credit hereunder, which failure will be an Event of Default under this Lease;
- (c) the failure of Tenant to make when due any payment of Base Rent, of any monthly installment of any Additional Rent, or pay any other monetary obligation within five (5) days after receipt of written notice from Landlord that the amount is due (unless the giving of notice to Tenant about such breach by Tenant is stayed or barred due to one of the events listed in subsection (a) above, in which case such five (5) day period shall run from the date the amount is due);

(d) the payment by Landlord of any sum to cure a failure by Tenant to comply with any non-monetary obligation hereunder which Tenant has not cured within thirty (30) days after notice thereof by Landlord (or, if Landlord is prevented from giving notice by application of the bankruptcy code's automatic stay, the payment of Landlord of any sum to cure a failure by Tenant to comply with any non-monetary obligation hereunder that Tenant has not cured within thirty (30) days from the date of the breach).

Draw Proceeds means the proceeds of any draw or draws made by Landlord under the Letter of Credit, together with any and all interest accruing thereon, if deposited in an interest-bearing account.

L/C Bank means J.P. Morgan Chase Bank or any other United States bank which is approved by Landlord in Landlord's sole discretion.

Letter of Credit means that certainone-year irrevocable letter of credit, in the Letter of Credit Required Amount, issued by the L/C Bank, as required under Section 62.2 and, if applicable, as extended, renewed, replaced or modified from time to time in accordance with this Lease, which letter of credit will be transferable and in substantially the same form as attached Exhibit L.

62.5 Transfer of Letter of Credit. The Letter of Credit shall not be mortgaged, assigned or encumbered in any manner whatsoever by Tenant. Tenant acknowledges that Landlord has the right to transfer or mortgage its interest in the Premises and the Building and in this Lease and Tenant agrees that in the event of any such transfer or mortgage, Landlord shall have the right, upon written notice to Tenant, to transfer or assign the Letter of Credit and/or the Draw Proceeds to the transferee or mortgagee to be held and drawn upon subject to the terms and conditions of this Lease, and in such event, following delivery of the Letter of Credit to such Transferee or mortgagee, Tenant shall look solely to such transferee or mortgagee for return of the Letter of Credit and/or the Draw Proceeds so transferred. Landlord shall pay all fees and charges of the L/C Bank with respect to any transfer of the Letter of Credit. Tenant shall, within five (5) business days of request by Landlord, execute such further instruments or assurances as Landlord may reasonably deem necessary to evidence or confirm Landlord's transfer or assignment of the Letter of Credit and/or the Draw Proceeds to such transferee or mortgagee.

62.6 Letter of Credit is Not Security Deposit Landlord and Tenant acknowledge and agree that in no event or circumstance shall the Letter of Credit, any renewal thereof or substitute therefor or the proceeds thereof be (i) deemed to be or treated as a security deposit within the meaning of California Civil Code Section 1950.7, (ii) subject to the terms of such Section 1950.7, or (iii) intended to serve as a security deposit within the meaning of such Section 1950.7. The parties hereto (A) recite that the Letter of Credit is not intended to serve as a security deposit and such Section 1950.7 and any and all other laws, rules and regulations applicable to security deposits in the commercial context (Security Deposit Laws) shall have no applicability or relevancy thereto and (B) waive any and all rights, duties and obligations either party may now or, in the future, will have relating to or arising from the Security Deposit Laws. Notwithstanding the foregoing, to the extent California Civil Code 1950.7 in any way: (a) is determined to be applicable to this Lease or the Letter of Credit (or any proceeds thereof); or (b) controls Landlord's rights to draw on the Letter of Credit or apply the proceeds of the Letter

of Credit to any amounts due under this Lease or any damages Landlord may suffer following termination of this Lease, then Tenant fully and irrevocably waives the benefits and protections of Section 1950.7 of the California Civil Code, it being agreed that Landlord may recover from the Letter of Credit (or its proceeds) all of Landlord's damages under this Lease and California law including, but not limited to, any damages accruing upon the termination of this Lease in accordance with this Lease and Section 1951.2 of the California Civil Code.

62.7 <u>Substitute Letter of Credit.</u> In the event the L/C Bank is declared insolvent by the FDIC or is closed for any reason, Tenant shall, within ten (10) business days thereafter, either (a) provide a substitute Letter of Credit meeting the requirements of this <u>Article 62</u> from another United States bank which is approved by Landlord in Landlord's sole discretion, or (b) if Landlord does not approve the issuer of the substitute Letter of Credit, deposit cash with Landlord in the Letter of Credit Required Amount to be held in the same manner, and under the same terms and conditions, as Draw Proceeds under this <u>Article 62</u>.

### ARTICLE 63 ADDITIONAL IMPROVEMENTS

63.1 Upon Tenant's written request, which must be delivered to Landlord within twenty-four (24) months after the Commencement Date, if at all, and subject to Applicable Laws, Landlord, at Tenant's sole cost and expense, will install twenty (20) electric vehicle charging stations at locations in the Parking Facilities reasonably determined by Landlord. Any parking space in which those charging stations are installed shall be counted against the number of personalized reserved parking spaces to which Tenant is entitled under this Lease and Tenant shall pay the monthly charge for each such space at the rate provided for in Section 49.3 for personalized parking passes. Tenant shall reimburse Landlord within thirty (30) days after written demand for the costs incurred by Landlord in connection with the installation, repair, maintenance, replacement and removal of such vehicle charging stations. Upon the expiration or earlier termination of this Lease, Landlord may require that some or all of such vehicle charging stations be removed and the parking areas affected thereby restored to a condition comparable to other parking areas in the Parking Facility without vehicle charging stations at Tenant's sole cost and expense.

63.2 Upon Tenant's written request, which must be delivered to Landlord within twenty-four (24) months after the Commencement Date, if at all, Landlord, at Tenant's sole cost and expense, shall install a rock climbing wall (the "Wall") outside of the new parking garage being constructed on or about the date of this Lease at 575 Anton on the terms and conditions contained in this Section 63.2. Landlord's obligation to construct the Wall is subject to determining the structural feasibility of its installation at that location, approval by all governmental authorities, limitations on the location and design of the Wall to reflect considerations as to safety and security and as to such other reasonable matters as Landlord may determine. The Wall shall be for the exclusive use of Tenant and its employees and invitees and Tenant shall be solely responsible for the control of the use of, and access to, the Wall and any and all liability, loss, cost or expense arising out of that use. Landlord shall have no obligation or duty to control or supervise the use of, or access to, the Wall. Tenant shall be responsible for the maintenance and repair of the Wall and any additional costs and/or additional improvements necessary to prevent use of the Wall by persons other than Tenant's employees and invitees.

Tenant shall reimburse Landlord within thirty (30) days after written demand for the costs incurred by Landlord in connection with the installation, repair, maintenance, replacement and removal of the Wall. Upon the expiration or earlier termination of this Lease, Landlord may require that some or all of the Wall be removed and the areas affected thereby restored to the condition existing prior to the installation of the Wall at Tenant's sole cost and expense. Without limiting the generality of Article 13, Tenant shall secure appropriate insurance to cover all liability arising out of the use and operation of the Wall.

IN WITNESS WHEREOF, Landlord and Tenant, acting herein through duly authorized individuals, have caused these presents to be executed as of the date first above written.

TENANT: VERITONE, INC., a Delaware corporation

By: /s/ Chad Steelberg

Chad Steelberg, Chief Executive Officer

# LANDLORD:

PR II/MCC SOUTH COAST PROPERTY OWNER, LLC, a Delaware limited liability company

By: PR II/MCC South Coast Venture, LLC, a Delaware limited liability company, its sole member

By: McCarthy Cook Ventures XIII, LLC, a Delaware limited liability company, its authorized member

By: /s/ Edward W. Cook III

Name: Edward W. Cook III

Title: Co-President

### EXHIBIT A

# THE PROJECT

The land situated in the City of Costa Mesa, in the County of Orange, State of California, and described as follows:

#### PARCEL A:

PARCEL 3 OF LOT LINE NO. 05-03, IN THE CITY OF COSTA MESA, COUNTY OF ORANGE, STATE OF CALIFORNIA, RECORDED SEPTEMBER 19, 2005 AS INSTRUMENT NO. 2005000738603, OF OFFICIAL RECORDS.

# PARCEL B:

PARCELS 2 AND 3 OF LOT LINE NO. 05-04, IN THE CITY OF COSTA MESA, COUNTY OF ORANGE, STATE OF CALIFORNIA, RECORDED SEPTEMBER 19, 2005 AS INSTRUMENT NO. 2005000738604 AND RE-RECORDED SEPTEMBER 26, 2005 AS INSTRUMENT NO. 2005000758094, BOTH OF OFFICIAL RECORDS.

### PARCEL C:

A PERPETUAL, NON-EXCLUSIVE, APPURTENANT EASEMENT FOR VEHICULAR INGRESS, EGRESS AND ACCESS AS SET FORTH IN THAT CERTAIN DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND MAINTENANCE OBLIGATIONS RECORDED OCTOBER 27, 1983 AS INSTRUMENT NO. 83-473268 OF OFFICIAL RECORDS OF ORANGE COUNTY, CALIFORNIA.

### PARCEL D

A PERPETUAL, APPURTENANT, NON-EXCLUSIVE EASEMENT FOR PROVIDING AND MAINTAINING VEHICULAR PARKING SPACES AND FOR PEDESTRIAN AND VEHICULAR INGRESS AND EGRESS AS SET FORTH IN THAT CERTAIN THIRD RESTATED AND AMENDED DECLARATION ESTABLISHING RECIPROCAL EASEMENTS RECORDED SEPTEMBER 19, 2005 AS INSTRUMENT NO. 2005000738612 OFFICIAL RECORDS OF ORANGE COUNTY, CALIFORNIA.

# PARCEL E:

A PERPETUAL, APPURTENANT, NON-EXCLUSIVE EASEMENT FOR INGRESS AND EGRESS PURPOSES AS SET FORTH IN THAT CERTAIN RESTATED GRANT OF EMERGENCY WATER FIRE SERVICE EASEMENT RECORDED SEPTEMBER 24, 1985 AS INSTRUMENT NO. 85-363132 OF OFFICIAL RECORDS OF ORANGE COUNTY, CALIFORNIA.

APN: 410-501-32; 410-501-33; 410-501-34; 410-501-35; 410-501-37; 410-501-38

Exhibit A

EXHIBIT B

<u>PREMISES</u>

(See Attached.)

[SPACE PLAN]

Exhibit B -1-

EXHIBIT B-1

TEMPORARY SPACE

[SPACE PLAN]

Exhibit B-1 -1-

EXHIBIT B-2

BALCONY AREA

(See Attached)

[BALCONY LAYOUT]

Exhibit B-2 -1-

EXHIBIT B-3

PATIO AREA

(See Attached)

[PATIO LAYOUT]

Exhibit B-3 -1-

### EXHIBIT C

# TENANT WORK LETTER

This Tenant Work Letter ("Tenant Work Letter") sets forth the terms and conditions relating to the construction of improvements for the Premises. All references in this Tenant Work Letter to "the Lease" shall mean the relevant portions of the Lease to which this Tenant Work Letter is attached as <u>Exhibit C</u>.

#### SECTION 1

### BASE, SHELL AND CORE

Landlord has previously constructed the base, shell, and core (i) of the Premises and (ii) of the floor(s) of the Building on which the Premises are located (collectively, the "Base, Shell, and Core"), and Tenant shall accept the Base, Shell and Core in its current "As-Is" condition existing as of the date of the Lease and the Commencement Date. Landlord shall install in the Premises certain "Tenant Improvements" (as defined below) pursuant to the provisions of this Tenant Work Letter. As used in this Lease, a "latent defect" is a design or construction defect or error in the Tenant Improvements which is not apparent upon an ordinary and reasonable inspection of the Premises. Landlord covenants that if any such latent defects appear during the Term, which resulted from faulty design, workmanship or materials, then Landlord shall cause the same, after receiving written notice thereof from Tenant, to be repaired and corrected with all reasonable speed at Landlord's expense. Except for the Tenant Improvement work described in this Tenant Work Letter and except for the Allowance set forth below, Landlord shall not be obligated to make or pay for any alterations or improvements to the Premises, the Building or the Project.

### SECTION 2

### CONSTRUCTION DRAWINGS FOR THE PREMISES

2.1 Selection of Architect/Construction Drawings. Landlord shall retain an architect/space planner (the "Architect") to prepare the "Construction Drawings," as that term is defined in this Section 3.1. Landlord shall retain Landlord's engineering consultants (the "Engineers") to prepare all plans and engineering working drawings relating to the structural, mechanical, electrical, plumbing, HVAC, lifesafety, and sprinkler work in the Premises. The plans and drawings to be prepared by Architect and the Engineers hereunder shall be known collectively as the "Construction Drawings." Notwithstanding that any Construction Drawings are reviewed by Landlord or prepared by its Architect, Engineers and consultants, and notwithstanding any advice or assistance which may be rendered to Tenant by Landlord or Landlord's Architect, Engineers, and consultants, Landlord shall have no liability whatsoever in connection therewith and shall not be responsible for any omissions or errors contained in the Construction Drawings, and Tenant's waiver and indemnity set forth in Article 10 of the Lease shall specifically apply to the Construction Drawings. Landlord hereby assigns to Tenant all rights against the Architect, Engineers and consultants for errors and omissions contained in the Construction Documents, which assignment shall be on a non-exclusive basis such that such

rights may be enforced by Landlord and/or Tenant. Landlord and Tenant agree to cooperate reasonably in any enforcement of such rights.

- 2.2 Final Space Plan. Prior to the execution of the Lease, Tenant has met with Landlord's Architect and provided Landlord's Architect with information regarding the preliminary layout and designation of all proposed offices, rooms and other partitioning, and their intended use and equipment to be contained therein (the "Information"). Tenant shall continue to communicate with Landlord's Architect as necessary concerning the Information to facilitate the design of the Tenant Improvements. Landlord and Architect shall, based on such Information (subject to changes reasonably required by Landlord), prepare the final space plan for Tenant Improvements in the Premises (collectively, the "Final Space Plan"), which Final Space Plan shall include a layout and designation of all offices, rooms and other partitioning, their intended use, and equipment to be contained therein, and shall deliver the Final Space Plan to Tenant's approval. Tenant shall approve or reasonably disapprove the Final Space Plan or any revisions thereto within five (5) business days after Landlord delivers the Final Space Plan or such revisions to Tenant; provided, however, that any delay due to Tenant's disapproval of the Final Space Plan (other than to the extent the same is not (subject to changes reasonably required by Landlord) in substantial conformance with the Information provided by Tenant to Architect (such nonconformity, a "Space Plan Design Problem")) shall constitute a Tenant Delay. Tenant's failure to disapprove the Final Space Plan for any Space Plan Design Problem or any revisions thereto by written notice to Landlord (which notice shall specify in detail the reasonable reasons for Tenant's disapproval pertaining to any Space Plan Design Problem) within said five (5) business day period shall be deemed to constitute Tenant's approval of the Final Space Plan or such revisions. The scope and design of the Tenant Improvements and all Construction Drawings shall be subject to Landlord's approval, which shall not be unreasonably withheld, conditioned or delayed. Without limiting the foregoing approval right, Landlord and Tenant agree that the overall look, feel and layout (specifically, the balance between open areas and enclosed offices/rooms) of the Tenant Improvements shall be substantially similar to the look, feel and layout of the tenant improvements existing in Suite 350 on the date of the Lease and the layout of the improvements shown on the preliminary plan for the Tenant Improvements dated July 12, 2017 (a copy of which is attached hereto as Exhibit C-2), including an interior stairwell as shown on those preliminary plans (provided that in any event the interior stairwell must be meet the requirements for Building Standard Stairs contained in Exhibit C-1 attached hereto) (collectively, the "Design Standards").
- 2.3 Final Working Drawings. Based on the Final Space Plan, Landlord shall cause the Architect and the Engineers to complete the architectural and engineering drawings for the Premises, and Architect shall compile a fully coordinated set of architectural, structural, mechanical, electrical and plumbing working drawings in a form which is complete to allow subcontractors to bid on the work and to obtain all applicable permits (collectively, the "Final Working Drawings") and shall submit the same to Tenant for Tenant's approval. The Final Working Drawings shall incorporate modifications to the Final Space Plan as necessary to comply with the floor load and other structural and system requirements of the Building. To the extent that the finishes and specifications are not completely set forth in the Final Space Plan for any portion of the Tenant Improvements depicted thereon, the actual specifications and finish work shall be in accordance with the Specifications. Tenant shall approve or reasonably disapprove the Final Working Drawings or any revisions thereto within five (5) business days

after Landlord delivers the Final Working Drawings or any revisions thereto to Tenant; provided, however, that any delay due to Tenant's disapproval of the Final Working Drawings (other than to the extent the same are not (subject to changes reasonably required by Landlord) in substantial conformance with the Final Space Plan (such nonconformity, a "Working Drawing Design Problem")) shall constitute a Tenant Delay. Tenant's failure to reasonably disapprove the Final Working Drawings or any revisions thereto by written notice to Landlord (which notice shall specify in detail the reasonable reasons for Tenant's disapproval pertaining to any Working Drawing Design Problem) within said five (5) business day period shall be deemed to constitute Tenant's approval of the Final Working Drawings or such revisions.

- 2.4 <u>Approved Working Drawings</u>. The Final Working Drawings shall be approved or deemed approved by Tenant (the "Approved Working Drawings") prior to the commencement of the construction of the Tenant Improvements. Landlord shall cause the Architect to submit the Approved Working Drawing to the applicable local governmental agency for all applicable building permits necessary to allow "Contractor," as that term is defined in Section 4.1 of this Tenant Work Letter, to commence and fully complete the construction of the Tenant Improvements (the "Permits"). No changes, modifications or alterations in the Approved Working Drawings may be made without the prior written consent of Landlord, which consent shall not be unreasonably conditioned, withheld or delayed.
- 2.5 <u>Time Deadlines</u>. Tenant shall use its best efforts to cooperate with Architect, the Engineers, and Landlord to complete all phases of the Construction Drawings and the permitting process and to receive the Permits, and with Contractor, for approval of the "Cost Proposal," as that term is defined in Section 4.2 below as soon as possible after the execution of the Lease and, in this regard, to the extent Landlord considers such meeting(s) to be reasonably necessary, Tenant shall meet with Landlord on a weekly basis to discuss Tenant's progress in connection with the same.
- 2.6 <u>Design Problem</u>. Notwithstanding anything to the contrary in this Tenant Work Letter, Landlord shall be deemed to have acted reasonably in disapproving plans or designs if Landlord determines in good faith that the matter disapproved constitutes or would create a Design Problem (as defined below). As used herein, a "Design Problem" shall mean (i) adverse effect on the structural integrity of the Building; (ii) more than a remote possibility of damage to the Building's systems; (iii) non-compliance with applicable codes; (iv) adverse effect on the exterior appearance of the Building; (v) creation of the potential for unusual expenses to be incurred in connection with the maintenance by Landlord of the alteration or improvement, unless Tenant agrees to pay for the incremental maintenance costs caused by the non-typical alterations, (vi) a material adverse effect on any other tenant or occupant of the Building, (vii) creation of an obligation to make other alterations, additions or improvements to the Premises or Common Areas in order to comply with applicable laws (including, without limitation, the Americans with Disabilities Act), (viii) adverse effect on the LEED rating of the Building or (ix) material variation from the look, feel and layout of the Design Standards.

### SECTION 3

### CONSTRUCTION AND PAYMENT FOR

### COSTS OF TENANT IMPROVEMENTS

3.1 Allowance. Landlord shall cause a contractor designated by Landlord (the "Contractor") to (i) obtain all applicable building permits for construction of the Tenant Improvements, and (ii) construct the Tenant Improvements as depicted on the Approved Working Drawings, in compliance with such building permits and all applicable laws in effect at the time of construction, and in good workmanlike manner. Landlord shall pay for the cost of the design and construction of the Tenant Improvements in an amount up to, but not exceeding, \$70.00 per rentable square foot of the Premises (i.e., up to \$2,651,250.00, based on 37,875 rentable square feet of the Premises (the "Allowance"). The cost of the design and construction of the Tenant Improvements shall include Landlord's construction supervision and management fee in an amount equal to the product of (i) three percent (3%) and (ii) the amount equal to the sum of the Allowance and the Over-Allowance Amount (as such term is defined below). Tenant shall pay for all costs in excess of the Allowance ("Over-Allowance Amount"), which payment shall be made to Landlord in cash within ten (10) days after Tenant's receipt of invoice therefor from Landlord and, in any event, prior to the date Landlord causes the Contractor to commence the actions described in the first sentence of this Section 3. In the event that after Tenant pays the Over-Allowance Amount Tenant requests any changes, change orders or modifications to the Approved Working Drawings (which Landlord approves pursuant to Section 2 above) which increase the cost to construct the Tenant Improvements, Tenant shall pay such increased cost to Landlord immediately upon Landlord's request therefor, and, in any event, prior to the date Landlord causes the Contractor to commence construction of the changes, change orders or modifications. Except as set forth below, in no event shall Landlord be obligated to pay for any of Tenant's furniture, computer systems, telephone systems, equipment or other personal property which may be depicted on the Construction Drawings; such items shall be paid for by Tenant. Tenant shall not be entitled to receive in cash or as a credit against any rental or otherwise, any portion of the Allowance not used to pay for the cost of the design and construction of the Tenant Improvements; provided, however, so long as the Tenant Improvements have been completed and all Allowance items have been paid, and so long as no Event of Default exists under the Lease, Tenant may use any unused portion of the Allowance not to exceed an amount equal to \$10.00 per square foot of rentable area in the Premises as a credit against Base Rent due under the Lease and/or toward Tenant's fixtures, furniture and equipment ("FFE"). Tenant shall present Landlord with paid invoices for the FFE, and Landlord shall reimburse Tenant within thirty (30) days after receipt thereof

3.2 <u>Tenant's Percentage</u>. Notwithstanding the provision of Section 3.1, so long as the Tenant is the Original Tenant, Tenant may pay the Over-Allowance Amount in accordance with the following schedule: (a) an amount equal to thirty-five percent (35%) of the Over-Allowance Amount (the "Initial Construction Deposit") shall by delivered to Landlord within thirty (30) days after the receipt from Landlord of an invoice therefor together with reasonable supporting documentation; and (b) the remaining sixty-five percent (65%) of the Over-Allowance Amount shall be delivered to Landlord from time-to-time within ten (10) business days after written request of Landlord (each, a "Funding Request") in amounts specified

by Landlord in each such Funding Request. Landlord may give multiple Funding Requests. Landlord agrees that the amounts specified in each Funding Request shall be Landlord's good faith estimate of Tenant's Percentage (as defined below) of the amounts necessary to make payment to third parties for the cost of the Tenant Improvement that have been incurred or will be incurred within thirty (30) days after the date of the Funding Request. Any failure of Tenant to deliver to Landlord any portion of the Over-Allowance Amount in cash as and when due as provided above shall constitute a Tenant Delay and, if not cured within five (5) business days after written notice, shall also constitute an Event of Default under the Lease. The Initial Construction Deposit shall be applied to Tenant's Percentage of the cost of the Tenant Improvements first coming due. For purposes of this Tenant Work Letter, "Tenant's Percentage" shall be equal one (1) minus the fraction determined by dividing the amount of the Allowance by the estimated budget for the Tenant Improvements. Landlord may from time to time re-determine the Over-Allowance Amount based on any changes, change orders or modifications to the approved Working Drawings (which Landlord approves pursuant to Section 2) which increase the cost to construct the Tenant Improvements and upon that re-determination modify Tenant's Percentage and adjust the amount of any Funding Request (or make additional Funding Requests) such that Tenant shall pay the entire Over-Allowance Amount and Tenant's funding of the Over-Allowance Amount at all times is equal to Tenant's Percentage of the estimated budget for the Tenant Improvements that has been expended through that date and that will be paid to third parties for the cost of the Tenant Improvement that have been incurred or will be incurred within thirty (30) days after the date of the most recent Funding Request.

### SECTION 4

# SUBSTANTIAL COMPLETION

#### OF THE TENANT IMPROVEMENTS

4.1 <u>Substantial Completion</u>. For purposes of this Lease, including for purposes of determining the Commencement Date, "Substantial Completion" of the Premises shall occur upon the last to occur of (a) the completion of construction of the Tenant Improvements in the Premises pursuant to the Approved Working Drawings, with the exception of any Punch List Items (as defined below) and any tenant fixtures, work-stations, built-in furniture, or equipment to be installed by Tenant or under the supervision of Contractor, (b) and the issuance of a temporary or permanent certificate of occupancy by the local building authority (or a reasonably substantial equivalent such as a sign-off from a building inspector), and (c) the Monday of the week following the week in which Landlord delivers possession of the Premises to Tenant in the condition described in clause (a) of this Section 4.1. Landlord will provide Tenant with thirty (30) days' prior notice of the date Landlord estimates in good faith Substantial Completion will occur. Landlord's incorrect estimation of the date of Substantial Completion shall not affect the determination of the Commencement Date or otherwise subject Landlord to any liability. Within thirty (30) days after the Commencement Date, Tenant shall give Landlord written notice of any incomplete work, unsatisfactory conditions or defects (the "Punch List Items") which do not materially interfere with Tenant's use of the Premises and were part of the work that is the obligation of Landlord to perform in the Premises under this Lease and Landlord shall, at its sole expense, complete said work and/or remedy such unsatisfactory conditions or defects as soon as possible. The existence of any incomplete work, unsatisfactory conditions or

defects as aforesaid shall not affect the Commencement Date or the obligation of Tenant to pay Base Rent, Additional Rent and all other charges hereunder.

- 4.2 <u>Delay of the Substantial Completion of the Premises</u>. If there shall be a delay or there are delays in the Substantial Completion of the Premises as a direct, indirect, partial, or total result of any of the following (collectively, "Tenant Delays"):
  - 4.2.1 Tenant's failure to timely approve the Working Drawings or any other matter requiring Tenant's approval;
  - 4.2.2 a breach by Tenant of the terms of this Tenant Work Letter or the Lease;
  - 4.2.3 Tenant's request for changes in any of the Construction Drawings that are deemed to be Tenant Delays pursuant to Sections 2.2 or 2.3;
- 4.2.4 Tenant's requirement for materials, components, finishes or improvements which are not available in a commercially reasonable time given the estimated date of Substantial Completion of the Premises, as set forth in the Lease;
- 4.2.5 changes to the Base, Shell and Core, structural components or structural components or systems of the Building required by the Approved Working Drawings;
- 4.2.6 any changes in the Construction Drawings and/or the Tenant Improvements required by applicable laws if such changes are directly attributable to Tenant's use of the Premises or Tenant's specialized tenant improvement(s); or
  - 4.2.7 any other acts or omissions of Tenant, or its agents, or employees;

then, notwithstanding anything to the contrary set forth in the Lease and regardless of the actual date of Substantial Completion, the Commencement Date shall be deemed to be the date the Commencement Date would have occurred if no Tenant Delays, as set forth above, had occurred.

# **SECTION 5**

### MISCELLANEOUS

5.1 Tenant's Entry Into the Premises Prior to Substantial Completion Subject to the terms hereof and provided that Tenant and its agents do not interfere with Contractor's work in the Building and the Premises, Contractor shall allow Tenant access to the Premises not less than thirty (30) days prior to the Substantial Completion of the Premises for the purpose of Tenant installing overstandard equipment or fixtures (including Tenant's data and telephone equipment) in the Premises. Prior to Tenant's entry into the Premises as permitted by the terms of this Section 5.1, Tenant shall submit a schedule to Landlord and Contractor, for their approval, which schedule shall detail the timing and purpose of Tenant's entry. In connection with any such entry, Tenant acknowledges and agrees that Tenant's employees, agents, contractors, consultants, workmen, mechanics, suppliers and invitees shall fully cooperate, work in harmony and not, in any manner, interfere with Landlord or Landlord's Contractor, agents or

representatives in performing work in the Building and the Premises, or interfere with the general operation of the Building and/or the Project. If at any time any such person representing Tenant shall not be cooperative or shall otherwise cause or threaten to cause any such disharmony or interference, including, without limitation, labor disharmony, and Tenant fails to immediately institute and maintain corrective actions as directed by Landlord, then Landlord may revoke Tenant's entry rights upon twenty-four (24) hours' prior written notice to Tenant. Tenant acknowledges and agrees that any such entry into and occupancy of the Premises or any portion thereof by Tenant or any person or entity working for or on behalf of Tenant shall be deemed to be subject to all of the terms, covenants, conditions and provisions of the Lease, excluding only the covenant to pay Rent (until the occurrence of the Commencement Date). Tenant further acknowledges and agrees that Landlord shall not be liable for any injury, loss or damage which may occur to any of Tenant's work made in or about the Premises in connection with such entry or to any property placed therein prior to the Commencement Date, the same being at Tenant's sole risk and liability. Tenant shall be liable to Landlord for any damage to any portion of the Premises, including the Tenant Improvement work, caused by Tenant or any of Tenant's employees, agents, contractors, consultants, workmen, mechanics, suppliers and invitees. In the event that the performance of Tenant's work in connection with such entry causes extra costs to be incurred by Landlord or requires the use of any Building services, Tenant shall promptly reimburse Landlord for such extra costs and/or shall pay Landlord for such Building services at Landlord's standard rates then in effect. In addition, Tenant shall hold Landlord harmless from and indemnify, protect and defend Landlord against any loss or damage to the Premises or Project and against injury to any persons caused by Tenant's actions purs

- 5.2 <u>Tenant's Representative</u>. Tenant has designated Jeff Coyne as its sole representative with respect to the matters set forth in this Tenant Work Letter, who shall have full authority and responsibility to act on behalf of the Tenant as required in this Tenant Work Letter.
- 5.3 <u>Landlord's Representative</u>. Landlord has designated Megan Allen as its sole representative with respect to the matters set forth in this Tenant Work Letter, who, until further notice to Tenant, shall have full authority and responsibility to act on behalf of the Landlord as required in this Tenant Work Letter.
- 5.4 <u>Time of the Essence in This Tenant Work Letter</u>. Unless otherwise indicated, all references herein to a "number of days" shall mean and refer to calendar days. If any item requiring approval is timely disapproved by Landlord, the procedure for preparation of the document and approval thereof shall be repeated until the document is approved by Landlord.
- 5.5 <u>Tenant's Lease Default</u> Notwithstanding any provision to the contrary contained in the Lease, if an Event of Default by Tenant under the Lease or any default by Tenant under this Work Letter has occurred at any time on or before the substantial completion of the Premises, then (i) in addition to all other rights and remedies granted to Landlord pursuant to the Lease, at law and/or in equity, Landlord shall have the right to cause Contractor to cease the construction of the Premises (in which case, Tenant shall be responsible for any delay in the Substantial Completion of the Premises caused by such work stoppage as set forth in <u>Section 4.2</u> of this Tenant Work Letter), and (ii) all other obligations of Landlord under the terms of this

Tenant Work Letter shall be forgiven until such time as such default is cured pursuant to the terms of the Lease (in which case, Tenant shall be responsible for any delay in the Substantial Completion of the Premises caused by such inaction by Landlord). In addition, if the Lease is terminated prior to the Commencement Date, for any reason due to an Event of Default by Tenant under the Lease or a default under this Tenant Work Letter, in addition to any other remedies available to Landlord under the Lease, at law and/or in equity, Tenant shall pay to Landlord, as Additional Rent under the Lease, within five (5) days of receipt of a statement therefor, any and all costs incurred by Landlord (including any portion of the Allowance disbursed by Landlord) and not reimbursed or otherwise paid by Tenant through the date of such termination in connection with the Tenant Improvements to the extent planned, installed and/or constructed as of such date of termination, including, but not limited to, any costs related to the removal of all or any portion of the Tenant Improvements and restoration costs related thereto.

5.6 <u>Preliminary Test and Pricing Plan Allowance</u>. Landlord shall provide Tenant a preliminary test and pricing plan allowance of up to \$0.15 per rentable square foot of the Premises (the "Preliminary Test and Pricing Plan Allowance"), which may be used only for the costs to prepare preliminary test and pricing plans for the Premises. Landlord will apply the Preliminary Test and Pricing Plan Allowance to payment of the fees charged by Landlord's architect for the preliminary test and price plans. Landlord shall be entitled to copies of all plans created utilizing the Preliminary Test and Pricing Plan Allowance and shall be the owner of all such plans. The Preliminary Test and Pricing Plan Allowance will be deducted from the Allowance.

# EXHIBIT C-1

# **BUILDING STANDARD STAIRS**

Building Standard Stairs shall be defined as a "between floor added access stairs" that link two floors of the Building together, in such a manner as to allow a single tenant to go between their occupied floors, without having to go into public elevators or stairwells.

Existing conditions: The structural steel structure at Building currently consists off-Beams (I shaped, generally smaller sized steel beams, with tapered flanges), W-Beam's ('wide flange' beams, heavier mid-sized beams), and H-Beam's (generally, the heavier "main" support beams that are directly supported by columns). The structure is then overlaid by steel decking, with light weight hard rock concrete.

Building Standard Stairs shall reflect minimal interference with building structure. As used in this Exhibit, "minimal interference" means that Building Standard Stairs shall not require removal of more than a single I-Beam from the Existing Building structure. Should the configuration desired by Tenant require removal and replacement of loading associated with the larger beams (W-Beams and H-Beams), or more than a single I-Beam, this shall NOT be considered Building Standard Stairs. The actual stairs configuration shall not dictate; as a "U" shaped", "L" shaped or "I" shaped stairs may all have their own specific constraints and issues. Similarly, the actual location shall not dictate; as each location has its own specific issues.

# EXHIBIT C-2

# PRELIMINARY PLAN

(See Attached)

[SPACE PLAN]

Exhibit C-2 -1-

# EXHIBIT D

# **RULES AND REGULATIONS**

Tenant shall faithfully observe and comply with the following Rules and Regulations. Landlord shall not be responsible to Tenant for the nonperformance of any of said Rules and Regulations by or otherwise with respect to the acts or omissions of any other tenants or occupants of the Project. In the event of any conflict between the Rules and Regulations and the other provisions of this Lease, the latter shall control.

1. No sign, placard, picture, advertisement, name or notice shall be inscribed, displayed or printed or affixed on or to any part of the outside or inside of the Building or any part of the Premises visible from the exterior of the Premises without the prior written consent of Landlord, which consent may be withheld in Landlord's sole discretion. Landlord shall have the right to remove, at Tenant's expense and without notice to Tenant, any such sign, placard, picture, advertisement, name or notice that has not been approved by Landlord.

All approved signs or lettering on doors and walls shall be printed, painted, affixed or inscribed at the expense of Tenant by a person approved of by Landlord.

If Landlord notifies Tenant in writing that Landlord objects to any curtains, blinds, shades or screens attached to or hung in or used in connection with any window or door of the Premises, such use of such curtains, blinds, shades or screens shall be removed immediately by Tenant. No awning shall be permitted on any part of the Premises.

- 2. No ice, drinking water, towel, barbering or bootblacking, shoeshining or repair services, or other similar services shall be provided to the Premises, except from persons authorized by Landlord and at the hours and under regulations fixed by Landlord.
- 3. The bulletin board or directory of the Building will be provided exclusively for the display of the name and location of tenants only and Landlord reserves the right to exclude any other names therefrom.
- 4. The sidewalks, halls, passages, exits, entrances, elevators and stairways shall not be obstructed by any of the Tenant Parties or used by Tenant for any purpose other than for ingress to and egress from its Premises. The halls, passages, exits, entrances, elevators, stairways, balconies and roof are not for the use of the general public and Landlord shall in all cases retain the right to control and prevent access thereto by all persons whose presence in the judgment of Landlord shall be prejudicial to the safety, character, reputation and interests of the Building and its tenants. No tenant and no employees or invitees of any tenant shall go upon the roof of the Building.
- 5. Tenant shall not alter any lock or install any new or additional locks or any bolts on any interior or exterior door of the Premises without the prior written consent of Landlord.
- 6. The toilet rooms, toilets, urinals, wash bowls and other apparatus shall not be used for any purpose other than that for which they were constructed and no foreign substance of any kind whatsoever shall be thrown therein and the expense of any breakage, stoppage or damage

resulting from the violation of this rule shall be borne by the tenant who, or whose employees or invitees, shall have caused it.

- 7. Tenant shall not overload the floor of the Premises or mark, drive nails, screw or drill into the partitions, woodwork or plaster or in any way deface the Premises or any part thereof.
- 8. No furniture, freight or equipment of any kind shall be brought into the Building without the consent of Landlord and all moving of the same into or out of the Building shall be done at such time and in such manner as Landlord shall designate. Landlord shall have the right to prescribe the weight, size and position of all safes and other heavy equipment brought into the Building and also the times and manner of moving the same in and out of the Building. Safes or other heavy objects shall, if considered necessary by Landlord, stand on a platform of such thickness as is necessary to properly distribute the weight. Landlord will not be responsible for loss of or damage to any such safe or property from any cause, and all damage done to the Building by moving or maintaining any such safe or other property shall be repaired at the expense of Tenant. The elevator designated for freight by Landlord shall be available for use by all tenants in the Building at no cost during the hours and pursuant to such procedures as Landlord may determine from time to time. The persons employed to move Tenant's equipment, material, furniture or other property in or out of the Building must be acceptable to Landlord. The moving company must be a locally recognized professional mover, whose primary business is the performing of relocation services, and must be bonded and fully insured. In no event shall Tenant employ any person or company whose presence may give rise to a labor or other disturbance in the Project. A certificate or other verification of such insurance must be received and approved by Landlord prior to the start of any moving operations. Insurance must be sufficient in Landlord's sole opinion, to cover all personal liability, theft or damage to the Project, including, but not limited to, floor coverings, doors, walls, elevators, stairs, foliage and landscaping. Special care must be taken to prevent damage to foliage and landscaping during adverse weather. All moving operations shall be conducted at such times and in
- 9. Tenant shall not employ any person or persons other than the janitor of Landlord for the purpose of cleaning the Premises, unless otherwise agreed to by Landlord. Except with the written consent of Landlord, no person or persons other than those approved by Landlord shall be permitted to enter the Building for the purpose of cleaning the Building or the Premises. Tenant shall not cause any unnecessary labor by reason of Tenant's carelessness or indifference in the preservation of good order and cleanliness.
- 10. Tenant shall not use, keep or permit to be used or kept any foul or noxious gas or substance in the Premises, or permit or suffer the Premises to be occupied or used in a manner offensive or objectionable to Landlord or other occupants of the Building by reason of noise, odors and/or vibrations, or interfere in any way with other tenants or those having business therein, nor shall any animals or birds be brought in or kept in or about the Premises or the Building. In no event shall Tenant keep, use, or permit to be used in the Premises or the Building any guns, firearm, explosive devices or ammunition.

- 11. No cooking shall be done or permitted by Tenant in the Premises, nor shall the Premises be used for the storage of merchandise, for washing clothes, for lodging, or for any improper, objectionable or immoral purposes. Notwithstanding the foregoing, however, Tenant may maintain and use microwave ovens and equipment for brewing coffee, tea, hot chocolate and similar beverages and cook in any kitchen area constructed in the Premises with Landlord's approval, provided that Tenant shall (i) prevent the emission of any food or cooking odor from leaving the Premises, (ii) be solely responsible for cleaning the areas where such equipment is located and removing food-related waste from the Premises and the Building, or shall pay Landlord's standard rate for such service as an addition to cleaning services ordinarily provided, (iii) maintain and use such areas solely for Tenant's employees and business invitees, not as public facilities, (iv) keep the Premises free of vermin and other pest infestation and shall exterminate, as needed, in a manner and through contractors reasonably approved by Landlord, preventing any emission of odors, due to extermination, from leaving the Premises, and (v) comply with all Applicable Laws with respect to such cooking, including that limitation installation of venting and cooking using only equipment permitted by Applicable Laws to be used in the Premises. Notwithstanding clause (ii) above, Landlord shall, without special charge, empty and remove the contents of one (1) 15-gallon (or smaller) waste container from the food preparation area so long as such container is fully lined with, and the contents can be removed in, a waterproof plastic liner or bag, supplied by Tenant, which will prevent any leakage of food related waste or odors; provided, however, that if at any time Landlord must pay a premium or special charge, shall cease.
- 12. Tenant shall not use or keep in the Premises or the Building any kerosene, gasoline, or inflammable or combustible fluid or material, or use any method of heating or air conditioning other than that supplied by Landlord.
- 13. Landlord will direct electricians as to where and how telephone and telegraph wires are to be introduced into the Premises and the Building. No boring or cutting for wires will be allowed without the prior consent of Landlord. The location of telephones, call boxes and other office equipment affixed to the Premises shall be subject to the prior approval of Landlord.
- 14. Upon the expiration or earlier termination of the Lease, Tenant shall deliver to Landlord the keys of offices, rooms and toilet rooms which have been furnished by Landlord to Tenant and any copies of such keys which Tenant has made. In the event Tenant has lost any keys furnished by Landlord, Tenant shall pay Landlord for such keys.
- 15. Tenant shall not lay linoleum, tile, carpet or other similar floor covering so that the same shall be affixed to the floor of the Premises, except to the extent and in the manner approved in advance by Landlord. The expense of repairing any damage resulting from a violation of this rule or removal of any floor covering shall be borne by the tenant by whom, or by whose contractors, employees or invitees, the damage shall have been caused.
- 16. No furniture, packages, supplies, equipment or merchandise will be received in the Building or carried up or down in the elevators, except between such hours and in such

elevators as shall be designated by Landlord, which elevator usage shall be subject to the Building's customary charge therefor as established from time to time by Landlord.

- 17. On Saturdays, Sundays and legal holidays, and on other days between the hours of 6:00 P.M. and 8:00 A.M., access to the Building, or to the halls, corridors, elevators or stairways in the Building, or to the Premises may be refused unless the person seeking access is known to the person or employee of the Building in charge and has a pass or is properly identified. Landlord shall in no case be liable for damages for any error with regard to the admission to or exclusion from the Building of any person. In case of invasion, mob, riot, public excitement, or other commotion, Landlord reserves the right to prevent access to the Building during the continuance of the same by closing the doors or otherwise, for the safety of the tenants and protection of property in the Building.
- 18. Tenant shall be responsible for insuring that the doors of the Premises are closed and securely locked before leaving the Building and must observe strict care and caution that all water faucets or water apparatus are entirely shut off before Tenant or Tenant's employees leave the Building, and that all electricity, gas or air shall likewise be carefully shut off, so as to prevent waste or damage, and for any default or carelessness Tenant shall make good all injuries sustained by other tenants or occupants of the Building or Landlord. Landlord shall not be responsible to Tenant for loss of property on the Premises, however occurring, or for any damage to the property of Tenant caused by the employees or independent contractors of Landlord or by any other person.
- 19. Landlord reserves the right to exclude or expel from the Building any person who, in the judgment of Landlord, is intoxicated or under the influence of liquor or drugs, or who shall in any manner do any act in violation of any of the rules and regulations of the Building.
- 20. The requirements of any tenant will be attended to only upon application at the office of the Building. Employees of Landlord shall not perform any work or do anything outside of their regular duties unless under special instructions from Landlord, and no employee will admit any person (tenant or otherwise) to any office without specific instructions from Landlord.
- 21. No food, soft drink or other vending machine or machines of any description shall be installed, maintained or operated upon the Premises without the prior written consent of Landlord, except those for use by Tenant's employees.
- 22. Subject to Tenant's right of access to the Premises in accordance with Building security procedures, Landlord reserves the right to close and keep locked all entrance and exit doors of the Building on Saturdays, Sundays and legal holidays and on other days between the hours of 6:00 P.M. and 8:00 A.M., and during such further hours as Landlord may deem advisable for the adequate protection of the Building and the property of its tenants.
- 23. Tenant, its employees and agents shall not loiter in or on the entrances, corridors, sidewalks, lobbies, courts, halls, stairways, elevators, vestibules or any Common Areas for the purpose of smoking tobacco products or for any other purpose, nor in any way obstruct such areas, and shall use them only as a means of ingress and egress for the Premises.

- 24. Tenant agrees to cooperate fully with Landlord to ensure the most effective operation of the Building's heating and air conditioning system, and shall refrain from attempting to adjust any controls. Tenant shall comply with and participate in any program for metering or otherwise measuring the use of utilities and services, including, without limitation, programs requiring the disclosure or reporting of the use of any utilities or services. Tenant shall also cooperate and comply with, participate in, and assist in the implementation of (and take no action that is inconsistent with, or which would result in Landlord, the Building and/or the Project failing to comply with the requirements of) any conservation, sustainability, recycling, energy efficiency, and waste reduction programs, environmental protection efforts and/or other programs that are in place and/or implemented from time to time at the Building and/or the Project, including, without limitation, any required reporting, disclosure, rating or compliance system or program (including, but not limited to any LEED ([Leadership in Energy and Environmental Design] rating or compliance system, including those currently coordinated through the U.S. Green Building Council).
- 25. Smoking (including vaping) is prohibited in the Premises, the Building and all enclosed Common Areas of the Project, including all lobbies, all hallways, all elevators and all lavatories. Without limiting the foregoing, Tenant must comply with the State of California "No-Smoking" law set forth in California Labor Code Section 6404.5, and any local "No-Smoking" ordinance which may be in effect from time to time and which is not superseded by such State law.
- 26. Tenant hereby acknowledges that Landlord shall have no obligation to provide guard service or other security measures for the benefit of the Premises, the Building or the Project. Tenant hereby assumes all responsibility for the protection of Tenant and its agents, employees, contractors, invitees and guests, and the property thereof, from acts of third parties, including keeping doors locked and other means of entry to the Premises closed, whether or not Landlord, at its option, elects to provide security protection for the Project or any portion thereof. Tenant further assumes the risk that any safety and security devices, services and programs which Landlord elects, in its sole discretion, to provide may not be effective, or may malfunction or be circumvented by an unauthorized third party, and Tenant shall, in addition to its other insurance obligations under this Lease, obtain its own insurance coverage to the extent Tenant desires protection against losses related to such occurrences. Tenant shall cooperate in any reasonable safety or security program developed by Landlord or required by law.
- 27. Tenant shall install and maintain, at Tenant's sole cost and expense, an adequate visibly marked (at all times properly operational) fire extinguisher next to any duplicating or photocopying machine or similar heat producing equipment, which may or may not contain combustible material, in the Premises.
- 28. The Premises shall not be used as an employment agency, a public stenographer or typist, a labor union office, a physician's or dentist's office, a dance or music studio, a school, a beauty salon, or barber shop, the business of photographic, multilith or multigraph reproductions or offset printing (not precluding using any part of the Premises for photographic, multilith or multigraph reproductions solely in connection with Tenant's own business and/or activities), a restaurant or bar, an establishment for the sale of confectionery, soda, beverages, sandwiches, ice cream or baked goods, an establishment for preparing, dispensing or

consumption of food or beverages of any kind in any manner whatsoever, or news or cigar stand, or a radio, television or recording studio, theatre or exhibition-hall, or manufacturing, or the storage or sale of merchandise, goods, services or property of any kind at wholesale, retail or auction.

29. Tenant shall not store any vehicle within the parking area. Tenant's parking rights are limited to the use of parking spaces for short-term parking, of up to twenty-four (24) hours, of vehicles utilized in the normal and regular daily travel to and from the Project. Tenants who wish to park a vehicle for longer than a 24-hour period shall notify the Building manager for the Project and consent to such long-term parking may be granted for periods up to two (2) weeks. Any motor vehicles parked without the prior written consent of the Building manager for the Project for longer than a 24-hour period shall be deemed stored in violation of this rule and regulation and shall be towed away and stored at the owner's expense or disposed of as provided by Applicable Laws.

Landlord reserves the right at any time to change or rescind any one or more of these Rules and Regulations, or to make such other and further reasonable Rules and Regulations as in Landlord's judgment may from time to time be necessary for the management, safety, care and cleanliness of the Premises, Building, the Common Areas and the Project, and for the preservation of good order therein, as well as for the convenience of other occupants and tenants therein; provided that no changes to the rules and regulations shall materially diminish Tenant's rights under this Lease or increase Tenant's financial obligations. Landlord may waive any one or more of these Rules and Regulations for the benefit of any particular tenants, but no such waiver by Landlord shall be construed as a waiver of such Rules and Regulations in favor of any other tenant, nor prevent Landlord from thereafter enforcing any such Rules or Regulations against any or all tenants of the Project. Tenant shall be deemed to have read these Rules and Regulations and to have agreed to abide by them as a condition of its occupancy of the Premises.

Evhibit D

### EXHIBIT E

# FORM OF TENANT'S ESTOPPEL CERTIFICATE

The undersigned, as Tenant under that certain Office Lease (the "Lease") made and entered into as of the undersigned, as Tenant, for Premises on the floor(s) of the office building located at , certifies as follows: , as Landlord, and

- 1. Attached hereto as Exhibit A is a true and correct copy of the Lease and all amendments and modifications thereto. The documents contained in Exhibit A represent the entire agreement between the parties as to the Premises.
- 2. The undersigned currently occupies the Premises described in the Lease, the Lease Term commenced on and the Lease Term expires on the undersigned has no option to terminate or cancel the Lease or to purchase all or any part of the Premises, the Building and/or the Project.
  - 3. Base Rent became payable on
  - 4. The Lease is in full force and effect and has not been modified, supplemented or amended in any way except as provided in Exhibit A.
  - 5. Tenant has not transferred, assigned, or sublet any portion of the Premises nor entered into any license or concession agreements with respect thereto except as follows:
  - 6. Tenant shall not modify the documents contained in Exhibit A without the prior written consent of Landlord's mortgagee.
- 7. All monthly installments of Base Rent, all Additional Rent and all monthly installments of estimated Additional Rent have been paid when due through

  The current monthly installment of Base Rent is \$
- 8. All conditions of the Lease to be performed by Landlord necessary to the enforceability of the Lease have been satisfied and Landlord is not in default thereunder. In addition, the undersigned has not delivered any notice to Landlord regarding a default by Landlord thereunder.
- 9. No rental has been paid more than thirty (30) days in advance and no security has been deposited with Landlord except the Letter of Credit in the amount of as provided in the Lease.
- 10. As of the date hereof, there are no existing defenses or offsets, or, to the undersigned's knowledge, claims or any basis for a claim, that the undersigned has against Landlord.
- 11. If Tenant is a corporation, limited liability company, partnership or limited liability partnership, each individual executing this Estoppel Certificate on behalf of Tenant

hereby represents and warrants that Tenant is a duly formed and existing entity qualified to do business in California and that Tenant has full right and authority to execute and deliver this Estoppel Certificate and that each person signing on behalf of Tenant is authorized to do so.

12. There are no actions pending against the undersigned under the bankruptcy or similar laws of the United States or any state.

, 201 .

Executed at

on the

day of

- 13. Other than in compliance with all applicable laws and incidental to the ordinary course of the use of the Premises, the undersigned has not used or stored any hazardous substances in the Premises.
- 14. All tenant improvement work to be performed by Landlord under the Lease has been completed in accordance with the Lease and has been accepted by the undersigned and all reimbursements and allowances due to the undersigned under the Lease in connection with any tenant improvement work have been paid in full.

The undersigned acknowledges that this Estoppel Certificate may be delivered to Landlord or to a prospective mortgagee or prospective purchaser, and acknowledges that said prospective mortgagee or prospective purchaser will be relying upon the statements contained herein in making the loan or acquiring the property of which the Premises are a part and that receipt by it of this certificate is a condition of making such loan or acquiring such property.

"Tenant":
a
By:
Its:
By:
Ito

Exhibit E

# EXHIBIT F

# COMMENCEMENT DATE MEMORANDUM

With respect to that certain lease ("Lease") dated ("Landlord"), and ("Tenant"), whereby Landlord leased to , 201 between Tenant and Tenant leased from Landlord approximately rentable square feet of that certain office building located at , California ("Premises"), Tenant hereby acknowledges and certifies to Landlord as follows: (1) Landlord delivered possession of the Premises to Tenant substantially complete on (2) The Lease commenced on ("Commencement Date") and Tenant's obligation to pay Rent commenced on (3) The Expiration Date of the Lease is rentable square feet of space; (4) The Premises contain (5) Tenant has accepted and is currently in possession of the Premises and the Premises are acceptable for Tenant's use; (6) Tenant's Building Percentage is (7) Base Rent per month is IN WITNESS WHEREOF, this Commencement Date Memorandum is executed this day of \_\_\_ "Tenant" By: \_\_\_ Its:

Exhibit F

Its:

# EXHIBIT G

# INCLUDED PERSONAL PROPERTY

# 575 – 9th Floor Furniture Inventory List

Photo:	Description:
[PHOTO 1	• 1 table
[PHOTO 2]	• 1 table
	• 2 chairs
[PHOTO 3]	• 1 table
	• 3 chairs
[PHOTO 4]	• 1 table
	• 7 chairs
[PHOTO 5]	• 1 table
[PHOTO 6]	• 1 table
	• 4 chairs
[PHOTO 7]	• 3 file cabinets (1 4-drawer cabinet, 2 3-drawer cabinets)
[PHOTO 8]	• 2 chairs
	Exhibit G
	-1-

# EXHIBIT H

# COVENANTS, CONDITIONS RESTRICTIONS

Third Restated and Amended Declaration Establishing Reciprocal Easements dated September 15, 2005 and recorded as Instrument Number 2005000738612 on September 19, 2005, in the Official Records of Orange County

Exhibit H

# EXHIBIT I

# JANITORIAL SPECIFICATIONS

(See Attached)

[JANITORIAL SPECIFICATIONS]

Exhibit I -1-

# EXHIBIT J

# HELIPORT LICENSE AGREEMENT

(See Attached)

Exhibit J

# 555 ANTON BOULEVARD HELIPAD ACCESS AGREEMENT

### 555 ANTON MET HELISTOP Costa Mesa, California

# HELISTOP LICENSE AGREEMENT AND APPLICATION

# A. GENERAL INFORMATION

• Site: 555 Anton Blvd, Costa Mesa, C.A.

Owner: PR II/MCC SOUTH COAST PROPERTY OWNER, LLC,

a Delaware limited liability company

Owner Location: c/o McCarthy Cook & Co.
 575 Anton Boulevard, Suite 350

Costa Mesa, California 92626

Owner's California Representative:

Location: McCarthy Cook & Co.
Address: 575 Anton Blvd., Suite 350
Costa Mesa, CA 92626

Telephone: (949) 427-5200 Facsimile:

Site Security Office:

Location: 575 Anton Blvd., Suite 350

Costa Mesa, CA 92626

Telephone: (949) 427-5200

# B. <u>LICENSE AGREEMENT</u>

- 1. THE AGREEMENT IS MADE BY AND BETWEEN PR II/MCC SOUTH COAST PROPERTY OWNER, LLC, A DELAWARE LIMITED LIABILITY COMPANY ("LICENSOR") AND VERITONE, INC., A DELAWARE CORPORATION ("LICENSEE").
- 2. LICENSOR GRANTS TO LICENSEE A LICENSE TO USE THE HELISTOP LOCATED ON THE SITE, SUBJECT TO ALL TERMS AND CONDITIONS OF THE LICENSE AGREEMENT AND APPLICATION, AND TO ALL THE TERMS AND CONDITIONS CONTAINED IN THE RULES AND REGULATIONS ATTACHED AS EXHIBIT A AND INCORPORATED BY REFERENCE HEREIN. IN CONSIDERATION FOR THE LICENSE GRANTED TO IT BY THE LICENSOR, LICENSEE AGREES TO THE FOLLOWING:

### Helistop License Agreement and Application

### 3. <u>INSURANCE</u>

- 3.1 Licensee agrees to maintain public liability, property damage and bodily injury insurance with Combined Single Limit of at least \$3 Million per passenger and \$10 Million per occurrence. Passenger liability converge must be specified as part of Licensee's public liability coverage;
- 3.2 Licensee must provide All Risk Aircraft Hull coverage;
- 3.3 All policies must contain a waiver of subrogation clause with respect to Licensor and its subsidiaries and affiliates and McCarthy Cook & Co. and its subsidiaries and affiliates;
- 3.4 Licensor and its subsidiaries and affiliates and McCarthy Cook & Co. and its subsidiaries and affiliates must be named as "additional insureds" on all policies;
- 3.5 All policies must contain a provision for thirty (30) day written notice of material change or cancellation;
- 3.6 All insurance coverage must include a cross liability provision and a stipulation that the Licensee's coverage is primary as to any other valid and collectible insurance; and,
- 3.7 The insurance required to be provided by Licensee may be provided by the owner of the helicopter using the helipad.
- 4. LICENSEE AGREES TO MAINTAIN WORKER'S COMPENSATION INSURANCE FOR STATUTORY LIMITS AND EMPLOYER'S LIABILITY COVERAGE WITH LIMITS OF \$1,000,000.00.
- 5. LICENSE AGREES TO INDEMNIFY, DEFEND AND HOLD HARMLESS LICENSOR, AND ITS SUBSIDIARIES AND AFFILIATES AND MCCARTHY COOK & CO. AND ITS SUBSIDIARIES AND AFFILIATES, AND THEIR RESPECTIVE PARTNERS, DIRECTORS, OFFICERS, AFFILIATES, REPRESENTATIVES, AGENTS AND EMPLOYEES (THE "INDEMNITIES") FROM ANY AND ALL JUDGMENTS, DAMAGES, CLAIMS, ATTORNEY'S FEES AND COSTS WHICH THE INDEMNITIES AND THEIR SUCCESSORS OR ASSIGNS MAY SUFFER OR INCUR ARISING OUT OF LICENSEE'S USE OF THE HELISTOP AND ACCESS THERETO REGARDLESS OF THE CAUSE THEREOF.
- 6. THIS AGREEMENT IS TERMINABLE AT WILL BY LICENSOR, WITHOUT PENALTY OR LIABILITY TO THE LICENSOR, IN THE EVENT ONE OR MORE TENANTS IN THE BUILDING COMPLAINS ABOUT THE HELICOPTER OPERATIONS. NOTWITHSTANDING THE FOREGOING, LICENSOR AGREES NOT TO TERMINATE THIS AGREEMENT AS PROVIDED IN THIS SECTION 6 UNTIL SUCH TIME AS LICENSEE HAS HAD A REASONABLE OPPORTUNITY TO RESPOND TO SUCH COMPLAINTS.

### Helistop License Agreement and Application

- 7. THIS LICENSE IS ISSUED WITH THE FOLLOWING LIMITATIONS WHICH ARE ATTACHED AS EXHIBIT A AND INCORPORATED BY REFERENCE HEREIN.
- 8. IN THE EVENT THE CITY OF COSTA MESA REQUIRES PROOF OF COMPLIANCE WITH THE CONDITIONS OF APPROVAL REGARDING THE NOISE LEVEL OF THE LICENSEE'S HELICOPTER, SUCH DOCUMENTATION SHALL BE PROVIDED TO CITY AND LICENSOR AT LICENSEE'S EXPENSE.
- 9. LICENSEE IS FAMILIAR WITH THE SITE, THE HELISTOP AND ALL EQUIPMENT, INSTRUMENTS AND OTHER ITEMS MADE AVAILABLE FOR LICENSEE'S USE IN CONNECTION WITH THE HELISTOP, AS WELL AS THE APPROVED FLIGHT PATH AND THE RULES AND REGULATIONS FOR THE HELISTOP WHICH ARE ATTACHED AS EXHIBIT A. LICENSEE HEREBY EXPRESSLY ACKNOWLEDGES AND AGREES THAT LICENSOR MAKES NO REPRESENTATIONS OR WARRANTIES WHATSOEVER CONCERNING THE CONDITION OF OR THE SUITABILITY FOR LICENSEE'S USE OR PURPOSES OF THE HELISTOP AND ANY SUCH EQUIPMENT, INSTRUMENTS AND OTHER ITEMS MADE AVAILABLE FOR LICENSEE'S USE AND LICENSEE ACCEPTS THE SAME "AS IS" AND AGREES THAT LICENSEES USE OF THE HELISTOP AND ALL SUCH EQUIPMENT, INSTRUMENTS AND OTHER ITEMS IS AT THE LICENSEE'S SOLE RISK.
- 10. LICENSE GRANTED HEREIN IS NON-EXCLUSIVE. OWNER RESERVES THE RIGHT TO GRANT, RENEW OR EXTEND SIMILAR LICENSES TO USE THE HELISTOP TO OTHERS. DUE TO THE LIMITATIONS ON THE NUMBER OF OPERATIONS PER DAY, AS SET FORTH ON EXHIBIT A, USE OF THE HELISTOP MAY NOT BE AVAILABLE TO LICENSEE FROM TIME-TO-TIME.

# C. LICENSE APPLICATION

APPLICATION

- 1. REQUIREMENTS
- · Applicant must complete License Application and sign it.
- Completion of this Agreement form does not constitute permission to use the facility. Pilots and machines will be approved on an individual basis. No substitutions will be allowed unless authorized in writing.

	10111011
2.1	NAME:
2.2	CATEGORY OF OPERATOR:
	Commercial Operator
	Corporate Operator
	Private Operator
	Government Agency

Helis	stop Li	cense Agreement and Application			
	2.3	ADDRESS:			
	2.4	TELEPHONE NUMBER:			
	2.5	FACSIMILE:			
	2.6	OFFICIAL IN CHARGE OF FLIGH	T OPERATIONS:		
		Name			
	2.7 H	IELICOPTER TO BE USED:			
Helico	opter	Manufacturer _	1	2	3
		Model			
Max	Gross V	Year Weight (MGW)			
141474	Gross	Reg. No ("N")			
Eng. TSO Total Time on Aircraft					
		Equipped with mufflers			
Equi	pped w	ith 2-way radios:			
		123.5 MHz			
3.	MAI	NTENANCE			
	3.1	Maintenance Organization:			
	3.2	Address:	_		
Telep	ohone N	Number: ()	_		
	3.3 N	Name of Supervisor:			

3.

PILOT QUALIFICATIONS. List the names of all pilots in your employ who will fly into this helistop. Provide the following information for each pilot. Attach additional pages as necessary. Airman's Certificate Number: 4.2 4.3 Current Ratings: \_\_\_ Total Helicopter hours DAY: \_\_\_\_\_ 4.4 Total Helicopter hours NIGHT: \_\_\_\_\_ 4.5 Hours in Equipment to be used DAY: \_\_\_\_\_ 4.6 Hours in Equipment to be used NIGHT: \_\_\_\_\_ 4.7 Rooftop Experience—DAY: \_\_\_\_\_\_NIGHT: \_\_\_\_\_ 4.8 LICENSEE SHALL PROVIDE LICENSOR WITH A COPY OF THE ACTUAL POLICY SHOWING ITS COMPLIANCE WITH THE ABOVE NOTED 5. REQUIREMENTS. Insurance Company \_\_\_\_\_ Agent \_\_\_ Address Telephone Number \_\_\_\_\_

**Helistop License Agreement and Application** 

Policy Number(s)

Expiration Date \_\_\_\_\_

Policy Term \_\_\_\_

Helistop License Agreement and Application			
6.	ACKNOWLEDGMENT OF APPLICANT		
	I hereby certify that I have read the above information and the foregoing is true and correct. In the event any of the above information is false or misrepresented, this shall constitute sufficient justification for immediate cancellation of the License Agreement.		

DATE \_\_\_\_\_

# **Helistop License Agreement and Application**

### Exhibit A

# 555 ANTON METRO CENTER HELISTOP

### RULES AND REGULATIONS

# A. CONDITIONS APPROVED BY COSTA MESA CITY COUNCIL (February 19, 1991)

Type of Permit: Conditional Use Permit PA-91-04 Summary:

- 1. Visual Flight Rules (VFR) only:
- 2. Hours of Operation 7:00 AM to 7:00 PM, except in emergencies;
- 3. <u>Number of Operations Per Day</u> not to exceed maximum average of 2 takeoffs and 2 landings;
- 4. <u>Flight Path</u> Ingress and egress to helistop should follow San Diego Freeway (Interstate 405) until an altitude where a noise free transition (avoiding flight over any noise sensitive areas) to the VFR helicopter routes can be made;
- 5. Aircraft Approved by City Council to Use Helistop
  - Agusta 109A Mark II."
  - Aerosaptiale 355E and F1
  - Aerospatiale 360C
  - Bell 205B
  - Bell 205L
  - McDonnell Douglas 500
  - Bell 222 (see Feb. 19, 1991 City Council Minutes)
  - 5.1 Helicopters used at this site shall be limited to those generating no more noise than the above models. If helicopters other than those listed above are proposed for use, the applicant shall show proof of compliance with this condition;
- 6. Noise. If residences are constructed within the Home Ranch project area, and the helicopter flight path generates noise that adversely impacts the residents, the CUP shall be returned to the Planning Commission for review, including but not limited to possible modifications of the flight paths;
- 7. <u>Annual Review</u>. Heliport subject to annual review by the Planning Commission.

# Helistop License Agreement and Application

### B. NOTIFICATION REQUIREMENTS FOR NEIGHBORING OFFICE BUILDINGS

Notification Procedure for 555 Anton Metro Center and Neighboring Two Towers 535 and 575 Anton.

- 1. The helicopter operator should send a fax or telephone the helicopter operations' schedule to Security 555 Anton.
- 2. Security at 555 Anton will immediately notify security at 535 and 575 Anton about the helicopter operation schedule.
- 3. Security at 555 Anton will keep an operations log of all helicopter activities in the form attached hereto a copy of which is available to the Licensee upon written request.

### C. OPERATIONAL RULES

- Licensee agrees to give the security office at 555 Anton a minimum of four (4) hours' notice prior to a helicopter operation to allow security to notify neighboring buildings.
- 2. Licensee agrees to keep a log of all helicopter operations in the log form attached hereto and a copy of which is available to the Licensor upon written request.
- 3. Since the helistop will permit the landing of only one helicopter at a time and there is no parking space, operators will be allowed on the helipad only for the purpose of picking up and dropping off passengers or packages. Parking for longer than one (1) hour will not be permitted.
- 4. All applicable federal, state and local regulations, laws and ordinances, as well as these rules will be strictly observed. Pilots will in every case lock controls and conduct passengers to and/or from the stairs at the main stairwell.
- 5. The pilot will review and acknowledge the information in the Pilot Information Sheet a copy of which is attached and incorporated herein.
- 6. Licensee must comply with the restrictions of the Federal Aviation Regulations (FARs) for Class C airspace.
- 7. The weight limitation is 12,000 lbs., which is designated on the corner of the helipad. The size of the touchdown area is 50 ft. in diameter and the takeoff and landing area is 65 ft. in diameter.
- 8. Licensee will instruct the pilot to use the HAI Fly Neighborly noise abatement flight procedures compatible with the helicopter and commensurate with the safe conduct of the flight mission.

#### **Helistop License Agreement and Application**

#### D. OPERATION ACCESSORIES

- 1. The emergency roof telephone is in the penthouse mechanical room.
- 2. Fire fighting equipment is adjacent to the helipad.
- 3. Approved lighting is available for night operations.

#### E. ACCESS TO THE HELIPAD

- 1. Access to the roof of the property around the helistop will be restricted to the following:
  - 1.1 Maintenance and security personnel, authorized inspectors and other employees of Licensee having duties requiring their presence in the area; and
  - 1.2 Aircraft crew.
- 2. Unless accompanied by the aircraft crew, the public will not be permitted beyond the gate. At the head of the stairway which will carry this sign: DANGER. NO ADMITTANCE. PASSENGER WILL WAIT HERE UNTIL ESCORTED TO HELICOPTER BY PILOT.
- 3. In moving from stairway to helipad, passengers will be cautioned to stay within the clearly defined pathway; to stay away from the tail rotor of the aircraft; and to follow the aircraft crew is instructions at all times.

Helistop License Agreement and Application

### FORM OF LOG

(See Attached)

-4-

EXHIBIT K

## FORM OF SNDA

(See Attached)

Exhibit K -1-

After Recording, Return to:

Gibson, Dunn & Crutcher LLP 200 Park Avenue New York, New York 10166 Attn: Aaron Beim, Esq.

#### SUBORDINATION, NON-DISTURBANCE AND ATTORNMENT AGREEMENT

This SUBORDINATION, NON-DISTURBANCE AND ATTORNMENT AGREEMENT (this "Agreement") is made as of , 20 , by and between [Affiliate of Blackstone Real Estate Special Situations Advisors L.L.C.], whose address for notice under this Agreement is c/o Blackstone Mortgage Trust, Inc., 345 Park Avenue, New York, New York 10154 (together with its successors and/or assigns, "Lender"), and VERITONE, INC., a Delaware corporation, whose address for notice under this Agreement is 575 Anton Boulevard, Suite 900, Costa Mesa, California 92626 ("Tenant").

#### Statement of Background

A. Lender has made a loan (the "Loan") to , a ("Landlord"), which Loan is evidenced by a promissory note (the "Note") made by Landlord to order of Lender and is secured by, among other things, a deed of trust and security agreement (the "Security Instrument") made by Landlord for the benefit of Lender covering the land (the "Land") described on Exhibit A attached hereto and all improvements (the "Improvements") now or hereafter located on the Land (the Land and the Improvements hereinafter collectively referred to as the "Property").

B. Tenant is the tenant or lessee under a lease dated as of , 2017 (which lease, as the same may have been amended and supplemented as of the date hereof, is hereinafter called the "*Lease*"), covering approximately 37,875 square feet of space located in the Improvements (the '*Premises*"). Landlord holds all rights of landlord or lessor under the Lease.

C. The parties hereto desire to make the Lease subject and subordinate to the Security Instrument in accordance with the terms and provisions of this Agreement.

#### Statement of Agreement

For and in consideration of the mutual covenants herein contained and other good and valuable considerations, the receipt and sufficiency of which are hereby acknowledged, and notwithstanding anything in the Lease to the contrary, it is hereby agreed as follows:

1. Lender, Tenant and Landlord do hereby covenant and agree that the Lease with all rights, options (including options to acquire or lease all or any part of the Premises), liens and charges created thereby, is and shall continue to be subject and subordinate in all respects to the Security Instrument and to any renewals, modifications, consolidations, replacements and extensions thereof and to all advancements made thereunder.

- 2. Lender does hereby agree with Tenant that, in the event Lender becomes the owner of the Premises by foreclosure, conveyance in lieu of foreclosure or otherwise, so long as no Event of Default (as defined in the Lease) has occurred and is continuing under the Lease, (a) the Lease shall continue in full force and effect as a direct Lease between the succeeding owner of the Property and Tenant, upon and subject to all of the terms, covenants and conditions of the Lease, for the balance of the term of the Lease, and Lender will not disturb the possession of Tenant, and (b) the Premises shall be subject to the Lease and Lender shall recognize Tenant as the tenant of the Premises for the remainder of the term of the Lease in accordance with the provisions thereof; provided, however, nothing contained herein shall prevent Lender from naming Tenant in any foreclosure or other action or proceeding initiated by Lender pursuant to the Security Instrument to the extent necessary under applicable law in order for Lender to avail itself of and complete the foreclosure or other remedy.
- 3. Tenant does hereby agree with Lender that, in the event Lender becomes the owner of the Premises by foreclosure, conveyance in lieu of foreclosure or otherwise, then Tenant shall attorn to and recognize Lender as the landlord under the Lease for the remainder of the term thereof, and Tenant shall perform and observe its obligations thereunder, subject only to the terms and conditions of the Lease. Tenant further covenants and agrees to execute and deliver upon request of Lender an appropriate agreement of attornment to Lender and any subsequent titleholder of the Premises.
  - 4. Tenant agrees that, in the event Lender succeeds to the interest of Landlord under the Lease, Lender shall not be:

or

- (a) liable for any act or omission of any prior landlord (including, without limitation, the then defaulting Landlord), except to the extent such act or omission relates to ongoing maintenance or repair obligations;
- (b) subject to any defense or offsets which Tenant may have against any prior Landlord (including, without limitation, the then defaulting Landlord), except for offsets expressly provided for under the Lease;
- (c) bound by any payment of rent or additional rent which Tenant might have paid for more than one month in advance of the due date under the Lease to any prior Landlord (including, without limitation, the then defaulting Landlord);
  - (d) bound by any obligation to make any payment to Tenant which was required to be made prior to the time Lender succeeded to any prior Landlord's interest;
  - (e) accountable for any monies deposited with any prior Landlord (including security deposits), except to the extent such monies are actually received by Lender;
    - (f) bound by any surrender, termination, material amendment or material modification of the Lease made without the consent of Lender.
- 5. Tenant acknowledges that Landlord has executed and delivered to Lender an Assignment of Leases and Rents (the "Assignment of Leases"), which assigns the Lease and the rent and all other sums due thereunder to Lender as security for the Loan, and Tenant hereby

expressly consents to such assignment. Tenant acknowledges that the interest of the Landlord under the Lease has been assigned to Lender solely as security for the purposes specified in said assignments, and Lender shall have no duty, liability or obligation whatsoever under the Lease or any extension or renewal thereof, either by virtue of said assignments or by any subsequent receipt or collection of rents thereunder, unless Lender shall specifically undertake such liability in writing or unless Lender or its designee or nominee becomes, and then only with respect to periods in which Lender or its designee or nominee becomes, the fee owner of the Premises. Tenant further agrees that upon receipt of a written notice from Lender of a default by Landlord under the Loan, Tenant will thereafter, if requested by Lender, pay rent to Lender in accordance with the terms of the Lease. Landlord shall have no claim against Tenant for any amounts paid to Lender pursuant to any such notice.

6. Tenant hereby agrees to give to Lender copies of all notices of Landlord default(s) under the Lease in the same manner as, and whenever, Tenant shall give any such notice of default to Landlord, and no such notice of default shall be deemed given to Landlord unless and until a copy of such notice shall have been so delivered to Lender. Lender shall have the right, but not the obligation, to remedy any Landlord default under the Lease, or to cause any default of Landlord under the Lease to be remedied, and for such purpose Tenant hereby grants Lender such additional period of time as may be reasonable to enable Lender to remedy, or cause to be remedied, any such default in addition to the period given to Landlord for remedying, or causing to be remedied, any such default. Tenant shall accept performance by Lender of any term, covenant, condition or agreement to be performed by Landlord under the Lease with the same force and effect as though performed by Landlord. No Landlord default under the Lease shall exist or shall be deemed to exist (i) as long as Lender, in good faith, shall have commenced to cure such default after receipt of notice and shall be prosecuting the same to completion with reasonable diligence, subject to force majeure, and Lender shall use reasonable efforts to complete same within ninety (90) days after notice thereof, provided that, if such default cannot reasonably be cured within such ninety (90) day period and Lender has diligently commenced to cure such default promptly within the time contemplated by this Agreement, such ninety (90) day period shall be extended for so long as it shall require Lender, in the exercise of due diligence, to cure such default, or (ii) if possession of the Premises is required in order to cure such default, or if such default is not susceptible of being cured by Lender, as long as Lender, in good faith, shall have notified Tenant that Lender intends to institute proceedings under the Security Instrument, and, thereafter, as long as such proceedings shall have been instituted and shall be prosecuted with reasonable diligence. In the event of the termination of the Lease by reason of any default thereunder by Landlord, upon Lender's written request, given within thirty (30) days after any such termination, Tenant, within fifteen (15) days after receipt of such request, shall execute and deliver to Lender or its designee or nominee a new lease of the Premises for the remainder of the term of the Lease upon all of the terms, covenants and conditions of the Lease, provided, however, such default shall be cured by Lender or its designee or nominee. Lender shall have the right, without Tenant's consent, to foreclose the Security Instrument or to accept a deed in lieu of foreclosure or to exercise any other remedies under the Security Instrument.

7. Lender shall have no obligation or incur any liability with respect to the construction or completion of the improvements in which the Premises are located or for completion of the Premises or any improvements for Tenant's use and occupancy. Lender shall

have no obligations nor incur any liability with respect to any warranties of any nature whatsoever, including, without limitation, any warranties respecting use, compliance with zoning, hazardous wastes or environmental laws, Landlord's title, Landlord's authority, habitability, fitness for purpose or possession. In the event that Lender shall acquire title to the Premises or the Property, Lender shall have no obligation, nor incur any liability, beyond Lender's then equity interest, if any, in the Premises, and Tenant shall look exclusively to such equity interest of Lender, if any, in the Premises for the payment and discharge of any obligations or liability imposed upon Lender hereunder, under the Lease or under any new lease of the Premises.

- 8. Tenant acknowledges, without limitation, that the subordinations provided hereby include a full and complete subordination by Tenant of any options it may have to purchase all or any portion of the Property, rights of first refusal or similar rights, whether such rights are provided in the Lease or elsewhere. Tenant hereby further agrees that any such option to purchase or right of first refusal shall be expressly inapplicable to any foreclosure of the Security Instrument or acquisition of the Property or any interest therein by Lender or any designee of Lender by conveyance in lieu thereof or similar transaction.
- 9. If any portion or portions of this Agreement shall be held invalid or inoperative, then all of the remaining portions shall remain in full force and effect, and, so far as is reasonable and possible, effect shall be given to the intent manifested by the portion or portions held to be invalid or inoperative.
  - 10. This Agreement shall be governed by and construed in accordance with the laws of the State in which the Property is located.
- 11. Lender shall not, either by virtue of the Security Instrument, the Assignment of Leases or this Agreement, be or become a mortgagee in possession. In addition, Lender shall not, either by virtue of the Security Instrument, the Assignment of Leases or this Agreement, be or become subject to any liability or obligation under the Lease or otherwise until Lender (or its nominee or designee) shall have acquired the interest of Landlord in the Premises, by foreclosure or otherwise, and then such liability or obligation under the Lease shall extend only to the person that so becomes the landlord under the Lease and then only to those liability or obligations accruing subsequent to the date that Lender has acquired the interest of Landlord in the Premises as modified by the terms of this Agreement.
- 12. All notices or other communications required or permitted to be given pursuant to this Agreement shall be in writing and shall be considered as properly given if
  (a) mailed by first class United States mail, postage prepaid, registered or certified with return receipt requested; (b) by delivering same in person to the intended addressee; or
  (c) by delivery to an independent third party commercial delivery service for same day or next day delivery and providing for evidence of receipt at the office of the intended addressee. Notice so mailed shall be effective upon its deposit with the United States Postal Service or any successor thereto; notice sent by a commercial delivery service shall be effective upon delivery to such commercial delivery service; notice given by personal delivery shall be effective only if and when received by the addressee; and notice given by other means shall be effective only if and when received at the office or designated address of the intended addressee. For purposes of notice, the addresses of

the parties shall be as set forth on the first page; provided, however, that every party shall have the right to change its address for notice hereunder to any other location within the continental United States by the giving of thirty (30) days' notice to the other parties in the manner set forth herein.

- 13. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, legal representatives, successors, successors-in-title and assigns. When used herein, the term "Landlord" refers to Landlord and to any successor to the interest of Landlord under the Lease, and the term "Lender" refers to Lender and to any successor-in-interest or assign of Lender under the Security Instrument.
- 14. This Agreement may be executed in any number of counterparts, each of which shall be effective only upon delivery and thereafter shall be deemed an original, and all of which shall be taken to be one and the same instrument, for the same effect as if all parties hereto had signed the same signature page. Any signature page of this Agreement may be detached from any counterpart of this Agreement without impairing the legal effect of any signatures thereon and may be attached to another counterpart of this Agreement identical in form hereto but having attached to it one or more additional signature pages.

[THE REMAINDER OF THE PAGE IS INTENTIONALLY BLANK]

IN WITNESS WHEREOF.	the parties hereto have exe	ecuted this Agreement under	seal to be effective as of	the date set forth in the f	irst paragraph hereof.

		LENDER:	
		[PARLEX 5 FINCO, LLC, a Delaware limited liability co.	mpany]
		By: Name: Title:	
A notary public or other officer completing this certificate vertruthfulness, accuracy, or validity of that document.	rifies only the identity of the individual wh	no signed the document to which	this certificate is attached, and not the
State of California County of Los Angeles	)		
On , before me, on the basis of satisfactory evidence to be the person(s) who his/her/their authorized capacity(ies), and that by his/her/their instrument.	. ,	nstrument and acknowledged to m	•
I certify under PENALTY OF PERJURY under the laws of t	ne State of California that the foregoing pa	aragraph is true and correct.	
WITNESS my hand and official seal.			

Signature \_\_

		TENANT:	
		VERITONE, INC., a Delaware corporation	
		By: Name: Title:	
A notary public or other officer completin truthfulness, accuracy, or validity of that of		e individual who signed the document to which this certificate is attached, and not the	
State of California County of Los Angeles	)		
	e to be the person(s) whose name(s) is/are subs	Notary Public, personally appeared , who prover the within instrument and acknowledged to me that he/she/they executed the einstrument the person(s), or the entity upon behalf of which the person(s) acted,	
certify under PENALTY OF PERJURY	under the laws of the State of California that t	ne foregoing paragraph is true and correct.	
certify under PENALTY OF PERJURY WITNESS my hand and official seal.	under the laws of the State of California that t	ne foregoing paragraph is true and correct.	

The undersigned Landlord hereby consents to the foregoing Agreement and confirms the facts stated in the foregoing Agreement.

## LANDLORD:

## PR II/MCC SOUTH COAST PROPERTY OWNER, LLC, a

	Delaware limited liability company	
	By: PR II/MCC South Coast Venture, LLC, a Delaware limited liability company, its sole member	
	By: McCarthy Cook Ventures XIII, LLC, a Delaware limited liability company, its authorized member	
	By: Name: Title:	
A notary public or other officer completing this certificate verifies only the identity of the truthfulness, accuracy, or validity of that document.	te individual who signed the document to which this certificate is attached, and not the	
State of California ) County of Los Angeles )		
On , before me, , a Notary Public, personally appeared , who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s) is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or the entity upon behalf of which the person(s) acted, executed the instrument.		
I certify under PENALTY OF PERJURY under the laws of the State of California that the	he foregoing paragraph is true and correct.	
WITNESS my hand and official seal.		
Signature		

#### Exhibit A

The land situated in the City of Costa Mesa, in the County of Orange, State of California, and described as follows:

#### PARCEL A:

PARCEL 3 OF LOT LINE NO. 05-03, IN THE CITY OF COSTA MESA, COUNTY OF ORANGE, STATE OF CALIFORNIA, RECORDED SEPTEMBER 19, 2005 AS INSTRUMENT NO. 2005000738603, OF OFFICIAL RECORDS.

#### PARCEL B:

PARCELS 2 AND 3 OF LOT LINE NO. 05-04, IN THE CITY OF COSTA MESA, COUNTY OF ORANGE, STATE OF CALIFORNIA, RECORDED SEPTEMBER 19, 2005 AS INSTRUMENT NO. 2005000738604 AND RE-RECORDED SEPTEMBER 26, 2005 AS INSTRUMENT NO. 2005000758094, BOTH OF OFFICIAL RECORDS.

#### PARCEL C:

A PERPETUAL, NON-EXCLUSIVE, APPURTENANT EASEMENT FOR VEHICULAR INGRESS, EGRESS AND ACCESS AS SET FORTH IN THAT CERTAIN DECLARATION ESTABLISHING RECIPROCAL EASEMENTS AND MAINTENANCE OBLIGATIONS RECORDED OCTOBER 27, 1983 AS INSTRUMENT NO. 83-473268 OF OFFICIAL RECORDS OF ORANGE COUNTY, CALIFORNIA.

#### PARCEL D

A PERPETUAL, APPURTENANT, NON-EXCLUSIVE EASEMENT FOR PROVIDING AND MAINTAINING VEHICULAR PARKING SPACES AND FOR PEDESTRIAN AND VEHICULAR INGRESS AND EGRESS AS SET FORTH IN THAT CERTAIN THIRD RESTATED AND AMENDED DECLARATION ESTABLISHING RECIPROCAL EASEMENTS RECORDED SEPTEMBER 19, 2005 AS INSTRUMENT NO. 2005000738612 OFFICIAL RECORDS OF ORANGE COUNTY, CALIFORNIA.

#### PARCEL E:

A PERPETUAL, APPURTENANT, NON-EXCLUSIVE EASEMENT FOR INGRESS AND EGRESS PURPOSES AS SET FORTH IN THAT CERTAIN RESTATED GRANT OF EMERGENCY WATER FIRE SERVICE EASEMENT RECORDED SEPTEMBER 24, 1985 AS INSTRUMENT NO. 85-363132 OF OFFICIAL RECORDS OF ORANGE COUNTY, CALIFORNIA.

APN: 410-501-32; 410-501-33; 410-501-34; 410-501-35; 410-501-37; 410-501-38

# $\frac{\text{EXHIBIT L}}{\text{FORM OF LETTER OF CREDIT}}$ [DRAFT LETTER OF CREDIT]

# CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

#### I, Chad Steelberg, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Veritone, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 8, 2017 /s/ Chad Steelberg

Chad Steelberg
Chief Executive Officer and Chairman of the Board
(Principal Executive Officer)

# CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

#### I, Peter F. Collins, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Veritone, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 8, 2017 /s/ Peter F. Collins

Peter F. Collins Executive Vice President and Chief Financial Officer (Principal Financial and Accounting Officer)

# CERTIFICATIONS OF THE CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

Each of the undersigned hereby certifies, in accordance with 18 U.S.C. 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, in his capacity as an officer of Veritone, Inc., that, to his knowledge, the Quarterly Report of Veritone, Inc. on Form 10-Q for the period ended June 30, 2017 fully complies with the requirements of Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 and that the information contained in such report fairly presents, in all material respects, the financial condition and results of operation of the company.

August 8, 2017 By /s/ Chad Steelberg

Chad Steelberg

Chief Executive Officer and Chairman of the Board

(Principal Executive Officer)

August 8, 2017

By /s/ Peter F. Collins

Peter F. Collins

Executive Vice President and Chief Financial Officer (Principal Financial and Accounting Officer)

A signed original of this written statement required by Section 906 has been provided to Veritone, Inc. and will be retained by Veritone, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.